

WL-TR-94-8033

NEXT GENERATION CONTROLLER SPECIFICATION FOR AN OPEN SYSTEMS ARCHITECTURE STANDARD

Martin Marietta Astronautics PO Box 179 Denver, CO 80201

28 September 1994



Final Report For the Period March 1989 - August 1994

Approved for Public Release; Distribution is Unlimited.

19950110 033

Manufacturing Technology Directorate Wright Laboratory Air Force Materiel Command Wright-Patterson Air Force Base, Ohio 45433-7739

DITO QUARTER AND AUTORNO 1

NOTICE

When Government drawings, specifications, or other data are used for any purpose other than in connection with a definitely related Government procurement operation, the United States Government thereby incurs no responsibility nor any obligation whatsoever; and the fact that the government may have formulated, furnished, or in any way supplied the said drawings, specifications, or other data, is not to be regarded by implication or otherwise as in any manner licensing the holder or any other person or corporation, or conveying any rights or permission to manufacture, use, or sell any patented invention that may in any way be related thereto.

This report has been reviewed by the Office of Public Affairs (ASC/PA) and is releasable to the National Technical Information Service (NTIS). At NTIS, it will be available to the general public, including foreign nations.

This technical report has been reviewed and is approved for publication.

DANIEL R. LEWALLEN Project Manager

MICHAEL F. HITCHCOCK Supervisor

Lesmusser

BRUCE A. RASMUSSEN, Chief Integration Technology Division Manufacturing Technology Directorate

"If your address has changed, if you wish to be removed from our mailing list, or if the addressee is no longer employed by your organization please notify WL/MTIA, Bldg. 653, 2977 P St., Suite 6, W-PAFB, OH 45433-7739 to help us maintain a current mailing list."

Copies of this report should not be returned unless return is required by security considerations, contractual obligations, or notice on a specific document.

REPORT DOCUMENTATION PAGE			FORM APPROVED OMB NO. 0704-0188
Public reporting burden for this collectior nstructions, searching existing data sou of information. Send comments regardir tions for reducing this burden, to Washin Davis Highway, Suite 1204, Arlington, V (0704-0188), Washington, DC 20503.	n of information is estimated to rces, gathering and maintainin ng this burden estimate or any gton Headquarters Services, A 22202-4302, and to the Off	o average 1 hour per ng the data needed, o other aspect of this Directorate for Inform fice of Management a	response, including the time for reviewing and completing and reviewing the collection collection of information, including sugges- nation Operations and Reports, 1215 Jeffers and Budget, Paperwork Reduction Project
1. AGENCY USE ONLY (Leave Blani	() 2. REPORT DATE	3. REPO	RT TYPE AND DATES COVERED
	28 September 1994	Final	March 1989 - August 1994
4. TITLE AND SUBTITLE			5. FUNDING NUMBERS
			C F33615-89-C-5706
Next Generation Controller	rchitecture Standard		PE 78011
			PR 3095
			TA 06
			WU 18
7. PERFORMING ORGANIZATION N	AME(S) AND ADDRESS(ES)		8. PERFORMING ORGANIZATION
Martin Marietta Astronautics PO Box 179 Denver, CO 80201			
9. SPONSORING MONITORING AGENCY NAME(S) AND ADDRESS(ES) Manufacturing Technology Directorate Wright Laboratory Air Force Materiel Command			10. SPONSORINGMONITORING AGENCY REP NUMBER WL-TR-94-8033
12a. DISTRIBUTION/AVAILABILITY STATEMENT 12b. DISTRIBUTION CODE Approved for Public Release; Distribution is Unlimited. 12b. DISTRIBUTION CODE			
13. ABSTRACT This report describes an open archite process control, and coordinate meas retrofit applications. A component-l blocks that address system requirem "application architecture" that addres using off-the-shelf, reusable "impler issues are addressed through the use Interface (POSIX) standard that is n aspects of the methodology on a typ components, requirements, and dom	acture framework for advance surement. The methodology of based approach is used that all ents and resulting functionali- sses specific system messagir mentation components" for the of system "profiles." a conve- ow gaining wide acceptance ical machine tool controller a ain models.	ed manufacturing con described is applicat lows system architec ty requirements. Ini- ng requirement. An e application softwar ention adopted form A number of examp pplication Appendi	ntrollers including machine tools, robotics, ole to both system design as well as system ctures to be developed from generic build itial architectures are used to produce an "application system" is then synthesized re. Hardware/software specific platform the IEEE Portable Operating system ples are provided demonstrating use of all ices provide details with respect to specific
14. SUBJECT TERMS			15. NUMBER OF PAGES
Open System Architecture, Computer Reusable Software, Hardware/Softwa	Numerical Control (CNC), I re Platform	Machine Tool Contro	ol. 315 16. PRICE CODE
17. SECURITY CLASSIFICATION	18. SECURITY CLASS	19. SECURITY CL	ASS 20. LIMITATION ABSTRAC
Unclassified	Unclassified	Unclassified	UL
			Standard Form 298 (Bey 2-8

Standard Form 298 (Rev 2-89) Prescribed by ANSI Std Z239-18 298-102

.

1

-

9

FOREWORD

The purpose of this document is to provide a description of the Next Generation Workstation/Machine Controller (NGC) program philosophy and objectives in sufficient detail so to be able to quickly and efficiently assess the impact on emerging system designs and concepts. The focus for this document is the Specification for an Open System Architecture Standard (SOSAS) that is the primary deliverable of the NGC program. The SOSAS will capture definitions, conventions, and standards as they relate to the final NGC family of controllers. The document presented here is not meant to be construed as the final SOSAS, but rather it represents the key elements of the SOSAS necessary to achieve the desired objectives relating to open systems, interchangeability, interoperability, portability, etc. It is fully anticipated that subsequent discussion of this document will provide some of the most important feedback with respect to the production of the final SOSAS document. In this respect, comments relating to this document are strongly encouraged.

This is the complete SOSAS document. An overview is also available entitled "Next Generation Controller Specification for an Open Systems Architecture Standard - Overview".



9

T/	٩B	LE	OF	CO	NT	ENTS	
							-

1.0	INTRODUCTION	1
Technolog	gy Infusion	1
Maintenan	100	2
Operator 7	Fraining	2
Document	Summary	3
2.0	NGC SYSTEM PHILOSOPHY AND STRUCTURE	6
2.1	Introduction	6
2.2	Industrial Control Domains	8
2.3	NGC Planning and Execution in the Manufacturing Context	10
2.4	NGC Development Process	13
2.5	Structure of a Working NGC	15
2.6	Timing Considerations	17
2.7	Open System Architecture Specification	18
2.8	Components	19
2.9	NGC Development and Validation Tools	19
2.10	Open System Environment Overview and Profiles	20
2.11	Conformance Overview	
2.12	Growth and Evolution	
3.0	DETAILED ARCHITECTURE	23
3.1	Reference Architecture	
3.1.1	Levels of Abstraction	
3.1.2	Components and Realization	
3.2	Integration Architecture	
3.2.1	Service Groups	31
3.2.2	Domain-Independent Components	32
3.3	NGC OSE Framework	32
3.3.1	NGC OSE Reference Model	33
3.3.2	NGC OSE Taxonomy	34
3.3.2.1	Application Programming Interface (API)	36
3.3.2.1.1	POSIX and ANSI C	
A	n Introduction	39
3.3.2.1.2	Platform Services API	42
3.3.2.1.3	Presentation Management Services (PM API)	44
3.3.2.1.4	Data Management Services (DM API)	48
3.3.2.1.5	Geometric Modeling Services (GM API)	51
3.3.2.1.6	Communications Services (Comm API)	52
33217	Device Input/Output Services (Device I/O API)	59

•

3.3.2.2	External Environment Interface (EEI)	
3.3.3	Profiles	
3.4	Development Process and Implementation	80
3.4.1	Integration Process	
3.4.2	Application Framework, Architecture, and Implementation	
3.4.3	Standardization and Document Vision	85
3.4.4	Implementation Component Library & Tools	88
3.4.5	Implementation	89
3.5	NGC Conformance	92
4.0	NGC Development Example	
<u>4</u> 1	Application Architecture for Machining	
4.1	Machining Scenario	101
4.2	Factory Interaction	103
4.2.1	Pactory Interaction	105
4.2.2		
4.2.3	JOD Kelease	
4.2.4	Part Program Setup	
4.2.5	Code Augmentation	
4.2.7	Tool Motion	
4.2.8	IEC 1131 Environment	
4.2.9	Implementation Component Messages	
4.3	Implementation	
4.3.1	Integration Process Review	
4.3.2	Implementation Example - Machining Center	
4.3.3	Implementation Example—Low End Controller	
5.0	SUMMARY AND RECOMMENDATIONS	132
5	.1 Summary	
5	.2 Synergistic Activities	
5	.3 Recommendation	135
6.0	GLOSSARY OF TERMS	136
APPEN	DIX A — REFERENCE REQUIREMENTS	A-1
A	A.1 General Requirements	A-1
A	A.2 Axis Motion Control Requirements	A-4
A	A.3 Auxiliary Machine Control Functions	A-5
A	A.4 Programming Requirements	A-7
APPEN	DIX B — PRIMITIVE COMPONENTS	B-1
APPEN	DIX C — ARCHITECTURE DESCRIPTION RULES	C-1

ø

•

	C.1	ADL Rules Accessing Multiple Primitive Components	. C-1 . C-2
	C.2 C.3	ADL Rules Accessing Single Primitive Components	. C-2
APPE	NDD	X D — DOMAIN MODELS	. D-1
APPE	NDD	K E — DOMAIN DICTIONARY	. E-1
	E.1	Bibliography	. E-2
	E.2	Glossary of Terms	.E-3
	E.3	Notation	.E-58
	E.4	Category Index	.E-61
APPE	NDD	K F ARCHITECTURE DESCRIPTION LANGUAGE (ADL)	. F-1
	The	Adaptive Semantic Language	.F-1

LIST OF FIGURES

Figure 2.1a.	Manufacturing Enterprise Hierarchy	7
Figure 2.1b	Integration Architecture	8
Figure 2.2.	Domains of Practice for Industrial Control	9
Figure 2.3-1.	Traditional Flow of NC Machine Tool Programming	10
Figure 2.3-2.	NGC Concept of Product Design and Production Planning	12
Figure 2.3-3.	The Concept of NGC Planning and Execution	12
Figure 2.4.	NGC Development Process	14
Figure 2.5.	Components and the Common Execution Environment (CEE)	15
Figure 2.6.	NGC Timing Domains	17
Figure 2.7	NGC Document Vision	22
Figure 3.1.1	NGC Abstraction Levels	25
Figure 3.1.2	NGC Realization Process	29
Figure 3.2-1	NGC Integration Architecture	30
Figure 3.2-2	NGC Integration Architecture & Related Standards	30
Figure 3.2.2	Domain-Independent Implementation Components	32
Figure 3.3.1	NGC OSE Reference Model	34
Figure 3.3.2	NGC OSE Taxonomy	35
Figure 3.3.2.1	NGC Architecture Perspective	38
Figure 3.3.2.1.3	Presentation Management Services—API Levels and Implementation	47
Figure 3.3.2.1.6-1	Communications Services—Trollius Example APIs Network Dependent	54
Figure 3.3.2.1.6-2	Communications Standards Architectural Hierarchy	56
Figure 3.3.2.1.6-3	CORBA-DCE IDL & RPCs	58
Figure 3.3.2.2-1.	NGC EEI System Context—Physical View	62
Figure 3.3.2.2-2.	NGC EEI Interoperability Standards-Physical Mapping	63
Figure 3.3.3-1	NGC Profiles—Template Structure	72
Figure 3.3.3-2	API Profile Suite	75
Figure 3.3.3-3	POSIX Profile Level 1	76
Figure 3.3.3-4	POSIX.1 Profile Level 2	77
Figure 3.3.3-5	POSIX.4 Profile Level 2	77
Figure 3.3.3-6	POSIX.4a Profile Level 2	78
Figure 3.3.3-7	POSIX.4b Profile Level 2	78
Figure 3.3.3-8	C Language Profile	79
Figure 3.3.3-9	EEI Profile Suite—Platform/Application	79
Figure 3.4.1	NGC Integration Process	81
Figure 3.4.2	From Reference Requirements to Implementation	84
Figure 3.4.3-1	NGC Document Vision	85
Figure 3.4.3-2	ANSI Standardization Process Model—ACS Method	87
Figure 3.4.4	Implementation Component Library & Tools—Structure &	00
	Contents	88
Figure 3.4.5-1	From Application Architecture to Implementation	90
Figure 3.4.5-2	Application Architecture	91

4

ş

Figure 3.4.5-3	Generic Implementation	92
Figure 4.0.	Horizontal Machining Center	96
Figure 4.1.	Example of an Application Architecture for a Machining Center	99
Figure 4.2a	Application Architecture Components and Messages-Part 1	102
Figure 4.2c	Application Architecture Components and Messages-Part 3	103
Figure 4.3.1.	NGC Integration Process	116
Figure 4.3.2-1	Platform Implementation Examples	118
Figure 4.3.2-2	Profile	119
Figure 4.3.2-3	Profile	119
Figure 4.3.2-4	Profile	120
Figure 4.3.2-5	Profile	121
Figure 4.3.2-6	Profiles	121
Figure 4.3.2-7	Platform Partitioning, Example 1	123
Figure 4.3.2-8	Platform Partitioning, Example 2	123
Figure 4.3.3-1a	From Application Architecture to Implementation (Simplified)	126
Figure 4.3.3-1b	From Application Architecture to Implementation (Detailed)	127
Figure 4.3.3-2	Application Architecture	127
Figure 4.3.3-3	Generic Implementation	128
Figure F-1.	A Tower of the n-Towers Model	F-4
Figure F-2.	A Factory Application of the n-Towers Model	F-5
Figure F-3.	Situational Dependency Nodes with Theta Relations	F-14
Figure F-4.	Epsilon Relations of Situational Dependency Nodes	F-15
Figure F-5.	Other Relations of Situational Dependency Nodes	F-16

1

.

~

ACRONYMS

ACLASCII	Cutter Location
----------	-----------------

A/D Analog-to-Digital

- ADA Ada programming language, in honor of Augusta Ada Byron, Countess of Lovelace (1815-1852), daughter of Lord Byron, was assistant and patron of Charles Babbage.
 - ADL Architecture Definition Language
 - AIS Application Interface Specification

ANSI American National Standards Institute

API Application Programming Interface

APT Automatically Programmed Tools

ASCII American National Standard Code for Information Interchange

AT IBM's 80286-based PC bus architecture

BCL..... Binary Cutter Location

BSD Berkeley Software Distributions

CAD Computer-Aided Design

CAM..... Computer-Aided Manufacturing

CAM-I Consortium for Advanced Manufacturing International

CAN Controller Area Network

CCITT International Consultative Committee on Telegraphy and Telephony

CEE Common Execution Environment

CLI Command Line Interpreter

CORBA Common Object Request Broker Architecture

COSE Common Open Systems Environment

COTS Commercial Off-the-Shelf

CNC Computer Numerical(ly) Controlled

CPU..... Central Processing Unit

CSMA/CD.. Carrier Sense Multiple Access/ Collision Detection

D/A Digital-to-Analog

DCE..... Distributed Computing Environment

DEC..... Digital Equipment Corporation

DFS	Distributed File Service
DM	Data Management
DNI	Detailed Network Interface
DOS	Disk Operating System

EDI Electronic Data Interchange EDIF Electronic Design Interchange Format EEI External Environment Interface EISA Extended Industry Standard Architecture

GDI..... Graphical Device Interface

GKS..... Graphical Kernel System

GM Geometric Modeling

GPIB..... General-Purpose Interface Board (IEEE 488)

GUI..... Graphical User Interface

HUI..... Human-User Interface

H/W Hardware

IEC International Electrotechnical Commission

IEEE Institute of Electrical and Electronics Engineers, Inc.

I/F Interface

IGES Initial Graphics Exchange Specification

I/O Input/Output

ISA Industry Standard Architecture

OR

Instrument Society of America

ISO International Organization for Standardization

September 23, 1994

xi

ISR Interrupt Service Routine

ITB Inside-the-box

LAN Local Area Network

LIS Language-Independent Specification

MCU..... Machine Control Unit

MIMD Multiple Instruction, Multiple Data

MIT Massachusetts Institute of Technology

MS Microsoft

NASREM ... NASA/NIST Standard Reference Model for Telerobot Control System Architecture

NC Numerical Control, Numerically Controlled

NDL Neutral Database Language

NFS Network File System

NGC Next Generation Workstation/ Machine Controller

OBIOS...... Open Basic Input/Output System

OMG Object Management Group

OS..... Operating System

OSE Open Systems Environment

OSF Open Software Foundation

OSI International Organization for Standardization (ISO) Open Systems Interconnection

PC..... Personal Computer

PDES Product Data Exchange using STEP

PEX PHIGS Extension (to MIT's X Windows

PHIGS Programmer's Hierarchical Interactive Graphics System

PII..... Protocol Independent Interface

PLC Programmable Logic Controller

POSIX Portable Operating System Interface

PWI Public Windows Interface

RDA Remote Data Access

RISC...... Reduced Instruction Set Computer

September 23, 1994

SOSAS Rev. 2.5

RPC Remote Procedure Call

RT.....Real-Time

RTOS Real-Time Operating System

SBX Specification for an I/O Expansion Bus (ANSI/IEEE 959)

SCO..... Santa Cruz Operation, Inc.

SCSI Small Computer Systems Interface (ISO 9316)

SERCOS Serial Data Link for Real-time Communication between Contols and Drives

SNI Simple Network Interface

SOSAS Specification for an Open System Architecture Standard

SQL Structured Query Language

STE..... Specification for the Exchange of Product Model Data

STEP Standard for the Exchange of Product Model Data

- S/W.....Software
- TCOS Technical Committee on Operating Systems (and Application Environments)

TCP/IP Transmission Control Protocol/ Internet Protocol

- TFA Transparent File Access
- UNIX Operating system based on Multics
- US PRO...... U.S. Product Data Association

UTA University of Texas at Arlington

VME...... Versatile Backplane Bus (IEC 821)

VMS Serial Subsystem Bus of VME Bus (IEC 823)

VSB Parallel Subsystem Bus of VME Bus (IEC 822)

XPG X/Open Portability Guide

¢

1.0 INTRODUCTION

The Next Generation Workstation/Machine Controller (NGC) program philosophy of systems that are interoperable, interchangeable, and portable is an outgrowth of a more general trend in all areas of systems development that stress "openness." Like many new ideas in engineering, the concept of an open system is one that has been much easier to describe qualitatively than it has been to establish rigorous design methodologies. In the arena of advanced manufacturing control systems, NGC and the Specification for an Open System Architecture Standard (SOSAS) are an attempt to merge the evolving methodology of open systems with a well-established technology base in machine tool, robotics, measurement, and process control. As in any emerging technology, growth and evolution do not come without some degree of controversy and argument. There is no lack of recognition on the part of the NGC development team that while the long-term benefits of an open approach to manufacturing controller design are obvious to virtually everyone, the burden of making this vision a reality will fall on a vendor community that is engaged in an intensive economic struggle and is faced, on a daily basis, with the necessity of looking at an unforgiving bottom line. The result is that the NGC development process has never been viewed as a "do it all over again" exercise, i.e., starts with a clean sheet of paper. Instead, it has been viewed as "given the space of open system solutions, pick the solution that is a closest fit to existing systems."

The NGC can be viewed in many different perspectives. While the control builder will ultimately be the one tasked with turning ideas and paper into working hardware and software, it is the end user of manufacturing control systems who will, in the long term, dictate the final form and structure of these evolving open systems. It is the end user who will apply the resulting systems in an attempt to produce products that are cheaper than the competitor and as a result, the end users that are successful in this process will, explicitly or implicitly, determine the marketplace forces that ultimately shape the technology. There are many complexities involved in defining what leads to a cheaper product from the standpoint of the end user. While the actual dollar cost of a manufacturing controller will always remain an important aspect of the overall metric, there are many other factors that influence the end user's manufacturing strategy.

Technology Infusion

The ability to rapidly integrate new technologies and practices has always been, and will always remain, a crucial factor in competitiveness. A new technology can appear in many different ways and impact practice, hardware, software, or all of these. A good example of this is the emerging trend towards machine tool controllers that makes use of temperature and/or vibration data at the spindle. This requires not only the ability to integrate new sensors into the system but also the ability to rapidly and efficiently modify existing software structures in order to get the right data to the right place in an appropriately timely fashion and implement newer and potentially more complex control laws that take advantage of the new data.

Maintenance

Key factors in the competitiveness of any system are the supporting elements necessary to keep the system running. In general, commonalty of system elements leads to lower system cost for a variety of reasons. These include a decrease in inventory size for spare components, the standardization of procedures for diagnosing and repairing systems, and the ability to use trained diagnostic and repair personnel across a wider variety of systems because of the decrease in specialization.

Operator Training

Commonalty of systems decreases the amount of additional training that is required to transition operators from one system to another. Major retraining can be supplanted by incremental training based upon knowledge of a core system structure. In this case, an open system architecture that utilizes standardized system features can achieve nearly the same desired commonalty as a mandate for all systems from one manufacturer. From the standpoint of the operator, common look and feel of interfaces is more important than the internal system that generates the look and feel.

While the issue of the specific attributes desired in a NGC controller were discussed at much greater length in both the Needs Analysis document and the Requirements Definition Document, it is obvious that the resulting NGC systems must help the end user in a number of ways; neither very high-technology nor extremely low-cost systems are in themselves the full answer.

Instead of attempting to mandate a specific point solution to the advanced manufacturing controller problem, a philosophy is embodied herein that leads to maximum flexibility with the respect to ability to modify both the hardware and the software in the system to achieve desired enhancements in capability as necessary.

Document Summary

Section 2.0 provides an overview of philosophy and structure of NGC. It describes the basic concepts that allow the freedom of action on the part of the designer to choose the proper mix of hardware and software to achieve an optimal solution to a given design problem.

To facilitate a standardized approach to developing the structural elements of a system, a component based approach has been adopted. The fundamental attributes of this approach are described including the use of a *reference architecture* that is comprised of *primitive components* that are used to delineate system requirements and structure in terms of generic "building blocks". From the reference architecture an *application architecture* is constructed that captures the functionality of the end system at an abstract level. While a level of abstraction above the final hardware and software system, the application architecture allows responsibilities and dependencies to be clearly established. Finally, through the selection of *implementation components* a final software structure is determined. The step from application architecture to implementation is a very difficult one because it involves the specifics of the system platform. The platform includes all system hardware as well other system software such as operating systems, communication software, etc.

Effectively dealing with platform issues requires dealing with the specifics of different types of potential hardware solutions as well as a large number of standards and conventions that have arisen with respect to communication, operating systems, device interfaces, graphics representations, etc. Because NGC has the goal of establishing a foundation for the introduction of open systems technology into the advanced manufacturing arena, it was felt that it would be a mistake for a small group to choose a relatively small set of possible standards and conventions that would then exclusively serve the NGC community. As a result, it was necessary to find a means of accommodating systems that could draw on a wide range of hardware and software solutions in arriving at a system implementation. This problem was dealt with through the use of a representation concept known as a profile. This concept is crucial to understanding how the NGC specification can achieve the open system objectives without overly constraining system designers. Profiles can be thought of as a means of classifying NGC-compliant systems. The concept of the profile was adopted from POSIX (Portable Operating System Interface) literature. (POSIX and it's role in NGC are discussed in greater depth in section 3.0.) The vendor will use profiles to succinctly and unambiguously describe the system structure. With a profile specified for a specific system, the end user will be able to determine whether or not the candidate system has the flexibility necessary for a given set of applications. A very simplified type of profile is now commonly used to describe software packages as being "Mac" or "PC" compliant. Similarly, the designation of "DOS" or "Windows" can be thought of as a subprofile under PC. This flexibility assessment could include the ability to operate with existing software packages, the ability to communicate with other factory systems with minor modifications, or the ability to expand hardware features such as input/output (I/O) or motor drivers. NGC does not preclude a vendor from building a system that is essentially non-open; it does, however, guarantee that a system purchaser can be assured of understanding precisely what degree of openness is being provided. Necessarily, standards play an important role in achieving the NGC goals. In addition to describing the basic architectural philosophy, Section 2.0 also provides a summary of the relevant standards that have been chosen for application to the NGC system and the rationale for their selection.

Section 3.0 includes both a more complete discussion of the component based approach to the design of the application architecture. Platform issues are discussed in depth from the standpoint of two basic elements; the *Application Program Interface* (API), and the *External Environment Interface* (EEI). Both are discussed in section 3.3. The full development process leading to implementation is discussed in section 3.4. Issues related to conformance are discussed in section 3.5.

A full design example illustrating all of the basic NGC concepts is presented in section 4.0 for a 5-axis machining center. The example includes the selection of requirements and primitive components leading to the application architecture. Implementation is considered from the standpoint of different realistic hardware and COTS software options.

Section 5.0 consists of a brief summary and further issues for the evolution of NGC. The primary issues cited with respect to the further evolution of NGC are the availability of libraries of NGC components (both primitive and implementation) that can be made widely available to the community at large for incorporation into emerging systems and the development and availability of tools that will facilitate the NGC development process. While an initial set of primitive components are provided for machining applications as an appendix to this report (see below), sufficient raw material for a robust initial implementation component library through the work of the National Institute of Standards and Technology (NIST) as an output of the Enhanced Machine Controller (EMC) program. Therefore, the key issue is the establishment of a repository for the initial library.

The issue of tools is more complex but not intractable. It is generally believed that the tool technology necessary for NGC currently exists through a variety of existing developmental and commercial packages. Ideally, a complete tool set could be integrated and made available to the general community. This would be the fastest way to spur the growth of NGC type systems. Unfortunately, the resources for such an activity have not yet been identified.

A set of appendices is provided that includes: Appendix A- Reference Requirements, Appendix B- Primitive Components, Appendix C- Architecture Description Rules, Appendix D- Domain Models, Appendix E- Domain Dictionary and Appendix F-Architecture Description Language.

2.0 NGC SYSTEM PHILOSOPHY AND STRUCTURE

2.1 Introduction

The open system concepts specified in this document establish a framework for the design and construction of a family of workstation/machine controllers for industrial machines. This framework addresses issues associated with both application software and the hardware software platforms on which this application software will run. This Next Generation Controller (NGC) specification is based where possible on de jure and de facto open standards in manufacturing, controls, and computing technology but provides a sufficiently robust structure such that evolving and new standards and technology can readily be incorporated. The NGC specification facilitates commonality of components across a complete range of machine controllers. These controllers are to be used in a wide variety of manufacturing operations that include machining, robotics, and inspection.

A NGC supports a wide range of processing and discrete part manufacturing applications, including machine tools of all types, robots, electronic assembly, material handling devices, inspection devices, and virtually all types of automated equipment in both manned and unmanned environments and networked and stand-alone configurations. Specifically, a NGC controls a manufacturing workstation, which is defined as a single material transformer and related material handling and inspection equipment. As shown in Figure 2.1, the manufacturing workstation is placed at the lowest level in the hierarchy of the overall manufacturing enterprise. Several manufacturing workstations, each controlled by an NGC, are managed by a cell. Multiple cells are managed by a center, and multiple centers are managed by a factory. At the top of the hierarchy, the enterprise manages multiple factories. While this specification focuses on the lowest level of the this hierarchy, the results and structure are immediately and easily extended to include the higher hierarchical levels on an evolutionary basis. Therefore, a foundation is established for much more complex Computer Integrated Manufacturing (CIM) systems that will retain the open systems structure described in this document.

The open systems concepts are evolving from requirements established by the NGC community of control builders, machine integrators, end users, members of standards organizations, and university researchers. It is flexible enough to cover the broad range

of manufacturing practice, and it is extendible to absorb future advances in technology while accommodating existing manufacturing and controller practice.

To produce an enduring standard, manufactured products and manufacturing processes must be described independently of specific equipment, methods, or topologies. An architecture, calling for a specific topological arrangement and interconnection of components, does not provide enough flexibility to control every application in discrete part manufacturing, nor does it accommodate technological advances over a period of decades. Moreover, several topologies may be required in order to develop a system and understand its operation under a variety of situations. The overall view of the system architecture that is used in the NGC process is captured in the *Integration Architecture*.



Figure 2.1a. Manufacturing Enterprise Hierarchy

The integration architecture, Figure 2.1b, combines manufacturing and control application components with a supporting infrastructure. It also acts as a framework for incorporating open and de facto standards. Applications components intercommunicate by messaging and can be adapted to a variety of component interconnection topologies.

Typical computing platform conventions, definitions, and capabilities are provided by the Open Systems Environment (OSE).



IMPLEMENTATION COMPONENTS

Figure 2.1b Integration Architecture

2.2 Industrial Control Domains

Manufacturing enterprises are complex, information-intensive environments. The practices associated with discrete part manufacturing can be grouped into three principal domains: (1) manufacturing practice, (2) controller practice, and (3) computing practice. The three domains and some associated concepts are shown in Figure 2.3. Terminology and representations from these domains are used in this specification. Existing and emerging standards, developed within the purview of these domains, are incorporated in this specification by reference. Thus the considerable investment and effort devoted to the three domains, especially computing practice, can be reused effectively to support manufacturing applications.



Figure 2.2. Domains of Practice for Industrial Control

Manufacturing practice covers the process of transforming raw materials into finished parts. The underlying concept for the discrete part manufacturing process is art-to-part. The process begins by capturing a part design using a computer-aided design (CAD) system. The resources for making the part are scheduled, and the machining, material handling, and measuring activities are planned using a computer-aided manufacturing (CAM) system. Motion paths are planned and used to drive the appropriate machinery, i.e., tool paths for machining, robot end-effector paths for manipulation, and probe paths for measurement. Some related manufacturing practice standards include those for feature-based part description, part programming, and measurement languages. Many of the steps in this process are done manually or disconnected. In the future, the art-to-part process is envisioned as a seamless information flow from the designer's concept to the finished part.

Controller practice covers the organization and control of manufacturing equipment. Equipment is controlled mostly by closed-loop controllers, which provide continuous (motion) control and discrete control.

Computing practice covers the computing and communication technologies required to support manufacturing and controller practice. The practice includes design and use of computer hardware and software. Computing practice is the target of intense open systems activity, which benefits NGC as it evolves and becomes sufficiently mature.

2.3 NGC Planning and Execution in the Manufacturing Context

The traditional manufacturing practice and general flow of numerically-controlled (NC) machine tool programming starts with a design review by a part programmer as shown in Figure 2.3-1. The part programmer's role has been to analyze the product design and determine how to make the part economically, the machining range of workpiece, the method of mounting the workpiece on the machine tool, the machining sequence of every operation, and the cutting tools and cutting conditions. The output of the part programmer has been a part programming manuscript that documents the logical order of machine operations. The part geometry and cutter location data are post processed to a specific machine-readable format and transferred to the machine control unit (MCU).



Figure 2.3-1. Traditional Flow of NC Machine Tool Programming

The MCU memory has an executive program, program reader, shop floor programming language, cutter line control, machine interface logic, program subroutines or canned cycles, and tool changer control. The program reader in the MCU converts the coded instructions and the MCU controller generates an output signal to servo mechanisms to drive and direct the machine tool. The machine tool resolver or feedback device determines the precise location of the workpiece relative to the cutter and returns this data to the machine controller and also to machine operator workstation display. The workstation display may display full operational and parametric data; display job setup instructions; and have the capability for program verification, editing and update, maintenance and troubleshooting, fault messages, and graphical representations of the workpiece, tools, and cutter paths. The responsibility of the machine operator is to monitor the machine operation, monitor workpiece loading and unloading, provide information feedback, and perform operator programming through manual data input. The intent of a NGC is to provide the product designers, process planners, NC programmers, machine operators, and factory managers with more flexibility in planning and committing factory resources. This increased flexibility and efficiency will be feasible through industrial machine controllers which allow open exchange of planning data, feature-based part representation, cutter location data, tooling and fixture data, and operations sequence data among different types machine tools without rewriting the part programs source code, and eliminating programming and processing for specific machines.

The concept of NGC product design and production planning is shown in Figure 2.3-2. The product designer is able to generate a CAD that specifies the geometric features and considers material handling, assembly requirements, and manufacturability. Part geometry and features are interpreted through a CAM system to develop the process plan and generate the machining program. A computer-aided process planning (CAPP) system would evaluate economical production based upon a part family data base that has a library of part features, geometric data representations, and machining processing data. The process planning system determines the machine routing and operations sequence, methods, fixtures and tooling, and setups. From the process plan and machining program, a NC program is sent to a NGC workstation controller.

The concept of NGC planning and execution is shown in Figure 2.3-3. The part program could be assigned to any machine in the factory that has the process capability and control configuration to accept feature-based part model input, such as Product Data Exchange Specification (PDES), or NC legacy code such as binary cutter location/ASCII cutter location (BCL/ACL) part programs, RS-274 part programs, or automatically-programmed tools (APT) part programs. The NGC operations planner creates an operations plan that specifies the logical order of workstation operations among the different machine mechanisms for part handling, machining, inspection, setups, and fixturing. The part model input and operations plan provides input data instructions to a task planner and path planner. The task planner and path planner specifies for each mechanism the logical order of motion tasks, tool changes, feeds and speeds, detailed motion geometry, and obstacle avoidance. The output from NGC operations planner, task planner, and path planner directs each mechanism to obtain the resultant machine tool action.



Figure 2.3-2. NGC Concept of Product Design and Production Planning



Figure 2.3-3. The Concept of NGC Planning and Execution

2.4 NGC Development Process

To achieve the goals of the NGC program, it is necessary that the SOSAS structure adequately addresses all of the primary issues associated with system structure, application software, and platform development. Without a roadmap that shows how all of these elements are accommodated by the SOSAS, it is difficult to appreciate how the SOSAS supports all of these areas: (1) the bottoms-up development of a fully integrated system, (2) development and integration of new system software, and (3) expansion of platform capabilities to accommodate new requirements. A key element of the NGC philosophy is the ability to achieve rapid and effective system modifications that address only the part that "needs to be fixed" and do not necessarily entail massive, expensive system re-design.

A roadmap for the SOSAS development process is shown in Figure 2.4, which illustrates the basic design and development paths for addressing system structural issues and platform issues. It is important to realize that any block on the diagram can be considered to be an entry point for the design process. Development of new system application software, for example, would entail only the "upper" path with the platform definition being considered fixed. Similarly, if only the platform structure is being considered for modification, then the application architecture elements can be considered as given from the standpoint of platform issues and therefore, definition of the new platform profile is the key issue.

The key to understanding Figure 2.4 is the realization that system structure is embodied in the specification of an **application architecture**. The "upper" design path results in the application architecture. To arrive at an application architecture, **reference requirements** are compared with **user needs** to derive a set of problem specific **application requirements**. By using a well-defined set of reference requirements as the basis for selecting the final system application requirements, a tremendous amount of consistency is brought to the entire requirements process, something not always achieved in ad hoc requirements derivation processes.



Figure 2.4. NGC Development Process

The Reference Architecture is the key to turning application requirements into the application architecture. The Reference Architecture consists of both primitive and aggregate components. Components are abstract building block elements that describe functionality and communication. The application architecture is built from these components. Although the Reference Architecture will necessarily be a "living" library that continually grows, the objective is to develop a set of components that "span the space" of desired functionality and communication for establishing a system structure. Therefore, in the system design process, when the Reference Architecture is found insufficient, new components are added to the reference architecture. The application architecture is a complete and consistent topology for the representation of the system.

The lower pathway of Figure 2.4 addresses the issue of how the system will be physically implemented from the standpoint of processors, buses, communication, I/O, etc. It is not an objective of the SOSAS to enforce a particular design philosophy with respect to the system platform structure. The NGC does not attempt to lead all system designers in the direction of a standard "box". What it does do, however, is to establish a methodology for accurately capturing what has been produced by a specific vendor.

Platform requirements are used to select the standards (or conventions) that the designer feels are necessary for a specific platform implementation based on heritage, cost, performance, etc. Once the basic platform structure has been established, the key elements of the design are documented in the form of a **platform profile.** The notion of the platform

14

profile was derived from the realization that (1) because of diverse market forces there will never be a "one platform does it all" to satisfy everyone, and (2) from the standpoint of achieving system openness with all of the associated "-ilities" it is of crucial importance that the user is able to understand what is being purchased in such a way that system expansion and extension can readily be assessed. The platform profile, by documenting the key standards and conventions used to construct the system, provides this insight.

The application architecture and platform design (as reflected in the profile) are unified in the **design application** phase of the process. Here, the **implementation component library** is used to take advantage of off-the-shelf software elements to instantiate the application architecture and insure consistency with the platform profile. Abstract functionality is replaced by actual software that includes both the functional aspects as well as the practical man-specific aspects necessary to accommodate the chosen platform profile.

2.5 Structure of a Working NGC

The NGC open system is comprised of application software exchanging information via data communication mechanisms and connected to the services provided by operating systems and hardware through a common interface called an application program interface (API). The software is implemented in the form of components that interchange messages in the process of carrying out their responsibilities. Components run under a Common Execution Environment (CEE) that provides for transparent peer-to-peer message exchange between the components as well as other services. Figure 2.5 shows this concept.



Figure 2.5. Components and the Common Execution Environment (CEE)

Each component is a single separate thread of execution within the CEE since a component must be able to send and receive messages independently from the other

components. A component encapsulates the data and functionality it needs to carry out its assigned responsibility. This approach is extremely flexible compared with older hard-wired systems, allowing the NGC system software to be reconfigured dynamically during operation by activating or deactivating agents. Dynamic reconfiguration does not cover the addition or removal of hardware; connecting or disconnecting hardware will require a system shutdown.

Components have the following attributes:

- Responsibility—The role the component plays, distinguishing it from the other components in the CEE, in the successful overall operation of the NGC system;
- Peer-to-peer Relationship(s)—The collaborative relationships the component has with other components as required to carry out its responsibility;
- Behavior(s)—The specific functionality encapsulated by the component, where each behavior is expressed as one or more operations to be performed in response to a message;
- Message(s)—The complete set of specific instructions necessary for evoking all of the behaviors encapsulated by the component These messages must be defined in enough detail to guarantee interoperability.
- Application Program Interface(s)—The interface(s) a component uses specifically to access services provided by the Open Systems Environment (OSE).

For a specific controller application, a configuration is an information structure that provides a description of all required hardware and software elements and their interconnections. The configuration contains the information needed to maintain safe and reliable operation of the controller in carrying out all of its intended responsibilities. The configuration includes, as a minimum, the description of all of the required components. A directory service component or group of components has the responsibility for assuring that all of the components needed for reliable controller operation are available and working properly in the CEE. A configuration is used by the directory service component(s) to maintain a roster of active components and facilitate message interchange among them. The roster of active components will change dynamically, for example, with a shift in the operational requirements of the controller as it carries out its various responsibilities.

2.6 Timing Considerations

- The NGC stands at the juncture between process planning for cell activities and controlling the equipment that actually fabricates parts or assembles products. Timing considerations are important, but not crucial, for the former types of activity. But controlling the equipment requires strict adherence to a time budget; failure to do so will often result in higher production costs due to flawed parts and/or damaged equipment.

The NGC spans three separate timing domains: non real time (non RT), real time (RT), and hard real time (hard RT). This partitioning of the timing requirements is shown in Figure 2.6. The non-RT domain of standard computing covers activities like compiling and linking software, reading data from a file, and reporting progress to the cell and other entities in the enterprise. The time required to complete an activity can be flexible, guided by the requirement to finish as early as possible. The standard computer domain covers a wide variety of environments and operating systems, e.g., UNIX, DOS, non-RT POSIX compliant systems, and Windows NT.



Figure 2.6. NGC Timing Domains

An activity in the RT domain is driven by a time budget, but the consequences of missing a deadline are not catastrophic, and the system can recover without serious loss. Some examples of RT activities are operations planning, task planning, path planning, cutter compensation, and a PLC's actuation of the coolant valve. Hard-RT activities have mandated timing deadlines; missing one of these deadlines can have serious consequences like a damaged part that must be reworked or discarded. Hard-RT constraints apply to servo loop closures and programmable logic controller (PLC) cycles that read values from sensors into memory and write values from memory to actuators.

- The need to consider the three timing domains will persist in spite of the ever-improving performance of available computing. As better computers are made, the increased throughput makes more complicated algorithms feasible within a fixed time limit. When they are deployed in advanced manufacturing applications, greater machining precision at higher speeds will be made possible. The result is better quality products at faster production rates. Thus the RT envelope will be pushed continually, and the relevance of the three timing domains will remain as more capable processors emerge.

Several existing controller products can be aligned with the three timing domains. One vendor links a transputer-based, RT platform to a UNIX front end through transputer links. The transputers drive an Industry Standard Architecture (ISA) expansion bus, and all motion servo loops are closed in the transputer using digital-to-analog (D/A) converters and encoder feedback. Another vendor combines standard computing and less demanding RT operations in a single PC AT running the Lynx operating system. The demanding RT operations, that is, motion and high-speed discrete control, are implemented in a separate ISA card. (The use of a separate dedicated function card for demanding RT operations on a DOS platform and distributes control via ARCNET to small proprietary nodes. All of these implementations fit within the timing partitions as long as the messages at the selected boundaries are conformant with the NGC messages.

NGC does not, at this time, directly address the issue of timing performance for resulting system implementations. In all likelihood, this will require dedicate tools that, in addition to insuring other NGC requirements are met, also evaluate the adequacy of the overall software/platform system with respect to performance/stability driven performance requirements. This issue is discussed further in sections 2.9 and 3.4.

2.7 Open System Architecture Specification

Controller designs are constrained from two aspects: the Reference Architecture and open standards (such as POSIX) and industry conventions (such as DOS). For a specific industrial application, controller primitive components are selected from the Reference

Architecture according to application-specific requirements and synthesized into an Application Architecture. The resulting Application Architecture constrains the design of the application system; that is, the system elements specifically required for manufacturing and control. The other constraining aspect supports the operation of the controller, which is based upon open and de facto standards of practice in manufacturing, controllers, computing, and data communication.

2.8 Components

All of the *primitive* components needed to cover the spectrum of machine controller applications over the NGC domain are maintained herein as the Reference Architecture. A component is an abstract encapsulation of functionality and information with an assigned responsibility and an abstract message interface definition. A component is specified as follows:

- Responsibility: The role the component plays, differentiating it from all other components in the Reference Architecture, in the successful overall operation of a controller;
- Message(s): The complete set of distinct instructions necessary for evoking all of the behaviors encapsulated by the component for carrying out its responsibilities. These messages are defined at an abstract level.

Depending on the application, the required primitive components are selected and arranged in an application architecture that will guide the actual design of the controller. An application architecture must contain, as a minimum, the components needed to fulfill all of the responsibilities of the application it is based on. Many such application architectures are possible over the NGC domain. Based on a single application architecture, a component may be implemented and delivered in a variety of ways. This activity will result in libraries of implementation components that will capture the functionality of the aggregated primitive components but contain the additional structure necessary for integration into specific platform implementations.

2.9 NGC Development and Validation Tools

The intense competitive pressures and risks associated with development and subsequent production of industry-fielded, open architecture, machine controllers necessitates the wise

and prudent use of development and validation tools, such as knowledge-based tools, libraries, and object-modeling tools. Tools should address system design and integration support; configuration management; event timing; profile mapping and applications supported; and the taxonomy and hierarchy of events, applications, standards, agents, and components. Metrics and measurement methodologies need to be defined, developed, and tested for validity and significance and then be evaluated for the benefits, capabilities, and features of a specific implementation. Performance testing should consider the related open system specifications, data exchange and information handling protocols, external and internal interfaces, and the explicit and implicit features of openness, i.e., portability, interchangeability, commonality of components, etc. The development and validation tools, and documentation of lessons learned, should be mapped to the capabilities and features of an open architecture machine controller.

Many software tools that would fulfill requirements for the NGC tool suite are either available on the market today or are being developed in other ongoing programs. The ARPA Domain Specific Software Architecture (DSSA) program is in the process of establishing many of the fundamental tools that would be required in the NGC requirements definition, design, and validation and verification activities. As NGC continues to evolve and mature, it will be essential to find mechanisms for capturing the tools legacy that exists and effectively integrating it into the NGC structure.

2.10 Open System Environment Overview and Profiles

The NGC is based on open standards. Open standards help to achieve a level of portability and interoperability between multi-vendor controller products that does not exist in controllers today. This "openness" of the controller facilitates the addition of new controller features and innovative technology with a relative ease that is unavailable in controllers that are closed. The benefits are two-fold: (1) the end-user has a controller that is adaptable to market dynamics and can be easily modified to incorporate the latest cost-saving technologies, and (2) third-party vendors are encouraged by the opportunities to develop new technologies and market niches for a new generation of controllers.

Central to the open theme is the NGC Open Systems Environment (OSE) framework. The OSE framework leverages open standards, both de jure and de facto, to specify an infrastructure for open controllers. The OSE framework embodies three general concepts: the reference model, the taxonomy, and profiles. The OSE reference model is the context for NGC open standards, the taxonomy is the logical grouping of these standards, and the profiles are selections of standards from specific groups. Rather than attempting to mandate a limited set of standards, the concept of the profile allows the vendor and the user to communicate key structural aspects of a system via a "snapshot" that, through the shorthand afforded by defined standards, allows system structure to be quickly determined. This is very similar to the common practice today of indicating whether products are "Mac" or "PC" compatible. The OSE framework is essentially the organized menu for this selection. While the possible number of permutations and combinations of profiles is overwhelming at this point, it is expected that as more vendors produce NGC systems, a smaller set of accepted profiles will emerge. These concepts are again introduced and described in more detail in Section 3.3.

2.11 Conformance Overview

NGC conformance is determined by adherence to a specific profile of standards. The NGC SOSAS does not attempt to specify a standard set of profiles. The market will drive out the set of profiles that are most practical to both the controller developer and end-user.

A NGC-conformant product must include a "NGC Disclosure Statement" that specifies the profile, component interfaces, and other conformance claimers where applicable. Profiling offers the controller developer options for product conformance at a variety of levels to satisfy user requirements at a competitive cost, with varying degrees of openness. Section 3.4 describes the disclosure statement requirements, levels of conformance, language documentation, statements of intent, and associated claimers in more detail.

2.12 Growth and Evolution

This document should be viewed as the initial document in what will be a growing set of companion documents that govern the full NGC development process. In section 3.4 a document hierarchy is discussed that parallels the structure used by IEEE to document the POSIX standard. The structure of this family of documents is shown in Figure 2.7. Like the POSIX standard, there is an overall document that describes basic system structure and philosophy. Other documents (the .1 and .2 documents) address basic issues associated with system framework and implementation. Finally, the architecture document set deals with domain specific considerations. As shown, it is presumed that this would initially include CNC, robotics, process control and PLC applications.

The issue of the criticality of tools to a mature NGC process was discussed in an earlier section. In addition to tools, success of NGC also hinges on the emergence of libraries (or repositories) of effective and reliable implementation components that allow the system developer to quickly satisfy both functionality and platform requirements in the implementation process. As in the case of tools, other ongoing programs have already developed much of the foundation for these libraries. This issue is discussed at greater length in section 5.0.



Figure 2.7 NGC Document Vision: Open Controller Standards

3.0 DETAILED ARCHITECTURE

This section specifies the details on the design, integration, and delivery of NGC conformant industrial controllers. Key elements of the NGC process, such as the Reference Architecture, Application Architecture, Open Systems Environment, etc. that were introduced in the previous - section are described in greater detail with respect to the methodology involved in the NGC design and development process.

Section 3.1 addresses the concept of the Reference Architecture and the evolution of system structure. Emphasis is placed on the basic process that is used to move from Reference Architecture to Application Architecture to a final system implementation as an Application System. It is shown that this process is equivalent to proceeding from a level of high abstraction that deals with general responsibilities to a final system that is implemented via specific choices of hardware and software, defined by appropriate APIs and standards. The NGC levels of abstraction facilitate flexible and extensible software designs and emphasize software reuse.

The NGC integration architecture is presented in Section 3.2 as a framework for incorporating services identified and defined in open standards of computing and data communication. These standards are relevant to NGC because they are likely to be supported as commercially available products from vendors of hardware and operating systems. The integration architecture is crucial to the overall NGC process because it provides a means visualizing the interplay between application software, standard services and operating system functions, and the underlying hardware platform.

Section 3.3 presents the NGC Open Systems Environment (OSE) framework and its role with respect to NGC applications and the Common Execution Environment (CEE). The OSE is standards based, having a taxonomy with application program interfaces (APIs) and external environment interfaces (EEIs) as main branches. Also, the profiling concept is introduced, where, by specifying a profile, the NGC user can select from among a competing set of de jure and de facto standards to suit a particular implementation environment.

Section 3.4 addresses the issue of carrying controller development through to implementation. An application framework guides the design and implementation of the controller, based on available products. Key features of this process are reuse of hardware and software components, adaptation of such components to related applications, and separate development of interoperating components by independent vendors. A strategy for evolving a family of open controller standards is also presented. This strategy parallels that employed by the IEEE for the
POSIX standard. This approach allows for independent documentation of different aspects of the NGC process and the emergence of domain specific companion volumes for specialized domains such as machine tools, robotics, process control, etc.

Section 3.5 addresses the issue of conformance for NGC systems. Because of the mechanism of - profiling, and the latitude it provides to the system designer, conformance becomes an issue of adequate documentation of all aspects of the final system. In addition to the profile suite for both API and EEI, the supplier will also be required to document component interface descriptions and language specifics.

3.1 Reference Architecture

This section describes the NGC Reference Architecture, the starting point for the design of a controller's application software, and its role in specifying the realization of an open controller. Stated in simplest terms, the Reference Architecture consists of a set of *primitive components*. The primitive components can be thought of as abstract building blocks for NGC applications. At the level of primitive components, the issues are those of function, responsibility, and generalized data flow rather than specifics of implementation via either hardware or software. Although it is somewhat of a simplification, primitive components can be thought of as a formalization of the elements normally used for generating system "block diagrams". The primitive components, as captured in the Reference Architecture form a language, of sorts, that can be used, later in the Application Architecture, for capturing overall system structure and requirements. The architectural levels of abstraction mark the stages in the process of implementing a controller from requirements analysis through implementation. Use of the Reference Architecture and a derivative Application Architecture constrains controller designs so that products developed independently by different suppliers can be expected to interoperate within a single controller implementation. Moreover, by publishing the architectural details of the controller design and its interfaces, a supplier is opening the system so that it can be readily extended in performance and/or capability. To realize a controller for a specific manufacturing application, an application architecture is synthesized from primitive components in the Reference Architecture, and the Application Architecture, in turn, is used to constrain the design of the application system.

3.1.1 Levels of Abstraction

For NGC, three levels of abstraction provide the flexibility to configure machines for a wide variety of discrete part manufacturing applications while enabling extensions of the standard into other industrial applications and/or controller implementation technologies. Figure 3.1.1 shows

these levels of abstraction. In decreasing degree of abstraction, they are: domain level (description of the controller in manufacturing and controller terminology), technology level (broad structural partitioning that constrains implementation), and implementation level (actual elements of a working instance of a controller).



Figure 3.1.1 NGC Abstraction Levels

The domain level captures the NGC in the terminology of manufacturing and controller practice. Multiple application scenarios from differing expert viewpoints have been analyzed to produce a common set of reference requirements (see Appendix A), a set of domain models (see Appendix D), and a Reference Architecture. These domain-level elements are independent of a specific technology or implementation. Part fixturing, for example, could be performed manually by a human operator or automatically by a robot. As another example, a controller can be implemented using an analog (continuous variable) design, or as in the NGC case, the implementation technology is digitally-based using computing platforms and programmable logic controllers (PLCs). NGC covers a wide variety of applications, and the controller elements actually used will vary from application to application. For example, the controller for a three-axis milling machine will no doubt have a spindle, but a spindle is meaningless as an element of a centerless grinder.

The Reference Architecture has two parts: the complete set of primitive components across all NGC applications (see Appendix B) and some architecture description rules (see Appendix C). Controller responsibilities are distributed among primitive components in the Reference

Architecture. Each primitive component represents a single responsibility. Architecture description rules are used with application requirements and primitive components to compose an application architecture.

The technology level holds the components of an Application Architecture. The Application - Architecture is synthesized out of primitive components following the analysis of the domain practices of a specific controller application. Interdependencies are made explicit through the definition of the component boundaries for the Application Architecture, and message interfaces, while still abstract, are specialized to those required by the specific application.

At the implementation level, the operational paradigm is *implementation components* exchanging messages within the Common Execution Environment (CEE) in order to fulfill their assigned responsibilities. An implementation component is an encapsulation of functions and data that interoperates concurrently with other implementation components within a computing environment by exchanging messages in a variety of contexts. A component from an application architecture represents one or more implementation components, but the actual number and composition of such implementation components are left as choices for an implementor. Consequently, an unlimited variety of designs can derive from a single application architecture component definition. This allows separate vendors to differentiate their products while retaining a correspondence to an application architecture at the technology level.

An Application Architecture component corresponds to an implementation component or a grouping of implementation components. Implementation components have non-overlapping responsibilities because they conform to the boundaries established by the Application Architecture. That is, no two Application Architecture components share an implementation component or group of implementation components, and no implementation component can be associated with more than one Application Architecture component. An implementation component is defined by its responsibilities, specific behaviors (functionality), exact message interfaces (messages that evoke behaviors), and APIs.

The NGC abstraction levels play an important role. During analysis, application requirements are selected from the reference requirements, and a range or a set of values are applied to them. A transition from the domain level to the technology level occurs when an application architecture is synthesized by applying the application requirements, domain models, and architecture description rules to selected primitive components. The Application Architecture and the integration architecture then guide the NGC design at the implementation level. A design is constrained by application architecture components' responsibilities and message interfaces.

The integration architecture facilitates portable software designs by defining standardized interfaces to services that are characteristic of a computing and data communications environment.

3.1.2 Components and Realization

The Reference Architecture contains all of the NGC primitive components (see Appendix B). A primitive component has only one assigned responsibility and the interface associated with that assigned responsibility; a primitive component's responsibility cannot be partitioned. A primitive component is an abstraction and, therefore, is not associated in the early system specification stage with either software or hardware. It is an abstract encapsulation of functionality and information with an assigned responsibility and an interface description. A primitive component in the Reference Architecture is specified as follows:

- Single Responsibility: The single indivisible role the primitive component plays, differentiating it from other primitive components, in the successful operation of an NGC system.
- Resources: The information the primitive component needs in order to carry out its responsibility.
- Products: The information the component, can provide.
- Temporal Information: The intended timing of the primitive component.

For a specific application, primitive components are selected from the NGC Reference Architecture and synthesized into the components of an Application Architecture. Such components may be composed into larger components, and in that sense, components of an Application Architecture are structurally recursive. Any combination of two or more components is called an aggregation. An aggregation carries the combined responsibilities of its constituents, and it must be able to respond to all of the constituents' abstract messages. A component of an Application Architecture is still an abstraction specifying functionality and responsibility, and it is specified as follows:

- Responsibility: The role the component plays, differentiating it from other components, in the successful operation of the overall controller application.
- Abstract Message(s): The complete set of instructions necessary for evoking all of the behaviors encapsulated by the component. These messages are defined at an abstract level.

The Application Architecture must satisfy all of the application requirements, and its components must interconnect properly, that is, the components' collective responsibilities must be sufficient to cover the application requirements, and abstract message definitions must align at component boundaries. Component interconnectivity is a necessary, but not sufficient condition to guarantee interoperability. Although Application Architecture component interconnectivity is verifiable at the abstract message interface, it will not assure that implementation components constrained by an application architecture are able to exchange messages cooperatively and unambiguously. Thus a component's abstract message definition must be specified exactly when the Application Architecture is realized as an implementation.

For a specific machine and process, the Application Architecture guides the implementation. One or more implementation components carry out an Application Architecture component's responsibilities. Implementation component boundaries conform to those set by the Application Architecture, and messages passed between implementation components correspond to the application architecture components' abstract message definitions. The process of realizing an application system from the Reference Architecture and a selected application architecture is shown conceptually in Figure 3.1.2.

There are as many different Application Architectures as needed to cover the wide range of NGC applications. Since this specification will evolve along with manufacturing and controller technology and be extended to new applications, the Reference Architecture is expected to grow and evolve continually. The variety of possible Application Architectures will also grow and evolve as the use of open industrial controllers becomes widespread.

ŧ



Figure 3.1.2 NGC Realization Process

3.2 Integration Architecture

The integration architecture shown in Figure 3.2-1 is a framework that supports interoperable, portable, scaleable, and interchangeable NGC component implementations, designed and developed by competing vendors. The integration architecture is a detailed refinement (representation) of the application system (see Figures 3.1.1 and 3.1.2). An implementation is divided into domain-dependent and domain-independent parts. The domain-dependent part is made up of the implementation components needed by a specific application and the CEE. The domain-independent part consists of the computing and data communication services that represent the underlying platform. The domain-independent services provide support to the implementation components and to the CEE.



Figure 3.2-1 NGC Integration Architecture

Services are based on computing practice and have a broader applicability than NGC, so that they are termed "domain independent." The integration architecture conveniently associates relevant open standards of practice with the services as shown in Figure 3.2-2. Services are provided through a specified API, and they can be invoked by implementation components and called by other services. NGC services form six groups that act as an abstraction layer between the implementation components and the native operating system, computing hardware, and peripheral devices.

IMPLEMENTATION (COMPONENTS
------------------	------------



Figure 3.2-2 NGC Integration Architecture & Related Standards

3.2.1 Service Groups

The NGC integration architecture provides six groupings of domain-independent services, which include *Presentation Management*, *Data Management*, *Geometric Modeling*, *Communications*, *Platform*, *and Basic Input/Output (I/O) services*. The services are captured as a set of behaviors and an API to those behaviors. The responsibilities assigned to the service groups are described below.

Presentation Management services provide standardized means for multiple applications to present data to a user in a common display representation and to receive input from shared input devices. Its responsibilities include displaying data in a common display (such as a display window); receiving, coordinating, and managing data from the user (possibly from multiple, simultaneous input devices); and creating and managing consistent graphic attributes and user operations.

Data Management services are responsible for the information resources of a manufacturing workstation, allowing applications to share data using an implementation-transparent mechanism. Information resources appear to applications as a centralized repository of data called the information base. The information base may be implemented in a variety of centralized or distributed configurations using data bases, knowledge bases, shared memory, or other forms of information storage.

Geometric Modeling services have responsibility for interactive and non-interactive solid modeling and geometric operations. A solid modeling system unambiguously represents threedimensional bodies using a combination of geometric and topological information. This capability is essential for NGC planning and execution utilizing feature-based part model inputs and geometric representation of part design.

Communication services are responsible for application-to-application communication via a protocol and mechanism independent interface. This facilitates interoperability among independently developed applications.

Platform services are a generic set of operating system and utility services that act as an abstraction layer over different operating system implementations, just like an operating system can be an abstraction layer over different hardware platforms. These services provide access to computing platform behaviors responsible for managing shared computing resources, and they encompass both RT and non-RT requirements.

The **Basic I/O** services implement the responsibility for interacting directly with a wide range of I/O devices. These services are used to initialize devices and transfer data to and from them.

3.2.2 Domain-Independent Components

- For the most part there is an intentional correspondence between primitive components in the Reference Architecture at the domain level and application architecture components at the technology level. Occasionally however, a required component may not be identifiable at the domain level because its role in a controller is considered to be *domain independent*. The *Display Manager, Exception Handler, Directory Services, Safety,* and *Help* services are examples of domain-independent components that have required implementation counterparts. They are shown as implementation components in Figure 3.2.2.



Figure 3.2.2 Domain-Independent Implementation Components

The Display Manager component illustrates the need for such domain-independent components. The Display Manager collects and organizes information intended for the video display of a digital controller implementation, but it is an artifact of computing practice, not manufacturing or controller practice. A Display Manager is obviously needed for the display area of a video monitor that must be shared among the independently-developed, components competing for display real estate during the operation of the controller.

3.3 NGC OSE Framework

The NGC is based on open system standards and conventions. While it would be preferable to deal only with official, documented standards endorsed by standards organizations, it is recognized that "conventions", such as DOS, are important technology elements in the advanced

Ĭ

manufacturing community. The complete set of open standards and conventions is referred to as the Open Systems Environment (OSE) framework.

This section and subsequent subsections introduces the concepts of the OSE reference model, taxonomy, and profiles. The OSE framework embodies all of these concepts, providing a general foundation upon which a multitude of NGC OSEs may be derived. The OSE reference model is the context for NGC open standards, the taxonomy is the logical grouping of these standards, and the profiles are selections of standards from specific groups. Loosely stated, the OSE framework may be viewed as the organized menu for this selection. The mechanism of *profiles*, adopted from POSIX, provides a means for determining very quickly what the overall structure of a system employs with respect to standards and platform elements.

The OSE framework includes relevant de facto and de jure standards of practice in manufacturing, controls, computing, and data communications. Much of the OSE framework emphasis is on the Common Execution Environment (CEE), the environment for NGC application software execution. This emphasis is primarily due to the general-purpose nature of computing and communications and the availability of a large base of existing standards to draw from. Therefore, the CEE can be viewed as the physical instantiation of an OSE profile.

3.3.1 NGC OSE Reference Model

The NGC Open Systems Environment (OSE) reference model is the basic context for categorizing the standards within the OSE framework. It is rooted in the reference model used in POSIX but is intended to accommodate a much larger scope. POSIX, the Institute of Electrical and Electronics Engineers (IEEE) portable operating system interface, originated, in part, from the UNIX operating system. The X in POSIX denotes this UNIX origin. This standard is discussed in greater detail in Section 3.3.2.

Figure 3.3.1 illustrates the OSE reference model. It includes two important interfaces: an application program interface (API) between the application software and platform, and an external environment interface (EEI) that supports the interface of the controller platform to such external devices as displays, file servers, networks, etc. The platform is viewed as the general-purpose computing engine, typically commercial off-the-shelf (COTS) hardware and software, that includes processors, memory, clocks, input/output (I/O) boards, and buses, as well as operating system software and other related, general-purpose software packages.



Figure 3.3.1 NGC OSE Reference Model

The focus of this specification is at the interface level. This allows for a variety of specialized controller platform implementations while still maintaining conformance with interfaces that provide the maximum benefit of open systems.

Open systems are characterized by the fundamental attributes of interoperability, portability, and user portability. A system and its components are interoperable if they are able to work properly together in accomplishing their responsibilities. Software is portable if it can be moved easily between computing platforms of different types with no more effort than recompiling. A system is user portable if its user interface is understandable, consistent in style, and tailorable.

The OSE framework supports system-level interoperability with the specification of EEIs. EEIs are essentially a collection of open standards in the areas of networking protocols, data interchange formats, device I/O, and distributed file systems. The OSE framework supports source-level portability through specification of the APIs. Call-level interface standards (C language bindings) are defined in the areas of general Platform, Device I/O, Communications, Data Management, Geometric Modeling, and Presentation Management services.

3.3.2 NGC OSE Taxonomy

The NGC Open Systems Environment (OSE) standards taxonomy categorizes the open standards referenced within the OSE framework. The NGC approach leverages existing standards in all

applicable areas without "re-inventing the wheel". These standards, many of which are applications independent and of a general-purpose computing nature, play a significant role in opening up the controller.

The standards taxonomy is used in deriving NGC profiles and provides one level of guidance for - NGC design. In previous sections, we specify the components that guide the design of the applications software, the system elements specifically required for manufacturing and control. The OSE taxonomy, related profiles, and component specifications together support the development of controller applications software that is both interoperable and portable.

A high-level inspection of the OSE taxonomy (Figure 3.3.2) reveals a close tie-in with the OSE reference model. The two major branches of the taxonomy map one-for-one with the two major interfaces identified in the reference model, specifically: application program interfaces (APIs) and external environment interfaces (EEIs). In review, the API branch deals with issues of source code portability and the EEI branch deals with system-level interoperability.



Figure 3.3.2 NGC OSE Taxonomy

It is unfortunate that today's available standards do not cover all the areas required for a fully open NGC. Many NGC application areas presently unsupported by existing standards are now being worked by standards committees and open systems consortiums.

This specification leverages such relevant draft standards work if it is sufficiently mature and promises to have market acceptance. This position is taken in lieu of the alternatives: either to define, by a much smaller committee, an independent solution, or to simply not address known gaps in the NGC open systems architecture. Existing standards and draft standards span most of the NGC open requirements space but still leave some gaps. These gaps are duly noted in the subsequent standards discussions.

3.3.2.1 Application Programming Interface (API)

API categories include: *platform*, *presentation management*, *data management*, *geometric modeling*, *communications*, and *device input/output (I/O)*. In each category, a set of open standards are identified to facilitate source code level portability. The essential ingredient for this level of portability is a well-specified, language-dependent, call-level interface description.

The NGC API definitions specify C language bindings. It is left up to the discretion of the NGC application designer as to whether or not he may wish to implement C++ "wrappers" (or some other object-oriented language wrapper) around such C language calls to access API services using a message-passing paradigm.

Wrapping maps the language interface of an application to the object interface. It places an object-oriented interface in front of one that is not. This concept goes hand-in-hand with the concept of a "container" object. Standard "C" applications may be placed inside container objects so that they may send messages through an object-oriented interface while preserving it's appearance as a single application.

The Object Management Group's (OMG) Common Object Request Broker Architecture (CORBA), for example, embodies many of these concepts. It is an open question as to whether there are more CORBA implementations worldwide today than there are Open Software Foundation (OSF) Distributing Computing Environment (DCE) implementations. Both standards are primarily focused on writing applications that communicate with each other. DCE's Remote Procedure Call (RPC) communications method provides node transparency using a client-server methodology. CORBA's API specification is designed around the object-oriented

methodology of message passing and supports the interoperability with other technologies via its "static" API interface, which compiles at runtime.

As evident in the NGC reference model and the standards taxonomy, a large part of the NGC API specification is rooted in the Institute of Electrical and Electronics Engineers, Inc., (IEEE) portable - operating system interface (POSIX). Many of the NGC concepts draw from IEEE/TCOS POSIX work. However, it should be clear that the NGC extends the notion of profiles defined for POSIX, as well as that for International Organization for Standardization/Open Systems Interconnection (ISO/OSI), to include a number of non-POSIX API standards. Purists within the open systems standards communities may have some difficulty with this approach but certain de facto standards have such large market support that they can not be ignored.

A common misconception is that POSIX is an operating system. POSIX describes the contract between the applications and the operating system. It defines the interface between applications and their libraries and says little about how to write those applications programs or how to write the operating system. The importance of adopting the interface is to enable portability of software applications across a variety of platforms. Any specific implementation of a NGC platform is acceptable provided the standard interface API, services, protocols, and associated behaviors are followed.

Therefore, workstation operating systems such as Digital Equipment Corporation (DEC) Ultrix, SMI SunOS, and Hewlett-Packard HP-UX; PC-based operating systems such as Santa Cruz Operation, Inc. (SCO) Xenix and MS Windows NT; micro-kernel-based systems such as Mach and Chorus; and finally real-time operating systems (RTOSs) such as Integrated Systems pSOS, Ready Systems VRTX, and Lynx LynxOS are all acceptable operating system implementations provided that they support at least one of the NGC platform API profiles. In terms of the computer hardware, platform implementations may include processors such as Intel, Motorola and Inmos; backplane busses such as VME, ISA/EISA, and NuBus; and displays such as ASCIIbased terminals and X-terminals. NGC platform implementations comprised of multiple processors and multiple operating systems are also viable solutions.

Many of the COTS operating system products already provide support for the NGC platform API. Others need to be adapted to support one or more of the NGC API profiles. This is especially true for those custom or proprietary operating systems that are not presently based on existing standards. The process of adapting COTS and proprietary products to the desired NGC API naturally leads to the notion of middleware.

Middleware is a term associated with distributed systems and communications. Within a client/server distributed database environment, middleware allows transparent client access of information over wide area networks. In the context of communications, it is a layer of code that sits between the operating system and the application. Middleware is designed to ease the development of peer-to-peer and client/server distributed systems. Middleware implements an API layer that shields distributed system and network designers from the details of specific programming environments. In other words, with middleware developers need not have to write to vendor-specific API; instead they may develop to a more generic, globally compatible standard interface.

The NGC extends the concept of middleware to support generic APIs for computer numerically controlled (CNC) device I/O, fieldbus and distributed control, window-based display management, etc. With this view, implementation of the NGC CEE may simply be considered a selection of computer hardware, a selection of COTS operating system and networking software, and a specific implementation of middleware to support one or several NGC API profiles. Figure 3.3.2.1 illustrates this perspective. The NGC support services layer is the middleware implementation layer.



Figure 3.3.2.1 NGC Architecture Perspective

The following discusses relevant standards within each major API category. POSIX is first introduced as a general background. Explanatory information and several examples are also provided to help clarify and bring into focus some of the key NGC concepts. A general background in computing, communications, and software is assumed.

- 3.3.2.1.1 POSIX and ANSI C: An Introduction

Growing in popularity is the commitment among operating system vendors, especially among the real-time operating system (RTOS) vendors, to support the POSIX interface and real-time (RT) extensions to POSIX. Since POSIX specifies an interface, and does not describe how the functions are implemented, it can be applied across a number of underlying COTS operating systems.

Some operating system vendors have leveraged their existing operating system products by simply adding to them a POSIX-compliant interface. To accomplish this, a translation element must be designed, typically in the form of a runtime library, that takes POSIX system calls into native operating system calls. Digital's OpenVMS with its POSIX C-language runtime library and added POSIX system services kernel is an example of this. This approach provides a migration path toward open solutions in the future while maintaining a level of legacy support for existing applications and their dependencies on proprietary interfaces. The burden of immediate conversion to a fully open system is somewhat eased, and can be achieved in a period of time to spread the cost of changeover.

Alternatively, a controller developer may internally develop a POSIX API middleware layer on top of a COTS operating system product or on top of a proprietary kernel for reasons of product discrimination such as improved controller performance, cost competitiveness, etc. The choice of the specific POSIX implementation is up to the developer.

Table 3.3.2.1.1 lists many of the POSIX related documents that are either current standards or at some level of draft release. This table provides a context for subsequent discussions of API services, profiles, and conformant implementations. Only the POSIX documents most applicable to this specification are identified.

The versions of designated draft standards are the latest available as of March 1993. It is recognized that this specification must be updated periodically if it is to accommodate the most recent standards developments. This is actually less of a problem than it may sound since many operating system vendors have already announced their intent to support the evolving POSIX

SOSAS Rev. 3.0

draft standards, especially in the area of RT POSIX extensions (POSIX.4). In that sense, modifications to the SOSAS in the area of POSIX will merely track known direction for commercial operating system products.

Designation	Title	Relevant NGC API
IEEE 1003.0 D15	Guide to POSIX Based Open System Architecture	All
none - guide	IEEE Standard Interpretations of IEEE Standard Portable Operating System Interface for Computer Environments (IEEE Std 1003.1-1988) Prod. No. SH15313-PVB ISBN: 1-55937-216-8	All
ANSI/IEEE 1003.1-1990	Portable Operating System for Computer Environments (same as ISO/ IEC 9945-1:1990)	All
ISO/IEC 9945-1:1990	Information Technology–Portable Operating System Interface (POSIX)- Part 1: System Application Program Interface (API) [C language] Ed. 1 356p. JTC 1	All
IEEE 1003.1a D7	Draft Revision to POSIX.1	All
IEEE 1003.1 LIS D3	System Application Program Interface	All
IEEE 1003.2a D8	POSIXPart II: Shell Utilities, User Portability Extensions	Pres.
IEEE 1003.2b D4	POSIXPart II: Shell and Utilities. Amendment 2	Pres.
ANSI/IEEE 1003.3-1991	Information Technology-Test Methods for Measuring Conformance to POSIX	All
IEEE 1003.3-1991	IEEE Standard for Information Technology- Test Methods for Measuring Conformance to POSIX Prod. No. SH14068-PVB ISBN: 1-55937-104-8 (same as ANSI/IEEE 1003.3-1991)	All
IEEE 1003.3.1 D14	Test Methods for Measuring Conformance to POSIXSystem Interfaces	All
IEEE 1003.3.2 D8	Test Methods for Measuring Conformance to POSIXShell & Utilities Interface	Pres.
IEEE 1003.4 D13	POSIXPart 1: Real Time & Related System API	All except Pres.
IEEE 1003.4a D6	Standards for Threads Interface to POSIX	Platform, Pres. Data
IEEE 1003.4b D4	Feb 92 POSIXPart 1: Real Time System API Extensions	Platform, Comm., Device I/O
IEEE 1003.8 D6	POSIX-Part 1: Network-Transparent File Access	Data
IEEE 1003.12 D1.2	POSIXProtocol Independent Interfaces	Comm
IEEE 1003.13 D5	POSIX Standardized Profile	Profiles-All
IEEE 1003.16 D3	POSIX C Language Bindings-Part 1	All
IEEE 1003.17 D4	POSIX Directory Service API	Comm., Data
IEEE 1003.21 D?	POSIX Real-Time Communications	Comm.
IEEE 2003.1	IEEE Standard for Information Technology-Test Methods for Measuring Conformance to POSIX SH15826	All
FIPSPUB151-2	Portable Operating System Interface (POSIX)-System Application Program Interface [C Language] 93 May 12	All

Table 3.3.2.1.1 POSIX Documents

The Federal Information Processing Standards Publications document, FIPSPUB-151, is a procurement profile specified by the government based on the POSIX.1 (ANSI/IEEE 1003.1) standard. This document is proof of government support of the POSIX standard, and is included here as a basis for defining future NGC government/military profiles.

Not included in the table are POSIX documents in the areas of system administration, language bindings for FORTRAN and ADA, system security, supercomputing, profiles for transaction

processing and multi-process platforms, and batch processing. Although these subjects may have some NGC relevance, they are considered secondary to the fundamental requirements for open system controllers. As open controller philosophies materialize in open controller prototypes and NGC products, these additional documents need to be revisited for SOSAS applicability.

POSIX.1 covers the basic operating system services. It had originated as an Institute of Electrical and Electronics Engineers, Inc., (IEEE) effort, became an American National Standards Institute (ANSI) standard (ANSI/IEEE 1003.1) in 1990, and is also an international standard (ISO/IEC 9945-1:1990). It is language-dependent, meaning the APIs are specified for C bindings (a calllevel interface description for the C language) and is perhaps the most essential document within the standard set. Current standards efforts include the re-write of this document as a language independent specification (LIS). The C language bindings are to be specified as a separate document, as for bindings to other languages such as FORTRAN or ADA.

The NGC POSIX profiles are based on POSIX.1 and are strongly dependent on ANSI C, officially designated ANSI/ISO 9899. This 1990 version of the C language standard is an international standard (ISO/IEC 9899:1990). It is a revision and redesignation of ANSI X3.159-1989. The corresponding government procurement profile to this standard is FIPSPUB-160.

Many of the NGC API services are supported through the adoption of the ANSI C standard libraries. POSIX.1 explicitly deals with C language-dependent services and defines two types of C language conformance: (1) C standard, and (2) common usage C. This emphasizes the strong connection between POSIX and ANSI C.

The adoption of ANSI C for use in non-POSIX operating system environments also satisfies a level of POSIX compliance. In such implementations, it is only when services are requested outside of the support ordinarily provided through standard ANSI C libraries that the applications are no longer conformant with POSIX. The designer's choices for NGC applications are then to use either the API extras and RT extensions defined within the POSIX standard and implement a layer of corresponding middleware, or to rely on the support of vendor-specific API libraries and de facto standards.

For the case where the developer chooses not to use POSIX, a NGC profile set is provided for using other system implementations and corresponding support libraries. This approach satisfies a need to support legacy systems and eases the migration toward future POSIX implementations. The developer may always choose, instead, to either select a commercial POSIX operating system platform or develop a POSIX middleware layer on top of a proprietary operating system.

3.3.2.1.2 Platform Services API



Platform services support many of the features commonly provided by commercial operating systems today. These services include, for example, process/task control, synchronization and scheduling, event and interrupt handling, time services, memory, and other resource management.

Table 3.3.2.1.2 identifies the relevant de jure standards in the area of Platform services. De facto standards in this area and their relevance to the NGC are discussed in Section 3.3.3, Profiles.

ISO/IEC 9945-1:1990	Portable Operating System for Computer Environments (same as ISO/ IEC 9945-1:1990)
IEEE 1003.4 D13	POSIXPart 1: Real Time & Related System API
IEEE 1003.4a D6	Standards for Threads Interface to POSIX
IEEE 1003.4b D4	Feb 92 POSIXPart 1: Real Time System API Extensions
ISO/IEC 9899:1990	Programming LanguageC

ANSI C, now an international standard, is identified in the table as a standard separate from POSIX but it should be recognized that it is also an integral part of the POSIX.1 specification. Most of the standard ANSI C library services fall under the umbrella of NGC platform services. This includes C library functions and corresponding API in the areas of character and string handling, localization, mathematics, non-local jumps, input/output, date and time, diagnostics, and general utilities. It is only in the I/O area that some of the C library functions relate to other service categories such as presentation management, data management, and device I/O.

POSIX.1 Platform services API include functionality for single and multi-processing, signals, and user groups. RT Platform services API, such as RT signals, semaphores, memory management, priority scheduling, and timers, are defined in POSIX.4. Thread services and reentrant functions API are covered in POSIX.4a. Spawning, central processing unit (CPU) time management, and sporadic server services API are specified in POSIX.4b. Note that the services mentioned are not the full breadth of the POSIX services, but only those services that relate

specifically to the platform category. POSIX also spans the areas of communications, device I/O, data management (file management), etc.

The draft nature of the RT API definition is not viewed as a major problem due to the current support of draft levels of POSIX.4 in a variety of RTOS products. For many NGC controllers, - these COTS operating system products may form an integral part of their total controller implementation solution.

The operating system industry acceptance and momentum towards open systems is evident in Microsoft's latest operating system product, Windows NT, which supports POSIX. The list of RTOS products with varying levels of in-progress and available POSIX.1 and RT support includes:

- Encore Computer (UMAX V R/T)
- Eyring (PDOS)
- Harris Computer Systems (CX/UX)
- Industrial Programming (MTOS)
- Integrated Systems (pSOS)
- JMI Software Systems (C Executive)
- Lynx Real-Time Systems (Lynx operating system)
- Microware Systems (OS-9000)
- Modular Computer Systems (8xxx)
- Precise Software Technologies (Precise/MQX, MPX)
- QNX Software Systems (QNX), RTMX-UniFLEX (RTMX/RN/RX)
- Spectron Microsystems (SPOX)
- Wind River Systems (VxWorks, MicroWorks)

Many other operating system vendors have announced their product plans to support POSIX.

3.3.2.1.3 Presentation Management Services (PM API)



Presentation Management services support the man-machine aspects of the NGC typically associated with a standard computer terminal. Supported hardware may include a display, keyboard, and trackball or mouse. Presentation Management services may be used to support control panel functionality, provided such capability is emulated on the NGC terminal and is not a hardware panel implementation. For example, one NGC controller product line may use a graphics windowing environment with specialized NGC menus, dialog boxes, and icons that implement (in software) many of the control panel features commonly supported today using hardware dials and buttons.

To clarify this point, Presentation Management services do not interface physical buttons dedicated to jog, feedhold, pause, and E-stop functions, nor do they manage jog handwheels, mode-select switches, and programmable function keys. However, API services have been identified to support these types of hardware interfaces and are described under Device I/O services API.

Table 3.3.2.1.3 lists the major presentation management standards. Not all the standards listed need to be adopted for NGC-conformant controllers and many of the standards shown overlap in functionality. NGC profiles for presentation management define the standards subsets as well as applicable de facto standards. In designing the controller, satisfying many of the standards listed in the table is easily accomplished through the use of existing COTS products.

In selecting relevant standards for this specification, the following general guideline has been applied: if the selection of a standard for a specific application involves a choice between the national (e.g., ANSI) standard and the corresponding international standard (e.g., ISO/IEC), the NGC selection is usually the international standard. Note the NGC selection for the ANSI C language is ISO/IEC 9899. Correspondingly, ISO/IEC standards are identified for Programmer's

Hierarchical Interactive Graphics System (PHIGS) and Graphical Kernel System (GKS) instead of existing ANSI standards, e.g., ANSI X3.124-1985 (R1991) GKS Functional Description.

ISO/IEC 9945-1:1990	Portable Operating System for Computer Environments (same as ISO/IEC 9945-1:1990)		
IEEE 1003.2a D8	POSIXPart II: Shell Utilities, User Portability Extensions		
IEEE 1003.2b D4	POSIX-Part II: Shell and Utilities, Amendment 2		
IEEE 1003.4a D6	Standards for Threads Interface to POSIX		
ISO/IEC 9899:1990	Programming Language - C		
MIT X Windows (X11R5)	MIT X Window System Version 11, Release 5		
MIT X Windows PEX extension	MIT X Window System PHIGS EXtension		
OSF/Motif 1.2	Open Software Foundation - Motif		
ISO/IEC 8651-4:1991	Graphical Kernel System (GKS) language bindings - Part 4: C		
ISO/IEC 8806-4:1991	Graphical Kernel System for Three Dimensions (GKS-3D) language bindings - Part 4: C		
ISO/IEC 9593-4:1991	Programmer's Hierarchical Interactive Graphics System (PHIGS) language bindings -		
	Part 4: C		
IEEE 1201.1	Uniform Applications Program Interface Graphical User, Rev. 3, DS01719 (Windowing		
	Toolkit API)		

Table 3.3.2.1.3 Presentation Management Services—Relevant Open Standards

Relevant portions of POSIX.1 deal with terminal identification and device/class specific APIs. POSIX.4a specifies reentrant functions for generating a terminal path name and determining terminal device names. Character and string based keyboard and display I/O functions defined in the C language (e.g., getchar(), printf(), etc.) are also applicable.

The POSIX shell utilities (POSIX.2x) are that part of the specification that deals with what is commonly referred to on most systems as the command line interpreter (CLI). It is the CLI that people identify with most when referring to a particular type of computer system and the expected behavior of that system from a user's perspective. The well-known "C, colon, backslash" (C:\) syntax is the CLI of the popular MS-DOS operating system found on many IBM PC compatible, Intel-based platforms. The POSIX shell is modeled after the user interface commonly found on many UNIX-based platforms.

A number of API open standards exist for graphics. The PHIGS and GKS international standards define 2-D and 3-D graphic programming interfaces. PHIGS is also the interface supported by the Massachusetts Institute of Technology (MIT) X Consortium for using the PEX 3-D graphics extension. It is interesting to note that the X Consortium is currently in the process of spinning out of MIT into a nonprofit, more market-driven organization. OpenLook, the chief graphical user interface (GUI) rival of Motif, is absent from the list due to the recent agreement by leading parties in the UNIX market to settle on the OSF/Motif interface for the Common Open Software Environment (COSE).

Other API strategies include X/Open's favorable intention to adopt the Public Windows Interface (PWI) in a future release of the Portability Guide (XPG). The PWI, developed by several COSE sponsors, is a specification of the proprietary Microsoft Windows interface. The plan is to pass the specification off to a vendor-neutral standards body (most likely X/Open) so that Windows-based technologies would no longer have to pay royalties to Microsoft. This is important for SunSelect's new "Wabi" interface product, which allows users to run Windows 3.1 applications on reduced instruction set computer (RISC) workstations running UNIX and X Windows. Should the PWI document survive the potential mitigation and become public domain, future revisions of this specification may include this standard as part of its OSE specification.

Also in this area is a current IEEE effort to standardize on a windowing toolkit API. This standard, IEEE 1201.1, is currently in draft form. It is premature to assess the resulting product support but it appears to carry enough momentum to qualify its candidacy in the NGC SOSAS.

Figure 3.3.2.1.3 illustrates some of the possibilities for implementing Presentation Management services at varying levels of capability and windowing support. Several different combinations of COTS products and/or middleware developed solutions are implied. As described through profiles, not all levels need be supported by any one specific implementation. This subject is expanded in the profile section.

The API standards identified provide for portable operator interface component source code. In addition, they also address issues related with user portability; that is, they provide a foundation for one of the major NGC benefits, common look-and-feel. Although the standards pave a direction, they are not, in themselves, the complete solution.

A display style guide to be followed in the development of a NGC operator interface component, and the user interface capability to customize and tailor the interface, are also essential ingredients. This allows one shop floor, for example, to tailor all their NGC controllers to suit their specific needs, yet also support a common user interface to minimize training and maintenance throughout their shop.



Figure 3.3.2.1.3 Presentation Management Services—API Levels and Implementation

3.3.2.1.4 Data Management Services (DM API)



Data Management services refer to the file management capabilities of the system. The commercial products available for data management today cover a broad spectrum from independent proprietary protocol implementations on a single host environment to open, transparent file access solutions for multi-processor configurations. The latter provides the most portability advantage for NGC applications.

From the perspective of the NGC as a system, Data Management services are concerned with both external as well as internal system file support. External file support may simply involve the use of a utility such as file transfer protocol (FTP) across a local area network (LAN) or Internet to a remote host file system or network. In terms of an ISO/OSI open system alternative, it may involve the use of file, transfer, access, and manipulation (FTAM) for remote file access. Both approaches are closely tied to the NGC communications services as well as Data Management services, and both are supported through NGC profiles. Internal to the NGC, Data Management services may also provide data storage access across loosely and tightly coupled multi-processor controller implementations as well as simple file access within a single processor environment.

The choice of a data management implementation is based on the intended application of the specific controller product line. It is conceivable that some low-cost NGC families may not feature a cell-level interface for external file support. The channel to the outside world for these controllers could be a simple serial interface for RS-274 input. In contrast, high-ended NGC controllers may not only support remote host file access across a shop floor network, but also support "lights-out" operation for a 24-hour period. This, of course, is the vision of the future.

Transparent remote data access (RDA), such as that provided from Sun's Network File System (NFS) product is the current trend and a key focus within open systems communities. In the UNIX environment, NFS makes remote file systems look the same as local ones. OSF's

Distributed Computing Environment (DCE) takes this a step further with the Distributed File Service (DFS) by providing all users a single view of all files (both UNIX and non-UNIX based) within an organization. Transparency deals with making the file resource more visible than the computer host system that supplies the resource.

Table 3.3.2.1.4 lists the standards most relevant to Data Management services. Relevant portions of the POSIX.1 standard include files and directory services, system database access services, and services defined for file descriptor manipulation and control operations on files. POSIX.4 covers file truncation, synchronization, and real-time files. POSIX.4a identifies reentrant functions for group/user database access, file lock/unlock synchronization, and thread-specific data key/data management. Standard C defines file input/output functions, e.g. fread(), fwrite(), fseek(), etc.

ISO/IEC 9945-1:1990	Portable Operating System for Computer Environments (same as ISO/IEC 9945-1:1990)
IEEE 1003.4 D13	POSIXPart 1: Real Time & Related System API
IEEE 1003.4a D6	Standards for Threads Interface to POSIX
IEEE 1003.8 D6	POSIXPart 1: Network-Transparent File Access
ISO/IEC 9899:1990	Programming Language - C
X/Open S203	Data Management: SQL Call Level Interface (CLI)
IEEE 1238.1	File Transfer Access & Management Applications Interface (Rev 2), DS02345
X/Open P206	FTAM High-level API (XFTAM)

Table 3.3.2.1.4 Data Management Services—Relevant Open Standards

The most interesting standard in the list is POSIX.8, Transparent File Access (TFA). TFA refines the file system services specified in POSIX.1. It defines full TFA and core TFA in a manner similar to the general NGC profiling approach. Full TFA provides all the file services defined in POSIX.1. Core TFA defines a minimum set so that TFA may be used with other non-POSIX file systems. Functional extensions to the core services are also defined for the flexibility of adding capability on an individual file basis. POSIX.8, Annex F includes informative profiles for several different file systems (e.g., Sun NFS, FTAM and PC/DOS, etc.).

Note the Structured Query Language (SQL) international standard is not referenced in the table, although it is quite relevant to data management. Many popular database management systems today (e.g., Oracle, Sybase, Ingress, Informix, etc.) support the SQL client/server protocol. It should be clear that the standards listed in the table focus on portability and relevant API. SQL is a protocol standard and, by definition, addresses interoperability concerns. Consequently, this standard falls under the external environment interface (EEI) branch of the Open Systems Environment (OSE) taxonomy and is discussed there.

Of lesser importance, X/Open's SQL call-level interface standard is included for completeness to allow portability of controller applications that have built-in remote file management using SQL. This may be an alternative to using a COTS proprietary data base API but may require a middleware bridge between the NGC applications and the leveraged commercial data base product. The other OSI/FTAM API standards are included for analogous reasons.

3.3.2.1.5 Geometric Modeling Services (GM API)



Geometric Modeling services define a standard programming interface to product modeling systems for extracting and operating on product definition data (PDD). The Geometric Modeling services category is unique in that it tends to be more application specific than the other API categories. However, the existence of open standards work in geometric modeling justifies its place as an API category. Since this specification is an anticipatory standard, it is hoped that other application-specific API categories will emerge in areas such as motion control, programmable logic control, etc.

Table 3.3.2.1.5 shows the only relevant open standard in the area of geometric modeling, the Consortium for Advanced Manufacturing International (CAM-I) Application Interface Specification (AIS). CAM-I is committed to maintaining compatibility with the PDES/STEP standards. The significance of the AIS is that it goes beyond data exchange by supporting operations on product data definition.

Table 3.3.2.1.5 Geometric Modeling Services-Relevant Open Standards

CAM-I AIS 2.0 Draft Standard	CAM-I Application Interface Specification (AIS) Revised 1991 Document Number
Vol II	R-90-PM-03 AIS 2.0 C Language Binding

The PDES/STEP product data exchange standards have an important future relevance to NGC and there will probably be a wealth of new products entering the market based on these standards as they continue to mature. They may be the key to the NGC art-to-part factory of the future by providing an industry standard that completely describes all part characteristics. As data standards, their NGC role is one of ensuring system and applications level interoperability and fall under the EEI branch of the NGC taxonomy.

3.3.2.1.6 Communications Services (Comm API)



Communications services API are perhaps the most important set of services of the NGC OSE. Communications API not only provides for portability of source code that interfaces the NGC to other external systems (e.g., networks), but also defines the set of callable interfaces that support peer-to-peer communications between the internal NGC component agents.

The transparency of the communications system, that is, the ability of the communications system to make resources more visible than the computer host that supplies the resource, is a key focus of the NGC. The Data Management services discuss the notion of transparency in the context of remote data access. The objective for file transparency is to make remote file systems resident on heterogeneous hosts look the same as local ones. Architecturally, the inherent transparency of a distributed file system is often a function of its underlying distributed communications system and directory services.

Table 3.3.2.1.6 lists the open standards most relevant to communications. Inter-process communications is achieved at various levels. One implementation may use a pipe to create an inter-process channel and the general read(), write() functions defined in POSIX.1. Other implementations may rely on POSIX.4 message queue functions and the corresponding blocking timeout services specified in POSIX.4b.

POSIX.12, protocol independent interface (PII), is perhaps the most important open standard of the list. It defines two networking interfaces for protocol-independent, process-to-process communication: the simple networking interface (SNI) and the detailed network interface (DNI). SNI is a simple applications interface that provides for co-operating process intercommunication without requiring details about underlying protocols. DNI offers protocol-independent mechanisms for manipulating protocol specific features of the underlying network. DNI will be specified with C language bindings for Berkeley Software Distributions (BSD) sockets and X/Open XTI. Both

connection-oriented, byte- or record-stream virtual circuits and connectionless channel datagrams are supported. Relevant OSI APIs are also referenced for completeness.

ISO/IEC 9945-1:1990	Portable Operating System for Computer Environments (same as ISO/IEC 9945-1:1990)
IEEE 1003.4 D13	POSIXPart 1: Real Time & Related System API
IEEE 1003.4b D4	Feb 92 POSIXPart 1: Real Time System API Extensions
IEEE 1003.12 D1.2	POSIX - Protocol Independent Interface
IEEE 1003.17 D4	POSIX Directory Service API
IEEE 1003.21 D	POSIX Real-Time Communications
ISO/IEC 9899:1990	Programming Language - C
IEEE 1238.0	Draft Common OSI Connection Management (Rev 4), DS01339 (API to Common OSI
	Connection Management and Support Functions Project)
IEEE 1224	OSI Applications Programmer Interface. Jan 93 (Rev 8) DS 01123
IEEE 1224.1	X.400 Electronic Messaging API (Rev 6) DS01362
IEEE 1224.2	Directory Services Applications Programmer Interface (Rev 5) DS02428
X/Open T003	X.400 (APIsand EDI Messaging) Vols. 1-6
X/Open P203	ASCE/Presentation Services API
X/Open C196	X/Open Transport Interface (XTI)
X/Open P210	The Common Object Request Broker: Architecture and Specification - X/Open in conjunction
	with Object Management Group (OMG)
OSF DCE 1.0.2	Open Software Foundation Distributed Computing Environment Version 1.0.2

Table 3.3.2.1.6 (Communications	Services—Relevant ()pen Stand	ards
-------------------	----------------	---------------------	------------	------

Directory Services (POSIX.17) is the transparency glue for resolving the physical location of objects, given globally defined logical names. This standard was mainly intended as an International Consultative Committee on Telegraphy and Telephony (CCITT) X.500 API but may also be used to access directory systems in existing practice. It offers one open system approach to NGC for network communications and physical node transparency.

A Directory services component is also defined for NGC to provide a level of communications transparency for peer-to-peer communications between agents, the processes and threads, within a NGC. This component hides the communications details (sockets, process ids, message queue descriptors, etc.) for single processor, multiple process environments. It also hides node id details for distributed, multi-processor NGC implementations.

Like a telephone book, the Directory services component is responsible for managing all the address information to globally named system entities. Every NGC component implementation must have at least one agent for sending and receiving messages. Each communications agent globally registers to the system symbolically, by its agent and corresponding component name set. Node residency and all other required information to communicate between agents may be obtained dynamically, during system operation, through the Directory services.

¥

Several existing products attempt to provide a simple and efficient communications interface in a loosely and tightly coupled multi-processor environment. Most of these products fall short in hiding node details or the underlying communications mechanisms.

For example, the Ohio State Trollius is an open architecture operating system for a concurrent - message passing machine and attached UNIX machines. This product, although it is quite powerful by today's standards, defines APIs that are network dependent. In addition, nodes are not transparent at an applications programming level, and node ids as well as event synchronization information must be supplied as arguments to the communications API service calls. Figure 3.3.2.1.6-1 illustrates the Trollius distributed memory multiple instruction, multiple data (MIMD) architecture and the dependency of the Trollius communications API on the four lower layers of the ISO/OSI model: physical, datalink, network, and transport layers.



Figure 3.3.2.1.6-1 Communications Services—Trollius Example APIs Network Dependent

An applications-level API for NGC must shield the programmer from these networking details, yet allow the programmer to selectively configure the implementation of the lower layers. This can be achieved with programming options during the link process of the application, so the desired communications implementation levels are built-in with the selection of the required software component libraries.

As Figure 3.3.2.1.6-1 illustrates, higher level send() and recv() API services would make network details transparent to the applications programmer, yet he may still have the option during the applications build process to select the level of protocol required to meet his specific performance needs. Link options to a variety of object libraries during applications build could facilitate such a flexible mechanism.

Two competing paradigms for communications are message passing and remote procedure calls (RPCs). Two consortiums, Open Software Foundation (OSF) and Object Management Group (OMG) are developing "middleware" in support of these different communications strategies. OSF's Distributed Computing Environment (DCE) implements RPCs. OMG's Common Object Request Broker Architecture (CORBA) standardizes distributed object-oriented computing using message passing, which may also use RPC mechanisms as its underlying implementation.

From an architectural perspective, the interrelationship between POSIX, DCE, and CORBA is somewhat hierarchical. This hierarchy is illustrated in Figure 3.3.2.1.6-2. This view is quite a simplified view of the relationships of the standards. The whole story is, in fact, quite complicated but the hierarchical view helps to understand some of the NGC communications options and the levels of transparency that are available or missing when a particular option is selected. Although CORBA and DCE both provide mechanisms for remote procedure calls (RPCs), CORBA is shown at the highest level of the hierarchy because it is also an object-oriented standard. DCE is function-oriented but there is nothing to preclude "wrapping" DCE functionality into an object-oriented implementation.

DCE has evolved through the integration and standardization of a number of vendor products, many of which were once separate stand-alone or bundled products. Although it is incorrect to make the claim that all of DCE is built around POSIX, there are portions of DCE that are truly based on POSIX. The DCE "threads" and the concept of "lightweight processes" to improve application performance through parallelism are based on POSIX threads. However, some DCE services are not layered on POSIX and potentially need not be supported by commercially available OS products, as the architectural hierarchy diagram also graphically depicts.

Within the set of POSIX standards there is also an implicit hierarchy of interrelated standards. The POSIX hierarchy shown in Figure 3.3.2.1.6-2 is somewhat arbitrary in that it is not the only hierarchy possible. Some of these standards are still in draft form. The resulting products will establish the true interrelationships when these standards mature.



Figure 3.3.2.1.6-2 Communications Standards Architectural Hierarchy

POSIX.8 is the POSIX standard for transparent file access (TFA). This standard, referenced in the DM services section and earlier in this section by inference in the discussion of remote file access, is shown at the top of the POSIX hierarchy. The rationale for its placement at the top is due to its inherent dependence on the underlying network and communications mechanisms for its implementation.

Potential communications mechanisms appear as the POSIX.12 SNI (Simple Network Interface) and POSIX.12 DNI (Detailed Network Interface). Both of these standards are specified in the POSIX.12 PII (Protocol Independent Interface) document and were briefly described earlier in this section. POSIX.12 SNI appears at a higher level than POSIX.12 DNI to illustrate that an "implementation" of a simple network interface may involve access to detailed network services.

The distributed services, illustrated as POSIX.17 DS in the hierarchy diagram, are accessed by both POSIX.12 SNI and DNI. In addition, IEEE 1238.0, the Open Network Interface (ONI) makes use of these same distributed services. The POSIX standards include not only those standards designated by IEEE 1003.x, but also a number of other standards such as the 1238 ONI standard. The IEEE 1238 standard defines API at an applications level to the ISO/OSI protocol stacks. As illustrated in the hierarchy, the 1238 services may be used to support network or file access functions required of the higher level POSIX standards.

To complete the POSIX hierarchy, POSIX.1 and POSIX.4 may provide all the service support required of the other POSIX standards at a local host level. Local communications services may range from pipes and signals to real-time semaphores, shared memory, mapped files, or message queues.

- The POSIX.12 SNI standard is highlighted (shadowed) in the hierarchy diagram because of its importance in specifying two communications API: sni_send() and sni_recv(). These two APIs approach a level of simplicity and communications transparency previously described as a necessary NGC feature yet is not available in many real-time communications products today. At an applications program level, these simple API may be used. The decision to use POSIX.4 shared memory or message queues for communications need not be made during applications development, but instead could be made during the applications build by selection from specific object libraries.

Another major point of distinction between the POSIX communications standards and the CORBA and DCE standards is that the POSIX standards deal with communications at the level of interprocess communications (IPCs). Both CORBA and DCE provide mechanisms for remote procedure calls (RPCs) which may be viewed to be at a higher level of abstraction than IPCs.

Figure 3.3.2.1.6-3 illustrates the development mechanisms for implementing CORBA or DCE RPCs. Within both the CORBA and DCE standards, an interface definition language (IDL) is specified. The applications developer defines his interfaces using IDL and then "translates" these interface definitions using an IDL compiler. The compiler generates header files to be included within the applications source code as well as client stubs and server skeletons which are "linked" into the executable client and server applications programs during the build process.

Both standards, CORBA and DCE, are currently viewed as most applicable at the external interface or workstation level of NGC communications, although there is nothing to prevent their use for peer-to-peer communications internal to NGC. The only restriction for their use in real-time applications is the current lack of real-time CORBA and DCE COTS products. Today's CORBA and DCE products are transaction-based and are oriented toward workstation level business applications. OMG's CORBA may provide the foundation for one implementation of an NGC external communications profile as well as perhaps OSF's DCE RPCs or even the TCP/IP capability of the underlying operating system. A real-time CORBA product, if one existed today, could have a tremendous impact on the approach to inter-process peer-to-peer communications within an NGC.



Figure 3.3.2.1.6-3 CORBA-DCE IDL & RPCs

The mechanism for message passing in NGC is Communications services. As mentioned in Section 3.3.2.1 (general API section) it is left up to the discretion of the NGC application designer as to whether or not he may wish to implement object-oriented wrappers around function-oriented calls to access API services using a message passing paradigm. Through the use of Communications services, NGC components may be implemented that embody the container object concepts defined for OMG's CORBA.

In summary, a set of communications standards and profiles are defined to enable the selection of a communications strategy most suitable for the specific application and real-time requirements of the NGC. As with all other API services, the Communications API focuses on NGC applications source code portability and does not necessarily define how the underlying services are to be implemented.

3.3.2.1.7 Device Input/Output Services (Device I/O API)



The Device I/O services are the peripheral management functions of the system. These services perform the logical-to-physical and physical-to-logical mappings between the device requests and the corresponding electrical signals to the device. The devices supported cover a broad spectrum from dedicated operator control panel buttons, handwheels and switches, to specialized sensor and actuator drive signals for motion control and tool change. Other devices may include communications devices, auxiliary storage devices, analog-to-digital (A/D) and digital-to-analog (D/A) converters, Centronics parallel interfaces, etc.

Table 3.3.2.1.7 identifies relevant device I/O open API standards. POSIX.1 covers general I/O primitives. These general open(), read(), write(), and close() primitives apply not only to files but also to devices. POSIX treats all I/O devices in the same general manner. File descriptors are general-purpose descriptors used for all device management. POSIX.4 defines asynchronous I/O primitives, e.g., aio_read(), aio_write(). Memory mapping functions may provide the required interface to memory mapped I/O devices. POSIX.4b specifies how to associate a user-written interrupt service routine (ISR) with an interrupt and also defines a control device function, devctl(), for a more direct path to the device driver.

ISO/IEC 9945-1:1990	Portable Operating System for Computer Environments (same as ISO/ IEC 9945- 1:1990)		
IEEE 1003.4 D13	POSIX-Part 1: Real Time & Related System API		
IEEE 1003.4b D4	Feb 92 POSIXPart 1: Real Time System API Extensions		
ISO/IEC 9899:1990	Programming Language - C		

Table 3.3.2.1.7	. Device	I/O Ser	vices—Rel	levant O	pen Standards
-----------------	----------	---------	-----------	----------	---------------

In 1991, The Real Time Consortium had been working on the Open Basic Input Output System (OBIOS) standard. The standardization effort, representing a collaborative effort of several key hardware and system software vendors in the real-time and embedded systems marketplace, was
focused on developing a call interface and associated client/device interaction model that provides a common abstract interface for a wide range of I/O devices.

The OBIOS is an ideal candidate for inclusion in the NGC SOSAS. Unfortunately, the consortium dissolved while the standard was still in the draft stage. This was the only standard - of its kind and identifies a standard gap in the NGC OSE. It may be of little value to include the contents from the OBIOS draft specification in the SOSAS since it appears at present to have little market interest. It is listed as a placeholder in the EEI section.

3.3.2.2 External Environment Interface (EEI)

The EEI includes API categories where interoperability issues are concerned and includes the following additional categories: *language, data exchange, device interface, and backplane bus.* In each category a set of relevant open standards are identified to facilitate both system-level interoperability and the interoperability of component-level implementations (applications software) as well.

Interoperability considerations include networking protocols, data interchange formats, distributed file systems, legacy support, and device interface. Each EEI category is concerned with addressing one or more of these considerations. The following summarizes the NGC approach to interoperability.

Networking protocols are defined in open standards such as IEEE 802.3 CSMA/CD, 802.4 token-passing bus, 802.5 token-passing ring, etc. These standards are supported by NGC profile implementations using de facto standards and products such as Ethernet and ARCNET, as well as the commonly used TCP/IP internet protocol found on many UNIX-based operating systems.

Standard data interchange formats tackle the problem of handling heterogeneous data, such as differences in byte ordering, data formats, or the padding of data items in heterogeneous host environments. The IEEE floating-point standard and OSF's DCE marshaling capability are examples in this area. Standard product data formats such as PDES/STEP are also key.

Distributed file systems support file exchange or shared file access in distributed environments requiring such access. Applications range from transferring files across a cell-level interface to the transfer of legacy IGES or RS274 files across a CAD/CAM network or RS-232 interface.

In the area of device interface, there is a wealth of de facto and de jure standards that deal with device interoperability.

Tables 3.3.2.2-1 through 3.3.2.2-5 list the EEI standards by category. The list of standards for EEI can be quite large, especially in the area of ISO/OSI standards which are not enumerated here for that reason. However, it must be remembered that subsets of these standards are selected through NGC profiling and that most of these standards are already accommodated with the use of commercial products.

There are some obvious overlaps of certain API standards within the EEI domain. For example, X Windows standards not only specify the X library and Xt intrinsics APIs, but also define an X protocol for client-server interoperability between the client application and an X terminal. Therefore, the X Windows standard also appears under the EEI branch of the taxonomy. In the area of data management, SQL is another protocol standard that has a corresponding API standard.

Even within the EEI categories there are functional overlaps. No matter what view is taken, there will probably be the potential for overlap. CAN bus, for example, may be classified as having fieldbus characteristics, yet may also be used as a remote keyboard interface. The primary focus for the NGC is on relevant standards for profiling. The NGC taxonomy organizes standards to facilitate the profiling process. A statement need only be supplied for use of a standard outside of its originally intended scope.

Figures 3.3.2.2-1 and 3.3.2.2-3 correlate some of the EEI standards into a more physical view of the NGC system context. At the highest level, the NGC has only three major physical interface types: a human-user interface, a cell or CAD/CAM interface, and a plethora of device interfaces. Relative to these physical interfaces, a number of standards are mapped that define protocols and/or data formats. Data format standards are grouped under the process heading in the diagram. These data standards could apply to all major physical interfaces.



Figure 3.3.2.2-1. NGC EEI System Context—Physical View



Figure 3.3.2.2-2. NGC EEI Interoperability Standards—Physical Mapping



Table 3.3.2.2-1. EEI Relevant Open Standards—Overlapping API Categories

Descentation Mamt FFI				
Presentation Night EE1	MET V Window System Version 11 Release 5			
MIT X Windows (XIIKS)	MIT X Window System Version 11, Reliance 5			
MIT X Windows PEX extension	MIT X WINDOW System PHIOS Extension			
OSF/Motif 1.2	Open Software Foundation - Moul			
IEEE 1201.1	Uniform Applications Program Interface Graphical User, Rev. 5, DS01715			
	(Windowing Toolkit API)			
ANSI X3.124-1985 (R1991)	Graphical Kernel System (GKS) Functional Description (includes Alver Abstraction)			
	1985)			
FIPSPUB120-1	Graphical Kernel System (GKS) 91 Jan 08			
ISO 7942:1985	Information Processing Systems - Computer Graphics - Graphical Keller System			
	(GKS) functional description			
ISO 8805:1988	Information Processing Systems - Computer Graphics - Graphical Kerner System for			
	Three Dimensions (GKS-3D) functional description			
ANSI/ISO 9592.1-1989	Information Processing Systems - Computer Graphics - Programmer's Hierarchical			
	Interactive Graphics System (PHIGS) - Part 1: Functional Description			
FIPSPUB153	Programmer's Hierarchical Interactive Graphics System 88 Oct 14			
ISO/IEC 9592-1:1989	Information Processing Systems - Computer Graphics - Programmer's Hierarchical			
	Interactive Graphics System (PHIGS) - Part 1: Punctional Description			
Data Management EEI				
IEEE 1003.8 D6	POSIX-Part 1: Network-Transparent File Access			
ISO 8907:1987	Information Processing Systems - Database Languages - NDL			
ANSI X3.133-1986	Database Language - NDL			
FIPSPUB126	Database Language NDL 87 Mar 10			
ISO/IEC 9075:1992	Information Technology - Database Languages - SQL			
ISO/IEC 9579-1	Information Technology - Database Languages - Remote Data Access - Part 1:			
	Generic Model, Service and Protocol			
ANSI X3.135-1989	Information Systems - Database Language - SQL with Integrity Enhancement			
ANSI X3.168-1989	Information Systems - Database Language - Embedded SQL			
FIPSPUB127-1	Database Language SQL			
X/Open C201	Structured Query Language (SQL)			
X/Open P205	SQL Remote Database Access			
X/Open 1301	RDA Mapping for TCP/IP			
Geometric Modeling FFI				
CAM-LAIS 20 Draft Standard	CAM-I Application Interface Specification (AIS) Revised 1991 Document Number			
Vol 1	R-90-PM-03 AIS 2.0 Functional Specification			

Table 3.3.2.2-1 (Continued)

Þ

27

Communications EEI			
IEEE 1003.12 D1.2	POSIX - Protocol Independent Interface		
IEEE 1003.17 D4	POSIX Directory Service API		
ISO 7498:1984	Information Processing Systems - Open Systems Interconnect - Basic Ref. Model		
X/Open T904	X/Open Portability Guide 4 (C202, C203, C204, G204)		
X/Open G212	X/Open Distributed Computing Services (XDCS) Framework		
X/Open P210	The Common Object Request Broker: Architecture and Specification - X/Open in		
•	conjunction with Object Management Group (OMG)		
OSF DCE 1.0.2	Open Software Foundation Distributed Computing Environment Version 1.0.2		
X/Open C210	Protocols for X/Open Interworking: XNFS, Issue 4		
ANSI/EIA 511-1989	Manufacturing Message Specification - Service Definition and Protocol		
ISO/IEC 9506-1:1990	Industrial automation systems - Manufacturing Message Specification - Part 1:		
	Service Definition		
ISO/IEC 9506-2:1990	Industrial automation systems - Manufacturing Message Specification - Part 2:		
	Protocol Specification		
ISO/IEC 9506-4:1992	Industrial automation systems - Manufacturing Message Specification - Part 4:		
	Companion Standard for Numerical Control		
ISO/IEC 9506-5:1990 (IEC 65A/	Industrial automation systems - Manufacturing Message Specification - Part 5:		
65B(Secretariat)111/138 Sept. '90)	Companion Standard for Programmable Controllers		
IEC DIS 1131-5 (IEC 1131-5 PC	Programmable Controllers - Part 5: PC Communications		
Comm Draft 1/15/93)			



Table 3.3.2.2-2. Language EEI—Relevant Open Standards

ISO/IEC 9899:1990	Programming Languages C		
FIPSPUB160	C 91 Mar 13		
ANSI X3.37-1987	Programming Language APT		
ANSI/EIA 494-B-1992	32-Bit Binary (BCL) and 7-Bit ASCII (ACL) Input Format for NCM		
ANSI/EIA 274-D-1980 (1988)	Interchangeable Variable Block Data Format for Positioning, Contouring, and		
	Contouring/Positioning Numerically Controlled Machines		
ANSI/EIA 267-C-1990	Axis and Motion Nomenclature for Numerically Controlled Machines		
ISO 841:1974	Numerical control of machines - Axis and motion nomenclature		
IEC 1131-3 (1993)	Programmable controllers - Part 3: Programming languages		
ISO 3592:1978	Numerical control of machines - NC processor output - Logical structure (and major words)		
ISO 4342:1985	Numerical control of machines - NC processor input - Basic part program reference language		
ISO 4343:1978	Numerical control of machines - NC processor output - Minor elements of 2000-type records (post-processor commands)		
ISO 6983-1:1982	Numerical control of machines - Program format and definition of address words - Part 1: Data format for positioning, line motion and contouring control systems		



Table 3.3.2.2-3. Data Exchange EEI—Relevant Open Standards

IGES 5.1	Initial Graphics Exchange Specification 5.1, October 1991 (US PRO P15.1)		
ANSI/ASME Y14.26M-1989	Digital Representation for Communication of Product Definition Data (IGES 4.0)		
FIPSPUB177	Initial Graphics Exchange Specification (IGES) 92 November 30		
ISO 10303 Initial Release	Standard for the Exchange of Product Model Data (STEP) - Product Data		
	Representation and Exchange (US PRO P10303.IR - incl. parts 1, 11, 21, 31, 41 - 44,		
	46, 101, 201, and 203)		
ANSI X3.4-1986 (R1992)	Coded Character Set - 7-Bit American National Standard Code for Information		
	Interchange (ASCII)		
ANSI X3.41-1990	Code Extension Techniques for Use with the 7-byte Coded Character Set of ASCII		
ANSI X3.42-1990	Representation of Numeric Values in Character Strings for Information Interchange		
ANSI X3.64-1979 (R1990)	Additional Controls for Use with the ASCII		
FIPSPUB1-2	Code for Information Interchange, its Reps., Subsets, and Extensions 84 Nov 14		
FIPSPUB86	Additional Controls for Use with the ASCII 81 Jan 29		
ISO/IEC 646:1991	ISO 7-bit coded character set for information interchange		
ISO 2022:1986	Information processing - ISO 7-bit and 8-bit coded character sets - Coded extension		
	techniques		
ISO 6093:1985	Information processing - Representation of numerical values in character strings for		
	information interchange		
ISO/IEC 6429:1992	Information technology - Control functions for coded character sets		
ANSI/EIA 227-A-1978 (1988)	One-inch Perforated Tape		
ANSI/EIA 358-B-1980 (R1990)	Subset of ASCII for Numerical Machine Control Perforated Tape		
FIPSPUB2-1	Perforated Tape Code for Information Interchange 84 Nov 14		
FIPSPUB26	One-inch Perforated Paper Tape for Information Interchange 73 June 30		
ANSI/ISO 8632-1990	Information Processing Systems - Computer Graphics - Metafile for the Storage and		
	Transfer of Picture Description Information (rev. & redesig. of X3.122-1986)		
FIPSPUB128-1	Computer Graphic Metafile, 93 May 11		
ISO/IEC 8632-1/2/3/4	Information Technology - Computer Graphics - Metafile for the storage and transfer		
	of picture description information - Part 1: Functional Specification, Part 2:		
	Character encoding, Part 3: Binary encoding, Part 4: Clear text encoding		
ANSI/ELA-548-1988	Electronic Design Interchange Format (EDIF)		
FIPSPUB161	Electronic Design Interchange (EDI) 91 Mar 29		

ľ



Table 3.3.2.2-4. Device Interface EEI-Relevant Open Standards

ANSI/EIA/TIA 232-E-1991	Interface between Data Terminal Equipment and Data Circuit-Terminating		
	Equipment Employing Serial Binary Data Interchange		
EIA 485	Standard for Electrical Characteristics of Generators and Receivers for Use in		
	Balanced Digital Multipoint Systems		
ANSI/IEEE 488.1-1987	Digital Interface for Programmable Instrumentation		
ANSI/IEEE 488.2-1987	ICodes, Formats, Protocols, and Common Commands		
ANSI X3.183-1991	Information Systems - High Performance Parallel Interface		
ANSI X3.131-1986	Small Computer Systems Interface (SCSI)		
ISO 9316:1989	Information processing systems - Small Computer System Interface (SCSI)		
SAE J1583	Controller Area Network (CAN) An In-vehicle Serial Communications Protocol		
IEC 44(secretariat)148	Serial Data Link for Real Time Communication Between Controls and Drives (SERCOS)		
ISA-S50.02-1992	Fieldbus Standard for Use In Industrial Controls Systems - Part 2: Physical Layer Specification and Service Definition		
ANSI/ELA 431-1992	Electrical Interface Between a Numerical Control and Machine Tools		
IEC 1131-1/2 (1992)	Programmable Controllers. Part 1: General information, Part 2: Equipment requirements and tests		
EIA 441-1979 (R1992)	Operator Interface Functions of Numerical Controls		
ISO 4336:1981	Numerical control of machines - Specification of interface signals		
OBIOS 1.15	OBIOS: Open Basic Input Output System Draft Specification - Real-Time Consortium		

,

ł



Table 3.3.2.2-5. Backplane Bus EEI—Relevant Open Standards

ANSI/IEEE 896.1-1992	Backplane Bus Specification for Multiprocessor Architectures: Futurebus+, Logical Layer Specification		
ANSI/IEEE 1296-1987	High performance Synchronous 32-Bit Bus: Multibus II		
IEC 796-1/2/3	Microprocessor System Bus - 8-bit and 16-bit data (MULTIBUS I) Parts 1-3:		
ANSI/IEEE 1196-1987	NuBus - A simple 32-Bit Backplane Bus		
ANSI/IEEE 959-1988	Specification for an I/O Expansion Bus: SBX Bus		
ANSI/IEEE 961-1987	8-bit Microcomputer Bus System (STD Bus)		
ANSI/IEEE 1000-1987	Specification for a Standard 8-Bit Backplane Interface (STE Bus)		
IEC 821 (1991)	VMEBus - Microprocessor system bus for 1 byte to 4 byte data		
IEC 822 (1988)	VSB - Parallel Subsystem Bus of the IEC 821 VME bus		
IEC 823 (1990)	Microprocessor System Bus (VMSbus) - Serial Subsystem bus of the IEC 821 bus		
ANSI/IEEE 1014-1987	Versatile Backplane Bus: VMEbus		
ANSIЛЕЕЕ 1096-1988	Multiplexed High-Performance Bus Structure (VSB)		
IEEE P1496	SBus		
IEEE P1754	SPARC		

¥

3.3.3 Profiles

The NGC standard supports varying levels of vendor-neutral open system conformance through the concept of profiles. Simply stated, a profile specifies a set of de jure and de facto standards that are adhered in the implementation of a specific controller. Thus, a profile characterizes a controller by defining its unique level of openness and functional capability.

Profiles may include all the open standards specified in the OSE taxonomy. In addition, profiles tie into accepted industry practice with provisions for including "de facto" standards. De facto standards, especially those that are commonly used in controller products today, can not be ignored. These standards are not always considered "open" in the purest sense. Many de facto standards are proprietary and they all lack formal specification by a recognized standards body. However, some de facto standards have had a tremendous impact on industry. One example of this is the PC ISA backplane bus which not only has gained wide popularity in the personal computer industry but is also growing in use in industrial controls applications.

Profiles provide a mechanism for "apples-to-apples" comparison of NGC applications software and platform environments to assess their relative compatibility. Through the use of profiles, the selection of controller software for execution on a specific platform implementation is a simple matter of comparing profiled capabilities. Compatible implementations may be interchanged since they support equivalent levels of system interoperability, component interchangeability, and portability. These "-abilities", facilitated through the use of profiles, enable the realization of the major benefits of NGC.

It is therefore the application of profiling that facilitates a new generation of controllers. These NGC standards-based controllers are distinguished by their relative ease in accommodating:

- (1) new features,
- (2) scaleable and upgradeable features,
- (3) interchangeable controller components,
- (4) portability of components to different platform implementations,
- (5) adaptable interfaces to a variety of hardware devices,
- (6) flexible peer and factory-level communications and networks,

- (7) enhancements to performance, and
- (8) consistent and tailorable user interfaces.

NGC profiles support both existing and emerging open standards. Many POSIX draft standards are referenced where mature standards are unavailable. As described earlier, de facto public and proprietary standards are also adopted where current market trends support such inclusion. Some NGC implementations will inevitably evolve based on profiles that specify only vendor-neutral open standards. It is hoped that completely vendor-neutral profiles will ultimately dominate the marketplace. NGC conformance implies the adoption of one or more NGC profiles in a controller implementation (see Section 3.5, Conformance).

NGC relevant standards are grouped into dimensions of similar purpose or functionality, as defined by the Open Systems Environment (OSE) taxonomy. For example, the Backplane Bus hardware dimension includes Multibus I/II, VME, Futurebus+, STD Bus, and NuBus. Backplane Bus profiles also include proprietary standards that are widely used, e.g., Industry Standard Architecture (ISA) and Extended Industry Standard Architecture (EISA) bus, when popularity of these standards merit their inclusion as de facto standards. Selection of one or more standards from each of the profile dimensions constitutes a specific profile. For example, the Backplane Bus dimension of an implementation profile can include the selection of both the VMEBus and the EISA bus standards.

Figure 3.3.3-1 illustrates the full set of NGC profile templates and their hierarchical relationship. The template organizational structure is strongly influenced by the OSE taxonomy with some subtle differences. One important difference is the introduction of de facto standards that are not identified in the OSE standards taxonomy.



Figure 3.3.3-1 NGC Profiles—Template Structure

Standards "levels" in the profile template structure is another departure from the OSE taxonomy organization. Template levels provide the obvious benefit of helping to organize and represent the tremendous amount of profile information on limited template space. However, this is only part of the rationale for specifying the POSIX profile templates at two levels of detail. Level 1 supports profiling of POSIX API options at a feature or API group level. This higher level satisfies the operating system developer's need to represent the compiler options used for a specific operating system product. Compiler options are essentially packaging/bundling options for tailored products that meet specific applications requirements and targeted hardware environments. Level 2 supports profiling of POSIX at an individual API level so that an applications developer may accurately reflect the service requirements of his controller component by reference to specific API calls.

A complete specification of all the profile templates represented by Figure 3.3.3-1 is beyond the scope of this specification. Figures 3.3.3-2 through 3.3.3-9, located at the end of this section, are sample profile templates used in defining implementation profiles. As the template headings suggest, the templates are dual-use and may be used to profile either a specific NGC platform or a specific NGC software application component or aggregate. To define a specific profile, one needs only to check mark the boxes corresponding to the desired template items.

Selection of some items may either preclude or mandate the inclusion of other items in a profile. The real-time (RT) or non-RT nature of the specific platform or application profile may also have a potential impact on the applicability of certain template items. (Note that hard-RT applications are not profiled since such applications are typically embedded or built-in hardware and fall outside the scope of general platform and applications portability issues.)

The profiling process described, and the impact of one profile selection on the use of another template, is best illustrated by example. The POSIX.13/D5 draft is a "Standardized Application Environment Profile--POSIX Realtime Application Support (AEP)". This draft specifies four POSIX RT profiles: MINimal embedded, CONtroller, DEDicated, and MULtipurpose. These profiles appear as selection items on the API Profile Suite template (Figure 3.3.3-2, upper right-hand corner). In specifying a platform profile, selection of one of these items negates the need to fill out the POSIX Profile Level 1 template (Figure 3.3.3-3) and the relevant POSIX Level 2 templates (Figures 3.3.3-4 thru 3.3.3-7) since these items are already predefined by the RT AEP selection. However, RT POSIX custom platforms, or extensions to one of the POSIX RT AEPs, will require the use of all the relevant POSIX profile templates.

The POSIX profiling example above illustrates the benefits to be gained from a user-friendly, interactive set of tools for profiling. Intelligent profiling tools can simplify the profiling task so that an NGC developer is presented with only the applicable options based on his previous profiling selections. These tools do not exist today. The availability of automated tools for NGC development and integration is considered a key enabling technology for bringing the NGC concept to a practical reality.

It is interesting to note that POSIX.13/D5 is based directly on existing small and/or RT (typically non-UNIX) kernel practice. Table 1-1 of the POSIX.13 draft specification offers a comparison of five products (VRTX32, pSOS, C Exec, MTOS, and VxWorks) based on small RT kernel features. The general approach of the standard specifies POSIX interfaces sufficient to deliver the functionality typical of current RT systems.

 Ongoing and future SOSAS work involves the tailoring and refinement of the profile suites. Standard AEPs that are more encompassing than the POSIX RT AEPs, that span all OSE dimensions, should evolve based on market-driven factors. It is premature (and unnecessary) to anticipate or guess at what these standard AEPs should be. Documented profiles and the benefit of time will illuminate the demand for certain profiles. Additional work is also required to identify component profiles or NGC-relevant subsets of the base standards that are selectable items in the profile suites. As with AEP definition, component profiling must also be sensitive to legacy and the projected product market.

Also related, but of a lower priority, is the feasibility of specifying a government/military AEP based on the Federal Information Processing Standards (FIPS) procurement standards (made available through the National Technical Information Service). Many of the PUBS documents refer to open standards that are identified in the NGC OSE. Each of these documents contain a "Specifications" section that is essentially a component profile of the referenced base standard. All additions to, modifications of, and deletions to the referenced standard are specified.

It is obvious that the POSIX standard has been leveraged to a large degree for its value-added standards in addition to its contribution to the fundamental philosophies of NGC and the OSE reference model. The adoption of POSIX.1 implies the adoption of the C language. Less obvious, perhaps, is that the adoption of ANSI C implies the adoption of POSIX.1. This connection to POSIX via the standard ANSI C library is often overlooked. It implies that much of the legacy software developed using the C language may be leveraged with little or no modification for POSIX compliance. POSIX-conformance is violated only if source-level API system calls exist within the C source code that fall outside of the POSIX specification. Therefore, when defining profiles for legacy C code, one should not be so quick to rule out its compliance to POSIX.

The IEEE/TCOS committee is also currently developing language-independent specification (LIS) versions of the POSIX standards. The LIS version of POSIX.1 is intended to replace the current C language-dependent version of the specification. FORTRAN and Ada language bindings to the POSIX.1 standard are also at various stages of draft release. As these standards mature and there is evidence of supporting products, they need to be integrated into the NGC OSE profile strategy to fill some the present language gaps that exist with the current NGC platform and POSIX approach.

Profiling offers complete flexibility in design to the developer. Many platform profiles will evolve by leveraging existing COTS products. It is possible to implement each profile dimension with a different product. Even the adherence to POSIX is optionally selectable by dimension. Little profiling constraints are placed on the developer other than to specify his unique profile. End user requirements and market demand will establish de facto standard profiles that will have the most impact on the next generation of controllers.

7

Target Processor (Platform Only)	API Profile Suite		POSIX RT Profile (RT Platform Only)	
CISC RISC DSP Other Supplier/Bd/Chip: Application: (Component/Ageregate Only)	Processing Requiremen	ls	CONtroller DEDicated MULtipurpose Custom Num DOSTV	
Development Language	Pr Command VF: Complian	esentation Mgmt.:		
□ Stand. C (opt. C++): □ Diagnostics □ FORTRAN □ Character Handling □ ADA □ Localization □ Non-POSIX binding □ Mathematics □ Pascal □ Non-Local Jumps □ Smalltalk □ Input/Output □ Lisp □ General Utilities □ PL/1 □ String Handling □ COBOL □ Date and Time □ Other:	□ POSIX □ Graphics: □ POSIX.2a □ 2D □ POSIX.2b □ 3D □ Other: □ PHIGS Prog. UF: □ Proprietary: □ POSIX □ HOOPS □ POSIX.1 □ OpenGL □ POSIX.4a □ GDI □ ANSI C □ Quickdraw (Applic. Only) □ Other:	Windows GUIs X Windows Xitb-library Xt-toolkit Non-Xt toolkit Andrew InterViews Other. PEX-PHIGS ex Motif Openlook IEEE 1201	 Non-X Win/GUIs: DOS Text/Graphics Windows OS/2 Windows NT (Win32) Mac Windows Other: 	
Platform:	Data Mgmt:	Comm.: C	Geom. Model: 1 CAM-I AIS	
□ POSIX □ Non-POSIX □ POSIX □ POSIX.1 □ UNIX □ POS □ POSIX.4 type: □ POS □ POSIX.4a □ DOS/Windows □ POS □ POSIX.4b □ OS/2 □ ANSI □ ANSI C □ NT (Ap (Applic. Only) □ MacOS □ SQL C □ Other □ FTAM □ IEE	POSLX 8 (IFA) IX.1 Core TFA IX.4 Full TFA IX.4 Profiled: C (RFS,NFS,FTAM, plic. Only) PC/DOS, other) I Other: E 1238.1 pen	D POSEX.21 D ANSI C I OSI APIs D - IEEE 1224 D -IEEE 1238.0 D -X/Open D CORBA D DCE D POSEX.12 D Other	Device I/O: 1 POSIX D POSIX.1 D POSIX.4 D POSIX.4b 1 ANSI C 1 Other:	

Figure 3.3.3-2 API Profile Suite

¥

Targ (Pla □ CISC □ R	et Processor fform Only) ISC 🗖 DSP 🗖	P(P 10ther	OSIX Prof (Shee Platform/A)	ile Leve et 1) pplication	el 1	Application:	ate Only)
Supplier/Bd./Chip	: <u> </u>		D Non-RT	DRT			
	POSE	X.1				POSIX.4a	
Platform:	Pres. Mgmt:	Data Mgmt:	Device I/O:	Platform:	1	Pres. Mgmt:	Data Mgmt:
 multi-process signals user_groups single_process 	□ terminals □ device_specific □ job_control	☐ file_system ☐ device_IO ☐ file_attr ☐ fifo ☐ system_database ☐ fd_mgmt	□ device_IO □ pipe	 threads process_s thread_p thread_p reentrant 	cheduling rio_inherit rio_protect _funcs	□ reentrant_funcs	□ reentrant_funcs
	PC	OSIX.4				POSIX.4b	
Platform:	Data Mgmt:	Comm.:	Device I/C).	Platform:	Comm.: .	Device I/O:
 realtime_signal semaphores memlock memlock_range mapped_files memory_protect shared_memory priority_schedut timers realtime_files 	s in mapped_files shared_memory synchronized_id a synchronous_i tion y_objects aling	D memory_ /_objects 0	passing □ synch □ async	nronized_io hronous_io	□ spawn □ cputime □ thread_ □ threads_	timeouts cputime sporadic_server	☐ device_control ☐ interrupt_control

Figure 3.3.3-3 POSIX Profile Level 1

September 23, 1994

-

Ĭ

Target Processor (Platform Only)	POSIX.1 Profile Level 2 Platform/Application	Application:
CISC RISC DSP Supplier/Bd./Chip:	□Other Processing Requirements □ Non-RT □ RT	(Component/Aggregate Only)
□ single_process	□ multi_process □ job_cont	rol 🛛 signals
□ uname() □ execl() □ sysconf() □ execv() □ main() □ execve() □ execve() □ execlp()	□ execvp() □ wait() □ sleep() □ setpgid() □ getpid() □ waitpid() □ alarm() □ tcgetpgr □ getppid() □ _exit() □ pause() □ tcgetpgr □ times() □ time() □ time() □ fork() □ getenv()) □ sigemptyset() □ kill() p() □ sigfillset() □ sigaction() p() □ sigaddset() □ sigprocmask() □ sigdelset() □ sigpending() □ sigismember() □ sigsuspend()
□ user_groups	terminals	file_system
getuid() setsid() geteuid() setgid() getgid() getgroups() getegid() getlogin() setuid() cuserid()	□ getpgrp() □ ctermid() □ opendir() □ geta □ ttyname() □ readdir() □ mk □ isatty() □ rewinddir() □ crea □ closedir() □ unl □ chdir() □ rma	cwd() □ rename() □ pathconf() □ umask() dir() □ stat() □ fpathconf() □ chmod() at() □ fstat() □ link() □ chown() ink() □ access() □ utime() □
□ fd_mgmt □ device_	IO 🛛 device_specific 🗖 system	m_database 🛛 pipe 🗂 FIFO
□ dup() □ open() □ dup2() □ close() □ lseek() □ read() □ fcntl() □ write()	□ cfgetispeed() □ tcsetattr() □ get, □ cfgetospeed() □ tcsendbreak() □ get, □ cfsetispeed() □ tcdrain() □ get, □ cfsetospeed() □ tcflush() □ get, □ tcgetattr() □ tcflow()	grgid() □ pipe() □ mkfifo() gmam() pwuid() pwnam()

Figure 3.3.3-4 POSIX.1 Profile Level 2

I Target Processor (Platform Only) CISC D RISC D DSP D Other Supplier:B4/Chip:	POSIX.4 Profile Level 2 Platform/Application Processing Requirements	Application: (Component/Aggregate (Daly)
□ realtime_signals □ sem □ sigwaint() □ sem_init() □ sigunedwait() □ sem_destroy() □ sigqueue() □ sem_open() □ sem_close() □ sem_unlink()	aphores I memlock I sem_wait() I mlockall() sem_trywait() I munlockall() sem_post() sem_getvalue()	memiock_range 🖸 mapped D mlock() D firunca D manlock() D mmap(D manmap(_files htc() 0 #P()
C fsync C shared_memory_objects C fsync() C shara_open() C C shara_open() C C shara_open() C C memory_protection C C mprotect() C	priority_scheduling sched_setparam() sched_get_priorit sched_getparam() sched_get_priorit sched_setscheduler() sched_rr_get_inte sched_getscheduler() sched_yield()	j time y_max() □ clock_settime() □ y_min() □ clock_gettime() □ crval() □ clock_getra() □ □ timer_create() □ □ timer_delete()	ers timer_settime() timer_gettime() timer_getoverum() nanosleep()
c realtime_files f_create() f_getbiocap() f_getbuo f_getatin() f_getaiocap() f_feebu f_getatiocap() f_getailocap() f_getailoccap() f_getailocancr() f_getcachecap() f_getiner()	□ synchronized_io □ asynchr f() □ fdatasync() □ aio_read() t uf() □ msync() □ aio_write t □ aio_iistio() t □ aio_error() □ aio_return()	anous_io □ me aio_cancel() □ mq_ope aio_suspend() □ mq_clo aio_fsync() □ mq_unl □ mq_sen □ mq_rec	mage_passing m() □ mq_notify() ec() □ mq_setatin() ink() □ mq_getatin() d() eive()

Figure 3.3.3-5 POSIX.4 Profile Level 2

-

Target Pro (Platform (CISC PRISC Supplier/Bd/Chip:	PC cessor] Dnly) DSP Other	SIX.4a Profile Leve Platform/Applicatio Processing Requirements Non-RT RT	el 2 n Application:
		threads	
 pthread_attr_init() pthread_attr_destroy() pthread_attr_setstacksize() pthread_attr_setstacksize() pthread_attr_setdetachstate() 	 pthread_self() pthread_equal() pthread_noce() pthread_matexattr_init() pthread_matexattr_destroy() 	pthread_mutex_trylock() pthread_mutex_unlock() pthread_condatir_init() pthread_condatir_destroy() pthread_condatir_getpshared()	Image: pitread_cond_wait() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_key_create() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_setspecific() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait() Image: pitread_setspecific() Image: pitread_cond_timedwait() Image: pitread_cond_timedwait()
 pthread_attr_getdetachstate() pthread_create() pthread_join() pthread_detach() pthread_detach() pthread_exit() 	 pthread_mutexatir_getpshared() pthread_mutexatir_setpshared() pthread_mutex_init() pthread_mutex_destroy() pthread_mutex_lock() 	 □ pthread_condattr_setpshared() □ pthread_cond_init() □ pthread_cond_destroy() □ pthread_cond_signal() □ pthread_cond_broadcast() 	□ ptinead_attr_setscope() □ ptinead_yield() □ ptinead_attr_getscope() □ sigwait() □ ptinead_attr_setisched()□ ptinead_kill() □ ptinead_attr_getisched()□ ptinead_cancel() □ ptinead_attr_setsched()□ □ □ ptinead_attr_setsched()□ □ □ ptinead_attr_setsched()□ □
 pthread_setintrtype() pthread_cestintr() pthread_clearnp_pusb() pthread_clearnp_pop() 	process_scheduling setpric() getpric() setscheduler() getscheduler() setscheduler() setschedule	 thread_prio_inherit ptiread_mutexattr_setprotoc ptiread_mutexattr_setprotoc ptiread_mutexattr_setprio_c ptiread_mutexattr_setprio_c 	□ thread_prio_protect blo □ pthread_mutex_getprio_ceiling() blo □ pthread_mutex_setprio_ceiling() eiling() seiling()
	Li yieldi)	C reentrant_funcs	
 readdir_r(): file_system localtime_r(): C date and time gratime_r(): C date and time asctime_r(): C date and time asctime_r(): C date and time ctermid_r(): terminals 	ctime_r(): C date and time getlogm_r(): user_groups tyname_r(): terminals getgrgid_r(): system_database getgrmam_r(): system_database	getpwnam_r(): system_database getpwnid_r(): system_database strok_r(): C string handling flockfile(): fmlockfile()	□ getc_unlocked(): C I/O - PM □ rand_r(): C general utilities □ getchar_unlocked(): C I/O - PM □ putc_unlocked(): C I/O - PM □ putchar_unlocked(): C I/O - PM □ tmpnam(): C I/O - DM

1. Figure 3.3.3-6 POSIX.4a Profile Level 2



Figure 3.3.3-7 POSIX.4b Profile Level 2

Target Processor (Platform Only) CISC CRISC DSP Other Supplier/Bd/Chip:	C Language Profile Platform/Application Processing Requirements	Key: "+" Indicates POSIX. 1 revised "*" Required by Standard C - not part of POSIX Application: (Component/Aggregate Only)
Standard C Compliance Stand. C C++ Option Common C C++ Option	□ Character Handling □ isalnum() □ islower() □ isxdigit() □ isalpha() □ isprint() □ tolower() □ iscntrl() □ ispunct() □ toupper() □ isdigit() □ isspace() □ isgraph() □ isupper()	 Localization setlocale()+ localeconv()* Diagnostics assert()
□ Mathematics □ acos() □ sin() □ exp() □ modf() □ asin() □ tan() □ freep() □ pow() □ atan() □ cosh() □ kdeep() □ sqrt() □ atan2() □ sinh() □ kog() □ cell() □ cos() □ tanh() □ kog(0) □ fabs()	□ Non-Local Jumps □ floor() □ setjmp() □ Platform 1 □ fmod() □ longimp() □ sprint □ sigsetjmp() □ sscant □ siglongimp() □ vsprint	□ Input/Output //O: PM I/O: f() □ getc() □ perror() □ puts() f() □ getchar() □ printf() □ scanf() f()* □ gets() □ putc() □ ungetc() □ putchar() □ vprintf()*
□ String Handling □ strcpy() □ strncmp() □ strspn() □ memcm □ strncpy() □ strch() □ strstn() □ memcm □ strncat() □ strcspn() □ strtok() □ memcm □ strncat() □ strphk() □ strlen() □ memset(□ strcmp() □ strch() □ memch()* □ strch()	00* □ stremor()* □ clearen() □ fgetc() 0* □ stremor()* □ fclose() □ fgets() 0* □ stremor()* □ fclose() □ fgets() 0* □ feof() □ fopen() □ feor() □ fopen() 0* □ feof() □ fopen() □ feor() □ fputc() 0* □ fflush() □ fputs() □ fflush() □ fputs()	DM I/O: fread() frintf() setbuf() fgetpos()* freopen() remove() tmpfile() fsetpos()* fseek() rename()+ tmpnam() setvbuf()* ftel() rewind() fileno() vfprintf()* fwrite() fseanf() fdopen()
□ General Utilities □ abs() □ srand() □ abort()+ □ atcoin()* □ a □ atof() □ calloc() □ coin() □ div()* □ a □ atoi() □ free() □ getenv()+ □ labs()* □ a □ atoi() □ malloc() □ becarch() □ kdiv()* □ a □ rand() □ realloc() □ quort() □ mblen()* □ a	Date a mbstowcs()* system()* mbtowc()* wcstombs()* actime() actime()* striod()* britoul()* britou	nd Time Gignals Inktime()+ Traise()* strftime()+ Signal()* tzset() clock()* difflime()*

Figure 3.3.3-8 C Language Profile

EEI Profile Suite Platform/Application							
Processing Requirements Target Processor Application: I Non-RT I RT Hard RT I CISC I RISC DSP Other Supplier/Bd/Chip							
Pres Manut D	hata Monte	Geom Model: (Comm.:	Lang.:	Data Exchange:	Device I/F:	Backplane Bus:
X Windows	NDL	CAM-I AIS	ISO OSI 7498.	ANSI C	IGES	RS 232, 485	VME
Protocol	SOLRDA	Func. Spec.	ASN.1 + others	APT	PDES'STEP,	ANSI X3.183	-VSB,VMS
Func Specs	Func Speca	I water opposit	(CCITT, IEEE)	ACL/BCL	EXPRESS	GPIB 488	SBX
Ture option	·		XPG4	RS 274	ASCII	SCSI	FutureBus+
			XDCS	IEC 1131-3	CGM	CAN	Multibus LTI
			XNFS		EDIF	SERCOS	STD
			MMS (NC, PLC)			ISA Fieldbus	STE
			IEC 1131-5			IEC 1131-1/-:	2 NuBus
			Func. Specs.			RS 431, 441	S Bus
						ISO 4336	SPARC
						OBIOS	

Figure 3.3.3-9 EEI Profile Suite—Platform/Application

Ì

4

ŧ

3.4 Development Process and Implementation

This section describes the NGC development process and illustrates the path from an architecture specification to an implementation solution. It is perhaps one of the most important sections for guidance to the NGC controller builder. Emphasis is placed on implementation, where the "rubber meets the road" in terms of NGC product development.

Up to this point, little guidance on "how to implement an NGC" is provided. The preceding sections describe a context, architectural framework, and supporting environment for a NGC. In some respect, they introduce the necessary background and terminology to better understand the NGC specification in preparation for development.

The remainder of this section focuses on those implementation aspects that will evolve a next generation of flexible controller products, bringing specification to reality. After all, the NGC vision will only be realized if there is a large base of existing NGC products available.

Open controller characteristics such as component interchangeability and system flexibility have fundamental significance at an implementation level. These characteristics bring the ultimate benefits to the end user. The ability to add new features, scale performance, and interconnect physical hardware and controller components are natural by-products of open controllers. It is the implementation approach to component development and the resulting interoperability of hardware and software products that enables the desired system flexibility. The NGC implementation approach is therefore key to bringing the controller flexibility desired by end users today into the capabilities of controllers tomorrow.

3.4.1 Integration Process

Figure 3.4.1 illustrates the NGC integration process with focus and greater detail on the design and implementation phase. The term "integration" is used in the title to highlight the notion that if controller development begins with a mature and robust implementation component archive, it is possible to construct a working controller system from existing COTS products without having to create any new components. Early NGC development work must naturally begin by populating this component repository with newly developed components and interactive tools for its access. A plan is currently underway to establish the controlling organization for this repository and develop a strategy for its required maintenance.



Figure 3.4.1 NGC Integration Process

The integration process diagram captures the general flow and process dependencies and is not necessarily a prescription for any one single approach to development. The developer may begin development starting from any stage in the process. For example, one organization may decide to follow the NGC integration process using concurrent engineering practices, assigning

engineering work groups to various stages of the process simultaneously. All feedback paths are implicit and do not appear on the diagram to avoid unnecessary complexity in illustration.

The integration process flow consists of two main parallel paths: applications development and platform development. The applications development path is illustrated as the upper path in the process diagram. The lower path addresses platform development. Both paths compliment and support each other and are interrelated in ways not easily captured in the process flow diagram.

For example, the selection of applications software is driven by the compatibility of the available software components and the platform selected for a specific controller implementation. Profiles provide the mechanism for contrasting software component implementations against selected platform implementations so their compatibility may be ascertained. The availability or lack of existing applications components for a specific platform implementation may drive the need to select an alternate platform implementation. Trades such as these, the "make-or-buy" choices, are implicit in the process flow.

The platform implementation tends to solve more of the general-purpose computation and communications requirements. These requirements are referred to as "domain-independent" requirements. Conversely, application component implementations address more of the application-specific or "domain-dependent" requirements to control and manufacturing. The dependence of the platform on existing open standards makes available to the developer a variety of implementation solutions and supporting COTS products. Every NGC platform implementation must be profiled. With every new NGC applications software product developed, a corresponding profile and interface description must be supplied.

To complete the integration story, selected and/or developed source-level software components are compiled and linked together with object-level component libraries to generate the executable controller applications for the target controller platform. Platform profiles and profiles supplied with applications software are used in the selection process so that only compatible components are selected. All hardware and executable software is then integrated into a fully functional NGC application system, or simply the controller.

3.4.2 Application Framework, Architecture, and Implementation

The architectural concept of a "framework" forms the underlying infrastructure and foundation of NGC. At the platform level, the NGC Open System Environment (OSE) framework embodies 3 concepts: (1) a "reference model", which is based on and extends the POSIX model; (2) a

"taxonomy" of de jure standards; and (3) "profiles" of de facto and de jure standards. It defines a set of language-dependent, function-oriented API callable services and facilitates the development of object-oriented wrappers.

In addition, at the application level, the NGC specification facilitates the generation of a variety of frameworks for specific application architectures, e.g., 5-axis machining center. Much of the information contained in this specification may be used to compose a "framework" for the development of an application architecture. An application framework is characterized by a selected set of primitive components and an implicit topology (resource-product associations). A framework is "composed" by selecting and specializing reference requirements with a specific application in mind, and by applying architecture description rules (ADRs) for primitive components, and architecture description rules (ADRs) used for composing an application framework is contained in the appendices of this document.

With this background, the process of creating an "application architecture" is better understood. This process is lightly touched upon in the upper left hand portion of the integration process diagram (Figure 3.4.1) and is further expanded in Figure 3.4.2. An application architecture is distinct from its corresponding framework in that it specifies the first major level of design detail.

Starting from an application framework, architectural components are defined consisting of primitive components and aggregates of primitive components that compose the framework. Language-independent message interfaces are specified between architectural components and between components and the external environment. The architecture specification must also take into consideration design requirements for system flexibility. By design, the architecture must accommodate the system's ability to integrate legacy motion controller boards, add new sensors, or increase the number of axis of control for a specific application requirement. The architecture may also be the first level for considering timing requirements. The "aggregation" or grouping of primitive components into larger architectural components for product packaging may be driven in-part by commonly shared timing requirements. The application architecture is therefore a top-level design manifestation of an application framework that supports all the - desired requirements for system flexibility.

SOSAS Rev. 3.0



Figure 3.4.2 From Reference Requirements to Implementation

Figure 3.4.2 illustrates the process of developing an implementation from reference requirements, showing the characteristic differences between an application framework, architecture, and the final system. Note that profiles and language issues are implementation concepts and are not architectural - concepts. This reinforces the previous discussion that component interchangeability, facilitated through the use of profiles, is fundamental at the implementation level.

Different architectures may potentially be spawned from one framework. Consistent with the profiling philosophy of this specification, the number of different architectures that evolve from

one framework and the number of specific frameworks that require standardization must be market-driven. The subsequent discussion on standardization and the document vision provides additional supportive rationale for separating the application framework from its corresponding architecture.

3.4.3 Standardization and Document Vision

Figure 3.4.3-1 depicts the NGC document vision for open controller standardization. This SOSAS specification, including its latest modifications, with the exception of its many appendices, may be viewed essentially as the "Guide to Open Controller Standards (OCS.0)". It introduces the standards and philosophies and provides a context and terminology for understanding by example the other standards documents. The adopted convention for designating the documents is similar to that used for the POSIX standards. For example, POSIX.0 is the guide to the POSIX standards.



Figure 3.4.3-1 NGC Document Vision: Open Controller Standards

The "Applications Framework Open Controller Standard (OSC.1)", is envisioned to contain most of the appendices from this SOSAS specification. In addition, it will contain an expanded set of reference requirements, primitive components, architecture description rules, and domain models to compose frameworks for application domains outside of its present focus on CNCs. Other application domains include robotics, programmable logic controllers (PLCs), process control, semiconductor manufacturing, etc.

Architecture specifications (OCS.3, OCS.4, etc.) are envisioned to evolve for specific applications based on market-driven factors. For traceability, these documents must specify the application framework upon which they are based. Each framework must be individually derivable from the OCS.1 document which will be expanded to accommodate additional application domains as market forces dictate.

Different documents satisfy different needs of the community. For example, from a CNC developer's perspective, the "CNC Application Architecture (OCS.3)" document and the "Implementation Guideline (OCS.2)" may be the only two specifications of interest. OSC.2 is envisioned to be an expansion of this SOSAS implementation section with detailed information on profiling, language binding, the component library, interactive development tools, and other system development and integration information. Standards bodies, on the otherhand, may have much more of an interest in the framework document, OCS.1, and its role in governing the evolving set of application architecture specifications. The developer of an NGC product may or may not care about the OCS.1 document. However, if a controller developer has a vested interest in standardizing his application architecture, he will not only have an interest in OCS.1 specification.

As a final point in understanding the NGC document vision and the document interrelationships, consider the ISO/OSI 7 layer reference model specification (ISO 7498) and its relationship to the Ethernet level 2 specification (IEEE 802.3). Full interoperability is only guaranteed when a specific physical implementation is determined (e.g. twisted pair, coax, etc.). Even the Ethernet specification is incomplete in assuring interoperability of physical components, but it certainly specifies a detailed level of protocols that is not specified by its parent ISO 7498 specification. Token ring and token bus are also defined level 2 protocols under the ISO/OSI reference model - yet neither will interoperate without converting bridges or gateways.

By analogy, the NGC applications framework is to a specific architecture as the ISO/OSI reference model is to Ethernet. Ethernet is to token ring as one NGC application architecture is

to another. Neither will interoperate. Just as physical characteristics are important to ensure interoperable Ethernet implementations, NGC component profiles and language binding considerations are of equal importance to ensure the interoperability and interchangeability of application architecture component implementations. These implementation issues must be considered to ensure interoperability yet, like the Ethernet specification, are not specified within the NGC architecture documents. The criteria for ensuring NGC implementation interoperability is discussed in the OCS.2 Implementation Guideline.

A formal standardization approach is required to bring the NGC documentation vision to reality. The following 3 methods are recognized by ANSI as establishing evidence of consensus: (1) Accredited sponsor using the canvas method, (2) Accredited standards committee method (ASC), and (3) Accredited standards developing organization method (ASDO). A full explanation of these methods is beyond the scope of this specification. Figure 3.4.3-2 is a graphic depiction of the process model for the second method, the ASC method. This process model provides some insight into the standardization process. All methods focus on capturing consensus and adequate representation (due process) and are equivalent in their final results. Therefore, any of these methods are acceptable for NGC document standardization.



Figure 3.4.3-2 ANSI Standardization Process Model—ACS Method

3.4.4 Implementation Component Library & Tools

Section 3.4.1, Implementation Process, introduced the notion of an implementation component archive that may be used to create a working controller system from an existing repository of controller parts. This, of course, is currently a vision and early NGC development work must naturally begin by populating this component repository with newly developed components and interactive tools for its access.

The NGC implementation component library is envisioned to be a shopping list of COTS software and hardware packages as well as a central archive of reuseable, publicly available (perhaps at a cost) aggregate software products. It is the repository for storage and retrieval of applications/platform product information, detailed interface descriptions, and profiles. To support this vision, interactive tools for creating, maintaining and browsing the library are needed that do not currently exist. Figure 3.4.4 presents a conceptual view of the component library structure and contents.



Figure 3.4.4 Implementation Component Library & Tools-Structure & Contents

Having user interactive tools to peruse available profile information on specific products facilitates "apples-to-apples" comparison of products. Product profiles may be used to assess if a specific

software component will run on a selected platform implementation. Profiles may also determine the relative compatibility of two different vendor applications for replacing one with the other to upgrade an existing NGC system. Creating a profile for a platform that was conceived with a specific set of vendor products in mind has the added potential benefit of exposing alternative implementation solutions that may not have been considered during the initial trade studies.

The unavailability of interactive library tools today implies that the first NGC controllers must break new ground. Early implementations will define platform and component profiles from clean templates. Much of the controller-specific applications software will have to be developed either from scratch or through the reuse and modification of legacy software. With every new NGC software product developed, a corresponding profile and interface description must be supplied.

The mature existence of an implementation library and interactive development tools for its use is believed to be the single, most critical enabling technology for NGC.

3.4.5 Implementation

Previous sections describe the process flow leading to NGC implementations. Sections 3.4.1 and 3.4.2 illustrate the integration process and provide a context for the implementation process. The NGC concepts of application framework and application architecture are reinforced and related to the implementation process. Section 3.4.3 ties specification to implementation and section 3.4.4 provides a deeper understanding of the structure and contents of the implementation component library.

Section 3.4.3, Standardization and Document Vision, discusses the need for an "Implementation Guideline" specification (OCS.2). Its contents must include all of the information described thus far in section 3.4 with more detail. It must address language issues such as language-dependent message interface specification. In addition, it must cover issues related with profiling, hardware-software partitioning, functional vs. object-oriented implementations, and the development and integration of various "types" of component packages from COTS to middleware to board support packages to specialized API service development. In depth coverage of these subjects are beyond the scope of this specification but is a requirement for interoperable, interchangeable component implementations.

It should now be apparent that an applications architecture specification forms the basis for an implementation but falls short of specifying the requirements for interoperable implementation components. Figure 3.4.5-1 illustrates some of the new design considerations in transitioning

ŧ

SOSAS Rev. 3.0

from an application architecture to an implementation. An application architecture is a high-level design that takes form in an implementation in terms of hardware and software components. Implementation software components support applications or platform functions, all of which execute on the hardware platform. Similarly, the platform hardware is composed of a number of hardware implementation components.



Figure 3.4.5-1 From Application Architecture to Implementation

Platform software may simply involve the installation of standards-based COTS packages. Where COTS packages fall short of the selected platform profile standards, middleware components may be developed or reused (if available) to bridge the API gap. Middleware may also provide the required access layer to board-level and embedded controller products.

At an applications software level, client/server stubs provide the required mechanisms for transparent communications between local and/or remote implementation components. These components are natural by-products of CORBA and DCE based implementations. They are generated by the interface definition language (IDL) compiler and implement remote procedure calls (RPCs).

Figures 3.4.5-2 and 3.4.5-3 expand the simplified architecture and generic implementation example of Figure 3.4.5-1 to introduce timing and multi-platform considerations. Component aggregation is illustrated in the expanded application architecture example. The primitive components of a specific application component aggregate are numerically identified. The intercomponent message sets are also numerically designated and directly correlate between the application architecture and the resulting generic implementation.



Figure 3.4.5-2 Application Architecture





3.5 NGC Conformance

NGC conformance is defined at several levels. At the system level, conformance is determined by following at least one of several NGC-specified profiles of EEI and API standards sets. An application is NGC-conformant with respect to a specific profile if that application always uses the standard API defined for that profile whenever it accesses functionality that is supported by that API.

The OSE reference model described in Section 3.3.1 explains the OSE framework and its impact on applications portability and systems-level interoperability. It is important to note that the OSE framework does not specifically deal with component-level interoperability issues other than to specify several applicable data standards for message interface. As described in Section 3.1, component messages and behaviors are specified to facilitate software interoperability, but component interfaces are not specified at a level that guarantees fully interoperable, fully interchangeable component implementations. However, specification of components, a reference topology, and definition of high-level message classes puts a natural limitation on the total design space of controller applications.

7

To ensure that independent component developers may develop complimentary products that will interoperate with other controller components, the following SOSAS requirement is added: a detailed description of all component interfaces shall be supplied with any controller or component that claims component-level interoperability.

This specification is a guide for an anticipatory standard. The fact that it is not a specification that simply documents existing practice presents an inherent obstacle to its acceptance by the controller industry. This partially explains the adopted position on component interoperability, that is, to keep the component interface descriptions at an abstract, implementation-independent level. In an ideal world, the SOSAS could specify component interfaces at a sufficient level of detail that would ensure full software interoperability. If such a detailed specification is to be adopted by the controller builder community, it would have to be sensitive to market drivers and de facto standards for component interfaces that may potentially evolve. To ensure the acceptance and future applicability of the SOSAS, these interfaces have been deliberately specified at a high level.

This NGC standards approach is analogous to the approach used for the ISO/OSI communications standard. The ISO/OSI reference model standard (ISO 7498) specifies a multi-level framework for communications at a sufficient enough level of abstraction to facilitate the evolution of several different network implementations (e.g., ethernet, token ring, token bus, etc.), yet it constrains the implementations to a specific hierarchy and specific sets of services and protocols. A possible scenario for the NGC is that supporting SOSAS documents may evolve that define the component interfaces at a level of detail that will facilitate fully interoperable implementations.

For a full NGC controller product, there are two basic levels of conformance: system level and component level. System-level conformance implies the disclosure of all profiling information specific to the platform (for all API/EEI categories). This same profiling information must also be disclosed for the applications software as a whole. Component-level conformance subsumes system-level conformance requirements and additionally requires the disclosure of detailed component interface specifications.

The SOSAS extends POSIX philosophies with regard to conformance. All conformance requirements within each individual POSIX.n specification shall apply if that specification is _ selected for inclusion in the developer's profile. Implementation conformance refers to the NGC system platform and associated services, or the execution environment for applications-level NGC software components. An NGC platform is implementation conformant if it conforms to a specific and complete profile in all dimensions of API and EEI standards, from communications

protocols and services to device interfaces and human-user interface (HUI) API. The NGC platform may include nonstandard extensions, provided such extensions are identified in the system documentation. For such platforms, an environment must be defined and documented in which applications may run with the expected behaviors specified by the corresponding standards of the profile suite.

Application conformance applies to an application component of the NGC claiming conformance to a specific platform implementation. A "strictly conforming" NGC application is one that requires only the facilities defined by the specific platform implementation for which it conforms. Additional documentation shall be supplied for "conforming" NGC applications that use non-standard extensions of a NGC platform implementation.

Software applications claiming conformance with the C language shall claim conformance with either "C Standard Language-Dependent System Support" or "Common Usage C Language-Dependent System Support". The former refers to conformance with the C programming language international standard, ISO/IEC 9899:1990. Applications conforming instead with the latter, common usage C, shall clearly document all variations from the international C standard. It is expected that much of this information can be acquired from the vendor of the specific C compiler product used for development.

POSIX.3, formally IEEE Standard 1003.3-1991: Test Methods for Measuring Conformance to POSIX, will be the NGC adopted standard for test suite validation of POSIX-compliant products. Since COTS products claiming POSIX compliance must follow this standard, the adoption of POSIX.3 for the NGC is simply a statement of current and expected future practice. Note the term "compliance" is used instead of the term "conformance". Compliance has stronger meaning and implies the existence of a test suite for measuring the conformance of an implementation. The application and extension of POSIX.3 test methods (beyond its intended POSIX scope) into other NGC standards domains is not a requirement. Other supporting test standards will apply as the market dictates.

Many POSIX draft standards are identified in the NGC OSE. Since there is an obvious lag of commercially available products for such standards, the NGC developer need only disclose a "statement of intent" for inclusion of such standards in future revisions of his product. This statement of intent applies especially to those implementations that may accommodate some level of the draft specification, but not necessarily the SOSAS referenced version, due to the development lag of supporting products.

×

7

For example, Wind River Systems' VxWorks 5.1 product literature advertises full ANSI C compliance. Several features from the real-time extensions proposed in the POSIX.4 draft were added to the existing POSIX.1 file and input/output system. Wind River's product literature further documents their commitment to full compliance with POSIX.1 and their offering of incremental compliance with POSIX.4 and .4a drafts, with full support once these standards are approved.

The specification of a profile suite, component interface descriptions, language documentation, statements of intent, and associated conformance claimers shall be supplied as applicable with each NGC-conformant product in the form of a "disclosure statement". An NGC product may include:

- Fully implemented platform;
- Platform product;
- Software applications (primitive component, component aggregates, module, and library implementations);
- NGC system.
4.0 NGC DEVELOPMENT EXAMPLE

The 5-axis horizontal machining center shown in Figure 4.0 requires a controller. This section addresses how to provide an open controller for this application, and it serves as an example of the open controller design process. We begin with a brief overview of the design and implementation process for this example. Salient parts of this process are elaborated in subsections.



Figure 4.0. Horizontal Machining Center

The first major step in the process is to create the application requirements for 5-axis machining by selecting all relevant Reference Requirements (see Appendix A) and making them specific to the application. Application requirements help to establish an application framework for 5-axis machining. These requirements drive the selection of the Reference Architecture primitive components (see Appendix B) that constitute the application framework. Collectively, the primitive component responsibilities must cover the application requirements. If no set of primitive components from the Reference Architecture can cover the requirements, then existing primitive components are modified and/or new ones are added until the revised set of primitive components entirely matches the requirements. If any of the new or modified primitive components prove to be generally useful and if they have unique responsibilities, they can be added to the Reference Architecture and used to compose future application frameworks.



During the next step, we group primitive components from the application framework into the components of the Machining Center Application Architecture, which is an open specification of

components and message interfaces. Suppliers will base their interoperable controller products on this published, freely available specification. These primitive components groupings are influenced by Architecture Description Rules (see Appendix C), by Domain Models (see Appendix D), and by design considerations and implementation technology, e.g., availability of off-the-shelf implementation components, timing capabilities, and information flow. The previous step focused on analysis, but during this step, design considerations become important. The behaviors of an application architecture component derive from the collective responsibilities of its grouped primitive components. A component's message set derives from its behaviors and the resources and products of its primitive components.

Then we try to match existing software implementation components from the library, in this example the AcmeTM software library, to the components of the application architecture. Again, wherever no implementation components match the component specifications of the application architecture, existing software must be modified and/or new implementation components must be added to the AcmeTM software library. After any needed modifications or additions, the resulting set of implementation components comprise an application system, which is one instantiation of the application architecture. The AcmeTM library can be updated by adding the new or modified software.

A controller for this 5-axis machining center becomes open when it is accompanied by a complete architectural description of its application system. This description includes, but is not limited to: the standards profile, the set of APIs used, the application architecture, the hardware configuration, the application software configuration, and the detailed definitions of the implementation components their message interfaces.

4.1 Application Architecture for Machining

Application requirements (see Table 4.1) are used in conjunction with the Architecture Description Rules and Domain Models to assemble the components of an application architecture for controlling the 5-axis machining center as shown in the example of Figure 4.1. Based on requirements, relevant primitive components are selected from the Reference Architecture, and their responsibilities and interfaces are aggregated into the components of this _ example. In Figure 4.1, each component is given a unique name descriptive of the role it plays in the application architecture. A component's constituent primitive components are indicated by number (see Appendix B). The lines connecting the components show the message paths.



Figure 4.1. Example of an Application Architecture for a Machining Center

.....

Reference Number	Application Requirement
A.1.3.1	The controller shall interface with a supervisory computer system.
A.1.3.2	The supervisory computer system shall exert complete control over the controller.
A.1.7.1	Block processing time shall be fast enough to prevent data starvation.
A.1.7.2	The controller shall automatically decelerate axes movement when the block execution time is
	less than the average block processing time.
A.2.1.1	The controller shall provide continuous path control of tool motion.
A.2.1.2	The controller shall provide point-to-point programming.
A.2.2.1	The controller shall perform linear interpolation in simultaneous motion of all axis drives.
A.2.2.2	The controller shall perform circular interpolation with 3-D motion control.
A.2.2.3	The controller shall perform helical interpolation.
A.2.7.1	The controller shall automatically control axis acceleration/deceleration.
A.2.8.1	The controller shall provide automatic compensation of lead screw errors (pitch error compensation).
A.2.8.2	The controller shall provide automatic compensation of backlash.
A.2.8.3	Lead screw error compensation shall be reprogrammable for wear compensation.
A.2.8.4	The controller shall perform cross compensation for axis alignment, beam sag and axis motion
	geometry errors.
A.2.9.1	The controller shall perform temperature growth compensation.
A.2.12.1	The controller shall support programmable feedrates.
A.3.1.1	Tool management shall have storage for the maximum number of tool data sets.
A.3.1.4	The tool shall be referenced by the tool identification code.
A.3.1.5	The tool shall be referenced by the tool location identification code.
A.3.1.6	The tool magazine shall provide random storage of tools with preassignment.
A.3.1.7	Tool slots, compartments, or other storage elements shall not have sensors for identification and location of each tool.
A.3.1.8	The tool magazine shall hold the complement of tools and spares for specific jobs.
A.3.1.9	The tool magazine shall not support removable tool cartridges.
A.3.1.10	The tool magazine shall not support sensors for identification of tool cartridges and their job associations.
A.3.1.15	Tool change sequence shall be initiated by a T code.
A.3.1.16	Tool change sequence shall not be initiated manually.
A.3.1.18	Tool length shall not be verified as part of the tool change procedure.
A.3.1.19	Tool diameter shall not be verified as part of the tool change procedure.
A.3.1.20	Tool form shall not be verified as part of the tool change procedure.
A.3.2.1	The controller shall identify pallets by non-contact sensors.
A.3.2.4	Pallet identification shall be used to automatically select part programs.
A.3.2.5	Pallet identification shall be used to automatically select fixture offsets.
A.3.2.6	Pallet identification shall be used to automatically select pallet offsets.
A.3.2.7	Multiple part programs shall be selected per pallet.
A.3.3.9	The controller shall measure engineering force levels on the spindle bearings.
A.3.3.10	The controller shall measure temperature of the spindle bearings.
A.3.3.14	The controller shall adjust feedrate to maintain constant cutting force.

Table 4.1. Application Requirements for Machining Example

(Continued)

_ .

7

ï

Reference No.	Application Requirement
A.4.1.5	Programming shall include feedrate programming: direct feedrate, inverse time feedrate, per revolution unit feed for turning controls.
A.4.1.11	Programming shall include fixture offsets for multiple pallet and table machines.
A.4.1.12	Programming shall include multiple program storage and management.
A.4.1.16	Programming shall include programmed tool change.
A.4.1.17	Programming shall include programming for cutting process control.
A.4.1.19	Programming shall include programmable adaptive control parameters.
A.4.1.23	Programming shall include preprogrammed (canned) cycles for drilling, boring, and tapping.
A.4.1.24	Programming shall include preprogrammed cycles for area milling, rectangular pocket milling, circular pocket milling, and bolt hole circles.
A.4.1.25	Programming shall include automatic chamfering and corner radiusing.
A.4.1.26	Programming shall include look-ahead cutter compensation.
A.4.1.27	Programming shall include coordinate system rotation.
A.4.1.28	Programming shall include work coordinate system setting.
A.4.1.30	Programming shall include constant surface speed programming.
A.4.1.31	Programming shall include tool center point programming for 5-axis machining.
A.4.2.2	Manual programming shall include automatic selection of cutting speed and feed.
A.4.2.3	Manual programming shall include cutter offset.
A.4.2.5	Manual programming shall include pallet or table change cycles.

Table 4.1. Application Requirements for Machining Example (Cont.)

4.2 Machining Scenario

The application architecture for a machining center controller facilitates its design and implementation. In this section we show how the application architecture, which consists of components and message interfaces, is also useful for verifying the operation of the controller in an example scenario. The example scenario covers mass production of parts, and it is better understood by referring to Figure 4.2 where the message interfaces of the application architecture in Figure 4.1 are shown in greater detail.

This is a partial example of an application architecture. For clarity, the functionality of the components and the number of messages have been limited intentionally just to cover the scenario. This cannot reflect the complete set of messages needed for all of the operating scenarios of a machining center, but it gives a flavor of the level of message specification for an application architecture.







Figure 4.2b Application Architecture Components and Messages—Part 2



Figure 4.2c Application Architecture Components and Messages-Part 3

In an application architecture, the components embody the responsibilities of the primitive components selected for the application, and the messages represent the flow of resources and products explicitly. However, messages between application architecture components may not be as specific as the interface definitions required for implementation components, e.g., function prototypes in C-language header files. This example will compare representative application architecture messages with their corresponding implementation messages.

In the following discussion, "the controller" refers to an open controller. Component names are all capitalized, e.g., JOB MANAGER. Message names, e.g., Allocate Tools, and the names of information types and structures in the messages, e.g., Tool Information, have leading caps. System elements outside the controller have no capitals, e.g., factory scheduling system.

4.2.1 Factory Interaction

The factory scheduling system resides above the shop floor and sends job instructions to the controller. These instructions are contained in the Perform Jobs message, which consists of a Job

List specifying what is to be done and a Tool List enumerating the tools required from the tool room. The Job List contains a sequence of Job Descriptions designated by job id. Each Job Description identifies a series of pallets to be processed, each having a pallet id, a pallet offset, and a list of parts. A Job Description is organized as follows:

job id

pallet id, pallet offset

part 1, stock id, part program id, part offsets, beta offset part 2, stock id, part program id, part offsets, beta offset part 3, stock id, part program id, part offsets, beta offset pallet id, pallet offset

part 1, stock id, part program id, part offsets, beta offset part 2, stock id, part program id, part offsets, beta offset part 3, stock id, part program id, part offsets, beta offset

etc.

factory scheduling systemPerform Jobs (FACTORY SCHEDULER, Job List, Tool List)FACTORY SCHEDULERAccept Status (factory scheduling system, Job Status)

The FACTORY SCHEDULER, a component of the controller, is responsible for communicating with the factory scheduling system. It passes the Job List to the JOB MANAGER. The Tool List is passed to the TOOL MANAGER, and the TOOL MANAGER sends the Tool IDs to the TOOL CHANGER. Through the FACTORY SCHEDULER, information about the status of jobs can be shared with the factory scheduling system because the JOB MANAGER maintains an awareness of the jobs that have been released and those which are still queued.

FACTORY SCHEDULER	,	Perform Jobs (JOB MANAGER, Job List)
FACTORY SCHEDULER	,	Allocate Tools (TOOL MANAGER, Tool List)
TOOL MANAGER	,	Load Tools (TOOL CHANGER, Tool IDs)
JOB MANAGER	,	Accept Status (FACTORY SCHEDULER, Job Status)

4.2.2 Resource Verification

When the JOB MANAGER receives the Job List, it asks the RESOURCE MANAGER to verify the availability of the needed resources. For each job in the Job List, the JOB MANAGER sends the RESOURCE MANAGER a Job Description, which contains the part program ids and stock ids.

JOB MANAGER ' Verify Resources (RESOURCE MANAGER, Job Description)

To verify that proper resources are available, the RESOURCE MANAGER asks other managers for applicable information. First, the FILE MANAGER is asked to send the RESOURCE MANAGER a Part Program. The FILE MANAGER responds with the Part Program File, or a Missing Program Error.

RESOURCE MANAGER	,	Provide Part Program (FILE MANAGER, Part Program ID)	
FILE MANAGER	,	Use Part Program (RESOURCE MANAGER, Part Program File)	
OF		Accept Error (RESOURCE MANAGER, Missing Program)	

After scanning the Part Program File for tool information, the RESOURCE MANAGER asks the TOOL MANAGER if the necessary tools (or adequate substitutes) are loaded. The TOOL MANAGER responds with Tool IDs or a Missing Tool Error.

RESOURCE MANAGER	,	Verify Tools (TOOL MANAGER, Tool Information)
TOOL MANAGER	,	Use Tools (RESOURCE MANAGER, Tool IDs)
ог		Accept Error (RESOURCE MANAGER, Missing Tool)

The RESOURCE MANAGER may need to further verify the ability of the machine to perform the intended job. It asks the MODEL MANAGER for the Machine Model, Stock Models, Part Models, etc. to compare them to the part program.

RESOURCE MANAGER	,	Provide Models (MODEL MANAGER, Machine Model, Stock Models, Part Models)	
MODEL MANAGER	,	Use Models (RESOURCE MANAGER, Model Information)	

Should any of these resources be missing, the RESOURCE MANAGER informs the OPERATOR which resources are missing, and it informs the JOB MANAGER that this job did

not pass. The JOB MANAGER then holds this job until the OPERATOR indicates something has been done to correct the problem. The JOB MANAGER then resubmits the job for verification to the RESOURCE MANAGER. This cycle continues until the RESOURCE MANAGER indicates to the JOB MANAGER that the required resources are present.

RESOURCE MANAGER	,	Find Resources (OPERATOR, Job Description, Missing Resources)
RESOURCE MANAGER	,	Accept Verification (JOB MANAGER, Resources)
or		Hold for (JOB MANAGER, Resource Verification)
OPERATOR	,	Release Job (JOB MANAGER, Job Description)

4.2.3 Job Release

After the OPERATOR has released the job for execution, the JOB MANAGER asks the PART HANDLER to load the first pallet with stock. The PART HANDLER loads a pallet and tells the JOB MANAGER the Pallet ID. The JOB MANAGER now knows which part to do first; it tells the CYCLE CONTROL to start running the first program; and it tells the ENHANCER the offsets (Pallet Offset, Part Offsets, Beta Offset) necessary to locate the part on the pallet. The JOB MANAGER marks the job as "in-progress" and waits for the CYCLE CONTROL to request the next program.

JOB MANAGER	,	Load Pallet (PART HANDLER, Stock IDs)
PART HANDLER	,	Use Pallet (JOB MANAGER, Pallet ID)
JOB MANAGER	,	Make Part (CYCLE CONTROL, Pallet ID, Part ID, Stock ID, Part Program ID)
JOB MANAGER	,	Use Offsets (ENHANCER, Pallet Offset, Part Offsets, Beta Offset)

4.2.4 Part Program Setup

The CYCLE CONTROL has been told to run the next part program, so it asks the FILE MANAGER to provide the program. If the CYCLE CONTROL can't hold the entire program, it periodically asks the FILE MANAGER to provide more code, until the entire program has been processed. The JOB MANAGER is informed when the end of the program has been reached so it can start processing the next part on the current pallet. If the next part uses the same program as the last part, then the file may not need to be accessed from the FILE MANAGER. If there

are no more jobs on the current pallet, the JOB MANAGER asks the PART HANDLER to change the pallet, and processing starts on the new pallet.

CYCLE CONTROL	,	Provide Part Program (FILE MANAGER, Part Program ID)
FILE MANAGER	,	Use Part Program (CYCLE CONTROL, Part Program File)
CYCLE CONTROL	,	Accept Finished Part (JOB MANAGER, Part ID)

To continue, CYCLE CONTROL feeds the Part Program to the PARSER, which analyzes each block of code to identify commands pertinent to the attached mechanisms. In our example, the mechanisms are the TOOL CHANGER, the PART HANDLER, and a coolant control device called COOLANT FLOW. The PARSER tells these devices which commands they will be expected to perform when given the appropriate messages. This gives the mechanisms a chance to do any necessary preliminary setups before the actual command is given.

CYCLE CONTROL	,	Analyze Part Program (PARSER, Part Program)
PARSER	,	Setup (TOOL CHANGER, Part Program Commands)
PARSER	,	Setup (PART HANDLER, Part Program Commands)
PARSER	,	Setup (COOLANT FLOW, Part Program Commands)

Any comments or operator cues for manual tasks are sent to the OPERATOR from the PARSER. When the OPERATOR completes the task, it informs the JOB MANAGER through the same pathway it used when adding missing resources. The JOB MANAGER tells the CYCLE CONTROL to continue with this cycle, and the CYCLE CONTROL passes the continue message on to the PARSER.

PARSER	,	Accept Comment (OPERATOR, Explicit Comment)
PARSER	,	Accept Cue (OPERATOR, Manual Task Description)
OPERATOR	,	Release Job (JOB MANAGER, Job Description)
JOB MANAGER	,	Continue Part (CYCLE CONTROL, Part ID)
CYCLE CONTROL	,	Continue Part (PARSER, Part ID)

4.2.5 Code Augmentation

When the interaction with the OPERATOR is concluded, the PARSER feeds the ENHANCER the Part Program, inserting the recipient or destination into each command. The ENHANCER augments motion command code from the PARSER. During the augmentation process, detail blocks or commands are inserted in place of canned cycles. The ENHANCER also translates all coordinate system data since it can derive the exact location of the part through the series of offsets provided by the JOB MANAGER.

PARSER ' Augment Code (ENHANCER, Part Program with Recipients)

Final augmentation done by the ENHANCER relates to the coordinated activities between mechanisms and motion. The ENHANCER inserts coordination points into the command sequence. Say a tool change needs to occur at some point. The ENHANCER would replace the tool change command in the program with a series of commands in the motion stream. These inserted commands might first bring the tool to a chosen location and wait for a continue message. Following the wait command would come commands to move the spindle to another location and wait again for a continue message. Finally, a rapid traverse is inserted after the second wait command.

This augmentation performed by the ENHANCER is highly dependent on the types of mechanisms and their functionality. In some cases, the system startup procedures may require the ENHANCER to ask the mechanisms for specialized augmentations for motion to perform whenever one of their commands is encountered.

ENHANCER	,	Provide Augmentations	(TOOL CHANGER)
ENHANCER	,	Provide Augmentations	(PART HANDLER)
ENHANCER	,	Provide Augmentations	(COOLANT FLOW)

After the ENHANCER inserts coordination points into the command sequence, it gives the COORDINATOR a complete copy of the augmented code, but it gives the TRAJECTORY GENERATOR just the Motion Command Sequence. This Motion Command Sequence consists of a series of points along with Feedrate, Spindle Speed, Acceleration, Deceleration, and any other part program information relevant to motion. The ENHANCER feeds the Motion Command Sequence to the TRAJECTORY GENERATOR in blocks and waits until the TRAJECTORY GENERATOR requests more.

ENHANCER		Use Part Program (COORDINATOR, Augmented Part Program)	
ENHANCER	,	Use Commands (TRAJECTORY GENERATOR, Motion Command Sequence)	
TRAJECTORY GENERATOR	,	Provide Commands (ENHANCER)	

4.2.6 Coordination Between Mechanisms

Before covering the details of motion, we consider coordination between mechanisms. The TRAJECTORY GENERATOR does not distinguish between the motion commands from the original Part Program and those commands inserted into the Motion Command Sequence by the ENHANCER to effect mechanism coordination. So, for example, if a tool change is the next operation, the next command the TRAJECTORY GENERATOR sees in the Motion Command Sequence is Wait. The TRAJECTORY GENERATOR informs the COORDINATOR, and then it waits.

COORDINATOR ' Proceed Until Wait (TRAJECTORY GENERATOR)

TRAJECTORY GENERATOR ' Affirm Wait (COORDINATOR, Trajectory Generator)

Since the COORDINATOR is in possession of the entire Augmented Part Program, it can understand and coordinate the activities of the mechanisms under control. The COORDINATOR knows that a tool change is next when the TRAJECTORY GENERATOR sends it a Trajectory Generator Waiting message. The COORDINATOR tells the TOOL CHANGER that it is time to initiate a tool change. When the TOOL CHANGER receives Change Tool, it verifies that the new tool is indeed the proper tool change, performs the change, and informs the COORDINATOR that the tool change is complete. The COORDINATOR directs the TRAJECTORY GENERATOR to continue, and the TRAJECTORY GENERATOR proceeds until it encounters the next wait.

> COORDINATOR ' Change Tool (TOOL CHANGER, Tool ID) TOOL CHANGER ' Accept Tool Change (COORDINATOR, Tool ID)

4.2.7 Tool Motion

To begin a description of the motion operations, the TRAJECTORY GENERATOR sends the Path Normal Vectors to the CUTTER COMPENSATOR and the SENSOR PROCESSOR. The RATE CONTROL and CUTTER COMPENSATOR components receive messages from the ENHANCER: the RATE CONTROL needs the Rate required by the program; the CUTTER COMPENSATOR needs the Plane Selection.

TRAJECTORY GENERATOR	,	Use Normals (CUTTER COMPENSATOR, Path Normal Vectors)
TRAJECTORY GENERATOR	,	Use Normals (SENSOR PROCESSOR, Path Normal Vectors)
ENHANCER	,	Use Rate (RATE CONTROL, Rate)
ENHANCER	,	Use Plane (CUTTER COMPENSATOR, Plane Selection)

The RATE CONTROL determines the Goal Rate by comparing the program requested Rate to the Tool Tooth Count from the TOOL MANAGER and the Spindle Speed from the SENSOR INTERFACE. (The SENSOR INTERFACE gets its information from the sensors.) The RATE CONTROL supplies the Goal Rate to the TRAJECTORY GENERATOR. The TRAJECTORY GENERATOR would prefer to operate at the Goal Rate, but the machine may be unable to handle the Goal Rate. The TRAJECTORY GENERATOR asks the MODEL MANAGER for the Axis Step Limits, and modifies the rate accordingly.

TOOL MANAGER	,	Use Tool Tooth Count (RATE CONTROL, Tool Tooth Count)
SENSOR INTERFACE	,	Use Spindle Speed (RATE CONTROL, Spindle Speed)
RATE CONTROL	,	Use Rate (TRAJECTORY GENERATOR, Goal Rate)
MODEL MANAGER	,	Use Axis Step Limits(TRAJECTORY GENERATOR, Axis Step Limits)

Since the CUTTER COMPENSATOR must adjust the path according to the specific dimensions and characteristics of the tool, it requests the tool's Offset Amount from the TOOL MANAGER, and calculates the Cutter Offsets and passes them to the TRAJECTORY GENERATOR.

CUTTER COMPENSATOR	,	Provide Offset Amount (TOOL MANAGER, Tool ID)
CUTTER COMPENSATOR	,	Use Offsets (TRAJECTORY GENERATOR, Cutter Offsets)

The last information TRAJECTORY GENERATOR requires is the corrections based on sensor information. The SENSOR PROCESSOR gets the Force and Torque from the SENSOR INTERFACE, which, of course, receives it from the sensors. The SENSOR PROCESSOR uses Force, Torque, and the path normal vectors to correct Position, Rate, and Spindle Speed for the TRAJECTORY GENERATOR.

SENSOR INTERFACE	,	Use Force (SENSOR PROCESSOR, Force)
SENSOR INTERFACE	,	Use Torque (SENSOR PROCESSOR, Torque)
sensors	,	Use Values (SENSOR INTERFACE, Sensor Values)
SENSOR PROCESSOR	,	Use Values (TRAJECTORY GENERATOR, Position Corrections, Rate Corrections, Spindle Speed Corrections)

The TRAJECTORY GENERATOR generates a Normalized Trajectory. It is normalized in the sense of precise time segments. The TRAJECTORY GENERATOR gives the Position to the ACCURACY ENHANCER.

TRAJECTORY GENERATOR ' Use Position (ACCURACY ENHANCER, Position)

The ACCURACY ENHANCER corrects the trajectory to compensate for predictable variations, which include thermal effects, axis non-linearities, etc. Thermal compensation is governed by accessing the Temperature from the SENSOR INTERFACE which is updated from a temperature sensor. Timing is provided by the system clock or by an internal timer. Tables and Polynomials of precalculated corrections are accessed from the MODEL MANAGER. From all of these information sources, the ACCURACY ENHANCER computes the corrections and passes them on to the COMPENSATOR.

SENSOR INTERFACE	,	Use Temperature (ACCURACY ENHANCER, Temperature)
system clock or timer	,	Use Time (ACCURACY ENHANCER, Timing)
MODEL MANAGER	,	Use Tables (ACCURACY ENHANCER, Tables)
MODEL MANAGER	,	Use Polynomials (ACCURACY ENHANCER, Polynomials)
ACCURACY ENHANCER	,	Use Corrections (COMPENSATOR, Corrections)

The COMPENSATOR receives the Normalized Trajectory from TRAJECTORY GENERATOR, makes corrections, and sends the Servo Command Positions to the AXIS CONTROL. Timing concerns will separate the COMPENSATOR from the ACCURACY ENHANCER: the former must operate continuously to direct tool motion; the latter is brought in when compensation or adjustments to the basic commanded motion path are needed.

TRAJECTORY GENERATOR ' Use Trajectory (COMPENSATOR, Normalized Trajectory) COMPENSATOR ' Position Servo(AXIS CONTROL, Servo Command Positions)

The AXIS CONTROL converts the Servo Command Positions to messages commanding the -DRIVES. It accesses the Kinematic Equations from the MODEL MANAGER and stores the Following Errors. If the Following Errors exceed the Following Error Limits, the AXIS CONTROL will shut down the machine and request maintenance. The interfaces from the AXIS CONTROL to the DRIVES and from the DRIVES to the MOTORS are simple position and velocity messages.

MODEL MANAGER	,	Use Kinematic Equations (AXIS CONTROL, Equations)
AXIS CONTROL	,	Use Following Errors (MODEL N	AANAGER, Following Errors)
AXIS CONTROL	,	Use Commands (DRIVES, Drive (Commands)
DRIVES	,	Use Commands (motors, Motor C	fommands)
motors	,	Use Feedback (DRIVES, Motor Fe	eedback)
DRIVES	,	Use Feedback (AXIS CONTROL,	Drive Feedback)

4.2.8 IEC 1131 Environment

The IEC 1131 graphical programming and execution environment provides for creating, debugging, and managing real-time programs that control the TOOL CHANGER, the PART HANDLER, the COOLANT CONTROL, and other discrete devices like the spindle. The environment supports control software coded either in C or in the programmable controller languages defined by IEC-1131 Standard for Programmable Controllers Part 3 - Programming Languages, which include:

- Sequential Function Chart,
- Ladder Diagram,
- Function Block Diagram,
- Instruction List, and
- Structured Text.

4.2.9 Implementation Component Messages

ŧ

The application architecture components exchange abstract messages, but when a controller implementation is realized from the application architecture, the abstract messages must be translated into an exact specification. For this example the exact message specification takes the form of a C function prototype similar to those found in C header files. The following exact message specifications are taken from the header files produced by the NIST Enhanced Machine Controller project.

<<<<More to go here --- Specific interface definitions from the NIST implementation are matched with selected component messages as shown below.>>>>

The abstract message from the COORDINATOR directing a tool change by the TOOL CHANGER is repeated below:

COORDINATOR ' Change Tool (TOOL CHANGER, Tool ID)

The NIST header file defines a message type $(nml_io_change_tool_msg_t)$ and a function prototype of an interface $(nml_io_change_tool)$ to the behavior that implements the tool change. This is a C-language implementation of the message interface to the TOOL CHANGER component. The first parameter of the message structure carries an integer, which is the identifier of the recipient of the message, TOOL CHANGER. The second message parameter is the integer Tool ID. The function prototype specifies the interface to the function that effects the tool change.

```
/* Message ID 313 */
typedef struct {
    int id; /* Message Recipient's ID */
    int 1; /* Tool ID */
} nml_io_change_tool_msg_t;
extern int nml_io_change_tool(int i);
```

Note that the application architecture message is made specific when a particular controller is implemented. The implementation interface will vary depending on the language chosen. In this case, ladder code can be used as easily as a C code, but the interface definitions will differ due to the differences in the languages. Also the way implementation components are identified may vary among implementations. However, interoperability can be assured among implementation components only if the interfaces are consistent.

Another example compares the application architecture message interface to the COOLANT FLOW with the NIST interface definition.

COORDINATOR ' Initiate Next Command (COOLANT FLOW, Tool ID)

The Initiate Next Command message to the COOLANT FLOW component expands into four interface definitions for the NIST implementation component

```
/* Message ID 314 */
typedef struct {
 int id; /* Message Recipient's ID */
} nml_io_mist_on_msg_t;
extern int nml_io_mist_on();
/* Message ID 315 */
typedef struct {
int id; /* Message Recipient's ID */
} nml_io_mist_off_msg_t;
extern int nml io_mist_off();
/* Message ID 316 */
typedef struct {
 int id; /* Message Recipient's ID */
} nml_io_flood_on_msg_t;
extern int nml_io_flood_on();
/* Message ID 317 */
typedef struct {
 int id; /* Message Recipient's ID */
> nml_io_flood_off_msg_t;
extern int nml_io_flood_off();
```

4.3 Implementation

The scenario described in the previous section walks through an example of a specific NGC application architecture for machining. It is important to emphasize that the example is merely one of many realizable controller architectures. This section continues this discussion with several examples to illustrate the process of how the NGC developer implements an abstract application architecture to generate a fully integrated NGC application system.

4.3.1 Integration Process Review

Figure 4.3.1 on the following page illustrates the NGC integration process and emphasizes the design and implementation phases. The integration process flow is first introduced in Section 3.4.1 and is repeated in this example section because of its importance in understanding the general flow and process dependencies. This figure should be referenced throughout the

ł

>

subsequent implementation discussions to provide an overall context for specific steps described in the examples. Only the general process flow is captured. The developer may start from any stage in the process and all feedback paths are intentionally omitted for the sake of simplicity, but these additional entry points and paths may be inferred.

- In brief review, two basic process paths are illustrated. Applications development appears as the upper path in the diagram and platform development is the lower path. Although each path may be traversed independently and concurrently, both paths are strongly interrelated and each path potentially feeds the other throughout the integration process. The two paths merge at the final development stages to produce a complete NGC system.

The unifying concept that joins the platform and applications paths is profiles. All NGC implementation components, whether they support platform or applications related functions of the system, are profiled. It is through the application of profiles that many of the implementation decisions are made.

Profiles provide a convenient mechanism for assessing the relative compatibility of complimentary products and for making "apples-to-apples" comparisons of similar products. They may be used to assess if a specific application software component will run on a selected platform implementation. They provide the developer with the information he needs to select the optimal mix of COTS products based on the availability of compatible applications and platform products. Profiles also clarify the "make or buy" choices that are of fundamental importance to developer organizations when they embark on the development of a new product line.

Frequently, the selection of certain implementation options are predetermined. The inclusion of a specific product in a system is often mandated by an organization for a variety of strategic reasons. In such cases, profiles provide the mechanism to compare and contrast available products and options based on previous selections. They expose the alternatives for component interchange and simplify the task of system upgrade.

September 23, 1994

SOSAS Rev. 3.0



Figure 4.3.1. NGC Integration Process

Creating a profile for a platform conceived with a specific set of vendor products in mind has the added benefit of surfacing new product solutions. Such solutions may not have been evident during the initial survey or trade studies yet they may offer comparable or even better alternatives than the original solution. Platform implementations that share a common set of profiles support the same set of services and equivalent levels of portability and interoperability by inference. These "compatible" implementations may offer significant cost saving alternatives that are not apparent prior to profiling. Products organized by profile category offer a powerful new dimension to the access of product information for both controller developers and end-users.

It may be useful to assume the existence of a mature NGC implementation component library while walking through the subsequent examples. This library, referenced in Section 3.4.1 and described in further detail in Section 3.4.4, is envisioned to be an on-line catalog of COTS hardware and software and an archive for publicly available NGC software products, all with a supporting set of interactive tools for creation, maintenance, and perusal. It is an information repository for the storage and retrieval of detailed component interface descriptions, standard profiles and templates, and product disclosure information used for NGC-conformant product development.

Q

Early NGC development work must populate this component repository with newly developed components and interactive tools for its access. NGC implementation may be accomplished without the existence of the component library. This is how the first NGC controllers must be built. However, the integration process flow can not be fully appreciated without consideration of the library as an integral part of the process. The component library brings the full advantages of open controllers to the end-users by making available a complete variety of off-the-shelf interchangeable, replaceable components for system upgrade and flexibility. Component "integration" and "reuse" become a major implementation emphasis. Less development effort need be spent on "reinventing the wheel" and efforts may be directed toward developing value-added features and enabling technology areas. The availability of a mature library, fully stocked with NGC-compliant products and user friendly interactive tools for product integration, will bring the major benefits of NGC to the end user. It is therefore critical that early NGC activities focus on developing the component library.

4.3.2 Implementation Example - Machining Center

The integration process is best understood by looking at specific implementations of the application architecture example for a machining center. To illustrate some of the key points, two specific controller implementations are described. Figure 4.3.2-1 portrays the real time (RT) and non-RT platforms, the major communications links, and profiling options of each system implementation. For lack of formal identification of these NGC controllers, they will be referred to as simply Sys-A and Sys-B. It should be recognized that there are many different possible implementations of the architecture example.



Figure 4.3.2-1 Platform Implementation Examples

Figures 4.3.2-2 and 4.3.2-3 contrast the platform and presentation management dimensions of the API profiles of the two platforms. Both systems are bi-polar, whereby the applications software is physically partitioned onto RT and non-RT system platforms, and both systems use a PC ISA-based (Intel 386/486) front end to support all non-RT functions. However, the operating system and graphical user interface (GUI) for the non-RT platform of Sys-A is a combination of MS DOS and Windows.

The Sys-A developer also opted to include Hoops as a strategic move. It was heard that AutoDesk, the maker of AutoCAD and one of Microsoft's key software vendors for Windows NT, recently acquired Ithaca Software, the maker of Hoops. Hoops also works with Microsoft's Graphical Device Interface (GDI).

The non-RTOS for the Sys-B platform is SCO UNIX running X Windows and Motif. SCO UNIX happens to be POSIX compliant and an XPG3 (X/Open Portability Guide 3) branded product. The X Windows PEX option was also purchased for three-dimensional, PHIGS-compatible graphics.

ł



Figure 4.3.2-2 Profile: Platform Dimension



Figure 4.3.2-3 Profile: Presentation Management Dimension

Sys-A uses a hybrid PC/transputer-based platform for its real-time functionality and runs SCO XENIX. Although this system is not POSIX compliant, it is a multi-user, multitasking UNIX system optimized for personal computers. Sys-B is VME Motorola 68030/40-based and uses the VxWorks operating system. VxWorks is POSIX.1 compliant and will incrementally support the POSIX.4 real-time extensions to POSIX that are currently at various stages of draft release.

Both systems support full ANSI C. The non-RT platform for Sys-B also supports C++. The Development Language profile is used to show language support (Figure 4.3.2-4).

Files are accessed on the MS-DOS/Windows platform (Sys-A) using ANSI C. The factory scheduling system external to the NGC is a SunSparc. Job lists are accessed using PC NFS which provides access to files on the Sun server transparently, as if they were stored locally on the PC disk (refer to Figure 4.3.2-5). System B, on the other hand, is much more elaborate and leverages the full capability of the POSIX file system. It also conforms with the POSIX.8 Sun NFS profile.



Figure 4.3.2-4 Profile: Development Language Dimension



Figure 4.3.2-5 Profile: Development Language Dimension

Figure 4.3.2-6 illustrates the communications and device I/O dimensions of the API profile suite. A data communications link ties the two platforms of each system together. Sys-A makes use of the onchip communications capability of transputers and an Ohio State University package called Trollius to achieve this inter-platform link. Sys-B uses the TCP/IP network capability of the two UNIX environments and specialized software to transfer data between the two environments.



Figure 4.3.2-6 Profiles: Communications and Device I/O Dimension

Standard networks are used to link the NGCs to the factory scheduling system. System B has a sophisticated set of communications alternatives. It has both simple and detailed network interfaces (SNI, DNI) for protocol independent interface (PII) that is fully conformant with POSIX.12. To facilitate global naming and remote access over the Internet, POSIX.17-

compliant Directory Services (X.500) software is installed. In addition, in preparation for the wave of soon-to-be-released object-oriented products, a C++ CORBA package, Expersoft's XShell, has been recently installed.

Sys-A uses specialized user-defined drivers to interface with NGC sensors, mechanisms, motor - drives, etc. Sys-B makes use of the available POSIX.4x device control API for standard I/O device communications (some of the latter features of Sys-B are exaggerated for illustrative purposes and do not necessarily reflect the present capabilities of the underlying operating system).

Having defined the profiles of the system platform in all dimensions, for both the API and EEI branches of the OSE taxonomy, the developer then begins to select/develop application components that implement various portions of the application architecture.

Figure 4.3.2-7 shows a possible partitioning of the application architecture primitive and aggregate components. The shaded boxes are non-RT components and the unshaded boxes represent RT components. The RT or non-RT nature of the components determine whether they operate on the RT or non-RT platforms of our two system example. Profiles for the software implementations of the architectural components are then compared against profiles of their respective platforms to determine their relative compatibility. Compatible implementations are then integrated into a final NGC system.

Figure 4.3.2-8 is a variation on the possible partitioning of the architecture components. In this example, the cross-hatched boxes indicate hard-RT components. In terms of the platform, this may signify that a specialized COTS motion controller board is incorporated into the system and subsumes all of the component functionality in firmware or hardware. It should be obvious from the numerous examples in this section that the number of implementation options is enormous.



Figure 4.3.2-7 Platform Partitioning, Example 1



Figure 4.3.2-8 Platform Partitioning, Example 2

4.3.3 Implementation Example—Low End Controller

The previous example describes the profiling process for two platform implementations. It focuses on platform development, the lower branch of the integration process diagram (Figure 4.3.1). The developer then selects or develops application components that are compatible with the profiled platforms to implement various portions of the application architecture. Similarly, the platform is implemented from a selection of off-the-shelf products and/or developed components. This section provides an expanded view of this implementation process.

Many of the implementation concepts presented herein have been introduced in subsections of Section 3.4, Development Process and Implementation. Figure 3.4.2.1 illustrates characteristic differences between architecture development and implementation. For example, profiling and language selection are concepts of implementation, and are not considerations during architectural development and specification. Section 3.4.3 introduces the notion of a specification of "implementation guidelines" to ensure the development of implementation components that are truly portable, interoperable, and interchangeable. Some of these guiding philosophies are described in this section. The "implementation component library", described in Section 3.4.4, offers an on-line central archive of reuseable, publicly available products and information. The existence of this library is viewed to be the enabling technology that will bring NGC from a conceptual vision to a reality.

The implementation process transitions from an architectural specification on paper to a set of implementation components that are integrated into a physical system. An example used for illustrating the process is introduced in Section 3.4.5. This section is a more thorough treatment of the use of profiles, interrelationships with the component library, timing considerations, and other language issues as they relate to the implementation process. Profiling is discussed as a logical continuation of the previous example and is followed with a detailed description of the example of Section 3.4.5. Relevant figures are repeated in this section for convenience.

To complete the profiling story of the previous example, it is useful to consider some of the more subtle implications of platform profiling. Given a mature implementation component library, standard profiles are made available. One possible scenario for platform implementation is to first choose a standard profile from the implementation component library and then select specific COTS products that, when packaged together, meet all the standards requirements of that profile. Platform profiling should not be viewed as that necessary step in the process that is performed after all the implementation choices are made, and specific COTS products are

selected. Although this second scenario is certainly a possibility, it does not leverage many of the inherent advantages of profiling.

Profiling is a selection from a large list of relevant standards that identifies the unique level of openness of a controller product. It is a statement of the commitment of the developer to support - specific standards in his product lines. In that sense, platform profiles are the fundamental requirements for platform implementation. Many developer organizations will make decisions to commit to a particular set of standards prior to making any specific implementation decisions. In such circumstances, platform profiling naturally precedes any consideration to implementation.

In practice, it is probably more realistic to assume that the platform profile for a specific product will evolve based on a number of mitigating factors. The integration process described in Section 3.4.1 and reviewed in Section 4.3.1 illustrates how profiles "unify" platform and applications development. Profiles clarify "make or buy" choices and provide the developer with the information he needs to select the optimal mix of COTS products based on the availability of compatible applications and platform products. Implicit in this statement is the advantage of profile flexibility in the development process. Profiles that are tailorable based on product availability help to achieve the best solution. Such a solution leverages all available products, meets cost, and satisfies many if not all of the initial objectives for controller openness with little compromise.

Figures 4.3.3-1 thru 4.3.3-3 illustrate the process of implementing a specific application architecture. Figure 4.3.3-1a is a simplification of the detailed architecture and generic implementation example of Figure 4.3.3-1b. For the sake of readability, Figure 4.3.3-1b is enlarged and divided into two diagrams, Figures 4.3.3-2 and 4.3.3-3, to better illustrate the detailed architecture and generic implementation examples respectively. These figures were introduced in Section 3.4.5.

Figure 4.3.3-1a shows the basic design considerations in transitioning from an application architecture to an implementation. In review, an application architecture is a high-level design that is implemented with hardware and software components. Hardware components support platform functions while software components may support either platform or applications level functions. Platform functions are the general-purpose service oriented functions that support software applications.

Implementation components fall into three general categories: platform hardware, platform software, and application software. As evident in figure 4.3.3-1a, these categories may be further subdivided. Although a complete list of categories is beyond the scope of this specification,

many of the categories may be inferred from the implementation component library figure of Section 3.4.4. The implementation component library is envisioned to have a directory structure to accommodate all of the various types of implementation components and associated information.



Figure 4.3.3-1a From Application Architecture to Implementation (Simplified)



Figure 4.3.3-1b From Application Architecture to Implementation (Detailed)



Figure 4.3.3-2 Application Architecture

-



Figure 4.3.3-3 Generic Implementation

Platform hardware categories include system hardware, peripheral devices, storage devices, communications devices, etc. Each of these categories are further subdivided. For example, the system hardware category includes PCs, chassis, backplanes, board-level products, and standard bus I/O. Platform software categories closely resemble those that appear in the OSE taxonomy. Operating systems and system software, communications/network packages, database packages, and window toolkits are example platform software categories.

The commercial availability of a product, or lack thereof, brings another level of dimensionality to the categorization of implementation components. Many platform profiles may be implemented using COTS products. However, where there is a deficiency in satisfying profile requirements with commercially available products, middleware must be developed. Middleware is strictly defined as a term associated with distributed systems and communications and is a layer of code that sits between the O.S. and the application. Middleware is more loosely defined in the context of NGC to represent software components that are developed expressly for bridging COTS interfaces to standard APIs referenced by specific profiles.

COTS hardware products include special-function boards, many of which provide services in support of embedded applications. Board support packages for board-level products and library packages that provide software interfaces for cross-development environments are often purchase options made available through third party sources. These support products are designed to minimize the difficulty of interfacing special-function boards and simplify their integration into a system. Such products today offer a variety of levels of software support and many do not necessarily adhere to open standards. They may be leveraged as with all other COTS platform products but may require a middleware component layer to satisfy profile API requirements.

At the applications level, Figure 4.3.3-1a illustrates a correlation between architecture components and implementation components that is not a one-to-one mapping. The SIC2/SIC3 component implements all the responsibilities that are defined for two components, AC2 and AC3, at the architecture level. SIC2/SIC3 is an example implementation component that represents the packaging or bundling implementation strategy of one developer. The client/server stubs are used for implementing transparent peer-to-peer communications and are typically object components that only have meaning to a link editor during the applications build process. Therefore, they are shown as an integral part of application component implementations. Client/server stubs have significance in CORBA and DCE implementations. They are generated by the interface definition language (IDL) compiler and implement remote procedure call mechanisms (RPCs). Refer to the discussion on CORBA and DCE IDL and RPCs in Section 3.3.2.1.6, Communications Services, for further details on client/server stub generation and the link process.

Client/server stubs, COTS products, board support packages and the anticipated set of component middleware products all have a place in the implementation component directory structure. From this viewpoint, they may all be considered implementation components. To keep the notation simple, these components are not numerically designated in the example implementation figures.

External message interfaces are reflected in the application architecture example as arrows that connect to a component from a single end with the other end disconnected. These messages are numerically designated and correlate with platform service requests at the implementation level. Intercomponent message sets at the architecture level are also numerically designated and translate into corresponding communications messages in the resulting implementation.

Figure 4.3.3-1b and its corresponding enlargements, Figures 4.3.3-2 and 4.3.3-3, expand the simplified example with an introduction to timing and multi-platform considerations. Also, the expanded application architecture example illustrates component aggregation. The primitive components of a specific application component aggregate are numerically identified.

The controller responsibilities, as defined by the primitive components included in this example, capture many of the basic requirements of the NGC controller at the low-end. This example concentrates on low-end requirements since it is anticipated that early NGC implementations will naturally have this focus. As technological advancements make higher-end capabilities more readily available in off-the-shelf products, it is believed that these capabilities will then be incorporated into future NGC products.

To illustrate the low-end capabilities of this architecture, consider the operator interface and manager component, designated as AC1 in Figure 4.3.3-2. The operator has responsibility for determining and sequencing each part program and its coordinate offsets for a given job (PC3) and ensures the availability of resources required by the jobs (PC5). Job sequences are initiated manually (PC62). Status is obtained interactively (PC7), and startup/shutdown procedures are also manually initiated (PC39,47). Unlike the the first implementation example, there is no support in this example for a factory scheduling system interface. Part programming and customization (PC53,54) is performed interactively using the manual data interface of the controller console. These operator responsibilities could be automated, as described by the first implementation example, but are instead handled by the operator interactively in this example. This type of stand-alone functionality, with perhaps a serial port interface for part program download, is typical of low-end controllers today.

The file access (and other service support) implicit in the operator interface and management functions is reflected abstractly in the application architecture as AM1b service message set. Display and keyboard management to/from the physical operator console is depicted as AM1a. These translate into IM1 service requests in the generic implementation example (Figure 4.3.3-3). Similarly, the AM12 communications message set between the operator interface component (AC1) and the parser, enhancer & coordinator component (AC2) at an architecture level maps to the IM12 message set of the implementation example.

Analogous to the display, keyboard and file management services that are required to support the operator *U*F component, similar *U*O services are required for implementing interfaces to motor drives and sensors. These are reflected as IM3 and IM4 service requests in the generic implementation and directly correlate with AM3 and AM4 message sets shown in the application architecture example. It is left as an exercise to the reader to use the appendices contained in this specification to research the definition of the primitive components identified in each of the other application architecture component aggregates and correlate message sets in the application architecture with specific communications messages and services in the generic implementation.

Applying the hard knock experience of implementing systems is crucial in specifying an architecture that may be feasibly implemented. System prototyping should play an essential role in the architectural specification process for its validation. This is especially important when the practicality of a new architectural approach remains unproven due to the lack of existing, working implementations.

Timing considerations required for implementation will naturally influence the aggregation of components specified at an architectural level. An architecture is a high-level design and can not be divorced from timing issues involved during implementation. The implementation of the architecture must be anticipated. Primitive components will be grouped into architectural aggregates such that their respective responsibilities will have "like" timing requirements in their implementation.

However, with the exception of considerations to the grouping of primitive components, timing issues need not be addressed at the architectural specification level. Most considerations to timing should be left to the details of implementation. This approach leverages the most benefit from the portability and interoperability characteristics inherent in NGC-compliant implementation component software. The timing of portable applications software will often only be limited by the capabilities of its underlying platform and supporting services. The timing characteristics of the applications software will also vary dramatically from platform to platform.

Implementation component aggregates having similar timing requirements may ultimately coreside on the same platform when they are implemented. The generic implementation example represents a multi-platform solution that has both RT and non-RT hardware platform components that are interconnected via a bridge. The two platforms have the effect of partitioning the applications software components such that SIC1 resides on the non-RT platform and the SIC2/SIC3 and SIC4 components reside on the RT platform. This is a realistic partitioning since most operator interface functions are easily managed by a non-RT support environment. For example, a one second response time is an acceptable response time to an operator (though possibly annoying), but is much too slow and potentially disastrous for sensor update to a motion control servo while cutting a part.

The RT platform also integrates a special-purpose sensor I/O board that is interfaced to the other components of the system via the SIC4 sensor I/F wrapper component. This particular component of the system takes on hard-RT timing characteristics because of the unmodifiable, applications-specific logic embedded in firmware on the product wrapped by SIC4.
5.0 SUMMARY AND RECOMMENDATIONS

5.1 Summary

This document has described a structure for the development of open architecture systems for advanced manufacturing applications. This structure accommodates both a wide range of software applications while at the same time permitting the use of a variety of potential platform solutions. The admissible platform solutions allow for a wide range of variability with respect to processor selection, buss selection, operating systems, and device interface and communication devices and standards.

The process of NGC system development has been formalized with respect to development of the overall system structure that satisfies system requirements through the use of a component based approach that allows the design process to begin at a level of abstraction that focuses on responsibilities and information flow within the system. This flow from the selection of primitive components from a reference architecture to the construction of an application architecture was chosen because of it's consistency with what is becoming common practice in the software community. This follows the same basic practices set forth in the Advanced Research Project Agency (ARPA) Domain Specific Software Architecture (DSSA) program.

The equally challenging issue of the system platform has been addressed through the concept of profiles, a concept adopted from the IEEE Portable Operating System Interface (POSIX) standard. The profile is essentially a means of declaring a specific implementation of a system by selecting standards and options from a large set of possible choices. The strategy of profiles was chosen as an alternative to selecting a small set of standards and conventions from a very large set and attempting to force this on the overall community. As stated in the introduction of this document, it will be the inevitable market forces that will result in a narrowing of design options for NGC systems and the emergence of a small set of standard implementations such as the PC and the Macintosh in the personal computer arena. Realistically, any attempt to guess which set of standards will win with respect to processors, busses, operating systems, device interfaces, etc. is futile. The profiling concept necessarily leads to the possibility of an extremely large number of NGC systems, at least initially. In some ways this can be viewed as no different from the current situation. There is, however, a crucial difference.

Profiling allows users to determine, with little effort, the degree of openness in the system that is being purchased. By declaring the fundamental structure of the system, the path is opened for other vendors to produce system elements that can easily be integrated into the resulting standardized systems. The ultimate benefit is to the user who, using the system profile, can draw on a wide range of related products for system expansion and improvement.

The profile system that has been developed addresses both the Application Program Interface (API) as well as the External Environment Interface (EEI). (Again, this is similar to POSIX). Taken together, the application architecture and the profiling structure form a firm foundation for NGC system development; they do not, however, represent a complete methodology that would support immediate development of a national, commercial NGC product base. There are two elements that are missing from the overall system that would facilitate the adoption of the NGC approach; *fully populated and widely accessible implementation component libraries* and a family of *tools for system design, integration, and validation*. Unfortunately, the difficulty lies in developing the initial libraries and tool set. Once initiated, it seems clear that they will themselves become important commercial products.

At the implementation component level (see section 3.3 and 3.4) abstraction is finally lost and issues of system interoperability are directly addressed. The implementation library, (or repository in DSSA terms), consists of elements of software that are actually linked and compiled into the final system. From the standpoint of the end user the truest manifestation of NGC is in the availability of implementation components, not in the abstract but necessary process that results in these components. It is not an over simplification to say that the implementation components will be the "currency" of the evolutionary NGC system.

Even with the availability of libraries of implementation components, computer based tools will be essential to the propagation and success of this technology. The tool set is what will be responsible for institutionalizing the full NGC process because it will provide a quantum leap in the way systems are developed and modified in the same way Windows and Macintosh interfaces represented a quantum leap with respect to humans interfacing with computers.

5.2 Synergistic Activities

There are a large number of ongoing activities that have the potential for filling in the remaining pieces of the NGC structure. These activities include controller development projects as well as more generalized real-time, distributed architecture projects. The controller projects provide much of the necessary detail needed to arrive at a complete and useful NGC structure. Specifically, they provide a set of primitive components, requirements definition, application architectures, but most importantly, specific messaging structures and an initial set of implementation components. It is also important to note that the identified controller projects are not limited to strictly machine tool control architectures. In addition to general machine tool applications, there are ongoing projects that also address inspection (coordinate measurement) and robotics (both autonomous and teleoperated.) Similarly, a number of ongoing real-time, distributed control projects are focusing on many of the "domain independent" communication issues that are important to NGC as well as the tool structure for both building and maintaining NGC-like systems.

The controller projects that could directly contribute to completing the overall structure of the NGC system include the NIST Enhanced Machine Controller, (EMC), the Unified Telerobotic Architecture Project, (UTAP), sponsored by the Aircraft Directorate at the San Antonio Air Logistics Center, (SA-ALC), with participation by NIST and JPL, the Air Force ManTech Title III activity, the Department of Energy Technologies Enabling Agile Manufacturing, (TEAM), as well as others. Even the European version of NGC, Open System Architecture for Controls within Automation Systems, (OSACA), is worthy of further study with respect to what it can offer a long term NGC activity. All of the programs described above offer elements that can be directly incorporated into the evolving NGC structure that was described in section 3.4. The NGC structure benefits from the standpoint of additional depth and completeness. Each of the individual programs also benefits from the ability to capture program legacy in the consistent framework of NGC.

Similarly, there are a number of projects that are addressing many of the domain independent and tools issues that are crucial to an effective and complete NGC system. The ARPA DSSA program is currently felt to be the most important among these programs. This program is working many of the issues essential to the advanced manufacturing domain. Issues that are currently at the forefront of this project include

requirements, system structural development, testing, and validation of systems. Most importantly, DSSA is focusing on a consistent set of tools that will make open, reusable systems a reality for all real-time control system applications. Even programs seemingly unrelated to NGC and advanced manufacturing such as the Reusable Software Architecture for Spacecraft, (RSAS), sponsored by the Air Force Phillips Lab, are in the process of working issues and developing tools that are directly relevant to NGC. Finally, the National Information Infrastructure Program (NIIP) is of direct interest to the NGC development process. This program has the goal of establishing and unifying many of the key information transfer standards necessary for NGC.

5.3 <u>Recommendation</u>

An effective overall structure has been developed for the NGC vision for advanced manufacturing systems. This structure facilitates the development of open, interoperable controllers for all advanced manufacturing operations. At the same time, the structure allows for an unprecedented growth in the vendor base that can field products relevant to this area. It is not an overstatement to say that this structure *could* serve as the foundation for a revolution in advanced manufacturing systems similar to that which has occurred in personal and workstation computing over the past ten years.

To make the NGC vision a reality, it will be necessary to continue the development of the overall system structure. This is not an overly daunting task because, as pointed out in the earlier parts of this section, the activity should focus on capture of technology from ongoing activities rather than additional development. It is safe to say that all the major elements required to complete the NGC system either currently exist, (e.g. NIST EMC), or are in the process of being developed (e.g. ARPA DSSA, Title III, TEAM, NIIP). The integration of these available elements is not an especially complex task, but it does require that some type of central repository site be identified and solid interfaces with these programs be developed.

6.0 GLOSSARY OF TERMS

ADL Rules: TBD

Aggregate Component: Two or more primitive components retaining notion of responsibilities and message requirements.

Application: The use of capabilities (services/ facilities) provided by an information system specific to the satisfaction of a set of user requirements.

Application Architecture: The result of applying specific application requirements to the selection of Reference Architecture components. Still an abstraction but messages become more specific. Many possible configurations of Reference Architecture components for any given applications.

Application Environment Profile (AEP): A profile, specifying a completed and coherent specification of the Open Systems Environment (OSE), in which the standards, options, and parameters chosen

are necessary to support a class of applications.

Application Platform: A set of resources that support the services on which application software will run. The application platform provides services at its interfaces that, as much as possible, make the specific characteristics of the platform irrelevant to the application software.

Application Program Interface (API): The interface between the application software and the application platform across which all services are provided. The API is primarily in support of application portability, but system and application interoperability are also supported by a communication API.

Application Software: Software that is specific to an application and is composed of programs, data and documentation.

Application System: One of many possible implementations of a specific application architecture. The computing platform is fully specified by a profile published by the control builder. Software module granularity conforms to chosen boundaries of the application architecture.

Architecture Description Language (ADL): TBD

Base Standard: A standard or specification that is recognized as appropriate for normative reference in a profile by the body adopting that profile, but is not a profile itself.

Component Profile: A profile that is made up of a defined subset of a single standard.

Conformance: Action or behavior in correspondence with current customs, rules, or styles. In particular, behavior in correspondence with SOSAS rules, requirements, and styles and documented by a SOSAS-coconsistent standardized profile.

External Environment Interface (EEI): The interface between the application platform and the external environment across which information is exchanged. The EEI is defined primarily in support of system and application interoperability. The primary services at the EEI comprise of human/ computer interaction services, information services, and communication services.

Hardware: Physical equipment used in data processing as opposed to programs, procedures, rules, and associated documentation.

Implementation Component: A hardware, software, or human entity that fulfills a specific set of responsibilities with a specific interface. The form of implementation is not specified.

Interface: The shared boundary between two functional units defined by functional characteristics and other characteristics, as appropriate.

Interchangeability: A characteristic of system components that makes it possible to replace one component with another component of equivalent functionality made by a different vendor.

Interoperability: The ability of two or more systems to exchange information and to mutually use the information that has been exchanged.

Open System Application Program Interface: A combination of standards based interfaces specifying a complete interface between an application program and the underlying application platform.

Open System: A system that implements sufficient open specifications for interfaces, services, and supporting formats to enable properly engineered applications software: to be ported with minimal changes across a wide range of systems, to interoperate with other applications on local and remote systems, and to interact with users in a style that facilitates user portability

Open Systems Environment (OSE): The comprehensive set of interfaces, services, and supporting formats, plus user aspects for interoperability or for portability of applications, data, or people as specified by information technology standards and profiles.

Performance: A measure of a computer system or subsystem to perform its functions; for example, response time, throughput, number of transactions per second.

Portability: The ease with which software can be transferred from one information system to another.

Primitive Component: A component with a single responsibility defined as an abstraction (no code) with an abstract

message interface and architecture constraints defined.

Process: A address space and one or more threads of control that execute within that address space and their required system resources.

Profile: A set of one or more base standards and, where applicable, the identification of chosen classes, subsets, options and parameters of those base standards necessary for accomplishing a particular function. A profile lists the choices of platforms and defines the interfaces for an implementation of SOSAS consistent NGC.

Protocol: A set of semantic and syntactic rules that determine the behavior of entities in performing communication functions.

Reference Architecture: The collection of all known components including both primitive and aggregate components.

Scaleability: The ease with which software can be transferred from one graduated series of application platforms to another.

Software: The programs, procedures, rules and any associated documentation pertaining to the operation of a data processing system.

Specification: A document that prescribes in a complete, precise, verifiable manner, the requirements, design, behavior or characteristics of a system or system component.

Standardized profile: A balloted, formal, harmonized document that specifies a profile.

Standards: Documents, established by consensus and approved by a recognized body, that provide for common and repeated use, rules, guidelines, or characteristics for activities or their results aimed at the achievement of the optimum degree of order in a given context.

Thread: A single flow of control within a process.

APPENDIX A — REFERENCE REQUIREMENTS

Reference requirements guide the grouping and specialization of responsibilities from the Reference Architecture when synthesizing an application architecture for a specific controller application. For any specific application, only a subset of the Reference Requirements are relevant, and when the relevant requirements are selected out of the full set of Reference Requirements, they must be made specific to the application. For example, if Reference Requirement A.1.2.2, "Availability shall be at ? percent." is selected for a given application, then it is made specific by changing it to "Availability shall be at 99 percent." The NGC Reference Requirements are listed below. The reference numbers provide a convenient grouping of the requirements.

A.1 General Requirements

A.1.1 Environmental

- A.1.1.1 The controller shall meet ? environmental standards.
- A.1.1.2 The controller shall be able to operate in ? temperature range.
- A.1.1.3 The controller shall/shall not be in an enclosure that keeps out dust-laden air.
- A.1.1.4 The controller shall/shall not be of modular construction.
- A.1.1.5 The controller shall/shall not automatically halt operation when the temperature within the control unit reaches ? degrees.
- A.1.1.6 The controller shall/shall not be equipped with an air conditioner that will assure operation in a temperature of ? and humidity of ?.

A.1.2 Reliability

- A.1.2.1 Availability shall/shall not be a function of mean time between failures and mean time to repair.
- A.1.2.2 Availability shall be at ? percent.
- A.1.2.3 The controller shall/shall not have a test program that will execute all control responsibilities implemented.

A.1.3 Factory Supervision

- A.1.3.1 The controller shall/shall not interface with a supervisory computer system.
- A.1.3.2 The supervisory computer system shall/shall not exert complete control over the controller.

A.1.4 Probing

- A.1.4.1 The controller shall/shall not incorporate measuring probes.
- A.1.4.2 The controller shall/shall not incorporate touch probes.
- A.1.4.3 The controller shall/shall not incorporate optical probes.
- A.1.4.4 The controller shall/shall not provide standard inspection (probing/gauging/measuring) cycles.
- A.1.4.5 The controller shall/shall not provide for user-defined measuring cycles.
- A.1.4.6 The controller shall/shall not measure the amount of excess or lack of workpiece material before operations.
- A.1.4.6 The controller shall/shall not adjust process parameters according to pre-operation measurements.
- A.1.4.7 The controller shall/shall not gauge the workpieces throughout operations to determine the necessity of tool changes.
- A.1.4.8 The controller shall/shall not gauge the workpieces throughout operations to determine the necessity of repeating operations.
- A.1.4.9 The controller shall/shall not gauge the workpieces throughout operations to determine the necessity of tool changes.
- A.1.4.10 The controller shall/shall not record the results of the inspection cycle.
- A.1.4.11 The controller shall/shall not generate digital coordinate point data describing any complex-shaped three-dimensional object.

A.1.5 Multitasking

- A.1.5.1 The controller shall/shall not support multiple concurrent tasks.
- A.1.5.2 The controller shall/shall not support background tasks of part programming.
- A.1.5.3 The controller shall/shall not support background tasks of program edit.
- A.1.5.4 The controller shall/shall not support background tasks of program loading.
- A.1.5.5 The controller shall/shall not support background tasks of program downloading.
- A.1.5.6 The controller shall/shall not support background tasks of program uploading.

A.1.6 User Interface

- A.1.6.1 The controller shall/shall not extract geometric information from program files.
- A.1.6.2 The controller shall/shall not graphically display geometric information in orthographic views.
- A.1.6.3 The controller shall/shall not graphically display geometric information in perspective views.
- A.1.6.4 The controller shall/shall not display a ?D simulation of the tool path.

- A.1.6.5 The controller shall/shall not display a ?D simulation of a cross sectional view of the part.
- A.1.6.6 The user interface design shall/shall not use colors and sounds.
- A.1.6.7 Colors and sounds shall/shall not be used sparingly and redundantly.
- A.1.6.8 The user interface shall provide access to information with a maximum of ? steps.
- A.1.6.9 The user interface shall have active ? windows at any one moment.
- A.1.6.10 The user shall/shall not be able to close all open windows with a single command.
- A.1.6.11 The user interface shall support ?-language presentation.
- A.1.6.12 Information shall/shall not be updated dynamically as it changes.
- A.1.6.13 The user interface shall provide timely feedback within ? ms for every input the user makes.

A.1.7 Block Processing Time

- A.1.7.1 Block processing time shall/shall not be fast enough to prevent data starvation.
- A.1.7.2 The controller shall/shall not automatically decelerate axes movement when the block execution time is less than the average block processing time.
- A.1.7.3 The controller shall have a look-ahead capacity of ? blocks.
- A.1.7.4 The average block processing time shall be ? milliseconds.
- A.1.7.5 The servo update time shall be from ? to ? milliseconds.

A.1.8 Software

- A.1.8.1 Software components shall be added or replaced incrementally with/without recompilation.
- A.1.8.2 I/O counts and types shall be expandable for ? type of equipment.
- A.1.8.3 The software shall/shall not preclude the use of a distributed system.
- A.1.8.4 The software shall/shall not preclude the use of a centralized system.
- A.1.8.5 The start-up sequence shall/shall not support appropriate start-up sequencing of all machine components.
- A.1.8.6 The start-up sequence shall/shall not support both a default and a user-defined idle state after start-up.
- A.1.8.7 The system shall/shall not come up in an operational mode following the start-up sequence.
- A.1.8.8 The system shall/shall not be orderly and responsible to emergency and catastrophic shutdowns.
- A.1.8.9 System shutdown shall/shall not result in graceful termination of process execution.

- A.1.8.10 A shutdown shall/shall not leave the system in a safe state with regard to the positions of valves, motors, and actuators.
- A.1.8.11 The system shall/shall not automatically save critical machine-state data.

A.2 Axis Motion Control Requirements

A.2.1 Continuous Path Control

- A.2.1.1 The controller shall/shall not provide continuous path control of tool motion.
- A.2.1.2 The controller shall/shall not provide point-to-point programming.
- A.2.1.3 The controller shall/shall not provide NURBS programming.
- A.2.1.4 The controller shall/shall not provide feature-based programming.

A.2.2 Interpolation

- A.2.2.1 The controller shall/shall not perform linear interpolation in simultaneous motion of all axis drives.
- A.2.2.2 The controller shall/shall not perform circular interpolation with 3D motion control.
- A.2.2.3 The controller shall/shall not perform helical interpolation.

A.2.3 Interpolation Resolution

- A.2.3.1 Interpolation resolution shall be a maximum of ? mm or ? inch in linear mode.
- A.2.3.2 Interpolation resolution shall be a maximum of ? mm or ? inch in circular mode.

A.2.4 Methods of Specifying Interpolation

- A.2.4.1 Circular interpolation shall/shall not be specified by radius and end point.
- A.2.4.2 Circular interpolation shall/shall not be specified by I, J, K parameters.
- A.2.4.3 Circular interpolation shall/shall not be performed in more than one quadrant in one command block.

A.2.5 Interpolation in Rapid Traverse

A.2.5.1 Rapid traverse motion shall/shall not be in linear interpolation mode in all axes.

A.2.6 System Resolution

- A.2.6.1 Control system least input increment shall be ? mm / ? inch and ? degree.
- A.2.6.2 Control system least command increment shall be ? mm / ? inch and ? degree.

A.2.7 Acceleration and Deceleration Control

- A.2.7.1 The controller shall/shall not automatically control axis acceleration/deceleration.
- A.2.7.2 The controller shall/shall not perform exponential acceleration for feed motions.
- A.2.7.3 The controller shall/shall not perform linear acceleration for rapid traverse.

A.2.8 Mechanical Error Compensation

- A.2.8.1 The controller shall/shall not provide automatic compensation of lead screw errors (pitch error compensation).
- A.2.8.2 The controller shall/shall not provide automatic compensation of backlash.
- A.2.8.3 Lead screw error compensation shall/shall not be reprogrammable for wear compensation.
- A.2.8.4 The controller shall/shall not perform cross compensation for axis alignment, beam sag and axis motion geometry errors.

A.2.9 Thermal Compensation

- A.2.9.1 The controller shall/shall not perform temperature growth compensation.
- A.2.9.2 The controller shall use ?# of thermal sensors.

A.2.10 Software Limitation of Axis Motion

A.2.10.1 The controller shall/shall not establish spatial zones (safe or forbidden zones) in addition to the fixed control limits provided by the machine builder.

A.2.11 Position Sensing

- A.2.11.1 The controller shall/shall not measure position with linear digit scales.
- A.2.11.2 The controller shall/shall not measure position with inductosyn devices.

A.2.12 Axis Velocity

- A.2.12.1 The controller shall/shall not support programmable feedrates.
- A.2.12.2 Programmable feedrates shall have a minimum of ? mm/min and a maximum of ? mm/min for the ? axis.
- A.2.12.3 Rapid traverse feedrates shall have a minimum of ? mm/min and a maximum of ? mm/min for the ? axis.

A.3 Auxiliary Machine Control Functions

A.3.1 Tool Management

- A.3.1.1 Tool management shall/shall not have storage for the maximum number of tool data sets.
- A.3.1.2 The number of data sets may/may not be equal to or greater than the number of tool changer pockets.
- A.3.1.3 The tool identification code shall have a minimum number of ? digits.
- A.3.1.4 The tool shall/shall not be referenced by the tool identification code.
- A.3.1.5 The tool shall/shall not be referenced by the tool location identification code.
- A.3.1.6 The tool magazine shall/shall not provide random storage of tools with/without preassignment.
- A.3.1.7 Tool slots, compartments, or other storage elements shall/shall not have sensors for identification and location of each tool.
- A.3.1.8 The tool magazine shall/shall not hold the complement of tools and spares for specific jobs.
- A.3.1.9 The tool magazine shall/shall not support removable tool cartridges.
- A.3.1.10 The tool magazine shall/shall not support sensors for identification of tool cartridges and their job associations.
- A.3.1.11 The tool length shall have a measurement of ? digits.
- A.3.1.12 Cutter diameter offset compensation shall have a measurement of ? digits.
- A.3.1.13 Cutter nose radius compensation shall/shall not be used with bull and ball nose cutters.
- A.3.1.14 Cutter taper compensation shall/shall not be supported.
- A.3.1.15 Tool change sequence shall be initiated by a ? code.
- A.3.1.16 Tool change sequence shall/shall not be initiated manually.
- A.3.1.17 Tool change operations shall/shall not automatically adjust to variations in spindle attachment geometries.
- A.3.1.18 Tool length shall/shall not be verified as part of the tool change procedure.
- A.3.1.19 Tool diameter shall/shall not be verified as part of the tool change procedure.
- A.3.1.20 Tool form shall/shall not be verified as part of the tool change procedure.

A.3.2 Pallet Changer

- A.3.2.1 The controller shall/shall not identify pallets by non-contact sensors.
- A.3.2.2 Pallet data shall consist of at least ? bytes.
- A.3.2.3 The controller shall track data for at least ? individual pallets.
- A.3.2.4 Pallet identification shall/shall not be used to automatically select part programs.
- A.3.2.5 Pallet identification shall/shall not be used to automatically select fixture offsets.
- A.3.2.6 Pallet identification shall/shall not be used to automatically select pallet offsets.

A.3.2.7 Multiple part programs shall/shall not be selected per pallet.

A.3.3 Cutting Process Control

- A.3.3.1 Cutting tool life shall/shall not be monitored based on user programmed time intervals/tool life time.
- A.3.3.2 A tool life data shall/shall not include: tool number, programmed tool life, accumulated cutting time, remaining tool life, and tools which have exceeded the programmed tool life.
 - A.3.3.3 The controller shall/shall not automatically select an alternative tool should a tool's life elapse.
 - A.3.3.4 Tool life data shall/shall not be passed to external devices or communication systems.
- A.3.3.5 The controller shall/shall not sense a broken tool.
- A.3.3.6 The controller shall/shall not sense significant changes in tool performance.
- A.3.3.7 The controller shall/shall not support a worn tap monitoring system for tapping speeds below ? RPM.
- A.3.3.8 The controller shall/shall not automatically change tools upon detection of a broken or worn tool.
- A.3.3.9 The controller shall/shall not measure engineering force levels on the spindle bearings.
- A.3.3.10 The controller shall/shall not measure temperature of the spindle bearings.
- A.3.3.11 The controller shall/shall not scan excessive loads for collision protection.
- A.3.3.12 The controller shall/shall not monitor for unbalanced spindle loads.
- A.3.3.13 The controller shall/shall not take appropriate actions when monitored spindle conditions exceed programmed tolerances.
- A.3.3.14 The controller shall/shall not adjust feedrate to maintain constant cutting force.
- A.3.3.15 The controller shall/shall not access work material parameter tables for automatic selection of cutting speeds, feeds, and depth of cut.

A.4 Programming Requirements

A.4.1 Automatic Programming

- A.4.1.1 Programming shall/shall not include high-level language part programming.
- A.4.1.2 Programming shall/shall not include inch or metric units.
- A.4.1.3 Programming shall/shall not include absolute or incremental programming.
- A.4.1.4 Programming shall/shall not include programmed dwell, stop, and optional stop.

- A.4.1.5 Programming shall/shall not include feedrate programming: direct feedrate, inverse time feedrate, per revolution unit feed for turning controls.
- A.4.1.6 Programming shall/shall not include direct spindle speed programming in RPM.
- A.4.1.7 Programming shall/shall not include block delete.
- A.4.1.8 Programming shall/shall not include program number search and sequence number search.
- A.4.1.9 Programming shall/shall not include conditional and unconditional jumps in the part program.
- A.4.1.10 Programming shall/shall not include mirror image.
- A.4.1.11 Programming shall/shall not include fixture offsets for multiple pallet and table machines.
- A.4.1.12 Programming shall/shall not include multiple program storage and management.
- A.4.1.13 Programming shall/shall not include safe zone definitions.
- A.4.1.14 Programming shall/shall not include parametric subroutines.
- A.4.1.15 Programming shall/shall not include custom macro routines.
- A.4.1.16 Programming shall/shall not include programmed tool change.
- A.4.1.17 Programming shall/shall not include programming for cutting process control.
- A.4.1.18 Programming shall/shall not include programmed tool life with designated replacement.
- A.4.1.19 Programming shall/shall not include programmable adaptive control parameters.
- A.4.1.20 Programming shall/shall not include programmable selection in tool failure detection modes.
- A.4.1.21 Programming shall/shall not include circular interpolation designated by radius.
- A.4.1.22 Programming shall/shall not include helical interpolation programming.
- A.4.1.23 Programming shall/shall not include preprogrammed (canned) cycles for drilling, boring, and tapping.
- A.4.1.24 Programming shall/shall not include preprogrammed cycles for area milling, rectangular pocket milling, circular pocket milling, and bolt hole circles.
- A.4.1.25 Programming shall/shall not include automatic chamfering and corner radiusing.
- A.4.1.26 Programming shall/shall not include look-ahead cutter compensation.
- A.4.1.27 Programming shall/shall not include coordinate system rotation.
- A.4.1.28 Programming shall/shall not include work coordinate system setting.
- A.4.1.29 Programming shall/shall not include inspection probe programming.
- A.4.1.30 Programming shall/shall not include constant surface speed programming.
- A.4.1.31 Programming shall/shall not include tool center point programming for 5-axis machining.

A.4.1.32 Programming shall/shall not include programming scaling.

A.4.2 Manual Programming Requirements

- A.4.2.1 Manual programming shall/shall not include provisions for conversational input.
- A.4.2.2 Manual programming shall/shall not include automatic selection of cutting speed and feed.
- A.4.2.3 Manual programming shall/shall not include cutter offset.
- A.4.2.4 Manual programming shall/shall not include programmable spindle orientation.
- A.4.2.5 Manual programming shall/shall not include pallet or table change cycles.
- A.4.2.6 Manual programming shall/shall not include program editing.
- A.4.2.7 Manual programming shall/shall not include the following setup capabilities: test run of part program, dry run of part program, graphic display of part position, graphic tool path display, execution of automatic probe routine to align program, axis inversion, and zero shift.
- A.4.2.8 Manual control shall/shall not include: emergency stop, program stop, manual axis select, manual positioning control, sequence number search and display, spindle power and RPM readouts, X, Y, Z, and other axis position readouts.

A.4.3 Status Record Requirements

- A.4.3.1 Status records shall/shall not include work order number: job id and quantity.
- A.4.3.2 Status records shall/shall not include job id number: part number and resource number or class number.
- A.4.3.3 Status records shall/shall not include job completion status: good, bad, rework, trail, active, unscheduled stop, program stop, and emergency stop.
- A.4.3.4 Status records shall/shall not include time stamp of start of job.
- A.4.3.5 Status records shall/shall not include time stamp of end of job.
- A.4.3.6 Status records shall/shall not include time stamp of last status record update for job cycle.
- A.4.3.7 Status records shall/shall not include elapsed time active for job cycle.
- A.4.3.8 Status records shall/shall not include data transfer enabled.
- A.4.3.9 Status records shall/shall not include edit mode enabled.
- A.4.3.10 Status records shall/shall not include block by block mode enabled.
- A.4.3.11 Status records shall/shall not include feedrate (percent override).
- A.4.3.12 Status records shall/shall not include spindle speed (percent override).
- A.4.3.13 Status records shall/shall not include programmable logic controller input, output, and status bits.

- A.4.3.14 Status records shall/shall not include machine activities: power on, start and stop, idle/running, remote or local, ready, hold, cycle start, controller stop, resume, emergency stop, test run, and dry run.
- A.4.3.15 Status records shall/shall not include tool status: list of tools, tool matrix data table (type of tool, number of pockets, pockets currently occupied, description), and tool object table (job number, T code, cutter compensation, tool offsets, gauge lengths, programmed tool life, pocket number).
- A.4.3.16 Status records shall/shall not include pallet status: number of pallets, pallet identification, list of pallet offsets, pallet object table (pallet type, pallet size, pallet priority levels, number of parts present, list of part coordinate reference) parts mounted, and parts machined.
- A.4.3.17 Status records shall/shall not include fixture status: type, id, and offsets.
- A.4.3.18 Status records shall/shall not include part program status: id, size, available memory size, create/modify data, restrictions on use, restrictions on modifications, running program is ?% over, program complete, and elapsed time.
- A.4.3.19 Status records shall/shall not include logical control parameters: feedrate override, spindle override, block delete, single block, and optional stop.
- A.4.3.20 Status records shall/shall not include diagnostic status: system hardware faults, tool magazine faults, tool changer faults, pallet magazine faults, spindle drive faults, and axes drive faults.
- A.4.3.21 Status records shall/shall not include event log: setting or resetting of conditions.

APPENDIX B — PRIMITIVE COMPONENTS

Primitive components represent indivisible responsibilities in the Reference Architecture. For each primitive component, a descriptive template represents temporal, resource, and product information. Temporal information explains the intended timing of the primitive component, such as start-up responsibilities. resource information explains what the primitive component will need in order to carry out its responsibility, such as coordinate system data. Product information explains what the primitive component can provide, such as translated coordinates. Shaded areas of each template represent information that is not appropriate at the reference architecture level. Once an application architecture is built, the determination of the contents of these areas brings the development process into the design phase. Below is the template including explanations for each entry. The remainder of this appendix contains the starter set of primitive components for NGC.

Responsibility: automatic descent relevance matter and		
Temporal Aspects	Anternation massages	Advance-Filmland Processe
Company destantion of company's Search of process during sharp activation a socialized	Cutore Messager Reported to	andreg to disposed metery to disposed metery to disposed
Local Resource		
Devroenen andere ander dans		1
Constructs Activity of Calls		
Source access of data		
Enternal Resources (nemerica)		Ι
Construction data sended from senders component		
		······································
and shall in the second state in the second state in the second state in the second state is the second state in the second state is the second state in the second state is the second st		† ·····
Broducts		
		1
Methoda Examples dep-by-dap present Autholog Ancientative if the companies	Sub-responsibilities: processes or methods complex shah segure latter dets	nter

B-1

Responsibility. NGC-1 H	andle operat	or moai	pations to part pro	gram statements
TemporalAspects	Incoming Messages:		Message-	
Initiation: modi%cation to part pro	gram necessar	ć		Initiated
Completion: program completed				- Process:
Span: prove-out		Chrige	mg wessages.	
Discrete / Continuous				
-		0		1
Local Besources		£		
Description:				
Attributes				· · · ·
Constraints:				
Smirno.				
External Resources				
Description: part program	machine mod	el	modi 1/cations	
Likely Source: workstation mgt	knowledge ba	ise	operator interface	
Beliability: weak	strong		weak	
Con1/acts: not likely	not likely		not likely	
Products				
Description: part program	error messa	ge		
Attributes: code: parameters	code: param	eter		
Constraints: dependent on machine	modi/cation	invalid		
Likely Recipients: wks mgt/plan inter	operator inte	rface		
Procedures/Methods• accept program hait program modivations• • modivy blocks • restart segm • send modivations•	am when s needed s if valid ent of prove-ou d blocks	Sub-r	esponsibilities: • halt program • modify blocks if valid • restart segment of p	rove-out

Responsibility: NGC-2 Pa	art program st	orage a	ctivities	
TemporalAspects		: 1007	ung Mossages:	Message-
Initiation system startup				- Dionates
Completion: system shutdown		Dutor	and lisesades	
Span: execution and prove-	out			
Discrete / Continuous				
1		2	3	4
Loop Pasa Intes				
Dean man	l			
deres ac	ļ			
Deversi e				
Same .				
External Resources				<u> </u>
Description:activity request	_ part program_			
Likely Source: workstation.mgt	storage mediu	m	· · · · · · · · · · · · · · · · · · ·	
Beliability:strong	strong			
ConVicts: not likely	not likely			ļ
Products	_			
Description: part program	program statu	s		1
Attributes: code: parameters	_code: parame	lers		
Constraints:				
Likely Recipients:workstation mgt	workstation m	at		
Procedures/Methods Example: • accept program activity	ty request	Sub-r	esponsibilities:	
access part program send part program or	program status	ł		

.

Responsibility: NGC-3	Determine ea	ch pa rs '	program and coord	linate offsets
TemporalAspects		Incom	ning Messages:	Message-
Initiation: system startup		1		Initiated
Completion: system shutdown		I	1	Process:
Span: execution and prove	-out	Outgo	ing Messages:	
Discrete / Continuous		T		
		·		
1		2		44
-Local Resources				
Aminutos				
Constraints	1	~~~~~		
Souther				
External Resources				
Likely Source: factory schedular				
Beliability: strong	<u> </u>			
Con1/acts not likely				
Products				
Description: part program id	nart.coordina	te offsets	<u></u>	
Attributes:	zero position	<u>S</u>		
Likoly Pociniontowerketetion met				1
	r workstation i			
Example: • accept job list • parse part program a • send part program ar	nd offsets nd offsets	500-1		
Responsibility: NGC-4 M	aintain inform	nation o	n available tools	
TemporalAspects		lincerr	ina Messages:	Message-
Initiation: system startup		I	-	hit atec
Completion: system shutdown		2.00	ina : /	H Process:
Span: execution and prove-	out	- ೧೯೯೯	n if is appelase	
Discrete / Continuous				;
1		2	3	
		C		
Descrator	1			
Atmostoc				
201.00				
External Resources	ļ			
Description: tool list	tool data updat	les	lebom.loot	
Likely Source: operator interface	operator interfa	ace	knowledge_base	
Heliability: strong	strong		_strong	
Des durate	not likely		not likely	+
	<u> </u>			1
Attributes	1			
Constraints:	[
Likely Recipients:knowledge base				
Procedures/Methods Example: • accept tool list, tool dat • access tool model • update model • send updated tool mode	ta updates Iel	Sub-r	esponsibilities:	

4

2

ì

Deepengibility				
nesponsibility: NGC-5 E	Ensure that re	esource	s required by the jo	bs are available
TemperalAspecto		Incom	ing Massages	Message-
Initiation:			inig meaadgea.	Initiated
Completion: system startep				Process:
Spap: execution and prove of		Outgo	ing Messages:	
Discrete / Continuous				
Discrete / Continuous	I			Ť
1		2	3	4
Lacal Resources				-
<u>Description</u>				
Attributes				
Obstants.				
Fistemal Deseurane				
Description: job description	tool machine	model	part program	
Likely Source: workstation mot	knowledge ba	ise	workstation mot	
Reliability: strong	strong		strong	
Con ¹ /cts ² not likely	not likely		not likely	
Products				
Description: veriveation results				
Attributes: code: parameters				
Constraints:			<u> </u>	
Likely Recipientsworkstation mgt	<u>I</u>	r	L	
Procedures/Methods		Sub-r	esponsibilities:	
Example: • accept job descri	iption machine models		 verify availability of re 	sources
verify availability	of resources			
send verilation	results	<u> </u>		
I				
Responsibility: NGC-6 F				
	Jetermine II r	next par	s program matches	s current past'progra
		next par	s program matche	s current pastprogra
TemporalAspects		incon	š program matche ning Messages:	Sourrent pastprogra
TemporalAspects Initiation: start of part program		incon	š program matche ning Massages:	s current past'progra
TemporalAspects Initiation: start of part program Completion: program completed		incon	š program matche ning Massages: pinn Mossages:	s current past'progra Message- licitiated Process:
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove-		incon Cutgo	š program matche ning Messages: bing Messages:	s current pastbrogra Nessage- Initiated Process:
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous	out	incon Cutgo	š program matche ning Messages: ping Messages: I	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		incon Crago	5 program matche hing Massages) bing Massages)	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		incon Cuigo 2	B program matche hing Massages) bing Massages;	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		ext par incon Outgo 2	S program matche ning Massages) ping Messages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		ext par incon Outgo 2	s program matches ning Massages: bing Messages:	s current past'progra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		ext par incom Outgo 2	s program matches ning Massages: bing Messages:	s current past'progra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		ext par incom Critge 2	s program matches ning Massages: bing Messages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous		Performance of the second seco	Sprogram matches ning Messages: bing Messages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Description: next part's program is Like Sources		Cuigo Cuigo 2	Sprogram matches hing Messages: bing Messages: c nid	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Contraction Description: next part's program is Likely Source workstation manager Peliability: strong	out	Citigo Citigo 2	s program matches ning Massages: ping Massages: ping Massages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	out out d current par nent plan interpr strong not likely	Cittige Cittige 2	Sprogram matches ning Massages: bing Massages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Contractions Description: next part's program is Likety Source workstation manager Reliability: strong Conflicts: not likety Products	out out d current par nent plan interp strong not likely	Cutgo Cutgo 2	Sprogram matches hing Massages: bing Massages: cing Massages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Contractions Starting External Resources Description: next part's program is Likety Source workstation manager Reliability: strong ConVicts: not likety Products Description: comparison result	out out d current par nent plan interp strong not likely	Cutgo Cutgo 2	Sprogram matches hing Massages: bing Massages: ping Massages:	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Contractions Starting External Resources Description: next part's program is Likety Source workstation manager Reliability: strong ConVicts: not likety Products Description: comparison result Attributes: value	out out 1 current par nent plan interp strong not likely	Cutgo Cutgo 2	Sprogram matches ning Massages: bing Massages: ping Massages: nid	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Image: Start of part program (completed) 1 Image: Start of part program (comparison managem) 1 Image: Start of part program (comparison result) 1 Image: Start of part part (comparison result) 1 Image: Start of part part (comparison result) 1 Image: Start of part (comparison result	out out d current par nent plan interp strong not likely	Cuitge Cuitge 2	Biprogram matches hing Massages: bing Massages: ping Massages: nid	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Span: execution and prove- Discrete / Continuous 1 Span: execution and prove- Discrete / Continuous 1 Span: execution Description: next part's program is Likely_Source workstation managem Beliability: Strong ConVicts: Description: comparison result Attributes: value Constraints: match / no match Likely Recipients:workstation man match	out out t current par nent plan interp strong not likely agement	Cuitge Cuitge 2	Biprogram matches hing Massages) bing Massages bing Massages hing Massag	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Discrete / Continuous 1 Description: next part's program is Likety Source workstation manager Beliability: Strong ConVicts: Description: companison result Attributes: value Constraints: match / no match Likety Recipients:workstation man Products Description: Description: companison result	out out t current par nent plan interp strong not likely agement	2 CE:190 CE:190 2 rt's progra	B program matches ning Massages: ping Massag	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Description: ext part's program is Likety Source workstation manager Beliability: strong ConVicts: not likety Products Description: companson result Attributes: value Constraints: match / no match Likety Recipients:workstation man Procedures/Methods Example: • accept next part's pro	out out t current pai nent plan interpi strong not likely agement rogram id	ext par incom Cutge 2 t's progra retation	B program matches ning Massages: ping Massages: ping Massages: a ping Massages: ping Mas	s current pastbrogra
TemporalAspects Initiation: start of part program Completion: program completed Span: execution and prove- Discrete / Continuous 1 Discrete / Continuous 1 Discrete / Continuous 1 Discrete / Continuous 1 External Resources Description: next part's program k Likety Source workstation manager Beliability: strong ConVicts: not likety Products Description: comparison result Attributes: value Constraints: match / no match Likety Recipients:workstation man Procedures/Methods Example: except current parts program k	out out t current par nent plan interpr strong not likely agement rogram id s program id	rest par incom Critica 2 rts progra	s program matches ning Massages: ping Massag	s current pastbrogra

Responsibility: NGC 7 P	atomino oto		otivo program	
NGC-7 L		aius 01 8 T	cuve program	
TemporalAspects		Incon	ning Messages:	Message-
Initiation: program startup				Initiated
Completion: program completed		0	Sina Massagar	Process:
Span: execution and prove	-out	1 Junge	ning iniessages.	
Discrete / Continuous				
		0		
		2		3
H-OCALHOSOUICES	}			
Attributoa				
Constraints:	1			
Source:				
External Resources				
Description: software status	1			
Likely Source: components				
Beliability: strong				
Con1/acts not likely			ļ	
Products				
Description: program status				
Attributes: code: parameters				
Constraints:	<u> </u>	·····		
LIKELY HECIPIENtSworkstation mgt	Į	1	l	
Procedures/Methods		Sub-r	esponsibilities:	
Example: • accept softwar	e status		•	
correlate for pr	ogram status	1		
• seno program	sidius			
	1	<u> </u>		
Posponsibilit <i>r</i>				
Responsibility: NGC-8 Re	equest additi	onal seg	gments of part pro	ogram
Responsibility: NGC-8 Re	equest additi	onal seg	gments of part pro	ogram Message-
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup	equest additi	onal seg	gments of part pro	ogram Message- Initiated
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown	equest additi	onal seg	gments of part pro	ogram Message- Initiated Process:
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove-co	equest additi	onal seg noor Outgo	gments of part pro ling Messages: ling Messages.	ogram Message- Initiated Process:
Responsibility: NGC-8 Re TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove-co Discrete / Continuous	equest additi	onal seg noon Outgo	gments of part pro ling Messages: ling Messages.	ogram Message- Initiated Process:
Responsibility: NGC-8 Re TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove-co Discrete / Continuous	equest additi	onal seg noor Outgo	gments of part pro ting Messages: bing Messages.	ogram Message- Initiatec Process:
Responsibility: NGC-8 Re TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous	equest additi	onal seg noon Outgo 2	gments of part pro ning Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous	equest additi	onal seg ndom Outgo 2	gments of part pro ning Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and proved Discrete / Continuous	equest additi	onal seg ndor Outgo 2	gments of part pro ming Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and proved Discrete / Continuous	equest additi	onal seg ndor Outgo 2	gments of part pro ming Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous	equest additi	onal seg noor Outgo 2	gments of part pro ming Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 Described Activities Construction Activities Sources External Resources	equest additi	onal seg noor Oungo 2	gments of part pro ang Messages: ping Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and proved Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional sec	onal seg occr Oungo 2	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional seguest	onal seg door Oungo 2	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional sep plan interpret strong	onal seg door Oungo 2 gments re ation	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional se	onal seg door Oungo 2 gments re ation	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional set strong not likely	onal seg door Oungo 2 ments re ation	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program stutdown Span: execution and prove-of Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional set strong pot likely	onal seg door Outgo 2 gments re ation	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Ref IemporalAspects Initiation: program startup Completion: program startup Completion: program startup Completion: program startup Completion: program stutdown Span: execution and prove-or Discrete / Continuous 1 Image: Products Description: part program segment Likely_Source: workstation mgt Reliability: strong Cont/acts: not likely Products Description: Description: part program request Attributes: program id	equest additional set	onal seg occr Outgo 2 gments re ation	gments of part pro ning Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program stutdown Span: execution and prove-of Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	s. additional seguest interpret	onal seg occr Outgo 2 gments re ation	gments of part pro ming Messages: bing Messages.	ogram Message- Initiated Process: 3 4
Responsibility: NGC-8 Ref Importal Aspects Initiation: program startup Completion: program startup Completion: program startup Completion: program startup Completion: program stutdown Span: execution and prove-or Discrete / Continuous 1 Construction: 1 Construction: 1 Construction: 1 Construction: 2 External Resources: 2 Description: part program segment Likely Source: workstation mgt Beliability: strong Cont/scts: not likely Products Description: Description: part program request Attributes: program id Constraints: Likely Recipientsworkstation mgt	equest additional set	onal seg occr Outgo 2 gments re ation	gments of part pro ning Messages: bing Messages.	An
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program stutdown Span: execution and prove-of Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	strong pot likely	onal seg Coord Coord 2 2 gments re ation	gments of part pro ing Messages: bing Messages. ping Messages. ping Messages. ping Messages.	A constraint of the second sec
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program sturtdown Span: execution and prove-of Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional set	onal seg Congo Oungo 2 gments re ation	gments of part pro ning Messages: bing Messages.	An
Responsibility: NGC-8 Re Temporal Aspects Initiation: program startup Completion: program sturtdown Span: execution and prove-of Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	equest additional set	onal seg Congo Oungo 2 gments re ation	gments of part pro ing Messages: bing Messages. ping Messages. ping Messages. esponsibilities:	An

emporalAspects		Incorr	ning Messages:	Message-
Initiation: part program loaded				Dragoog
Completion: part program started	-	Outor	ing Messages:	1 - 100 - 53.
Span: execution and prove	-out	Ourâr	mig motologiot.	
Discrete / Continuous				
4		2	3	
ocal Resources				
Description:				
Attributes:				
Constraints:				[
Source				
xternal Resources				
Description: cycle start comman	ł			
Likely Source: operator interface				
Reliability: strong				
Contracts: pot likely				
Products				
Description: commands for starti	ha			
Attributes: code: narameters	·			
Constraints.				
Likely Recipientsplan interpretation				
and the second sec		Sub r	esponsibilities:	
Responsibility: NGC-10 Ini	itiate end of p	part pro	gram activities	
Responsibility: NGC-10 Ini	itiate end of p	eart pro	gram activities	Message-
Responsibility: NGC-10 Ini emporalAspects	itiate end of p	ert pro	gram activities n ng kicesagos:	Message- Initiated
Responsibility: NGC-10 Ini emporalAspects Initiation: program startup Completion: program shutdown	itiate end of p	eart pro	gram activities	Message- Initiatec Process:
Responsibility: NGC-10 Ini emporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove	itiate end of p	eart pro	gram activities nong Mossagos: o ng Mossagos:	Message- Initiatec Process:
Responsibility: NGC-10 Ini emporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	itiate end of p	eart pro	gram activities norgil/ossagos: oongil/ossagos:	Message- Initiatec Processi
Responsibility: NGC-10 Ini emporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	itiate end of p	oart pro Incon Outge 2	gram activities nong Monsagos: oong Mossagos: 3	Message- Initiatec Process:
Responsibility: NGC-10 Ini iemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	itiate end of p	eart pro	gram activities h ng Mossagos: o ng Mossagos: 3	Message- Initiatec Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	-out	eart pro	gram activities norgilifessages: origilifessages: 3	Message- Initiatec Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	-out	Part pro	gram activities h rg klossagos: o ng klossagos: 3	Message- Initiatec Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	-out	Part pro	gram activities nong Mossagos: o ng Mossagos: 3	Message- Initiatec Process:
Responsibility: NGC-10 Ini iemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	-out	Part pro	gram activities norgialessages: oingialessages: 3	Messago- Initiatec Process:
Responsibility: NGC-10 Ini iemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous	-out	Part pro	gram activities nong Mossagos: o ng Mossagos: 3	Messago- Initiateo Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	-out	Part pro	gram activities norgialessages: ongialessages: 3 machine model	Messago- Initiateo Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program startup Discrete / Continuous 1 1 1000 Reserves Discrete / Continuous 1 1 1000 Reserves Discrete / Continuous	-out	Part pro	gram activities nong Mossagos: o ng Mossagos: 	Messago- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program startup Completion: program startup Discrete / Continuous 1 coa Reserves Discrete / Continuous	next.command plan_interpreta	Part pro	gram activities h tyliciossagos: o ng Mossagos: 3 machine model koowledge hase strong	Messago- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program startup Completion: program startup Discrete / Continuous 1 coa Russer Discrete / Continuous	next.commany not likely	Part pro	gram activities h typiciossagos: o ng Mossagos: 	Message- Initiated Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 coal Resources Description: part program Likety Source: workstation mgt Reliability: strong ConVicts: not likety Products	next.commany not likely	Part pro	gram activities n	Messago- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 coa Reserve Discrete / Continuous 1 coa Reserve Continuous 1 coa Reserve Continuous 1 coa Reserve Continuous Continu	next.commany not likely	Part pro	gram activities n my lefessages: o ng Messages: 	Messago- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 coa Reserve Discrete / Continuous	-out	Part pro	gram activities n my loossages: o ng Mossages: 	Message- Initiated Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 coa Reserve Discrete / Continuous 2 contraints containts coa Reserve Discrete / Continuous	-out	Part pro	gram activities a rg k/ossagos: a rg k/ossagos: 3 machine model koowledge hase strong not likely	Message- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 coa Resources Discrete / Continuous	itiate end of p	Part pro	gram activities	Message- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 000 Reserve Discrete / Continuous 1 000 Reserve Contraints: 1 000 Reserve Discrete / Continuous 1 000 Reserve Discrete / Continuous 1 000 Reserve Discrete / Continuous 1 0 000 Reserve Discrete / Continuous 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	next.commany not likely	Dart pro	gram activities	Message- Initiated Process:
Responsibility: NGC-10 Ini TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	next.commany not likely	Dart pro	gram activities	Message- Initiatec Process:
Responsibility: NGC-10 Ini imporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove Discrete / Continuous 1 000 Reserve Discrete / Continuous 1 000 Reserve 1 000 Reserv	next step	Dart pro	gram activities	Message- Initiatec Process:

.

Responsibility: NGC-11 D	Determine wh	ere to s	send part program c	odes
TomporalAccorta		Incor	ing Macenne	Messane-
Initiation: program startup		нсон	шіў меррацер.	Initiated
Completion: program shutdown		Autor	View Mannagan	- Process:
Span: execution and prove-or	ut	Uuugu	ning messages.	
Discrete / Continuous		l l		
		-		
		2	3	4
Lccal Bosources				
Description:				
Attributes:	ļ			
Constraints:			1	
Source:	ļ			
External Resources				
Description: part program	machine mod	el		
Likely Source: workstation mgt	knowledge ba	ISB		
Beliability: weak	strong			
Con1/acts: not likely	not likely	·····		
Products	l			
Description: modil/ed part program	I			
Attributes: code: parameters				
Constraints:	ļ			
Likely Recipients plan interpretation	l			
Procedures/Methods • accept progr	am .	Sub-r	esponsibilities:	
Example:	nine model		a dotorming recipient of a	ach
• determine re	cipient of each		component	
aurophica • auroment coo	ie with destination	ns	 augment code with dest 	tinations
• send modi/e	d program			
				······································
Deeneneihiliku				-
Responsibility: NGC-12 Ir	nterpret part j	program	<u> </u>	
TomporalAchard		incon	ing Messages:	Messace-
heitistics:	n ctroom	11.001		Initiated
Initiation: beginning of program	ii stream			Process
Completion: end of program strea	am	Outdo	ing Messages:	1 103020.
Span: execution and prove	-out			
Discrete / Continuous				1
4	1	2	3	4
Ator we are				
	i			
Source	1			
External Hesources				
Description: part program Ve	macbine.	nodel		
LIKELY SOUTCE: workstation manage	ment_knowledg	e base		
Beliability: weak	strong			
CONVICTS: not likely	not likely			
Products	 			
Description: parsed part program	error mes	egsa	mechanism commands	
Attributes: code: parameters	code: par	ameters	code: parameters	
Constraints:				
Likely Recipients plan interpretation	operator i	nterface	mechanisms	
Procedures/Methods		Sub-r	esponsibilities:	
Fxample: • accent program the			• validate onde for mach	ine
			parse codes for motion	n / mechanism
validate code for machin	ne / mechanism		•	
	/ 11100410183111			

-

		Incom	ing Messages	Message-
Initiation: program startup		0 100011	ang moodugee.	Initiated
Completion: program shutdown		~ .:		Process:
Span: execution and prove-	out	Ontão	ing Messages:	
Discrete / Continuous				
1		2		3
Local Besources				
Description:				_
Attributes				
Constraints:				
Source				
External Hesources				
Description: part program codes	recipierii	ion		
Likely Source: plan interpretation	strong	.011		
Contents:	not likely			
Draduato				
Products Description: modil/ed part program	n			
Attributes' onde: parameters				
Constraints:				
Likely Recipientsplan interpretation				
Example: • accept part program C • update codes with rec • send modil/#d code	odes and recipie apients			
example: • accept par program of • update codes with rec • send modiled code Responsibility: NGC-14 Inf	odes and recipie ipients	of par	t program commer	nts
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf	odes and recipie ipients	of par	t program commen	nts Messago-
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup	odes and recipie	of par	t program commen	hts Message- In. atec
Example: •accept par program C •update codes with rec •send modived code Responsibility: NGC-14 Inf TemporalAspects Initiation: program startup Completion: program shutdown	odes and recipie	of part	t program commer hing Massages	nts Mossago- In., atec Process:
Example: •accept par program C •update codes with rec •send modiled code Responsibility: NGC-14 Inf TemporalAspects Initiation: program startup Completion: program shutdown Span: execution and prove-c	odes and recipie ipients	of part	t program commen hing Messages oing Messages.	hts Messago- In. atec Process:
Example: •accept par program c •update codes with rec •send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous	odes and recipies	of part	t program commen ning Messages cing Messages.	nts Messago- h., atec Process:
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous	odes and recipies	of part shoor Outga 2	t program commen ning Messages cing Messages.	nts Messago- In atec Process: 3
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 L cont Elesio, troos	odes and recipies	of part uncorr Outgo 2	t program commen hing Messages ping Messages.	nts Mossago- In., atec Process: 3
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf TemporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Possi Biosources Descrete of	odes and recipie spients	of part uncorr Outgo 2	t program commen hing Messages ong Messages.	nts Mossago- In., atec Process: 3
Example: • accept par program C • update codes with rec • send modiled code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 Const Elesion program Startup Description: 1 Const Elesion program Startup	odes and recipies	of part uncorr Outgo 2	t program commen hing Messages ong Messages.	nts Message- h., atec Process: 3
Example: • accept par program C • update codes with rec • send modil/#d code Responsibility: NGC-14 Inf TemporalAspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 Const Bosocuroes Descention Attracted	odes and recipies	of part	t program commen hing Messages ping Messages.	nts Messago- h., atec Process: 3
Example: • accept par program for • update codes with rec • update codes with rec • send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 • accept par program send modived code	odes and recipies	of part	t program commen hing Messages ong Messages.	nts Mossago- h., atec Process: 3
Example: •accept par program for • update codes with rec • update codes with rec • send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 • cont Ecolo prose Description: Attracted External Resources	orm operator	of part	t program commen	nts Mossago- h., atec Process: 3
Example: •accept par program c •update codes with rec •send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program startup Completion: program startup Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1	part requirement	of part	t program commen	nts Mossago- h., atec Process: 3
Example: • accept par program c • update codes with rec • send modil/ed code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Incel Bosocuroos Description: part program Likely Source: workstation mgt Reliability: starton	part requirement integring	of part	t program commen	nts Messago- In. atea Process: 3
Example: • accept par program c • update codes with rec • send modi/kd code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Post Hose unces Description: part program Likely Source: workstation mgt Reliability: strong Con Verts: part likely	part requirement engineening	of part	t program commen	nts Messago- In. atea Process: 3
Example: • accept par program c • update codes with rec • send modil/ed code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Construction: program shutdown Span: execution and prove-c Discrete / Continuous 1 Construction: program shutdown Span: execution and prove-c Discrete / Continuous 1 Construction: program shutdown Span: execution and prove-c Discrete / Continuous 1 External Resources Description: part program Likely Source: workstation mgt Reliability: strong Con/dcts: not likely Products	part requirement engineening weak not likely	of part	t program commen	nts Messago- In. atec Process: 3
Example: • accept par program Completion: • update codes with rec • update codes with rec • send modi/kd code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program shutdown Span: • execution and prove-c Discrete / Continuous 1 Prodit Bosocuroes Description: part program Likely Source: workstation mgt Beliability: strong Con/dcts: not likely Products Description: comments to operation	part requirement engineening weak not likely	of part	t program commen	nts Messago- In. atea Process: 3
Example: • accept par program C • update codes with rec • send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Products: not likely Products: not likely Products: code: parameters	odes and recipies ipients	of part	t program commen	nts Messago- In. atea Process: 3
Example: • accept par program C • update codes with rec • send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program startup Completion: program shutdown Span: execution and prove-c Discrete / Continuous 1 Accel Bosocuros Description: part program Likely Source: workstation mgt Beliability: strong Convicts: not likely Products Description: comments to operato Attriburies: code; parameters Constraints: pnonty levels	odes and recipies ipients	of part 	t program commen	nts Messago- In. atea Process: 3
Example: • accept par program C • update codes with rec • send modived code Responsibility: NGC-14 Inf Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove- Discrete / Continuous 1 Accel Bosources Description: part program Likety Source: workstation.mgt Beliability: strong Convicts: not likety Products Description: comments to operator Attriburtes: code: parameters Constraints: pronty levels Likety Recipientsoperator interface	part requirement mot likely	of part 	t program commen	nts Messago- In. atec Process: 3

.

Aessage- nitiated Process:
Process: 4
4
4
4
4
· · · · · · · · · · · · · · · · · · ·
tight coordination legree of coord. g
lessage- hitiated
rocess:
4
·
equired

.

• .

_

d

	nterpret augn	nented o	code	
TemporalAspects		Incom	ing Messages:	Message-
Initiation: program startup		1	<i>Q</i>	Initiated
Completion: program shutdown			Las Managana	-Process:
Span: execution and prove	-out	I Ouigu	ning wessages.	
Discrete / Continuous				
		2		34
Local Rosources				
Bescription:				
Constraints				
External Besources	1			
Description: augmented program	status report:	5	machine model	
Likely Source: plan interpretation	motion/mech	anism	knowledge base	
Beliability: strong	strong		strong	
Con Victs not likely	not likely		not likely	
Products	l			
Description: coordination comma	nds			_
Annunes: code: parameters				
Likely Recipients motion/mechanism	1			
Responsibility: NGC-18 De Temporal Aspects	etermine wai	t/continu	ue commands for https://essages.	coordination
<u>Completion: program shutdown</u>		0.00	ine Messages	- Process:
Span: execution and prove- Discrete / Continuous				1
1		2		3
	1			
			······	
				· · · · · · · · · · · · · · · · · · ·
Extend Besources				
External Resources Description: program status	part program		machipe model	
External Resources Description: program status Likely Source: plan interpretation	part program plan interpreta	100n	machine model knowledge base	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong	part program plan interpreta strong	tion	machine.model knowledge.base strong	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong Con Victs: not likely	part program plan interpreta strong not likely		machine model knowledge base strong not likely	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong ConVicts: not likely Products:	part program plan interpreta strong not likely	1.tion	machine model knowledge base strong not likely	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong ConVicts: not likely Products Description: coordination comman Attributes: code: narameters	part program plan interpreta strong not likely	1.bon	machine model knowledge base strong not likely	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong Convacts: not likely Products Description: coordination comman Attributes: code: parameters Constraints:	part program plan interpreta strong not likely		machine model knowledge base strong not likely	
External Resources Description: program status Likely Source: plan interpretation Beliability: strong ConVicts: not likely Products Description: coordination comman Attributes: code: parameters Constraints: Likely Recipientsmotion/mechanism	part program plan interpreta strong not likely	ition	machipe.model koowledge.base strong not.likely	

Responsibility: NGC-19 S	wap current	tool wit	h required tool	
TemporalAspects		Incom	ning Messages:	Message- Initiated
Completion: tool change accompl	ished			- Process:
Span: execution and prove	-out	Outgo	ing Messages:	
Discrete / Continuous				
1		2	3	4
Lecal Besources				
Arribotom				
Gonstraints:				
Source:				
External Besources				
Description: current tool id	next tool id	ation	swap command	
Reliability: strong	strong		strong	
ConVicts: not likely	not likely		not likely	
Products				
Description: swap commands				-
Announes: code: parameters				
Likely Recipientstool mechanism				
Procedures/Methods Example: • determine location of • track location of curre • send swap command	tool ids next tool ent tool Is to mechanism	Sub-r	esponsibilities:	
Responsibility: NGC-20 M	ap tool id to a	actual to	ool location	
TemporalAspects		lincom	ning Messages:	Message-
Initiation: system startup		1		Initiated
Completion: system shutdown		Girar	ine Vessages	- Process:
Span: execution and prove- Discrete / Continuous	out	l Conge		
4		2		4
Loca Bosourzos				
	 			
Saurea				
External Resources				_ <u>_</u>
Description: required tool id	tool model			
Likely Source: plan interpretation	strong	158		
Convicts: not likely	not likely			
Products				
Description:tool location				
Attributes:	·}			
Likely Recipients and changer				
Procedures/Methods Example: • accept required tool id • access tool model • determine tool location		Sub-r	esponsibilities:	

•

TemporalAspects		Incon	ning Messages:	Message-
Initiation: system startup			~ ~ ~	Initiated
Completion: system shutdown		<u></u>		-Process:
Span: setup		Ouigu	my wessayes.	
Discrete / Continuous				
		0		
· · · · · · · · · · · · · · · · · · ·		<u> </u>		3
<u>Department</u>				
Atributes				
Constraints:				
Source:				
External Besources				
Description: tools required	tools supplied		tool model	
Likely Source: factory scheduler	operator inter	face	knowledge base	
Reliability: strong	weak		strong	
ConVicts: not likely	not likely		not likely	
Products				
Description: modiled tool model				
Attributes:				
Constraints:	ł			
Likely Recipients:knowledge base	f		1	1
e access tool model update model with ne send modi//ed tool model	w tool data odel			
eccess tool model update model with ne send modil/led tool model	w tool data odel ack tool locati	ions		
	ack tool locat	ions	nina Messaces:	Nessage-
	ack tool locat	ions Incorr	ning Messages:	Message-
	ack tool locat	ions Incorr	ning Messages:	Message- Initiated Process:
	ack tool locat	ions Incorr Outga	ning Messages: bing Messages:	Message- Initiated Process:
	ack tool locati	ions Incorr Outgo	ning Messages: bing Messages:	Message- Initiated Process:
	ack tool locat	ions Incorr Outgo	ning Messages: bing Messages:	Message- Initiated Process:
	ack tool locat	ions Incom Outga 2	ning Messages: bing Messages:	Message- Initiated Process: 3
e access tool model e update model with ne e send modil/led tool mod responsibility: NGC-22 Tra Tra TemporalAspects Initiation: system startup Completion: system startup Completion: system startup Discrete / Continuous 1	ack tool locat	ions Incorr Outga 2	ning Messages: bing Messages:	Message- Initiated Process:
e access tool model e update model with ne e send model wi	ack tool locati	ions Incom Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
eacess tool model eupdate model with ne e send model with	ack tool locati	ions Incom Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
e access tool model e update model with ne e send model wi	ack tool locati	ions Incom Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages:	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: next tools location knowledge base strong not likely	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2	ning Messages: bing Messages: 	Message- Initiated Process: 3
	ack tool locati	ions Incorr Outgo 2 ion Sub-r	ning Messages: bing Messages: next tools location knowledge base strong not likely.	Message- Initiated Process: 3

٠

~

Responsibili	ty: NGC-23 S	wap current	part wil	h next part	
TemporalAspe	octe	·····	Incon	ninn Messanes	Message-
Initiation	nart %turing mecha	nism started	100-00	and woordgoor	Initiated
	part 1/during mecha	niem sturted			Process
Completion:	part muning mecha	nisin shut down	Outa	bing Messages:	1.100000
Span:	program execution				
Discrete / Cor	ntinuous	<u></u>			1
			2	3	4
Local Resour	ces				
<u>Attributes:</u>					
Constraints:		1			
Sourcer					
External Reso	ources				
Description:	part 1/xturing comma	hds machine r	nodel		
Likely Source	 mechanism control 	knowledg	e base		
Beliability:	strong	strong			
Con ¹ /cts	not likely	not likely			
Products					
Description:	command ack	actuator s	ionals		
Attributos	code: parameter	voitage			
Constraints:	success of executio	h device tvr	es		
Likely Recinie	ntemechanism contro	actuators			
Example:	 convert to actuato send signals send command action 	knowledgment	 -		
Responsibili	ty: NGC-24 Id	lentify curren	t pallet		
TemporalAspe	cts		Incon	ning Messages:	Message-
Initiation:	pallet %turing mecha	anism started			Initiated
Completion:	pallet %turing mecha	anism shut dow		ing Magagana	- Process:
Span:	execution		Cuige	ong wessages.	
Discrete / Cor	ntinuous		•		
	1		2	3	4
Loon Resource	105				
<u>Description:</u>					
<u>amenderation</u>					
<u>Constraints</u>					
Enterna -1 D					
External Heso	urces				
Description:	sensor data				
Description:	sensor data sensor interface				
Description: Likely Source Beliability:	sensor data sensor interface weak				
External Heso Description: Likely Source Beliability: ConVicts:	sensor data sensor interface weak not likely				

•

Description: Attributes:

Procedures/Methods Example:

pallet id

Constraints: Likely Recipientsworkstation mgt

code: parameters

accept sensor data
identify pallet
send pallet id

Sub-responsibilities:

-

1

÷.

TemporalAspec Initiation:	ts start of part program)	Incom	ing Messages	:	Message- Initiated
Span: Discrete / Conti	execution and prove	-out	Outgo	ing Messages		
	4		0		2	
Local Resource	\$+					
Description:	-					
<u>Attrinutes</u>						
Source:						
External Resour	rces					
Description:	command rate	mechanism co	mmands			
Likely Source:	motion	plan interpreta	tion			
Beliability:	strong	strong				
Con1/scts	not likely	not likely				
Products	monhaniam comme					
	code: parameters					
Constraints	altered for smooth 14	w				
Likely Recipien	ts:mechanism					
Example: • acc • ana • mo • ser	cept command rate, c alysis for continuous 3 dfly yew according to ad modil/éation comma	ommands winformation rate ands	• ana	alysis for continuous	; %əw	information
Responsibility	ept command rate, c alysis for continuous 3 dify J&w according to ad modi leation commi- Ministry (NGC-26 Cc to	ommands dwintormation rate ands ontrol acceler	• ana	alysis for continuous	5 %9 w	information
Responsibility	tept command rate, c alysis for continuous 3 dify yew according to a modivention commi Minodivention commi Minodivention commi Minodivention Minodi Minodivention Minodivention Minodi M	ommands winformation rate ands ontrol acceler	•ana ration ai incom	nd deceleration	; %	Messago-
Responsibility Iemporal Aspec Initiation:	tept command rate, c alysis for continuous 1 dify Jew according to dimodil/tation commi NGC-26 Cc ts	ommands winformation rate ands ontrol acceler	• ana	nd deceleration	; %	Messago- Initiated Process:
Responsibility Temporal Aspec Initiation: r Span: Cont	ept command rate, c alysis for continuous 3 dfy J&w according to ad modi leation commi- d modi leation commi- d modi leation commi- station commi- ts notion started notion started notion stopped execution and prove-co- inuous	ommands dwintormation rate ands ontrol acceler	•ana ration at incom Outgo	nd deceleration Ing Messages Ping Messages	: Xow	Messago- Initiated Process:
Responsibility Iemporal Aspec Initiation: r Completion: r Span: c Discrete / Cont	tept command rate, c alysis for continuous 3 dify Jew according to ad modil/teation commi NGC-26 Cc ts notion started notion started notion stopped execution and prove-c inuous	ommands winformation rate ands ontrol acceler out	•ana ration a incom Outgo	nd deceleration and Messages bing Messages		Messago- hitiated Process:
Responsibility Temporal Aspec Initiation: r Completion: r Span: e Discrete / Cont	ept command rate, c alysis for continuous 1 diffy We according to ad modil/eation commi- Microsoft NGC-26 Cc ts notion started notion started notion started execution and prove-c inuous	ommands winformation rate ands ontrol acceler put	•ana ration al incom Outgo 2	nd deceleration ang Messages ang Messages	; /øw	Messago- Initiated Process:
Responsibility Temporal Aspect Initiation: r Completion: r Discrete / Cont	ept command rate, c alysis for continuous 1 dify J&w according to ad modil/tation commi- motion started notion started notion started notion stopped execution and prove-co inuous 1	ommands wintormation rate ands ontrol acceler out	•ana ration al incom Outgo 2	nd deceleration ang Messages ping Messages	3.	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Span: e Discrete / Cont	ept command rate, c alysis for continuous 1 dify J&w according to ad modil/tation commi- motion started notion started notion started notion stopped execution and prove-c inuous 1	ommands wintormation rate ands ontrol acceler out	•ana ration at incom Outgo 2	nd deceleration ang Messages bing Messages	3.	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Discrete / Cont	Pept command rate, c alysis for continuous 3 dify Mew according to ad modil/tation commi- motion started notion started notion started notion stopped execution and prove-c inuous 1	ommands dwintormation rate ands ontrol acceler out	•ana ration at incom Outgo 2	Ind deceleration	3	Messago- hitiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspect Initiation: r Completion: r Span: e Discrete / Cont	ept command rate, c alysis for continuous 1 dify V& according to ad modi vation commi- motion started notion started notion started notion stopped execution and prove-c inuous 1	ommands dwintormation rate ands ontrol acceler out	•ana ration at incorr Outgo 2	nd deceleration ang Messages bing Messages	3 	Messago- hitiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Span: e Discrete / Cont Local Resource Description: Atmoutes Constraints: Sei rec: External Resource	rces	ommands winformation rate ands ontrol acceler out	• ana ration at incom Outgo 2	nd deceleration aing Messages bing Messages	3 	Messago- hitiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Span: ¢ Discrete / Cont Discrete / Cont Local Resource Description: c Saurce: External Resource	rces	ommands wintormation rate ands ontrol acceler out 	• ana ration al incom Outgo 2	nd deceleration ang Messages bing Messages	3	Messago- hitiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspect Initiation: r Completion: r Span: e Discrete / Cont Local Resource Description: c Likely Source: f	rces rces	ommands wintormation rate ands ontrol acceler out out axis step limit knowledge ba	• ana ration al incom Outgo 2	nd deceleration ang Messages bing Messages	3	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Span: e Discrete / Cont	rept command rate, c alysis for continuous to alysis for continuous to aly was according to aly was according to all modily attended to the second of the motion started motion started mo	ommands wintormation rate ands ontrol acceler out out axis step limit knowledge ba strong pot lipot:	• ana ration al incom Outgo 2	alysis for continuous and deceleration aing Messages bing Messages	3 3	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspect Initiation: r Completion: r Span: e Discrete / Cont	rept command rate, c alysis for continuous to alysis for continuous to alysis for continuous to alysis for continuous to alysis and continuous to motion started notion sta	ommands wintormation rate ands ontrol acceler out out axis step limit knowledge ba strong not likely	• ana ration and incom Outgo 2 s sse	Ind deceleration and deceleration and Messages bing Messages	3 3	Messago- Initiated Process:
Example: • acc • and • mo • ser Responsibility Temporal Aspect Initiation: r Completion: r Span: e Discrete / Cont	rces	ommands wintormation rate ands ontrol acceler out out axis step limit knowledge ba strong not likely	• ana ration at incom Outgo 2 s sse	Ind deceleration and deceleration and Messages bing Messages		Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspect Initiation: r Completion: r Discrete / Cont Discrete / Cont	rces rces	ommands wintormation rate ands ontrol acceler out out axis step limit knowledge ba strong not likely	• ana ration at incom Outgo 2 s sse	Ind deceleration and deceleration and Messages bing Messages	3 3	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Iemporal Aspect Initiation: r Completion: r Span: e Discrete / Cont Discrete / Cont Discrete / Cont Constraints: Ser rec: External Resource Description: c Likely Source: t Reliability: s Convicts: r Products Description: c	rces rces	ommands wintormation rate ands ontrol acceler out axis step limit knowledge ba strong not likely	• ana ration at incorr Outgo 2 S	Ind deceleration and Messages bing Messages	3 3	Messago- Initiated Process:
Example: • acc • ana • mo • ser Responsibility Temporal Aspec Initiation: r Completion: r Span: e Discrete / Cont Discrete / Cont Discrete / Cont Lised Resource Description: c Likely Source: t Reliability: s Cont/acts: r Products Description: c Attributes: Constraints: Likely Recipien	rces rces rces rces raiectory generator strajectory generator strajectory generator	ommands winformation rate ands ontrol acceler out axis step limit knowledge ba strong not likely	• ana ration at incom Outgo 2	Ind deceleration and deceleration and Messages bing Messages		Messago- Initiated Process:

.

nesponsibility. NGC-27 De			·	-	Managan
Temporal Aspects		Incom	ing Messages:		Initiated
Completion: motion stopped			···· » • • • • • • • • • •		Process:
Span: execution and prove-o	out	Outge	ing wessages:		·
Discrete / Continuous					
1		2		31	
Local Resources					
Description:					
<u>Attributos:</u>					
<u>Constraints:</u>					
Source:				-+	
External Resources					
Description: trajectory					
Likely Source: trajectory generator					
Heliability: strong					
Convects not likely				\neg	
Products					······································
Description: normals					
			······································		
Likoly Pociniontermienter concerter					
		0			
Procedures/Methods		Sub-r	esponsibilities:		
Example: • accept trajectory					
calculate normals					
• calculate normals • send normals					
• calculate normals • send normals	rmalize traie	ectory w	ith respect to time	e	
• calculate normals • send normals Responsibility: NGC-28 No Temporal Aspects Initiation: motion started	rmalize traje	ectory w	ith respect to tim	e	Message- Initated
	rmalize traje		ith respect to tim	e	Message- Initated Process:
	rmalize traje	ectory w Incom Outgo	ith respect to tim ling Messages: bing Messages:	e	Message- Initated Process:
	rmalize traje	ctory w incom Outgo	i th respect to tim hing Messages: bing Messages:	e	Message- Initated Process:
	rmalize traje	ectory w Incom Outgo	i th respect to tim ling Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje ut	ectory w Incom Outgo 2	i th respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w Incom Outgo 2	i th respect to tim hing Messages: bing Messages:	e 3	Message- Init.ated Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Ourtgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Ourtgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 	Message- Init.atec Process:
	rmalize traje	ectory w incorr Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.ated Process:
	rmalize traje	ectory w incorr Outgo 2 bel ase	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:
	rmalize traje	ectory w incom Outgo 2 Jel ase	ith respect to tim hing Messages: bing Messages:	e 3	Message- Init.atec Process:

Responsibilit	Y: NGC-29 Tr	anslate moti	on com	mands to trajectory	
TemporalAspe	cts start of part program	· · · · · · · · · · · · · · · · · · ·	Incom	ning Messages:	Message- Initiated
Completion:	program completed		0.000	Jan Mannagar	Process:
Span:	execution and prove-	out	l Onrâc	ning messages.	
Discrete / Con	itinuous				
	1		2	3	4
Local Besourc	ē\$				
Attributes					
Constraints:					
Shime					
External Reso	urces				
Description:	motion command	cutter offs	sets	position corrections	rate corrections
Likely Source	plan interpretation	motion		motion	
Beliability:	strong	weak		weak	weak
Con1/acts	not likely	not likely		not likely	
Products					
Description:	trajectory information				
Attributes:	setpoints: parameter	β			
Constraints:	per coordinate trame				
Likely Recipie	ntsactuator control	I			
Procedures/Me Example:	ethods. accept progra analysis for d access corre o determine sei send trajecto	am lestination inforr ctions tpoints and para ry	Sub-r nation ameters	 esponsibilities: analysis for destination determine setpoints an 	information d parameters

Responsibility: NGC-30 Co	ordinate with	n mecha	nisms	
Temporal Aspects Initiation: program started		hrcom	ing Messages:	Message- Initiated
Completion: program shutdown Span: execution and prove-or Discrete / Continuous	ut	Outgo	ing Messages:	
1		2	3	4
50.000				+
External Resources Description: augmented code Likely Source: plan interpretation Reliability: strong Con Victs: not likely	continue comm plan interpretat strong not likely	hand		
Products				
Description:wait indicator Attributes:code: parameter Constraints: Likely Recipients:plan interpretation				
Procedures/Methods Example: • accept augmented cod • execute codes • if code is to wait, send • if waiting, accept contin	le wait indicator vue command	Sub-r	esponsibilities:	

Г				
Responsibility: NGC-31 A	woid data sta	arvation	l	
TemporalAspects		incon	nino Messades:	Message-
Initiation: start of part program		1		Initiated
Completion: program completed	······	T		Process:
Span: execution and provo-or	uit .	† Outge	oing Messages:	
Discrete / Continuous		† [°]		
	1		1	<u>í</u>
1		2	3	4
Lacal Resources				
Description:				
Attributes	ļ			
Constraints:	ļ			
Source:				
External Resources				
Description: motion commands	machine mo	del		
Likely Sourceplan interpretation	knowledge t	Dase		
Beliability: strong	strong			-
Due due te	not likely			
	<u> </u>			
Attributoe: code: parameters			[
Constraints: time (memory limitation	e			
Likely Recipients plan interpretation	,			
	a	Cub	eenensibilities:	
Procedures/Methods		Sub-r	esponsibilities.	
• accept motion comma • analysis for look-ahea	d information	•	analysis for look-anead i	niormation
 send look-ahead resp 	onse to control			
rate of command tr	anster	L		
ſ <u>·····</u>				
Responsibility: NGC-32 De	etermine rate	of mov	rement	
				Videeearo.
lemporalAspects		1.001	೫ ೮ . ನಿರತಿಶಿಷಕ್ಷಿತಶಿ	Linic seage
Initiation: motion started		<u> </u>		- Procasa
Completion: motion stopped		Quice	oind Massades:	- r 0.0000.
Span: execution and prove-c	but	1		
Discrete / Continuous	r			<u> </u>
1		2	3	4
Forn Persoutens				
Description				
<u>Amo (196)</u>	l			
Constraints				
<u>Sec. (co)</u>				
External Resources				L
Description: rate commands	spindle speed		tool.model	
Likely Source: trajectory generator	sensor interfac	;e	knowledge_base	
Reliability: strong	weak		strong	
ConVicts: not likely	not likely		pot likely	
	1		1 [.]	1
Products				
Products Description:				
Products Description:				
Products Description:				
Products Description:goaLrate Attributes: Constraints: Likely Recipients:trajectory generate				
Products Description:goaLrate Attributes: Constraints: Likely Recipients:trajectory generate Procedures/Methods)r	Sub-r	esponsibilities:	
Products Description:goaLrate Attributes: Constraints: Likely Recipients:trajectory generate Procedures/Methods Example: • accept rate commands	ar	Sub-r	esponsibilities:	
Products Description:goaLrate Attributes: Constraints: Likely Recipients:trajectory generate Procedures/Methods Example: • accept rate commands • access spindle speed, • determine anorganate	pr tool model roal rate	Sub-r	esponsibilities:	

.

1

.

Responsibility: NGC-33 Ad	djust trajecto	ry for to	ol deviations	
Temporal Aspects		Incom	ing Messages:	Message- Initiated
Completion: start of part program				Process:
Span: execution and provo of		Outgo	ing Messages:	
Discrete / Continuous	<u>Jui</u>	•		
Discrete / Continuous				
		2	3	4
Local Pascuras				
Bescrintton				
Attributes				
Constraints:				
Source				
External Resources				
Description: pormals to trajectory	tool offset	amounts	plane selection	
Likely Source: trajectory generation	knowledge	base	plan interpretation	
Beliability strong	strong		stroog	
Con1/acts: not likely	not likely		not likely	
Products				
Description: autter offsets				
Attributes: values				
Constraints: tool offset accuracy				
Likely Recipientsmotion				<u></u>
oetermine cutter send new trajecto	compensation ory			
Responsibility: NGC-34 D	etermine traj	ectory	corrections from pre	dictable variations
Temporal Aspects		noon	nng Messages:	Message-
Initiation: machine startup				Dropener
Completion: machine shutdown		Class	ting Mossages:	1 F. UUGSS.
Span: execution and prove-	out			
Discrete / Continuous		l	Y	
		2	2	4
External Recourses				
Description: themai data	temporal o	lata	tables, polynomials	position
Likely Source: sensor interface	operating	system	knowledge base	sensor interface
Reliability weak	strong		strong	weak
Contácts: likely	not liketv		not likely	not likely
Products				
Description: trajectory corrections				
Attributes: code parameters				
Constraints:				
Likely Recipientstrajectory generation	h			
Procedures/Methods. access mach Example: • access temp • access them • determine ne	ine model oral data hal data ocessary correct	Sub-i	responsibilities:	

6

۴

.

Responsibility: NGC-25 D	etermine trai	ectory	corrections from ser	sed variations
Temporal Aspects	olerance	incom	ing Messages:	Message- Initiated
Completion: sensor data with teler	2000 2000	•		Process:
Completion. sensol data with toler		Outgo	ing Messages:	
Disercto / Continuous	Jui		-	
Discrete / Continuous	[]			
1		2	3	4
Local Basauroas				
Description				
Attributos:				
Constraints:				
Source:				
External Resources				
Description: sensor data (force, to	rque)normals to	trajector	sensor tolerances	sensor ranges
Likely Source: sensor interface	trajectory_c	eneratio	plan interpretation	knowledge base
Beliability: weak	strong		strong	stiony
Con1/acts likely	not likely_		not likely	
Products				anindle correction
Description: error message	position com			<u>somole contenaion</u>
Affribules code: parameters		andant	machine denendent	machine dependent
Constraints: bad path / had sensor	F machine.der	nomion	trajectory generation	trajectory generation
Example: • compare sen tolerance • send correcti	onitor sensors isor data agains s rection on	Sub-It	continually monitor sen calculate correction	sors
Responsibility: NGC-36 A	djust trajecto	ory with	corrections general	ed by sensor data
TemporalAspects		Incon	iing Messages:	Message-
Initiation: start of part program				Disaleu
Completion: program completed		Outer	ing Messages:	Trubess.
Span: execution and prove	-out	Caugo		
Discrete / Continuous				! r
1		2	2	4
The second second	1			
	1			
Source				
External Besources				
Description: trajectory	sensed co	rrections		
Likely Source: trajectory generation				
Reliability: strong	strong			
ConVicts: not likely	not likely_			
Products	Į			
Description:modiled_trajectories	<u> </u>		-	
Attributes:setpoints: parameter	*			
Constraints: non-permanent prog	am_change			
Likely HecipieritSactuator control	L	Suber	esponsibilities.	
Frocedures/Methods		Sub-r	eshousinges.	
• accept trajectory • accept sensed corr • modify trajectory as • send new trajectory	ections s per tolerances			

4

emporalAspec	ts start of part program		Incom	ing Messages:	Message- Initiated	
Completion:	program completed				Process:	
Span:	execution and prove-	out	Outgo	ing Messages:		
Discrete / Conti	nuous					
	1		2		4	4
Local Resource	Ş					
Description:						
Attributes						
Constraints:						
Saura						
External Resou	rces					
Description:	servo command pos	tions kinematic	equations	feedback		
Likely_Source:	trajectory generation	knowledg	e base	servo interface		
Beliability:	strong	strong		weak		
Con%cts	not likely	not likely		not likely		
Products						
Description:	movement command	s following	error			
Attributes:	voltage	value				
Constraints:	accuracy of models	accuracy	of encode	rs		·····
Likely Recipien	ts axis drives	machine i	nodel			
Example:	interpolate between and setpoint convert to command send voltage	current position voltage		interpolate between convert to command	current position and s voltage	etpoir
Example:	interpolate between (and setpoint convert to command send voltage	current position		interpolate between convert to command	current position and s voltage	etpoin
Example: Responsibility	interpolate between (and setpoint convert to command send voltage /: NGC-38 C	current position voltage Control motor	rs to pro	interpolate between convert to command	current position and s voltage	
Example: Responsibility Temporal Aspec	interpolate between (and setpoint convert to command send voltage /: NGC-38 C	current position voltage Control motor	s to pro	interpolate between convert to command duce movement mng Messages:	voltage Nessago- Initiated	
Example: Responsibility Temporal Aspect Initiation:	interpolate between (and setpoint convert to command send voltage /: NGC-38 C tts receipt of voltage co feedback provided	current position voltage Control motor mmands	s to pro	interpolate between convert to command duce movement mng Messages:	Messago- Initiated Procass:	
Example: Responsibility Temporal Aspec Initiation: Completion: Snan:	interpolate between (and setpoint convert to command send voltage NGC-38 C ts receipt of voltage co feedback provided execution and prove	current position voltage Control motor mmands	s to pro		Messago- Initiatec Process:	
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous	current position voltage Control motor mmands	s to pro	 interpolate between i convert to command duce movement aing Messages: bing Messages. 	Messago- Initiated Process:	
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous 1	current position voltage Control motor mmands out	s to pro	 interpolate between i convert to command duce movement ung Messages: bing Messages: 	Messago- Initiated Process:	etpoin
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command convert to command converto command conver	current position voltage Control motor mmands	s to pro	• interpolate between • convert to command duce movement hing Messages: bing Messages.	Messago- Initiated Process:	etpoin
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous 1	current position	s to pro	• interpolate between • convert to command duce movement aing Messages: bing Messages.	Messago- Initiatec Process:	
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous 1	current position	s to pro	• interpolate between • convert to command duce movement aing Messages: bing Messages.	Messago- Initiatec Process:	
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous 1	current position	s to pro	• interpolate between i • convert to command duce movement aing Messages: bing Messages.	Messago- Initiatec Process:	etpoir
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between and setpoints interpolate between and setpoint convert to command send voltage NGC-38 receipt of voltage co feedback provided execution and prove inuous	current position	s to pro	• interpolate between i • convert to command duce movement hing Messages: bing Messages.	Wessago- Initiated Process:	etpoir
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co feedback provided execution and prove inuous 1	current position	s to pro	• interpolate between i • convert to command duce movement hing Messages: bing Messages.	Messago- Initiatec Process:	etpoir
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between (and setpoint convert to command send voltage /: NGC-38 /: NGC-38 /: NGC-38 /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: /: NGC-38 /: /: /: /: NGC-38 /: /:	current position	s to pro	 interpolate between i convert to command duce movement aung Messages: bing Messages:	Messago- Initiatec Process:	etpoir
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont	interpolate between and setpoints interpolate between and setpoint convert to command send voltage /: NGC-38 /: NGC-	current position	s to pro	 interpolate between i convert to command duce movement ang Messages: bing Messages: 	Messago- Initiatec Process:	4
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Description: Likely Source: Reliability:	interpolate between and setpoints interpolate between and setpoint convert to command send voltage /: NGC-38 /: NGC-	current position	s to pro	 interpolate between i convert to command duce movement ang Messages: bing Messages: 	Messago- Initiatec Process:	4
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Description: Likely Source: Reliability: Con ¹ /2cts:	interpolate between (and setpoint convert to command send voltage /: NGC-38 /: NGC-38 /: NGC-38 /: NGC-38 /: NGC-38 /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: /: NGC-38 /: /: /: NGC-38 /: /: /: /: NGC-38 /: /: /: /: NGC-38 /: /: /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: /: NGC-38 /: /: NGC-38 /: /: NGC-38 /: /: /: /: NGC-38 /: /: /: /: NGC-38 /: /: /: /: /: NGC-38 /: /:	current position	s to pro	 interpolate between i convert to command duce movement ang Messages: bing Messages: 	Messago- Initiatec Process:	4
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Beliability: Con Vacts: Products	interpolate between and setpoints interpolate between and setpoint convert to command send voltage /: NGC-38 C tts receipt of voltage co teedback provided execution and prove inuous 1 Inces movement comman actuator control strong not likely	current position	s to pro	interpolate between i convert to command duce movement ang Messages: bing Messages:	Messago- Initiatec Procass:	4
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Beliability: ConVacts: Products Description:	interpolities interpolities interpolities convert to command send voltage /: NGC-38 C ts receipt of voltage co teedback provided execution and prove inuous	current position	s to pro	 interpolate between i convert to command duce movement ung Messages: bing Messages: 	Messago- Initiatec Procass:	4
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Beliability: ConVacts: Products Description: Attributes:	interpolate between (and setpoint convert to command convert to command convert to command receipt of voltage receipt of voltage co feedback provided execution and prove inuous 1 	current position	s to pro	 interpolate between i convert to command duce movement ung Messages: bing Messages: 	Messago- Initiatec Procass:	
Example: Responsibility Temporal Aspect Initiation: Completion: Span: Discrete / Cont Discrete / Cont External Resource: Reliability: ConVacts: Products Description: Attributes: Constraints:	interpolate between (and setpoint convert to command send voltage /: NGC-38 C ts receipt of voltage co teedback provided execution and prove inuous //	current position	s to pro	 interpolate between i convert to command duce movement ang Messages: bing Messages: 	Messago- Initiatec Procass:	

Procedures/Methods accept movement commands Example: • send voltage • accept leadscrew feedback • convert for actuator control • send feedback

September 23, 1994

9

•
Responsibility: NGC-39 II	nitiate startup	proced	lures	
TemporalAspects		Incom	ing Messages:	Message-
Initiation: system power on		I	S	Initiated
Completion: startup completed		I		- Process:
Span: prior to all operations	s	Ontão	ing Messages:	
Discrete / Continuous		t		
	T	0		A
		2	3	4
Attributes	-			
Gonstraints:				
Source:				
External Resources	1			
Description: reference setpoints	machine more	del		
Likely Source: operator interface	knowiedae b	ase		
Reliability: strong	strong			
ConVicts: not likely	not likely			L
Products				
Description: diagnostic results	axes coordin	ate zero i	ositions	
Attributes: codes: parameters	codes: parar	neters		
Constraints: device types	axes			
Likely Recipientsop if / maintenance	a trajectory ge	neration		
• determine a • send axes z	xes zero position ero positions	s		
Responsibility: NGC-40 C	Coordinate wi	th facto	ry scheduler/contro	ol system
TemporalAspects		com	ning Messages.	i.lessage-
Initiation: system start up		1		init ated
Completion: system shut down		0.00	no Mossagas:	- Process:
Span: no program executir	ŋ		,	
Discrete / Continuous		1		1
1		2	3	4
a a constant				
Najar stat				
So				
External Resources				
Description: inhinstructions	machine_mod	tel	heartbeats	
	• • • • •	ase	i all componente	-
Likely Source: tactory system	knowledge b	M WW	all components	
Likely_Source:_tactory_system Beliability:strong	knowledge b strong	M VV	strong	
Likely_Source:_factory_system Beliability:strong Con1/acts:ot_likely	knowledge b strong not likely	4vv	strong not likely	
Likely_Source: factory system Beliability: strong Con1/cts: not likely Products	knowledge b strong not likely		strong not likely	
Likely Source: factory system Beliability: strong ConVicts: not likely Products Description: health information	knowledge b strong not likely job_list		tool list	job_status
Likely Source: factory system Beliability: strong Con%cts: not likely Products Description: health information Attributes: code; parameters	knowledge b strong not likely job list id, pallet/part	id. offsets	tool list	job_status _code; parameters
Likely Source: factory system Beliability: strong ConVicts: not likely Products Description: health information Attributes: code: parameters Constraints;	knowjedge b strong not likety job_list id, pallet/part	id. offsets	tool list	job_statuscode; parameters
Likely Source: Jactory system Beliability: strong ConVicts: not likely Products Description: health information Attributes: code; parameters Constraints: Likely Recipientstactory system	knowjedge b strong not likely job_list id, pallet/part workstation r	id. offsets	tool list tool id_location workstation mgt	job_status _code; parameters factory system

4

Ì

Responsibility: NGC-41 Pa	ath planning	based o	on part features, su	face model, etc.
Temporal Aspects Initiation: beginning of model-ba	ased program	Incom	ing Messages:	Message- Initiated
Completion: program completed Span: execution and prove-or Discrete / Continuous	out	Outgo	ing Messages:	Process:
1		2	3	4
Local Resources			-	,
Attributae				
<u>- 00 37 63 63</u>				
External Resources	machine n	nodel	stock description	
Likely Source: workstation mgt	knowledge	a base	knowledge base	
ConVicts: not likely	not likely		not likely	
Products Description: path plan				
Constraints: machine dependent				
Procedures/Methods • accept part de Example: • access machi stock desc • determine pat • send path	escription ine model, pription th for motion	Sub-ro	esponsibilities: • determine path for mot	ion given models
				<u></u>

NGC-42 MC	pan pro	ogram b	ased on sensor da	ta
Temporal Aspects Initiation: sensor data beyond tol	lerance	.ncon	ing Messages:	Message- Initiated
Completion: sensor data within toler	rance	~		Process:
Span: execution and prove-or	ut l	Clitgo	ing Messages:	
Discrete / Continuous				ł
1		2	3	4
an Loss and				
1				
External Besources				
Description: sensor data	Dart program	m]	sensor tolerances	sensor rannes
Likely Source: sensor interface	workstation	_mm	plan interpretation	knowledge base
Beliability: weak	strong		stroon	strong
Convicts: likely	not likely		not likely	not likely
Products				
Description: part program	ATTOLIDASS	a0e]		Į
Attributes: blocks of codes	code: paran	neters		<u></u>
Constraints: machine dependent	bad program	m / bad s	ensors	
Likely Recipientsworkstation planning	operator int	ertace		
Procedures/Methods accept program continually more Example: • compare sense tolerances • adjust part prov • send new part	m nitor sensors or data against gram program	Sub-ri	esponsibilities: • continually monitor sen: • adjust part program	sors

.

Responsibility: NGC-43 S	Switch betwee	en posit	ion control to force	control
TemporalAspects		Incon	ning Messages:	Message-
Initiation: start of part program			and thorse and soot	Initiated
Completion: program completed		F	· • • • ·	Process:
Span: execution and prove-	out	Outgo	ung Messages:	
Discrete / Continuous		t		
	1	<u>.</u>		T
11		2	3	4
Local Resources	l			
Description:				
Attrinutes:				
	. [
External Hesources		- 1	not model	aument perities 0 force
Description: switch command	_ machine mode	∋i	knowlodge base	current position & force
Likely Source: Irajectory generation	L KIIOWIECIGE ba	se	strong	weak
Contácto: actilization		••••••	not likely	not likely
Draduata				
Description: activate command		nmand		
Attributes: ode: parameters	Code, parame	ters		
Constraints:	A CONTRACTOR OF THE PERSON OF	xw1.w		l
Likely Recipientenotion	motion			
Dressdurge/Mathede		Cub -	eenensihilitiee:	
• intermediate conversion be • send deactivate to position • send activate to force/position	etween control v/force compone tion component	nt		
Responsibility: NGC-44 P	lan for multi-	axis inte	eractions	
TemporalAspects		moon	are Mossages:	iliossage-
Initiation: start of part program		ļ		
Completion: program completed		0.00	Ver Masseres	
Span: execution and prove	-out			
Discrete / Continuous		I		1
4		0	^	Л
1 contractor	†		_	4
Deverone	<u> </u>			
2.00-00.000	1	·		
Constranger	I			
Sauraa	1			
External Resources	1		· - · · · · · · · · · · · · · · · · · ·	
Description trajectory	machine n	nodel		
Likely Source motion	knowledge	e base		
Beliability: strong	strong			
ConVicts: not likely	not likely			
Products				
Description: movement command	s error mes	sage		
Attributes: code: parameters	code: para	ameter		
Constraints: time limitations	interaction	invalid		· · · · · · · · · · · · · · · · · · ·
Likely Recipientsmotion	operator in	ntertace		
Procedures/Methods		Sub-r	esponsibilities:	
(Flocedules/Methods			esponsionnies.	
Example: • accept trajectory			• perform look-ahead to	determine possible
Example: • accept trajectory • perform look-ahead to dete	ermine possible		perform look-ahead to multi-axis coordinate	determine possible on optimization

\$

٠

Ŧ

			1	in the second	Marezon.
Initiation: setup of participation			niconning messages.		Initiated
Completion: tool in positi	on				Process:
Span: prove-out			l Outgo	ing Messages:	
Discrete / Continuous					
Discrete / Continuous				·····	
	1		2		3
Local Basauroas					-
Description:					
Attributes:					
<u>Constraints:</u>					
Source:					
External Resources					
Description: trajectory ro	ough termin	nation point		sensor data	servo teedback
Likely Source: trajectory g	eneration			sensor interface	naroware interface
Heliability: strong	}			Weak	Strong
CONVICIS: not likely					
Products					
	nands				
Anribules: Voltage					
Likoly Poolisiontowis	uracy				
 move to termina continue move servos pass to 	ation point until senso olerance	ors or			
• move to termina • continue move servos pass to Responsibility: NG(ation point until senso olerance	erform sense	or fusion	n ning Messages:	Message-
• move to termina • continue move servos pass to Responsibility: NG(Temporal Aspects Initiation: start of par	ation point until senso olerance C-46 Pe	ors or erform sense	or fusion	n ling Messages:	Message- Initiated
• move to termina • continue move servos pass to Responsibility: NGC Temporal Aspects Initiation: start of par Completion: program co	ation point until senso olerance C-46 Pe t program ompleted	ors or erform sense	or fusion	n ling Messages:	Message- Initiated Process:
• move to termina • continue move servos pass to Responsibility: NGC TemporalAspects Initiation: start of par Completion: program co Span: prove-out	ation point until senso olerance C-46 Pr t program ompleted and execut	erform senso	or fusion Indon: Outgo	n ling Messages: bing Messages:	Message- Initiated Process:
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC TemporalAspects Initiation: start of par Completion: program of Span: prove-out of Discrete / Continuous	ation point until senso olerance C-46 Pe tt program ompleted and execut	ors or erform sense	or fusion Indon: Outgo	n ling Messages: ping Messages:	Message- Initiated Process:
• move to termina • continue move servos pass to Responsibility: NGO Temporal Aspects Initiation: start of par Completion: program co Span: prove-out a Discrete / Continuous	ation point until senso olerance C-46 Pe t program ompleted and execut	ors or erform sense	or fusion Froom Outgo	n ning Messages: ping Messages:	Message- Initiated Process:
• move to termina • continue move servos pass to Responsibility: NGC TemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous	ation point until senso olerance C-46 Point t program ompleted and execut	ors or erform sense tion	or fusion Cricon Outgo 2	n ning Messages: ping Messages:	Message- Initiated Process: 3
move to termina continue move servos pass to Servos pass to Responsibility: NGC Temporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous	tion point until senso olerance C-46 Point t program ompleted and execut	erform senso	or fusior Lindon Gutgo 2	n ning Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Responsibility: NGC TemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous	tion point until senso olerance C-46 Point triprogram ompleted and execut	erform senso	or fusior Fridom Outgo 2	n ling Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Responsibility: NGC IemporalAspects Initiation: start of par Completion: program of Span: prove-out of Discrete / Continuous	tion point until senso olerance C-46 Point the program ompleted and execution	erform sense	or fusion Froom Outgo 2	n ling Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Responsibility: NGC IemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous	tion point until senso olerance C-46 Point t program ompleted and execut	erform sense	or fusion Froom Outgo 2	n ling Messages: bing Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Responsibility: NGC IemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous External Resources	tion point until senso olerance C-46 Point triprogram ompleted and execution	erform sense	or fusion Froom Outgo 2	n ling Messages: bing Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Responsibility: NGC IemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous Discrete Servers Description: sensor dat Description: sensor dat	ta ta ta ta ta ta ta ta ta ta	erform sense	or fusion Congo Quego 2	n ling Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC IemporalAspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous Continuous Continuous Discrete / Continuous	ta erface	tion machine knowledg	or fusion Congo 2 2 model je base	n ling Messages: ping Messages:	Message- Initiated Process: 3
move to termina continue move servos pass to servos pass to remporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous Sector of Discrete / Continuous Sector of Sector	ta erface	tion machine (knowled strong	or fusion Congo Quego 2 model je base	n ling Messages: bing Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC Iemporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous Consistent External Resources Description: sensor dat Likely Source: sensor move Reliability: weak Con Victs: likely	ta erface	tion machine (knowledg strong not likely	or fusion Congo 2 2 model je base	n ling Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC Iemporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Continuous Consistent External Resources Description: sensor dat Likely Source: sensor into Reliability: weak Cont/acts: likely Products	ta erface	tion machine i knowledg strong not likely	or fusion Congo 2 model je base	n ling Messages: bing Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Persponsibility: NGC Iemporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous External Resources Description: sensor dat Likely Source: sensor unto Reliability: weak Cont/scts: likely Products Description: sensor unto Reliability: weak	ta erface ormation	erform senso tion machine i knowledg strong not likely	or fusion Congo Quitgo 2 model je base	n ning Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC Iemporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous External Resources Description: sensor dat Likely Source: sensor unto Reliability: weak Cont/scts: likely Products Description: sensor unto Reliability: weak Cont/scts: likely Products Description: sensor unto Attributes: values Constraints: sensor unto	ta erface ormation	erform senso tion machine i knowledg strong not likely	or fusion Congo Congo 2 model je base	n ning Messages: ping Messages:	Message- Initiated Process: 3
• move to termina • continue move servos pass to Servos pass to Responsibility: NGC Iemporal Aspects Initiation: start of par Completion: program of Span: prove-out a Discrete / Continuous Discrete / Continuous External Resources Description: sensor dat Likely Source: sensor unto Reliability: weak Cont/scts: likely Products Description: sensor unto Attributes: values Constraints: sensor act	ta erface commation curracy curacy curracy curracy curracy curracy curracy curracy curracy cur	erform senso tion machine i knowledg strong not likely	or fusion Congo Congo 2 model je base	n ning Messages: ping Messages:	Message- Initiated Process: 3

ŧ

ę

Responsibility: NGC-47 In	itiate shutdov	wn proc	edures	
TomporalAspects		Incom	ing Messages:	Message-
Initiation: System start UP				Initiated
Completion: system shut down			· 2 *	Process:
Span: always running		Outgo	ing Messages:	
Discrete / Continuous				
Discrete / Continuous	_			
1		2	3	4
Local Resources				
Description:				
Attributos				
Gonstraints:				·
Sources				
External Resources				
Description: watchdog warning	machine r	nodel		
Likely Source: watchdog system	knowledg	e base		
Beliability: strong	strong			
Con1/acts not likely				
Products				
Description: Shudown continant				
Attributes: code. parameters				
LIKEIV Recipients an components		 Cub #	energibilities:	
Example: • accept watchdog warning • determine safe shutdown • send shutdown command	procedures			
Responsibility: NGC-48 C	ue the opera	tor for r	nanual tasks	
		Incor	und Messades:	Message-
	mands			Initiated
	SA SA			- Process:
Completion. Valid operator respon		- Outge	ang Messages:	
Discrete / Continuous		•		
Discrete / Continuous	I			
1		2	3	4
an Ascourges				
Decorrect				
2				
<u></u>	ļ			
NC 02	Į			
External Resources				
Description: part program	machine m		operator response	
Likely Source: workstation manager	pent knowledge	Dase	code: parameters	
Beliability: weak	strong		not likoly	
Convicts: not likely	not likely		TIOLIIKEIY	
Producis	 		error message	
Lescoption: naticontinue cues	1	<u>-</u>	code: parameters	l
AUTIDUTES: code: parameters	<u> </u>			1
UNISILAIIIIS.	operator interfac	ce	operator interface	
	mblocks	Sub -	esponsibilities.	
I Procedures/Methods	no model	i gan-i	E2DOH2IDIIIIIE2.	

¢

Responsibil	ity: NGC-49 H	andle operat	or com	mand interactions	
TemporalAspe	ects		Incom	iing Messages:	Message-
Initiation:	system start up				Initiated
Completion:	system shut down		Outro	inn Messanes	- Process:
Span:	always running		00.90	nng mosokgeo.	
Discrete / Co	ntinuous				-
			2	3	4
Local Resour					
Attributors					
Constraints:					
Source					
External Reso	ources				
Description:	part program	manual inte	raction		
Likely Source	workstation managem	ent operator int	ertace		
Beliability:	weak	Weak			
Convers:		not likely			
Producis	execution commands				
Attributes:	code: parameters	code: parar	neter		
Constraints:	non-permanent chang	e interaction	nvalid		
Likely Recipi	entelan interpretation	operator in	erface		
Procedures/M Example:	lethods accept progra halt program v interaction • insert executio if valid • send executio	m when is needed on commands in commands	Sub-r	esponsibilities: halt program insert execution comma	ands if valid
Responsibil	ity: NGC-50 H	landle modiø	ations o	of offset values by c	operator
TomporalAco	orts		cuon	nno liessages:	Message-
Initiation:	operator signal	·····	1		Inifated
Completion:	part program start or	shutdown	i		- Process:
Span:	always running		l Cattoria	olno Wessages.	

Span: atways running		u ingo	.ng ., esseges.	
1		2	3	4
Level Zoon rear				
Dagaration				
1				
Charles and				
2 /m				
External Resources Description: offset values Likely Source: operator interface Reliability: strong Con'acts: ont likely Products Description: modived machine.mo Attributes: ondes: parameters Constraints: operator capabilities Likely Recipientskoowledge base	machine r knowledg strong not likely dels_operator k code: par memory s operating	podels e_base 0g ameter system		
Procedures/Methods • accept opera e log command • modify mach • send new modify	ator offsets ds ine models odels	Sub-r	esponsibilities:	

۶

	landle opera	tor inter	face for manual int	erpolation command
TemporalAspects		Incom	ning Messages:	Message-
Initiation: system start up				initiated
Completion: system shut down	····•	Outor	Sinn Macconae	Process:
Span: always running		Conge	ning messages.	
Discrete / Continuous				
1		2	3	4
Local Basauroas		_		
Descrintion:				
Attributes:				
Constraints:				
Source:	<u></u>			
External Resources				
Description: trajectory	manual int	erpolation	command	
Likely_Source: trajectory generation Reliability: strong	operator ir strong	nterface		
Con1/acts: not likely	not likely			
Products				
Description: servo commands	halt comm	ands		
Attributes: voltage	code: para	meter		
Constraints: accuracy of operator				
Likely Recipientsaxis motor	trajectory	generatio	h	
Responsibility: NGC-52 F	nsure safety			······································
	noure saleiv			
TemporalAspects Initiation: system start up	,	incer	ling Messages:	Message- Initiated
TemporalAspects Initiation: system start up Completion: system shut down			ning Messagas:	Message- Init.atec Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: atways running		Incert Outgo	ling Messagas: ling Messages:	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous Start down		Incert Outgo	ling Messages: ling Messages:	Message- Initated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: atways running Discrete / Continuous		incom Outgo 2	ling Messages: ling Messages: 3	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1		incom Outgo 2	ing Messages: ing Messages: 3	Message- Initatec Process:
TemporalAspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1		Incom Outgo 2	ing Messagas: ing Messages: 3	Message- Initatec Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Coal Loop urose 1 Asystem start 1		Incom Outgo 2	ing Messagas: ing Messages: 3	Message- Initatec Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 1 1		Coutgo	iing Messagas: ing Messages: 3	Message- Init.atec Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: atways running Discrete / Continuous 1 Constraints 1 Constraints 2 Attractions 2 Constraints 2		Incom Outgo 2	iing Messagas: ing Messages: 	Message- Initated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 2000/2000/2000 1 2000/2000/2000 2000/2000/2000 2000/20000/2000/2000/2000/2000/2000/2000/2000/20000/2000/200		Coutgo	iing Messagas: ing Messages: 	Message- Initated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Concernance 1 Concernance 2 External Resources Description: Description: voltages Likely Source understart	beartbeats	Dutgo	ing Messages: ing Messages: 	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: atways running Discrete / Continuous 1 1 1 2 1 2 2	beartbeats	Cutgo Outgo 2	ing Messages: ing Messages: 	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 1 1		Dutgo	ing Messages: ing Messages: 	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 1 1	beartbeat: 	Dutgo	ing Messages: ing Messages: 	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: atways running Discrete / Continuous 1 1 1	beartbeat: 	Dutgo	ing Messages: ing Messages: 	Message- Init.ated Process: 4
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Construction: 1 Construction: 1 Construction: 2 Amount 2 Source: 1 Construction: 2 Amount 2 Amount 2 Amount 2 Amount 2 Construction: 2 Construction: 2 Construction: 2 Source: 1 Construction: 2 Source: 1 Construction: 2 Source: 1 Description: 2 Description: 2 Description: 2 Description: 2 Codes: 2	beartbeat: 	Coutgo 2	ing Messages: ing Messages: 	Message- Init.ated Process: 4
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Image: State Contract 1 Image: State Contract 1 State Contract 1 Image: State Contract 1	beartbeat: 	Coutgo 2	ing Messages: ing Messages: 	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Constraints 1 Constraints 1 Constraints: 0 Constraints: out of tolerance	beartbeat: 	Cutgo Outgo 2	ing Messages: ing Messages: 	Message- Init.ated Process: 4
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Constraints 1 Constraints 2 Amount 2 Constraints 1 Constraints 1 Constraints 1 Constraints 0 Attributes: codes: parameters Constraints: out of tolerance Likely Recipients shutdown compon Procedures/Methods	beartbeat: 	Cutgo Outgo 2	ing Messages: ing Me	Message- Init.ated Process:
Temporal Aspects Initiation: system start up Completion: system shut down Span: always running Discrete / Continuous 1 Constraints 1 Constraints 1 Constraints: 0 Description: votages Likely Source: hardware interface Reliability: strong Constraints: out of tolerance Likely Recipients shutdown compon Procedures/Methods Example: • accept votages	beartbeats all compor strong not likely ent	Cutgo Outgo 2 Sub-re	ing Messages: ing Messages: in	Message- Init.ated Process:

ý

mesponsibility: NGC-53 G	ienerate a pa	art progr	ram	
TemporalAspects		incom	ing Messages:	Message-
Initiation: part design complete	đ		Q	Initiated
Completion: program developed			·	Process:
Span: product development		Outgo	ng Messages:	
Discrete / Continuous				
	[
1		2	3	4
Lecal Resources		·		
Description				
Attributes:				
Constraints:				
Source				
External Resources				
Description: part model	stock model		operations available	machine model
Likely Source: knowledge base	knowledge.ba	15 8	knowledge_base	knowledge_base
Beliability: strong	strong		strong	strong
ConVicts: not likely	not likely		not likely	not likely
Products				
Description: part program				
Attributes: code: parameters				
Constraints:				
LIKELY RECIDIENTS workstation mot	I		L	l

Responsibility: NGC-54 Cu	stomize par	t progra	m for particular mac	hine
TemporalAspects		lincom	nno Messages:	Message
Initiation: new part program				linit ated
Completion: part program customiz	zed	13.000	van Macanaas	Hrccess:
Span: product development		v ⊂ . i., u), 'g iv caseigua.	
Discrete / Continuous	<u> </u>		(·
1		2	3	
<u> </u>				
A 4 ye in the gal				
To more presente o				
So, con:			<u> </u>	
	t			
	manhina mada		{	
Likoh Souroo watatata at				
Poliability dross dross	t NINMIGOGE Das	×		
Conlácte:	Det likeh:			
Droducto				
Products	t			
	PUL	······	<u>ا</u>	-
Attributes: orde: parameter				
Attributes:code; parameters Constraints:Likely Recipients				

ÿ

.

~

TemporalAspects	ted	Incom	ling Messages:	Initiated
Completion: part program re-sequ	Jenced	†		- Process:
Span: product developmen	t	1 Outgo	ning Messages:	
Discrete / Continuous		1		
<u> </u>				
Rectinion:				
Atributes:				
Constraints:				
Saurae				
External Resources				
Description: part program	part model		stock model	
Likely Source: workstation mgt	knowledge b	ase	knowledge_base	
Beliability: strong	strong		strong	
ConVicts not likely	not likely		not likely	
Products	ļ			
Description:resequenced part pr	ogram			
Attributes code: parameters	<u> </u>		ļ	
Constraints:	1			
Likely Recipientsworkstation mot	I		l	
• analyze nmoram for r	Jucio			
send resequenced participation of the send resequenced participat	optimality art program	ļ		
• send resequenced participation of the send resequenced participation	mulate part p	program	execution	
• send resequenced participation of the send resequence of the sen	mulate part p		execution	i Mossaço-
• send resequenced participation initiated	mulate part p	program	execution	Message-
	mulate part p	program	execution	Mossaço- Initiatoc Process:
	mulate part p	program noam Outge	execution ang Mossuges: ang Messages:	Mossaço- Initiatoc Process:
• send resequenced participation in the send resequence is send resequences. The send resequences is send resequences is send resequences in the send resequences in the send resequences is send resequences in the send resequences in the send resequences in the send resequences is send resequences in the send res	mulate part p	orogram near Outge	execution ang Mossuges: Sing Messages:	Message- Initiated Process:
	mulate part p	orogram roar Outge	execution Mrg Wossages: Sing Messages:	Mossaçe- Initiatoc Process:
	mulate part p	orogram near Outge	execution ang Mossuges: ang Messages: 3	Moesaçe- Initiated Process:
	mulate part p	orogram near Outge	execution ang Mossuges: ang Messages: 3	Moesaçe- Initiated Process:
	mulate part p	Drogram Tream Outge	execution ang Mossuges: aing Messages:	Mossaçe- Initiated Process:
	mulate part p	Drogram Tream Outer 2	execution ang Mossuges: Sing Messages:	Messaçe- Initiated Process:
	mulate part p	Drogram Treat Outge 2	execution ang Mossages: Sing Messages:	Message- Initiated Process:
• send resequenced participation of the send resequence of the send res	mulate part p	Drogram Troat Duter 2	execution ang Mossages: Sing Messages:	Message- Initiated Process:
	mulate part point start comma	Drogram Toar Outer 2	execution arg Messages:	Message- Init atec Process:
	mulate part point part program	Duter 2 nd rtace	execution arg Messages: bing Messages: hardware models knowledge base stmon	Mossage- Initiated Process:
	start comma start comma optimality art program	Duter Duter 2	execution arg Messages: bing Messages: hardware models knowledge base strong met likely	Mossage- Initiated Process:
	mulate part program	Duter Duter 2	execution arg Messages: bing Messages: hardware models knowledge base strong not likely	Mossage- Initiated Process:
	mulate part program	Duter Duter 2	execution arg Messages: bing Messages: hardware models knowledge base strong not likely	Mossage- Initiated Process:
	mulate part program	Duter Duter 2	execution arg Messages: arg Messages: hardware models knowledge base strong not likely	Mossaço- Initiatos Process:
	mulate part program	Duter Duter	execution arg Messages: arg Messages: hardware models knowledge base strong not likely	Mossaço- Initiatos Process:
	mulate part program	Duter Duter	execution arg Messages: arg Messages: hardware models knowledge base strong not likely	Mossaço- Initiatos Process:
	mulate part program	Duter Duter	execution arg Messages: arg Messages: hardware models knowledge base strong not likely	Mossaço- Initiatos Process:
	mulate part program	Drogram Trocar Outer 2 nd nd nface	execution	Mossaço- Initiatos Process:
	start command	Drogram Trocitor Outgram 2 nd nd nface	execution	Mossaçe- Initiates Process:

۵

	imulate softw	vare lau	Its	
TemporalAspects		incom	ing Messages:	Message-
Initiation: simulation initiated			с .	Initiated
Completion: simulation complete			· 8 4	- Process:
Span: maintenance		l Ontão	ing wessages:	
Discrete / Continuous				
		2		3
Local Reserves	,			
Description:				
Attriputes:				
External Resources	software area	m model		
Description: part program	sonware errol	rs moder		
		196		
	not likoly			
Draduata				
Products				
Attributos: code: parameters	315			
Constraints' erms inserted				
Likely Recipientsknowledge base				
Responsibility: NGC-58 Si	mulate hardv	vare fau	llts	
TemporalAspects		.ncor	nng Mossages:	Niessage-
Initiation: simulation initiated		1		Init atec
Completion: simulation complete			Ner Masser Le	Process:
Span: maintenance		1	. ್ರ ೧೯೮೨ನಲ್ಲಿರು.	
Discrete / Continuous		1		
		1	1 · · · · · · · · · · · · · · · · · · ·	
1		2		3
1		2		3
1		2		3
1 		2		3
1 		2		3
1 		2		3
1		2		3
1 External Resources Description: part program	hardware en	1 2		3
1 External Resources Description: part program Likely Source: workstabon mgt Reliability	hardware en knowledge b	2 rors mode ase		3
1 External Resources Description: part program Likely_Source:workstabon mgt Beliability: strong Cont/erts: anti-	hardware en knowledge b strong not likely	2 rors mode ase		3
1 External Resources Description: part program Likely Source: workstabon mgt Beliability: strong ConVicts: not likely Products	hardware en knowledge b strong not likely	2 rors mode ase		3
1 External Resources Description: part program Likely Source: workstabon mgt Beliability: strong Con!/acts Products Description: motiled batteres	hardware en knowledge b strong not likely	2 rors mode ase		3
1 External Resources Description: part program Likely_Source: workstabon mgt Beliability: strong Con'icts: products Description: motilized barrivate mot Attributes:	hardware en knowledge b strong not likely	2 rors mode ase		3
1 External Resources Description: part program Likely Source: workstabon mgt Beliability: strong Con'dcts: products Description: motived barrivare	hardware en knowledge b strong not likely	2 rors mode ase		3
1 External Resources Description: part program Likely Source: workstabon mgt Beliability: strong Contracts: not likely Products: part workstabon mgt Description: moduled bardware model Attributes: code, parameters Constraints: errors inserted Likely Recipientsknowledge base	hardware en knowledge b strong not likely del	2 rors mode ase		3

4

۶,

	Create a part	aesign		
TemporalAspects		Incorr	ning Messages:	Message-
Initiation: part identil/ed				Initiated
Completion: design completed				-Process:
Span: product development	an: product development		nng Messages:	
Discrete / Continuous		Ι		
1		2		4
Lacel Resources				
Description				
<u>Affrihules:</u>				
External Resources		onte	aburrical laws	
Description:part description	engineering		knowledge base	
Poliability: weak	weak		strong	
Contacts: not likely	not likely		not likely	
Products	1			
Description: part design model				
Attributes features				
Constraints:				
Likely Recipientsmodel manageme	ent		<u> </u>	
Responsibility: NGC-60 N	lotify mainten	ance wi	nen parameters ex	ceed tolerances
Temporal Aspects	•••••	CCCB.	nch menergas	Initiates
Completion: system shutdown				- Process:
Span: execution and prove	-out	I Oriá:	ang messayas.	
Discrete / Continuous		I		
	ıl	2		3 4
<u></u>				
 A star production of the star star star star star star star star				
Amenicosi Deceneration				
Attractory Attractory Constructor Sautor				
American Operations Sources				
Attended Construction: Sources Description: machine.model	machine statu	s	sensed status	
External Resources Description:machine.model Likely_Source:knowledge.base	machine statu	s	sensed status	
External Resources Description:machine_model Likely_Source:knowledge_base Beliability:strong	machine statu components weak	s	.senseri status .sensor.iotedace. .weak	
External Resources Description: machine model Likely Source: knowledge base Beliability: strong Con Victs: not likely	machine statu components weak not likety	S	sensed status sensociotedace weak not likely	
External Resources Description: machine model Likely Source: knowledge base Reliability: strong Cont/cts: not likely Products	machine statu components weak not likely	S	sensed status sensor iotedace weak not likely	
External Resources Description: machine model Likely Source: knowledge base Reliability: strong ConVicts: not likely Products Description: exception not Vicition	machine statu: components weak not likely	S	sensed status sensor iotedace weak not likely	
Attributes: Constraints Description: machine.model Likely.Source:knowledge.base Reliability: strong ConVicts: not likely Products Description: exception.notVeation Attributes: code: parameters Constraints	machine statu: components weak not likely	S	sensed status sensor iotedace weak not likely	
Attributes: Constraints: Description: machine model Likely Source: knowledge base Beliability: strong ConVicts: not likely Products Description: exception not Viction Attributes: code: parameters Constraints: Likely Bocinients:	machine statu: components weak not likely	s	sensed status sensor iotedace weak not likely	
Attributes: Constraints: Description: machine model Likely Source: knowledge base Beliability: strong ConVicts: not likely Products Description: exception notivation Attributes: code: parameters Constraints: Likely Recipients:tactory control	machine statu components weak not likely	s	sensed status sepsor jotedace weak not likely	

¥

ŧ

J

Responsibility: NGC-61 N	laintain mode	eis				
TemporalAspects		Incom	ing Messages:	ĮN	Message-	
Initiation: system startup]			itiated	
Completion: system shutdown				۲ ۲	rocess:	
Span: execution and prove-out						
Discrete / Continuous					1	
4		2		2		
Local Resources						
Description:						
Attributes:						
Constraints:						
Spaner						
External Resources	1					
Description: updates to models	models					
Likely Source:_op_if/mechanism	hion_knowledg	je hase				
Reliability: strong	strong					
Draduoto	not likely		· · · · · · · · · · · · · · · · · · ·			
Producis				-1-		
Attributes:						
Constraints:	1					
Likely Recipients: knowledge base						
Procedures/Methods Example: • accept updates to m	nodels	Sub-r	esponsibilities:			
		1				
 access models validate updates send updated model 	ls					
access models validate updates send updated mode	ls					
• access models • validate updates • send updated mode Responsibility: NGC-62 Ir	ils hitiate job sec	uence				
eaccess models validate updates send updated mode Responsibility: NGC-62 Ir TemporalAspects	nitiate job seq	uence	າະເຊ ພໍ່ມຣຣລເງອຣ:	1 1	essage-	
e access models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup	itiate job seq	uence	ang dessayes:		iessage- tated	
e access models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system shutdown	nitiate job seq		ning libeschost		iessage- n tated Process	
eaccess models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove	nitiate job seq e-out	uence noon Outgi	ang dessayes: ang dessayes:		iessage- ntated Process	
eaccess models validate updates validate updates send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous	nitiate job seq e-out	uence noon Outgi	ang dessayes: ang dessayes:		iessage- htated Processi	
e access models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous	hitiate job sec e-out	uence noon Outpi	ang dessayes: ang Messayes:	3	iessage- tated Process	
eaccess models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous 1	hitiate job seq e-out	uence noon Outor 2	ang dessayes: ang dessayes:	3	essage- tated Process	
eaccess models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous Income Hospurses Completion:	hitiate job seq e-out	uence noon Outor 2	ang dessayes: ang dessayes:	3	iessage- tated Process	
eaccess models validate updates validate updated mode send updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	hitiate job seq	uence noon Outor 2	ang dessayes: ang dessayes:	3	iessage- tated Process	
eacess models • validate updates • validate updates • validate updated mode Responsibility: NGC-62 Ir TemporalAspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove Discrete / Continuous 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	hitiate job seq	uence noor Outor 2	ong dessages: ong dessages:	3	iessage- ntated Process	
eaccess models • validate updates • validates • v	e-out	uence noon Outpr	cing dessayes: cing dessayes:	3	iessage- ntated Process	
eaccess models • validate updates • validates •	e-out	uence noon Outor 2	ວະກຽ ທີ່ສະຣະເທີອຣ: ວະກຽ ທີ່ອອະສະນຸອຣ:	3	iessage- ntated Process	
	e-out	uence noon Outor 2	ວະກຽ ທີ່ສະຣະເທີອຣ: ວະກຽ ທີ່ອອະສຽອຣ:	3	iessage- ntated Process	
	e-out	uence noon Outor 2	ວະກຽ ທີ່ສະຮະດູອະ ວະກຽ ທີ່ອະຊຽອະ:	3	iessage- ntated Process	
	e-out	uence noon Outor 2	ວະກຽ ທີ່ສະຮະດູອຣ: ວະກຽ ທີ່ອະຣະດູອຣ:	3	iessage- ntated Process	
	e-out tus_job_list workstation strong	uence noor Outor 2	cing disseques: cing desseques:	3	iessage- ntated Process	
	e-out		ning dessayes: ning dessayes:	3	iessage- tated Process	
	e-out		ning dessages: ning dessages:	3	essage- tated Process	
	els		nung libssaljes: ping libesages:	3	essage- tated 2rccess	
	els		nung dessages: ning dessages:	3	essage- tated Process	
	els hitiate job sec e-out tus_job.list workstation strong not likely turycle		ning dessayes:	3	iessage- tated Process:	
	hitiate job sec e-out this_job.list workstation strong not likely t.cycle ce	uence noon Outor 2 	responsibilities:	3	essage- rtated Process	

1

۰

٠

~

			and the second secon	
Responsibility: NGC-63 V	erify proper (part pro	ogram and revision	number for each part
TemporalAspects	71	Incon	ning Messages	Message-
Initiation: system startup		a containg incoderged.		Initiated
Completion: system shutdown		*		Process:
Span: execute and prove-o	Span: execute and prove-out		oing Messages:	
Discrete / Continuous		4		
	T	1	1	T ⁱ
1		2		4
Lecal Resources				
<u>Description:</u>				
A tributos				_
Sources				
Estemal Deseuress				
External Resources	- nort program		inter the d	
Likely Source: workstation met	workstation m	nat	IOD IISt	
Reliability: strong	strong	<u></u>	stopo	
Con ¹ /cts: not likely	not likely		not likely	
Products				
Description: veri/cation noti/cation	h			
Attributes:	·			
Constraints:				
Likely Recipients:workstation mot				
Procedures/Methods		Subr	esponsibilities:	
Example: • accept part id and part	t program	Guba	esponsionnies.	
access job list	, program			
• venty that all match • send vent/cation poti/	tation			
		L		
Responsibility: NGC-64 Ide	entify current	part		
Responsibility: NGC-64 Ide	entify current	part	me Macanese	: Mossauc-
Responsibility: NGC-64 Ide	entify current	part Incor	ing Messages:	, Mossaço-
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown	entify current	part Feor	ing Messages:	Mossage- Litates Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: executes and provide	entify current	part Incor	eng Messages: eng Mossages.	Mossago- Litates Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Proor Outgo	rng Messagas: Sng Mossages.	Mossago- Liitiateo Process:
Responsibility: NGC-64 Ide TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Theory Dutge	rng Messages: ong Mossages.	Mossage- Litifated Process:
Responsibility: NGC-64 Ide TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Incor Outgo 2	rng Messages: mg Mossages. 3	Mossage- Litates Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Incor Outgo 2	mg Messages: mg Mossages. 3	Mossage- Litates Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Incor Outgo 2	mg Messages: mg Mossages. 3	Mossage- Lifates Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Incor Outgo 2	ing Messages: ong Mossages. 3	Mossago- Lifateo Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Detge 2	mg Messages: mp Mossages. 3	Mossage- Lifatee Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous	entify current	part Detge	mg Messages: mp Mossages. 3	Mossage- Lifatee Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction Span: 1 Construction External Resources Description: request for part id	entify current	part Treor Dutge	mg Messages: mg Mossages. 3	Mossage- Lifatee Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: request for part id Likely: Source: workstation mot	entify current	part Treor Dutge	rng Messages: xng Mossages. 3	Mossago- Liitateo Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: request for part id Likely Source: workstation mgt Reliability: strong	entify current	part Croom Dutgo 2	mg Messages: mg Messages. 3	Mossago- Lifateo Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: request for part id Likely Source: workstation.mgt Beliability: strong Convers: not likely	entify current	part Detge	rng Messagas: eng Messages. 3	Mosaaço- Liifatea Process: 4
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: execution and prove-o Discrete / Continuous	entify current	part Detge	rng Messagas: xng Messages. 3	Mosaago- Liitatea Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: execution and prove-o Discrete / Continuous 2 Construction: execution and prove-o Discrete / Continuous 2 Construction: execution and prove-o Discrete / Continuous Construction: execution and prove-o Discrete / Continuous Construction: execution and prove-o Discrete / Continuous Construction: execution and prove-o Construction: execution and prove-o Construct	entify current	part Croom Dutgo	rng Messagas: xng Messages. 3	Mosaaço- Liitatea Process: 4
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: execution and prove-o Discrete / Continuous Construction: execution and prove-o Discrete / Continuous Construction: execution and prove-o Construction: execution and prove-o Constructi	entify current	part Detge	rng Messagas: 2ng Messages. 3	Mosaaço- Liitatea Process: 4
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Constraints: not likely Products Description: part id Likely Source: workstation mgt Beliability: strong ConVicts: not likely Products Description: part id Attributes: code, parameters Constraints:	entify current	part Detge	rng Messagas: 2ng Messages. 3	Mossage- Litatea Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: equest for part id Likely Source: workstation mgt Beliability: strong ConVicts: not likely Products Description: part id Attributes: code, parameters Constraints: Likely Recipients:workstation mgt	entify current	part Detge	rng Messagas: 2ng Messagas: 2003 2003 2003 2003 2003 2003 2003 200	Mossage- Liitatea Process:
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 2000	entify current	part Detge 2	eng Messagas: eng Messagas: 3	Mossaço- Luitatea Process: 4
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: execution and prove-o Discrete / Continuous External Resources Description: request for part id Likely Source: workstation mgt Procedures/Methods Example: execut request for part	id	part Detge 2	eng Messages: 2002 Mossages. 3 3 3 4 3 4 3 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5	Mossaço- Luitatea Process: 4
Responsibility: NGC-64 Ide Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and prove-o Discrete / Continuous 1 Construction: execution and prove-o Discrete / Continuous External Resources Description: request for part id Likely Source: workstation mgt Reliability: strong ConVicts: not likely Products Description: part id Attributes: code; parameters Constraints: Likely Recipients: workstation mgt Procedures/Methods Example: execution and prove-o execution and prove-o Startion and prove-o Starti	id	part Detge 2	eng Messages: 2009 Micesages. 3 3 3 4 3 4 3 4 3 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5	Mossaço- Luitatea Process: 4

Responsibility: NGC-65 F	nsure sancti	v of saf	e zones		
		Incoming Messages		Message-	
Initiation:				Initiated	
Completion: system shutdown				- Process:	
Span: execution and prove-out		Outgoing Messages:			
Discrete / Continuous					
2100.0107 0011110000					
1		2		3 4	
Local Resources					
Description					
<u>Attributas</u>					
Courses					
External Resources					
External Resources	nart model		machine model		
Likely Source: trajectory generator	knowledge ba	ise	knowledge base		
Reliability: strong	strong		strong		
Contácts: not likely	not likely		not likely		
Products					
Description: modi/ed trajectory					
Attributes: setpoints: parameter	\$				
Constraints:					
Likely Recipients:trajectory generate	dr	r			
send modiled trajecto	xv				
• send modilisid trajecto Responsibility: NGC-66 Re	eassignment	of tasks	s between machin	e and operator	
• send modified trajecto Responsibility: NGC-66 Re Temporal Aspects	eassignment	of tasks	s between machin	e and operator	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup	eassignment	of tasks	s between machin hing Mossages:	e and operator Message- Initiated	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown	eassignment	of tasks	s between machin hing Mossages:	e and operator Message- Initiated Processi	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove-4	eassignment	of tasks near	s between machin hing Mossages: hing Messages:	e and operator Message- Initiated Processi	
send modified trajector Responsibility: NGC-66 Re Initiation: system startup Completion: system shutdown Span: execution and prove- Discrete / Continuous	eassignment	of tasks near Outgr	s between machin hing Mossages: bing Messages:	e and operator Message- Initiated Processi	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove- Discrete / Continuous	eassignment	of tasks neer Outgr	s between machin hing Mossages: bing Messages:	e and operator Message- Init.atec Process:	
• send modified trajector Responsibility: NGC-66 Ref TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and prove- Discrete / Continuous	eassignment	of tasks neer Outgr	s between machin hing Mossages: bing Messages:	e and operator Message- Initiated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1	eassignment	of tasks near Outgr	s between machin hing Mossages: b.ng Messages:	e and operator Message- Initiated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1		of tasks near Outgr	s between machin hing Mossages: b.ng Messages:	e and operator Message- Initiated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1		of tasks near Outgr 2	s between machin hing Mossages: b. ng Messages:	e and operator Message- Initiated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1	eassignment	of tasks near Outgr 2	s between machin hing Mossages: b. ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1 Line and the sources	eassignment	of tasks near Outgr 2	s between machin hing Mossages: b. ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1 Logical Aspects Logical Aspect	eassignment	of tasks near Outgr 2	s between machin hing Mossages: b. ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system shutdown Span: execution and proved Discrete / Continuous 1 External Resources Description: part program Likely Source: plan interpretation	eassignment	of tasks near Outgr 2 signments	s between machin ning Mossages: b. ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system startup Completion: system startup Completion: system startup Discrete / Continuous 1	eassignment	of tasks near Outgr 2 signments tace	s between machin ning Mossagas: b. ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re TemporalAspects Initiation: system startup Completion: system startup Completion: system startup Completion: system startup Discrete / Continuous 1	eassignment	of tasks near Outgr 2 signments ace	s between machin hing Messages: b.ng Messages:	e and operator Message- Initated Process:	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system startup Completion: system startup Completion: system startup Discrete / Continuous 1 External Resources Description: part program Likely_Source: plan interpretation Reliability: strong Con Victs: not likely Products	eassignment	of tasks recor Outgr 2	s between machin hing Mossages: b.ng Messages:	e and operator Message- Initated Processi 3	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system startup Completion: system startup Completion: system startup Discrete / Continuous 1 External Resources Description: part program Likely Source: plan interpretation Reliability: strong Con Victs: not likely Products Description: modived part program	eassignment	of tasks reer Outgr 2	s between machin hing Mossages: b.ng Messages:	e and operator Message- Initated Processi 3	
send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system startup Completion: system startup Completion: system startup Discrete / Continuous 1 External Resources Description: part program Likely Source: plan interpretation Reliability: strong ConVicts: not likely Products Description: modived part program Attributes: code: parameters	eassignment	of tasks neer Outgr 2	s between machin ning Mossages: b. ng Messages:	e and operator Message- Initated Processi 3	
• send modified trajector Responsibility: NGC-66 Re Temporal Aspects Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove- Discrete / Continuous 1 External Resources Description: part program Likely Source: plan interpretation Reliability: strong Confficts: not likely Products Description: modified part program Attributes: code: parameters Constraints: Likely Recipients plan interpretation	eassignment	of tasks neer Outgr 2 signments tace	s between machin ning Mossages: b.ng Messages:	e and operator Message- Initiated Process:	
• send modified trajector Responsibility: NGC-66 Re Initiation: system startup Completion: system startup Completion: system shutdown Span: execution and prove- Discrete / Continuous 1 External Resources: Description: part program Likely Source: plan interpretation Reliability: strong Confficts: not likely Products: Description: modified part program Attributes: code: parameters Constraints: Likely Recipients: plan interpretation Procedures/Methods Example: • accept operator reassignm	eassignment	of tasks near Outgr 2 signments face	s between machin ning Mossages: b.ng Messages:	e and operator Message- Initated Processi 3	

.

÷

4

•

Responsibility: NGC-67 C	coordinate sr	indle a	ear changes with	servo control
		Incoming Messages:		Message-
Initiation: motion started		and any webbuget.		Initiated
Completion: motion stands		-		-Process:
Span: execution and prove-out		 Outgoing Messages: 		
Discrete / Continuous		•		
Discrete / Continuous	I			
1		2		3 4
Local Resources				
<u> </u>				
Constraints				
Serves:				
	movement.cc	mmande	machina model	
Description: spinole dear state	motion	mmanus	knowledge base	
Deliability store	strong			
	strong		situliy	
Draduata				
	e			
Attributoo	2			
Antibules:				
Likely Register actor				
			11 1111	
modify movement comr send modil/ed moveme	nands according nt commands	to spindi	e gear	
Responsibility: NGC-68 Re	e-evaluate co	ordinat	e zeros for alterna	ate tool choices
TemporalAspects		Licom	rne Massades.	Messago-
Initiation: system startup				Heit ateo
Completion: system shutdown				Propess:
Span: execution and prove-	out	ಂಡಂ	ang Massages.	1
Discrete / Continuous				
1		2		3 4
nor: Exprisons		<u> </u>		
Deservice				
Almo tea				
Construction				
Rourrow				
External Besources				
Description: tool substitution	coordinate svs	tem mode		
Likely Source: mechanism	knowledge bas	;е	knowledge base	
Beliability: strong	strong		stroog	
ConVicts: not likely	not likely		. not likely	
Products				
_Description:new coordinate zeros				
Annoures:				
LIKELY HECIPIENTS plan interpretation	1			
Procedures/Methods Example: • accept tool substitution • access tool and coordin • determine new coordin	nate system mo nate seros	Sub-ri	esponsibilities:	

¥

ŧ

NG	10-09 II		in case	na Báncenene:	Messane-
emporalAspects	art program		incomi	ng wessages.	Initiated
Completion: program completed				-Process:	
Completion. plogram c	pan: execution and prove-out		Outgoing Messages:		
Discrete / Continuous					
Discrete / Continuous			T		
	1		2		3
acal Resources					
Description:					
Attributes.					
				······································	
Second Decouroon					
	mineton	NURBS	labo		
Description. Nones (knowledge	hase		
		strong			
Reliability subrig		pot likely			
Dreducto					
Producis motion (mmands				
Attributes: setopints	: parameter	 S			
Constraints:		·			
Likely Recipientsactuato					
			Sub	esponsibilities.	
example: • accent NL	JRBS model	1			•
determine send new	e setpoints al rajectory	nd parameters			
• determine • send new Responsibility: N(GC-70 M	nd parameters	ed cycle	es	
• determine • send new Responsibility: No	Bestpoints all trajectory	nd parameters lanage cann	ed cycle	es ning Messages.	Message-
• determine • send new Responsibility: No Temporal Aspects Initiation: start of pi	GC-70 M art program	nd parameters lanage cann	ed cycle	es ning Messages.	Message- Initiated Process
• determine • send new Pesponsibility: No Temporal Aspects Initiation: start of program	GC-70 M art program completed	nd parameters lanage cann	ed cycle	es ning Messages:	Message- Initiated Process:
• determine • send new Pesponsibility: NO Temporal Aspects Initiation: start of program Completion: program Span: executor	GC-70 M art program completed n and prove	nd parameters lanage cann out	ed cycle incom Outgo	es ang Messages: bing Messages:	Message- Initiated Process:
• determine • send new Presponsibility: NO Iemporal Aspects Initiation: start of program Completion: program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove-	nd parameters	ed cycle incom Outgo	es ang Messages: bing Messages:	Message- Initiated Process:
• determine • send new Presponsibility: NC Temporal Aspects Initiation: start of program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove-	nd parameters	ed cycle incom Cutgo 2	es ang Messages: ping Messages:	Message- Initiated Process:
• determine • send new Presponsibility: NC Temporal Aspects Initiation: start of program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove	anage cann	ed cycle incom Cutgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new Send new Responsibility: N(Initiation: start of pi Completion: program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove	anage cann	ed cycle incom Outgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new Send new Responsibility: N(Iemporal Aspects Initiation: start of pi Completion: program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove	anage cann	ed cycle incom Outgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new Send new Responsibility: N(Initiation: start of pi Completion: program Span: executor Discrete / Continuous	GC-70 M art program completed n and prove	anage cann	ed cycle incom Outgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new	GC-70 M art program completed n and prove	anage cann	ed cycle incom Outgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new	GC-70 M art program completed n and prove	anage cann	ed cycle incom Cutgo 2	es hing Messages: bing Messages:	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Iemporal Aspects Initiation: start of pi Completion: program Span: executor Discrete / Continuous Send new S	GC-70 M art program completed n and prove	nd parameters	ed cycle incom Cuitgo 2	es hing Messages: bing Messages: capped cycle le	Message- Initiated Process: 3
determine send new send new Responsibility: N(Iemporal Aspects Initiation: start of pi Completion: program Span: executor Discrete / Continuous External Resources Description: part program Ikely Source: workstat	GC-70 M art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2	es wing Messages: bing Messages: caooed cycle le knowledge base	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Initiation: start of p Completion: program Span: executor Discrete / Continuous External Resources Description: part program Likely Source: workstat Reliability: weak	GC-70 M art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang i/essages: ping N'essages: canned cycle le knowledge base strong	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Initiation: start of pi Completion: program Span: executor Discrete / Continuous External Resources Description: part program Likely Source: workstat Reliability: weak Con Victs: pot likely	art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2 model e hase	es ling i/essages: ping N'essages: canned cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Iemporal Aspects Initiation: start of pi Completion: program Span: executor Discrete / Continuous External Resources Description: part program Likely Source: workstat Reliability: weak Con ½cts: not likely Products	art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang i/essages: bing N'essages: caoned cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Initiation: start of pi Completion: program Span: executor Discrete / Continuous External Resources Description: part program Likely Source: workstat Reliability: weak ConVacts: not likely Products Description: detailed	art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2 model e hase	es ling i/essages: ping N'essages: canned cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Initiation: start of p Completion: program Span: executor Discrete / Continuous External Resources Description: part program Likely Source: workstat Reliability: weak ConVacts: not likely Products Description: detailed Attributes: codes: c	GC-70 M art program completed n and prove- 1	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang i/essages: ping N'essages: canned cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Send new Responsibility: N(Initiation: start of p Completion: program Span: executor Discrete / Continuous External Resources Description: part prog Likely Source: workstat Reliability: weak ConVacts: not likely Products Description: detailed Attributes: codes: p	GC-70 M art program completed n and prove- 1 1 prommanager kon manager kon manager kon manager kon manager	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang i/essages: bing N'essages: canoed cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Initiation: start of p Initiation: start of p Completion: program Span: executor Discrete / Continuous External Resources Description: part prog Likely Source: workstat Reliability: weak ConVicts: not likely Products Description: detailed Attributes: codes: p Constraints: limited.c	GC-70 M art program completed n and prove- 1 1 prommanager kon manager kon manager kon manager kon manager kon manager	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang i/essages: bing N'essages: canned cycle le knowledge base strong not likely	Message- Initiated Process: 3
determine send new send new Send new Responsibility: N(Iemporal Aspects Initiation: start of p Completion: program Span: executor Discrete / Continuous External Resources Description: part program Secure Source: workstat Reliability: weak ConVacts: not likely Products Description: detailed Attributes: codes: p Constraints: limted.c Likely Recipientsraject	GC-70 M art program completed n and prove 1 1 gram program comparent program compare	nd parameters	ed cycle incom Cutgo 2 model e hase	es ang Messages: bing Messages: canned cycle le knowledge base stmng not likely	Message- Initiated Process: 3

•

~

Initiation: start of part program Completion: program completed Span: execution and prove	Temporal Aspects			Messade-
Completion: program completed Span: execution and prove	Initiation: start of part program			Initiated
Span: execution and prove	Completion: program completed		*	- Process:
	Span: execution and prove-out		ing Messages:	
Discrete / Continuous				
		2	3	
- <u>geal B</u> esources				
<u>)escription:</u>				
Concintor			·····	
Por coor				
Tada manda Dina a sa sa sa				
-xternal Hesources				
_Description: trajectory serboints	machine n		sensor data	error descriptions
Polichility Strong	i Knowledge	e base	sensor interface	knowledge base
Conlácts: not likely				Strong Det likely
Products				
Description: error notilization	halt comm	and		
Attributes: code: parameters	code: para	meters		
Constraints: accuracy of sensors	; <u> </u>			
Likely Recipientsoperator interface	all compor	nents		
Procedures/Methods		Sub	cooncibilities	• • • • • • • • • • • • • • • • • • • •
xample.		Sub-le	sponsibilities.	mm avacutos
model machine as prograi compare model against ei	m executes		compare model agains	error descriptions
 send halt command if mat 	ch		. 5	•
emporalAspects Initiation: automatic feedrate/	orce activated	incom.	ng Messagos:	Mossage- Initiated
Completion: automatic teedrate/	orce deactivated	Cutto	nd Messaces:	- Process:
Discrete / Continuous	11001			
Discrete / Continuous	l			:
1		2	3	4
an Secono				
Deserves				
• • • • • • • • • • • • • • • • • • •				
	1	1		
	+			
xtemal Resources				
xternal Resources Description: feedrate / feedforce	value sensor dat	ia	machine model	
xternal Resources Description: feedrate / feedforce Likely Source: motion	alue sensor dat sensor inte	ta erface	machine model knowledge base	
xternal Besources Description: feedrate / feedforce Likely_Source: motion Beliability: strong	alue sensor dat sensor inte weak	la erface	machine model knowledge base strong	
xternal Besources Description: feedrate / feedforce Likely Source: motion Beliability: strong Convects: not likely	value sensor dat sensor inte weak likely	la erface	machine model knowledge base strong not likely	
xternal Besources Description: feedrate / feedforce Likely_Source: motion Beliability: strong ConVicts: not likely 'roducts	value sensor dat sensor inte weak likely	ta erface	machine model knowledge base strong not likely	
xternal Besources Description: feedrate / feedforce Likely_Source: motion Beliability: strong ConVicts: not likely 'roducts Description: servo commands	value sensor dat sensor inte weak likely	a orface	machine model knowledge base strong not likely	
xternal Besources Description: feedrate / feedforce Likely_Source: motion Beliability: strong ConVicts: not likely Products Description: servo commands Attributes: voltage	value sensor dat sensor inte weak likely	a erface	machine model knowledge base strong not likely	
xternal Besources Description: feedrate / feedforce Likely_Source: motion Beliability: strong ConVicts: not likely Products Description: servo commands Attributes: voltage Constraints: accuracy of machine	value sensor dat sensor inte weak likety model	ta	machine model knowledge base strong not likely	
xternal Besources Description: feedrate / feedforce Likely Source: motion Beliability: strong ConVicts: not likely Products Description: servo commands Attributes: voltage Constraints: accuracy of machine Likely Recipients axis motor	alue sensor dat sensor inte weak likely model	ta interest of the second seco	machine model knowledge base strong not likely	

-

TemporalAspects		Incoming Messages		Message-
Initiation: start of part program			and an an gener	Initiated
Completion: program completed	program completed		(as Managan)	- Process:
Span: execution and prove-	execution and prove-out		ing wessages.	
Discrete / Continuous				
1		2	3	4
Local Pesources Rescription:				
Atrinotae				
External Resources Description: discrete commands Likely Source: plan interpretation Reliability: strong ConVicts: not likely Products Description: actuator commands Attributes: voltage Constraints: device types Likely Recipients:hardware interface Procedures/Methods	machine knowledo strong not likely	model le base	esponsibilities:	
Example: • accept discrete command: • convert for appropriate vol • send voltage	s Itage per device			
• send voltage Responsibility: NGC-74 Ev	valuate part v	vith prot	De	

nesponsibility. NGC-74 Evalua	ate part w	vith prot)e	
Temporal Aspects Initiation: program startup Completion: program shutdown Span: execution and prove-out		linooming lifessages		Message-
		3	inn Magnanan	- ***CD3531
		Ulitgoing Wessages.		,
Discrete / Continuous		t		i
1		2	3	4
Y Electronic				
Dessesses				
				l
1 × 280	-			
External Besources				
Description: part model	ere	adings	machine model	
Likely Source: knowledge base	sensor in	nterface	knowledge base	
Reliability: strong	weak		strong	
Contácts: not likely	not likely	·]	not likely	
Products				
Description: probe selection	probe_co	mmands	movement commands	probe_results
Attributes: code: parameters	code: pa	rameters	_code: parameters	code: parameters
Constraints:				
Likely Recipientstool change mechanism	1 sensor ir	nterface	motion	workstation mgt
Procedures/Methods Example: • access part, machine mode • send command to change t • measure part • send probe results	els lool to probl	Sub-r	esponsibilities: • measure part	

•

٠

Responsibility: NGC-75 E	valuate tool	with pro	be	
TemporalAspects		Incoming Massages		Message-
Initiation: program startup		I		Initiated
Completion: program shutdown	Completion: program shutdown		* · · · * # . · · · · · · · · · · · · · · · · · ·	Process:
Span: execution and prove-	-out	Outgo	ing Messages:	
Discrete / Continuous		T		
	1	•		
		2		4
Local Hesources				
Attributor				
Constraints				
Source:				
External Besources				
Description: tool model	probe reading	IS	machine model	
Likely Source: knowledge base	sensor interfa	ice	knowledge base	
Beliability: strong	strong		strong	
Con1/acts: not likely	not likely		not likely	
Products				
Description: probe selection	probe cor	nmands	movement commands	probe results
Attributes: code: parameters	code: par	ameters	code: parameters	code: parameters
Constraints:				
Likely Recipients:toot change mech	nism sensor in	erface	motion	workstation mgt
Procedures/Methods Example: • access tool, machine models • send command to change tool to probe • measure tool • send probe results				
Responsibility: NGC-76 Au	tomatically le	earn ma	chine'kinematics	
TemporalAspects		incom	ing Messages:	Message-
Initiation: system startup				Initiated
Completion: system shutdown		12 incie	ing Macanasa	Process:

Initiation: system startup		1		Initiated	
Completion: system shutdown		0.000	·	- Process:	
Span: execution and prove-out		l Churgo	រកថ្ម សេខឲនឧត្តខន		
Discrete / Continuous		Ī			
		· · · · · · · · · · · · · · · · · · ·			
1	ļ	2	3	4	
LANCE POLA PRAS					
Depotion					
			· · · · · · · · · · · · · · · · · · ·		
Construction					
Se ree					
External Besources					
Description: machine status	sensed statu	IS	kinematic equations	kin eq. history	
_1 ikely Source: components	sensor interf	ace	knowledge base	knowledge base	
Beliability: strong	weak		strong	strong	
Contacts: not likely	likety		not likely	not likely	
Products					
_Description: modi/ad kinematics					
_Attributes: matrices					
Constraints:					
Likely Recipientsknowledge base					
Procedures/Methods		Sub-re	sponsibilities:		
Example: • access kinematic equa • monitor machine status	itions and histor	Y •	analyze history and stat	us	
analyze history and sta modify and send equat	atus tions				

APPENDIX C - ARCHITECTURE DESCRIPTION RULES

The following rules guide the construction of the components of an application architecture. Some of them are based on requirements, others are based on primitive components. The rules are expressed in simple English, although the architecture description language (ADL) explained in Appendix F could be used for ease in computation. Sequence numbers were added for ease of reference.

Most Important Structural Rule:

If a primitive component (PC NGC-#) needs a resource, you need some other PC that has that resource as a product.

C.1 ADL Rules Accessing Multiple Primitive Components

C.1.1	If you want to control an axis, you need PC NGC-38, Control motors to produce movement, you need PC NGC-49, Handle operator command interactions.
C.1.2	If you have PC NGC-29, you need PC NGC-28, Normalize trajectory with respect to time, you need PC NGC-37, Translate servo commands into drive readable movement commands.
C.1.3	If you want to maximize speed, you need PC NGC-26, Control acceleration and deceleration, you need PC NGC-32, Determine rate of movement.
C.1.4	If you want to optimize cutting, you need PC NGC-4, Maintain information on available tools, you need PC NGC-26, Control acceleration and deceleration, you need PC NGC-32, Determine rate of movement, you need PC NGC-35, Determine trajectory corrections from predictable variations, you need force sensors, you need torque sensors.
C.1.5	If you want to control tool changes, you need PC NGC-21, Load and unload tool, you need PC NGC-20, Map tool id to actual tool location,

- you need PC NGC-19, Swap current tool with required tool,
- you need PC NGC-22, Track tool locations.

- C.1.6 If you want to control pallet changes, you need PC NGC-23, Swap current part with next part, you need PC NGC-24, Identify current part.
- C.1.7 If you have PC NGC-15, you need PC NGC-17, Interpret augmented code, you need PC NGC-18, Determine wait/continue commands for coordination.
- C.1.8 If you are coordinating mechanisms and motion through a high-level language, you need PC NGC-11, Determine where to send part program codes, you need PC NGC-12, Interpretation of part program, you need PC NGC-13, Augment code with component destinations.
- C.1.9 If your high-level language commands are embodied in a part program, you need PC NGC-14, Inform operator of part program comments, you need PC NGC-9, Initialize start of part program activities, you need PC NGC-10, Initialize end of part program activities, you need PC NGC-48, Cue the operator for manual tasks, you need PC NGC-49, Handle operator command interactions.
- C.1.10 If you want to work multiple parts with multiple programs as a job, you need PC NGC-3, Determine each part's program and coordinate offsets, you need PC NGC-7, Determine status of active program, you need PC NGC-5, Ensure that resources required by the jobs are available, you need PC NGC-6, Determine if next part's program matches current part's program, you need PC NGC-62, Initiate job sequence.
- C.1.11 If you want to create new parts, you need PC NGC-59, Create a part design, you need PC NGC-53, Generate a part program. you need PC NGC-1, Handle operator modifications to part program statements, you need PC NGC-54, Customize part program for particular machine, you need PC NGC-55, Schedule processes in part program for optimal coordination.
- C.1.12 If you want to simulate operations, you need PC NGC-56, Simulate part program execution, you need PC NGC-57, Simulate software faults, you need PC NGC-58, Simulate hardware faults.

C.2 ADL Rules Linking Primitive Components

- C.2.1 If you have PC NGC-34, you need PC NGC-36, Adjust trajectory with corrections generated by sensor data.
- C.2.2 If you have PC NGC-33, you need PC NGC-4, Maintain information on available tools.

	C.3	ADL Rules Accessing Single Primitive Components
	C.3.1	If you want the operator to customize the part program, you need PC NGC-1, Handle operator modifications to part program statements.
-	C.3.2	If you are going to store part programs for automatic access, you need PC NGC-2, Part program storage activities.
	C.3.3	If you want to execute a job list of multiple arts and programs, you need PC NGC-3, Determine each part's program and coordinate offsets.
	C.3.4	If you have dynamic tool data, you need PC NGC-4, Maintain information on available tools.
	C.3.5	If you want to execute programs with various resources, you need PC NGC-5, Ensure that resources required by the jobs are available.
	C.3.6	If you want to execute various programs, you need PC NGC-6, Determine if next part's program matches current part's program.
	C.3.7	If you need to keep track of the executing program, you need PC NGC-7, Determine status of active program.
	C.3.8	If you cannot load an entire program at once, you need PC NGC-8, Request additional segments of part program.
	C.3.9	If you need to perform activities at the start of every program, you need PC NGC-9, Initialize start of part program activities.
	C .3.10	If you need to perform activities at the end of every program, you need PC NGC-10, Initiate end of part program activities.
	C.3.11	If you need to distribute code to various components, you need PC NGC-11, Determine where to send part program codes.
	C.3.12	If you need to parse code for motion and mechanisms, you need PC NGC-12, Interpretation of part program.
	C.3.13	If you need to track recipients of code, you need PC NGC-13, Augment code with component destinations.
	C.3.14	If you want the operator to follow the program execution, you need PC NGC-14, Inform operator of part program comments.
	C.3.15	If you want to coordinate mechanism control with motion control, you need PC NGC-15, Augment code for coordination between motion and mechanisms.

C.3.16	If you have multiple coordinate systems, you need PC NGC-16, Translate coordinate systems.
C.3.17	If you need to track program execution, you need PC NGC-17, Interpret augmented code.
C.3.18	If you need to coordinate notion and mechanisms, you need PC NGC-18, Determine wait/continue commands for coordination.
C.3.19	If you have multiple tools, you need PC NGC-19, Swap current tool with required tool.
C.3.20	If you have dynamic tool location assignments, you need PC NGC-20, Map tool id to actual tool location.
C.3.21	If your operator places tools in the tool changer, you need PC NGC-21, Load and unload tools.
C.3.22	If you the tool changer swaps tool locations, you need PC NGC-22, Track tool locations.
C.3.23	If the part handler can automatically change parts, you need PC NGC-23, Swap current part with next part.
C.3.24	If you need to verify the loaded pallet, you need PC NGC-24, Identify current part.
C.3.25	If you want to control a continuous flow device, you need PC NGC-25, Maintain control of feedrate-dependent operations.
C .3.26	If you want to ensure the trajectory follows the machine's physical limitations, you need PC NGC-26, Control acceleration and deceleration.
C.3.27	If you need to modify the trajectory, you need PC NGC-27, Define normals to path.
C.3.28	If you need to follow a trajectory based on setpoints, you need PC NGC-28, Normalize trajectory with respect to time.
C.3.29	If you want to enter high-level commands instead of axis commands, you need PC NGC-29, Translate motion commands to trajectory.
C.3.30	If you need motion and mechanisms to coordinate, you need PC NGC-30, Coordinate with mechanisms.
C.3.31	If you want to avoid witness marks, you need PC NGC-31, Avoid data starvation.
C.3.32	If you want the goal rate to adjust for a chip per tooth rate, you need PC NGC-32. Determine rate of movement.

C.3.33	If you want to adjust for tool variations, you need PC NGC-33, Adjust trajectory for tool deviations.
C.3.34.1	If you want to adjust for predictable variations in axis linearity, you need PC NGC-34, Determine trajectory corrections from predictable variations.
C.3.34.2	If you want to adjust for predictable variations with respect to temperature, you need temperature sensors, you need to track time of machine powered up, you need PC NGC-34, Determine trajectory correction from predictable variations.
C.3.35	If you want to adjust for unpredictable variations in trajectory following, you need PC NGC-35, Determine trajectory corrections from sensed variations, you need force sensors, you need torque sensors.
C.3.36	If you want to quickly modify the trajectory with pre-analyzed corrections, you need PC NGC-36, Adjust trajectory with corrections generated by sensor data
C.3.37	If you need to interpolate between current position and setpoints, you need PC NGC-37, Translate servo commands into drive readable movement commands.
C.3.38	If you need a closed loop motion control, you need PC NGC-38, Control motors to produce movement.
C.3.39	If you need diagnostics and zero positions determined, you need PC NGC-39, Initiate startup procedures.
C.3 .40	If you want automatic job control by the factory scheduling system, you need PC NGC-40, Coordinate with factory scheduler/control system.
C.3.41	If you want to generate paths with as-is / to-be algorithms, you need PC NGC-41, Path planning based on part features, surface model, etc.
C.3.42	If you want sensor data to be used for program modifications, you need PC NGC-42, Modify part program based on sensor data.
C.3.43	If you can control axes by both force and position, you need PC NGC-43, switch between position control and force control.
C.3.44	If you have more than one axis, you need PC NGC-44, Plan for multi-axis interactions.
C.3.45	If you want automatic setup of parts, you need PC NGC-45, Touch off for automatic setup, you need touch sensors.
C.3.46	If you have multiple sensors providing one piece of information, you need PC NGC-46, Perform sensor fusion.

à

•

_

C.3.47	If you want to gracefully shutdown form exception conditions, you need PC NGC-47, Initiate shutdown procedures.
C.3.48	If you expect the part program to require interaction with the operator, you need PC NGC-48, Cue the operator for manual tasks.
C.3.49	If you expect the operator to control the program execution, you need PC NGC-49, Handle operator command interactions.
C.3.50	If you want the operator to enter offsets, you need PC NGC-50, Handle modifications of offset values by operator.
C.3.51	If you expect the operator to perform manual interpolation, you need PC NGC-51, Handle operator interface for manual interpolation commands.
C.3.52	If you need a system watchdog, you need PC NGC-52, Ensure safety.
C.3.53	If you want to generate programs with as-is / to-be algorithms, you need PC NGC-53, Generate a part program.
C.3.54	If you need to execute the same program on various machines, you need PC NGC-54, Customize part program for particular machine.
C.3.55	If you want to improve the efficiency of the part program, you need PC NGC-55, Schedule processes in part program for optimal coordination.
C.3.56	If you want to test the system without cutting a part, you need PC NGC-56, Simulate part program execution.
C.3.57	If you want to introduce software errors into the system for testing, you need PC NGC-57, Simulate software faults.
C.3.58	If you want to introduce hardware errors into the system for testing, you need PC NGC-58, Simulate hardware faults.
C.3.59	If you want to generate a manufacturable part design from a concept, you need PC NGC-59, Create a part design.
C .3.60	If you want maintenance to be informed automatically of problems, you need PC NGC-60, Notify maintenance when parameters exceed tolerances.
C.3.61	If you want to keep machine, part, tool (etc) models in a central knowledge-base, you need PC NGC-61, Maintain models.
C.3.62	If you want to setup multiple executions in one job list, you need PC NGC-62, Initiate job sequence.

.

Y

C.3.63	If you want to ensure the right program is loaded on the right machine for the right
part,	you need PC NGC-63, Verify proper part program and revision number for each part.
C.3.64	If you need to verify the part in the fixture, you need PC NGC-64, Identify current part.
C.3.65	If you want to modify the trajectory to stay within safe zones, you need PC NGC-65, Ensure sanctity of safe zones.
C.3.66	If you want to allow the operator to perform some machine functions, you need PC NGC-66, Reassignment of tasks between machine and operator.
C.3.67	If you want to coordinate motion with the spindle gear setting, you need PC NGC-67, Coordinate spindle gear changes with servo control.
C.3.68	If you allow substitute tools with different offsets, you need PC NGC-68, Re-evaluate coordinate zeros for alternate tool choices.
C.3.69	If you want to follow complex curves, you need PC NGC-69, Translate NURBS trajectories into motion commands.
C.3.70	If you want to program using canned cycles, you need PC NGC-70, Manage canned cycles.
C.3.71	If you want to warn of impending movement errors, you need PC NGC-71, Notification of errors such as singularity, lashup, unreachable points.
C.3.72	If you want the system to maintain a constant rate or force, you need PC NGC-72, Automatic feedrate / feedforce at point of contact.
C.3.73	If you want to control discrete actuators, you need PC NGC-73, Executing plans targeted at discrete actuators.
C.3.74	If you want to measure for part accuracy, you need PC NGC-74, Evaluate part with probe.
C.3.75	If you want to measure for tool accuracy, you need PC NGC-75, Evaluate tool with probe.
C .3.76	If you want the machine to learn its own physical nature, you need PC NGC-76, Automatically learn machine's kinematics.

.

APPENDIX D — DOMAIN MODELS

Domain models describe the domain. They include a description of, actions of, relationships to, communications among, and constraints of entities. The implementation may follow an objectoriented paradigm or a functional or other paradigm. Some representations of domain models include object diagrams, task diagrams, topology diagrams, and interaction diagrams. Some domains may be better expressed in specialized models, such as algorithm models or hybrid control models.

While domain models contain a full view of the domain, the models' appearance changes to support multiple viewpoints. Particular objects, subsets of attributes and services, and certain relationships are only relevant to certain views. Viewpoint also drives the collection of responsibilities into components and the relevance of reference requirements. The domain models for NGC are listed below.

D-1



è

۵













h.

۵

.



.

September 26, 1994

¥



١,



¥

D-9


An Example of a Controller Topology



Topology for a Less-Sophisticated Controller



an Interpretation							
an Interpretation							
				and a second and	an an frankriger frank		
				and the second sec			
G code		int brace and barrent at	معدا ليبد تشايينتوردي	متبدع والمتباد والمت	فعسو سكماتهم وراتيه	are de artes a	
determine trajectory					2 2 2 2 2	er yester te gjit torson og en er en er	1. 1. 19.27 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
Childre Competention		Sector					
	nath segments	and the second	wall lessed for	nellinge de fra		<u>h</u> tin 17 <u>51 - 15</u> 53	
mickenterssenies of points							
controllaws				AN 1997 1997	and a stand	and a strengt	
	joint move command			، القاسم الشائلة	1.51 63-74	E AL	
lachine Interface	vol	lage					
ditiv				12 10 10 10 10 10 10 10 10 10 10 10 10 10			
lachine					劉劉	劉國	樫
molor							
					20 2007 20		

TIME STEPS

(not to scale)

An Interaction Diagram Illustrating a Tool Change

Block of RS-274 code										
IDE IN SIE	<u></u>	<u>المنحد منحد ،</u>	نه . <u>س</u> نهست					and the second second	and a second second second	
Plan Interpretation	all said and an and an and an and an and a second s									
Commences Construction	- 7								kar 2000 ang 1	
A CONTRACTOR OF CONTRACTOR	Long to the Lancestonia	شدهده، و مردین سند. محمد مرد ریست. سمره								75
	Grodes							مذهبيتين وتروحكن		
determineurajector									an formal a transmission	
Trajectory Generator	7,0	1.1.1								114
culler compensation										A al
		nath saor	mente I I				4		والتد تسديعه	曲马
THE CONSTRUCTION OF THE STATE	<u></u>			an said	S-4690.40	New York		alora al		
Actuator Control						1.			1.0	調調
CONTONE .					a e . g					井크
			16 CCW	nanas n	i di internet d Internet di internet di inte		11 12 - 12 - 12 - 12 - 12 - 12 - 12 - 12	nt from 21 Acres	1.15. C S. 15	-
Machine Interface						است فر اس ک	1		1997 g 100 1	
				voilage	┧╼╄╼╄	┝┥╍╿┥╍┥	<u>↓</u>	╉╍╄╍╄╍	<mark>┥╼╌╄╌┿╾</mark> ┦┺╌╄	dd ar a sha a s
ALL SALES								EBARE		町名
Alachine		- 21-1	-	Sector					包地到	
								SELE		
1690200X		C					50		could be stated	
าราวาราวารา	1									
		1 2 13035	routies.	Fanzing.		ter far		70101		
		t postions	<u></u>	intoolehi	and an air an an an air an	10000000000		- position		
									\rightarrow	

TIME STEPS

(not to scale)

ŧ.

غ



APPENDIX E — DOMAIN DICTIONARY

This dictionary contains the terminology and definitions from the domain of manufacturing in general and in the domain of the Next Generation Controller in particular. The dictionary is divided into several sections: a bibliography, a glossary of terms, notation, and a category index. The bibliography lists the information sources used. Sources include books, documents generated by NCMS members, commercial brochures, and notes from knowledge acquisition sessions. The glossary lists all terms (which are boldfaced) and their definitions alphabetically. Each definition includes a bibliographical annotation of the form [#] to indicate its source. Where a term has definitions from more than one source, each definition is annotated separately. The notation section includes tables of symbols, nomenclature, and units. The category index lists all terms divided into meaningful categories. Although a term may apply to more than one category, Most are listed only once in one category.

E.1 Bibliography

- [1] Allen Bradley, PAL Programmer's Guide
- [2] Asimov, Isaac & Karen A. Frenkel, Robots: Machines in Man's Image, Nightfall Inc. (1985). [TJ 211.A83]
- [3] Automation Intelligence, IntelliPost Command Reference Manual for Turning Equipment (Mar 93).
- [4] Hurco, "NGC Architecture and Modeling Project: Scenarios" (Jun 93).
- [5] IBM, "Dictionary for the ADAGE of the DSSA Project" (31 Jul 92).
- [6] IBM & Northrop, "Enterprise Integration Framework Next Generation Workstation/ Machine Controller Glossary" (24 Jan 91). This document contains many other software engineering and manufacturing terms that were not included here.
- [7] Martin Marietta, "NGC SOSAS Volume I" (Mar 92 draft).
- [8] New England Affiliated Technologies, product catalog.
- [9] Walker, John R., Machining Fundamentals (1977). [TJ 1160.W25]
- [10] Wizdom, sales brochures.
- [11] Dictionary of the English Language: 2nd Edition unabridged, Random House (1987). [PE 1625.R3]
- [12] Webster's New Universal Unabridged Dictionary: 2nd Edition, Simon & Schuster (1983).
- [13] Webster's 3rd New International Dictionary: unabridged (1981). [PE 1625.W36]
- [14] The Dictionary of Computing, Oxford University Press (1983). This reference was given in [6].
- [15] Knowledge Acquisition Session with Mike Wright, Hurco (20-21 May 93).
- [16] Knowledge Acquisition Session with Lester Godwin (6 Jun 93).
- [17] Knowledge Acquisition Session with Whitey Simon & Dan Ma, GM, and Tom Carbone, Hughes (10-11 Jun 93).
- [18] Knowledge Acquisition Session with Mike Nelson, Mayday (8 Jul 93).
- [19] Knowledge Acquisition Session with Automation Intelligence (25-26 Aug 93).
- [20] Knowledge Acquisition Session with Ken Stoddard, Trellis.
- [21] Knowledge Acquisition Session with Wizdom.
- [22] Computer Graphics: Principles and Practice: 2nd Edition.

E.2 Glossary of Terms

2.5 D

a 2D representation (topographical map) of a 3D object. 2.5D cannot be rotated, because the necessary 3D information is only from one perspective, so there is insufficient information to produce a 3D model. [16]

abbe error

a linear positioning error caused by a combination of an angular error in the ways, and an offset between the precision determining element (leadscrew, feedback device, etc.) and the actual point of interest. [8]

abort

to cease normal operations [5]

abrasive waterjet machine (AWJ)

a machine which performs material removal via a stream of water containing an abrasive solute at high pressure.

absolute

applies to measurements, in a standard, fixed reference, as opposed to moving reference; compare with relative [5]

absolute value

common math function; synonym for magnitude; standard notation lxl; for vectors, absolute value means length (magnitude) of the vector [5]

abstract data type

1 a non standard, user defined data type. [6]

2 a class of objects together with a set of operations which can be applied to them. The objects are entirely characterized through the external behavior of the operations. Neither the storage representation nor the implementation of the operations should be known to the user. [6]

AC

abbreviation for alternating current.

acceleration

rate of change of velocity; either scalar or vector, often with subscripts to denote the coordinate frame; time derivative of velocity; time integral of jerk; standard symbol a, A; standard units ft/s^2 , g; primary units L/θ' [5]

accelerometer

an inertial device for measuring acceleration, usually in three orthogonal axes (lateral X, longitudinal Y, and vertical Z); accelerometers usually consists of a mass, spring, and damper, accelerometers are usually included in inertial sensors [5]

AC induction motor

a device that uses alternating current to produce mechanical energy. These motors do not have brushes and are simple to operate as well as inexpensive to produce. However, they are very difficult to control. DC brushless motors are used as AC servo motors. [4]

accuracy

1 measure to exactness, possibly expressed in percent, for example, position measured +10%; compare with precision. [5]

-2 extent of agreement of measured/reported value with true magnitude. [7]

3 extent to which a given value for a measurement agrees with the standard for that measurement. The degree of correctness of a quality or expression. [11]

activity

a repetitive, well-defined set of tasks that have common functional characteristics. [7] actuator

a motor or transducer that converts electrical, hydraulic, or pneumatic energy to motion. [7]

adaptive control

a response to time-varying characteristics of the machinery, material, and process involving an adjustment of elements in the feedback control loop. [7]

ADL

abbr for Application Development Language.

AGV

abbr for Automated Ground Vehicle.

AI

abbr for Artificial Intelligence.

aiding

a process by which one or more sensors provide data to another sensor to produce results better than any single sensor. Aiding occurs at the physical device level, depending upon specific implementation of the device. Aiding is automatically controlled by software without input from an operator [5]

algorithm

1 a sequence of mathematical operations that precisely performs a specific task. [4]

2 a finite set of well defined rules for the solution of a problem in a finite number of steps. [6]

American National Standards Institute (ANSI)

an organization consisting of producers, consumers, and general interest groups, that establish the procedures by which accredited organizations create and maintain voluntary industry standards in the United States. [6]

AMICE

abbr for European Computer Integrated Manufacturing Architecture (in reverse)

analog input/output

continuous data input or output presented as a measurable quantity such as voltage and current. [7]

angle

synonym for angular position; standard units rad, deg; primary units 1 [5]

angular acceleration

rate of change of angular velocity, either scalar or vector, often with subscripts to denote the coordinate frame; time derivative of angular velocity; time integral of angular acceleration; standard symbol α ; standard units rad/s'; primary units 1/ θ ' [5]

angular position

amount of rotation about an axis, either scalar or vector, often with subscripts to denote the coordinate frame; time integral of angular velocity; synonym for angle; standard symbol θ ; standard units rad, deg; primary units 1 [5]

angular velocity

rate of change of rotation about an axis, either scalar or vector, often with subscripts to denote the coordinate frame; time derivative of angular position; time integral of angular acceleration; see tachometer; standard symbol ω ; standard units rad/s, rpm; primary units 1/ θ [5]

ANSI

abbr for American National Standards Institute.

aperiodic

a process that executes based on events rather than a fixed rate, it is not synchronized to other processes of interest; compare with periodic [5]

API

abbr for Application Programming Interface.

application program

a program that, when executed on a NGC computing platform, performs designated functions. [7]

application programming interface (API) an interface that provides modules access to vendor components, shared components, and computing platform services. [7]

application-specific integrated circuit (ASIC)

a custom hardwired chip that performs a desired function.

application viewpoint

a DSSA view of a system.

approach vector

machine's normal approach of moving to the piece. Can be dynamically changed. [20]

architecture description language (ADL)

an ADL describes components' structures and specifies their protocols.

architecture viewpoint

a DSSA view of a system.

Arcnet

Arcnet is classified as a deterministic, high speed 2.5 Mb/sec, peer-to-peer token passing LAN. Arcnet is deterministic in that is allows the user to calculate the worst case latency of the network based on the number of nodes in use. Arcnet provides the software with a high-level interface that minimizes software overhead. Packets are written to the Arcnet controller along with the destination address and a single command is given to begin message transmission. The completed transmission and reception of a packet can be signaled with an interrupt.

Arcnet can be used by a number of network configurations. It can be used in a star configuration which uses active and passive hubs to grow the network in any shape required. Arcnet also supports a bus - configuration which uses a high impedance version of the network transceiver on each node, and enables several nodes to be connected to a bus section of cabling by short cable spurs. Star and bus configurations can be mixed to provide the most effective layout for any installation.

Arcnet supports multiple media. Physically, the connection can be made with coaxial, fiber optic, and twisted pair cabling including 485. typically, the network is tied together with inexpensive RG62 coaxial cable, making it highly immune to electrical noise. A network can be up to 4 miles long. the maximum node-to-node or node-to-hub distance is 2000 feet.

Arcnet was designed to allow nodes to be added or removed during live everyday use while the network is running. There is no need to shut down. The protocol will automatically reconfigure itself within 60 milliseconds of a node being removed or failing. [10]

arctangent

common math function; standard notation arctan x, atan x, $\tan^{-1}x$, $\tan^{-1}(y,x)$; $\tan^{-1}(y,x)$ means two-argument arctangent of y/x [5]

area clear milling

method of surface finishing in which the tool follows a simple back-and-forth pattern across the part surface. [15] an interconnected set of links and articulated joints between an end effector and its support structure. [7]

¥

artificial intelligence (AI)

the ability of a process to perform functions normally associated with human intelligence, such as reasoning, learning, and self-improvement. A technology or field of study that encompasses all attempts to emulate or reproduce human neurological functions, such as cognition, perception and action, through symbolic means. [14]

ASCII

abbr for American Standard Code for Information Interchange.

ASIC

abbr for Application-Specific Integrated Circuit

assembly

a group of parts and/or subassemblies that are put together. An assembly may be an end item or a component of a higher level assembly. [6]

asset

any expendable resource within the manufacturing workstation. Asset does not include the machines under control. [7]

async abbr for Asynchronous.

asynchronous

arm

tasks or events that occur at non-fixed time intervals and are large independent of one another; compare with synchronous [7]

attribute

- 1 data element that represents a characteristic used to describe an object. [7]

2 a quality or characteristic element of an entity, having a name and a value; an item of information about an entity; properties which describe data objects. [6]

autonomous control

independent control over the sequence, options, and schedule of tasks. [7]

availability

the measure of time that a manufacturing workstation is free to carry out its designated functions. [7]

averaging filter

a filter for combining multiple data sources, usually of the same type, by adding with weighted averages; a simple average of the data sources; compare with complementary filter, Kalman filter.

Averaging filters are often designed as:

$$y = \sum_{1}^{n} \alpha_{ixi}$$

where α_i are usually computed at run time such that

[5]

$$\sum_{i=1}^{n} \alpha_{i} = 1$$

AWJ

abbr for Abrasive WaterJet machine.

September 26, 1994

axis

1 one direction in an orthogonal reference frame [5]

2 Used to locate or move the part. An axis is mathematically described as one of the reference lines of a coordinate system, such as Cartesian coordinates (x, y, and z). There can be x, y, z, a, b, c, u, v, and w axes. [4]

3 Any line used as a fixed reference in conjunction with one or more other references for determining the position of a point or of a series of points. [11]

4 the center line, real or imaginary, passing through an object about which it could rotate. a point of reference. [9]

backlash

the amount of free play between a leadscrew and nut, or worm and worm gear. It is determined by measuring the range of angular movement of the driven shaft which results in no motion. [8]

band-pass filter (BPF)

a filter that allows frequencies between two cutoff frequencies to pass while attenuating frequencies outside the cutoff frequencies; a band-pass filter can be constructed as the composition of a low-pass filter and a highpass filter [5]

band saw

a type of saw machine in the form of a continuous steel belt running over pulleys. [13]

batch execution

a condition that occurs when the controller is in the normal production mode. Interaction may be required initially, but file-based operations are generally thought of as being "automatic". This operation is - characterized by the execution of internally represented plans. and facilitates part/process production while performing planning operations. A file typically consists of pre-generated batched control plan or process plan information that is executed automatically from start to finish. [7]

BCD

abbr for Binary Coded Decimal.

BCL

abbr for Binary Cutter Location.

binary cutter location (BCL)

a numeric control programming and interpretation scheme that is based on a generic machine reference. This eliminates the need for post processing the CNC program for a specific machine or type of machine. CNCs designed for BCL interpretation can execute the BCL instructions directly. [6]

bias

an offset applied to a measurement for error correction; standard engineering term synonym for offset [5]

bed

machine's cutting surface, also referred to as the "table"; the part is attached to this surface with a clamping device. [4]

bed mill

a.k.a. mill or machining center. A type of machine that is more powerful than a knee mill and can be used to cut large parts in large quantities. The bed can travel horizontally or vertically on a bed mill, and the cutting area is usually enclosed. [4]

block

a group of instructions from a part program to control machine actions. A block usually describes a discrete movement or action such as cutting a bolt circle. [4]

block cycle time

time it takes to execute one line of executable RS-274 code [15]

blue a finishing task.

BOC

abbr for Buick, Oldsmobile, Cadillac; a now defunct GM organization.

bolt

a type of fixture; any of several types of strong fastening rods, pins, or screws, usually threaded to receive a nut. [11]

bore

a machining task; to make a hole in a material by turning a tool.

BPF

abbr for Band-Pass Filter [5]

break chip

temporarily stop z-motion of the tool to break off the chip. [15]

broach

¥

1 v. a machining task; to cut with a broach tool.

2 n. a cutting tool for removing material from metal or plastic to shape an outside surface or hole that has been previously formed, consisting of a bar of suitable length provided on its surface with a series of cutting edges or teeth that increase in size from the entering or starting end. [13]

brushless motor

an "inside-out" DC motor, with a permanent magnet rotor, and electrical coils in the stator. Commutation of current in the windings is typically achieved via external switching transistors, and Hall effect detectors. This avoids the limited life of brushes and their radiated EMI. [8]

CAD

abbr for Computer-Aided Design.

calibration

1 an interactive condition that occurs when the controller is in the maintenance mode; starts processes unique to a specific machine or process that enables the operator to fine tune the controller. This mode shall be used to adjust control law gains, modify axis soft limits, input positional offsets, verify setpoint accuracy, adjust overshoot and balance process parameters, or adjust other known model inaccuracies though the use of error correction tables and/or other methods. It is assumed that this mode will rely heavily on information provided by the operator interactively. It is also assumed that manual operation of the machine may be required in this mode. [7]

2 calibrate sequence is used to determine a reference point or zero location on a part.[4]

CAM

abbr for Computer-Aided Manufacturing.

CANBus

an I/O bus used by numeric controllers.
 [4]

2 CAN could be classified as a Small Area Network (SAN). It runs on an inexpensive twisted pair bus and provides an interface to I/O and other peripherals, peer-to-peer networking of controllers, deterministic prioritized message passing, small efficient packet sizes, robust error detection, and a reduced wiring solution. CAN is an industry standard protocol supported by multiple vendors. Packets are written to the bus controller along with priority and the destination address. A single command is given to begin message transmission. The completed transmission and reception of a packet can be signaled with an interrupt. The data transmission rate can be up to Imb/sec for networks less than 130 feet in length. Lower data rates can support longer networks. Up to 32 nodes can be attached to a given network.

CAN is a deterministic network with prioritized message passing. Worst case message transmission latency in a CAN network can be calculated based on the number of nodes in use and the priority of the message. The use of prioritized

messages guarantees that critical messages are transmitted ahead of lower priority ones. [10]

canned cycle

- common machining operations, such as drilling, tapping, pecking, and boring and reaming, that are pre-programming into a CNC. These cycles allow the operator to specify a type of operation and define only the necessary variable information. Then the system makes additional calculations to define the operation fully. [16]

cache

 an architected area of computer main storage for system and/or application use.
 [6]

2 the virtual storage within which objects are presented to methods. [6]

3 an architected area made up of control structures and pointers that point to a variable number of Logical Views. The Logical Views may contain internal and/or external references to the same/other Logical views which are resolved as relocatable addresses when loaded into The entire cache computer storage. structure may be treated as a unique entity and may be modified, saved and restored as a means of maintaining the context of a user or unit of work between user logons or automatic activation of enterprise activities. [6]

CALS

abbr. for Computer-aided Acquisition and Logistics Support. [6]

canonical

conforming to a standard form of semantics and syntax. [6]

CCLW

abbr for Counterclockwise.

cell

1 a manufacturing unit that has the capacity to manufacture a family of products that are a subset of a product type within a fixed domain of operations. It may contain several manufacturing workstations. [7]

2 a manufacturing unit which has the capacity to manufacture a family of products which are a subset of a product type (e.g. sheet metal, composites, circuit card, etc.) and which share the same manufacturing operations with minor changes. Therefore, a cell is a manufacturing unit which has the ability to produce a family of parts within a fixed domain of operations. [6]

center

a manufacturing unit consisting of two or more cells and the materials transport and storage buffers that interconnect them. [7]

centerless grinding

a type of grinding.

CEP

abbr for Circular Error Probability [5]

chatter

the quick repeated sounds resulting from a machine tool vibrating against the workpiece while the machine is in operation. [7]

checkpointing

a recovery mechanism that take a snapshot of everything pertinent to system state, producing a virtual memory page, and forces this page periodically to disk. [7]

chip

\$

a small residue piece of material remaining from cutting, shaping or finishing a part with a machine tool. [7]

chuck

a type of fixture; a device for centering and clamping work in a lathe or other machine tool. [11]

CIM

abbr for Computer Integrated Manufacturing

circular error probability (CEP)

a probability that a percentage of twodimension measurements will lie within a circle of given radius, with the circle centered at truth or mean of the measurements; compare with radial error probability, spherical error probability

CEP specifies test cases for measurement errors of sensors of two dimensions, such as X and Y. For example, a velocity error of 1 ft/s (50% CEP) means that for any given measurement of velocity

then

$$p(\left|\vec{V}_i - \vec{V}_{\mu}\right| \le 1) = 0.5$$

 $\bar{V}i = \{VE, VN\}$

where V_{μ} is the average of all V_i or V truth, depending upon context. [5]

clamp

a type of fixture; an appliance with opposite sides or parts which may be adjusted or brought closer together to hold or compress something. [11]

class

1 the object-oriented paradigm's abstract data typing mechanism that groups objects with commonalty of attributes and services.

2 a type of data object. There are primitive classes or types such as fixed, character, float, and bit. There are also more complex types that are constructed from the primitive types. [6]

3 a group of objects which carry the same instance variables, the same methods, and respond to the same messages. [6]

class hierarchy

defines superclass and subclass relationships and supports the is-a relationship. [6]

climb milling

milling in which the tool motion is against the tool rotation, such that feedrate + rotation = actual speed of cutting tool; compare with conventional milling [15]

clockwise (CLW)

positive rotational direction.

close dancing

two robots working on the same thing independently without interfering with each other. Matching parts takes 12 degrees of freedom (position, velocity, etc.), 6 degrees of freedom on each of 2 parts (force, torque, etc.). [19]

closed loop

control achieved by measuring the degree to
which actual system response (feedback) conforms to desired system response and using the difference to drive the system in conformance. Also called feedback control. [4]

CLW

abbr for Clockwise.

CMM

abbr for Coordinate Measuring Machine.

CNC

abbr for Computer Numerical Control.

coefficient of friction

the ratio of the force required to move a given load to the magnitude of that load. [8]

command

a signal that actuates a device. [7]

commercial off-the-shelf package (COTS)

a software system that is commerciallyavailable and not specifically targeted toward NGC system needs. [7]

commit

the act of writing buffered data to permanent storage upon completion of a transaction. Commits are usually carried out periodically for groups of data. [7] communications viewpoint

a DSSA view of a system.

complementary filter

a filter in which the complement of the filter is desired, for example {EMBED Equation I}, giving the effect of a high-pass filter implementing a low-pass filter; a filter for combining multiple data sources, usually of different types, by adding filtered values, where the sum of the filters in the frequency domain is unity; a Kalman filter with fixed gains; compare with averaging filter, Kalman filter λ

Complementary filters are often designed in the frequency domain as

$$Y(s) = \sum_{1}^{n} F_{i}(s) X_{i}(s)$$

where $F_i(s)$ are filters determined at build time such that

$$\sum_{i=1}^{n} F_i(s) = 1$$

For example, with velocity sources V_x and V_y , one might choose

$$V_{sys}(t) = f_{LPF}(V_x(t) + f_{HPF}(V_y(t)))$$

where the cutoff frequency of the LFP is equal to that of the HPF. [5]

component viewpoint

a DSSA view of a system.

computer-aided design (CAD)

the use of computers in interactive engineering drawing and storage of design. Programs complete the layout, geometric transformations, projections, rotations, magnifications, and interval (cross-section) view of part and its relation with other parts. [6]

computer-aided manufacturing (CAM)

1 the use of computers to program, direct, - and control production equipment in the fabrication, assembly, and distribution of manufactured items. [6]

2 the application of a computer in a manufacturing environment to bridge various systems and connect them into a coherent, integrated whole. For example, budgets, CAD/CAM, process controls, group technology systems, MRPII, financial reporting systems, etc., would all share data in an integrated environment. [6]

3 the application of information systems technology to increase the productivity and responsiveness of the organization. [6]

4 a business philosophy aimed at reducing the development time of industrial products, increase its quality and create an environment where the production of goods will be based on market requirements. Use of (C)omputer technology to enable us to work faster and more precisely in dealing with complex problems. (I)ntegration means that we make our enterprise work as an undivided unit. Transferring information about our products, plans, methods and decisions effortlessly and without risk of introducing errors. (M)anufacturing refers to all industrial activities necessary for design. manufacture, installation and maintenance of products. [6]

5 a computer-based technology that helps define and create a program to drive a CNC machine. CAM generates tool path. [4] in a periodic, cyclical computer system, the most basic, fastest timing loop [5]

computer numerical control (CNC)

controlling machine tools using 1 microcomputers attached to the machines in some manner. The computer controls the machining sequence of a machine to provide faster, more accurate production and the capability to make parts that are essentially impossible to create by hand control. Within the machine industry, the icontrolî is usually the enclosure and the computer boards within it that contain the machine software and handle controlling communications between the machine and the operator console. [4]

2 a type of machine controller designation, in which the computer stores and recalls programs. [16]

computing platform

the processing infrastructure of the NGC manufacturing workstation consisting of processor(s), related hardware, system, and system utility software, and communication mechanisms. [7]

concurrency

proper control of concurrent updates made on shared data so that inconsistent data does not result. The results produced by multiple concurrent transactions are equivalent to results that would be produced by serial transactions. [7]

configuration

the arrangement of a manufacturing workstation as defined by the nature, number, interaction, and main characteristics of its functions. The features,

computer cycle

customer options, software modules and engineering specifications that collectively define the functionality of the manufacturing workstation. [7]

- configuration management

establishing a process for managing and controlling changes to functional and physical characteristics during the life cycle; this provides tracability and status of change activities to previous configurations. [7]

conformance

action or behavior in correspondence with current customs, rules, or styles. In particular, behavior in correspondence with SOSAS rules requirements, and styles. [7]

conformance class

a set containing members having common behavior in correspondence with a defined set of (SOSAS) rules, requirements, and styles. [7]

consumable

material that is physically used or transformed by a production process in the completion of that process. It includes items such as coolants and lubricants. [7]

context

the conditions that bound and relate discrete events temporarily, spatially, or semiannually. [7]

continuous path control

motion control of a device that is not planned to stop at intermediate points along a path. [7]

continuous time

time which can have any point expressed as a real quantity, without regard for any specific interval or processing rate; compare with discrete time [5]

continuous-time equation

a mathematical relationship to describe a function of time, expressed in terms of continuous time; compare with difference equation, differential equation, discrete-time equation, Laplace transform, Z transform; see first order filter, second-order filter, unit functions for examples.

Example standard notation for continuoustime equations is: [5]

$$y(t) = \frac{-5}{16}e^{-t} + \frac{5}{16}e^{3t} - \frac{1}{4}te^{-t}$$

control law

the mathematical definition of a system used to control or to change the dynamic response of a system. [5]

control plan (CP)

a plan that is executable by a specific machine within the workstation. [7]

controls standardized application (CSA)

provides both continuous and discrete control of an individual machine. The SA receives machine specific task commands from the task execution SA, and produces effector commands to support all modes of operation. Process control commands are also decomposed to a sequential set of discrete effector commands and sent to SESA. Coordination between machine specific motion and process commands is provided by this SA. [7]

conventional machine tool

a machine tool that is not controlled by - computers. [7]

conventional milling

milling in which the tool motion is with the tool rotation; compare with climb milling [15]

conversational part programming

method implemented by software within the control that allows part programming in a question/answer format on the operator console. This method of part programming uses multiple choice and fill-in-the-blank questions, as well as clearly worded operator prompts. It is the method not the software. [4]

coolant

a cutting fluid that keeps the tool and part cool to prevent reduction in hardness and resistance to abrasion. It therefore helps prevent distortion of the work. [4]

coolant mechanism

delivers coolant in either a flood, tap, or mist fashion to prevent the tool or part from overheating.

coordinate

any of the magnitudes which serve to define the position of a point in terms of a fixed reference frame such as coordinate axis. [17]

coordinated joint

a joint that is coordinated to begin and end at prescribed, controlled, programmed points with special tolerances on the (interpolated) path in between them. [7]

coordinate measuring machine (CMM)

a machine that uses a probing tool to determine coordinate points on a part.

correlation

the degree to which two or more attributes or measurements on the same group of elements show a tendency to vary together. [11]

cosine

common math function; standard notation $\cos x$, cx [5]

COTS

abbr for Commercial Off-The-Shelf package

counterbore

a machining task; to form a flat-bottomed enlargement in the mouth of a cylindrical bore [13]

counterclockwise (CCLW)

negative rotational direction.

countersink

a machining task; to cut a funnel shaped enlargement at the outer end of a drill hole, usually for the reception of a screw, bolt, or rivet head. [13]

CP

abbr for Control Plan

CPC

abbr for Chevrolet, Pontiac, CM of Canada.

- CPU

abbr for Central Processing Unit

CSA

abbr for Controls Standardized Application.

cutoff frequency

the frequency at which the gain of a filter is at an edge of a band, usually taken to be when gain is 0.5, or Å-3.01 dB; the frequency at which the output of a filter is half the power of the input; see band-pass filter, high-pass filter, low-pass filter, standard symbol ω_c ; standard units rad/s, Hz; primary units 1/0} [5]

cutter compensation

tool path is programmed for a tool of zero diameter. Cutter compensation uses algorithms to determine the offset of the tool path to compensate for the actual diameter of the tool. [4]

cutter compensation plane

plane in which cutter compensation is measured. Either the xy-, yz-, or xz-plane.

cutter compensation radius offset

attribute of a cutting tool.

damped frequency

the frequency of oscillation of an underdamped second-order filter

$$\omega = \omega_n \sqrt{1 - \zeta^2}$$

where ω_n is the natural frequency and ζ is the damping ratio; see second-order filter; standard symbol ω ; standard units rad/s, Hz; primary units 1/0 [5]

damping ratio

see second-order filter; standard symbol ζ ; standard units 1; primary units 1 [5]

data

a representation of facts, concepts, or instructions in a formalized manner suitable for communications, interpretation, or processing. [6]

database

a structured repository for information, with well defined methods for storing and accessing data. Instances of objects and relationships as defined by a schema. [7]

database management system (DBMS)

a software system (often COTS) for managing one or more, centralized or distributed data bases. Data management facilities typically provide mechanisms for data entry, update, retrieval, and storage. [7]

data bus

communication path between one or more CPUs and related hardware. [7]

data dictionary

a centralized repository of information about data such as its meaning, relationships to

other data, origin, usage, format, and business rules governing the data. It assists management, database administrators, system analysts, and application programmers in planning, controlling, and evaluating the collection, storage and use of - data. [6]

data starvation

finish executing a block of code before next block has been analyzed. Eliminated with sufficient look-ahead. [15]

data structure

a formal representation of information for communication and processing purposes without regard to actual storage configuration. [7] DBMS

abbr for Database Management System

DC

abbr for Direct Current.

DC motor

such a motor uses direct current. There are two types of DC motors: brushed and brushless. The brushed DC motors use brushes for commutation. Brushless motors are externally commutated. [4]

deadband

the deadband range is adjustable and lets the programmer select the appropriate error range both above and below setpoints outside of which the output will not change. With the deadband, the programmer can closely match the process variable to the setpoint without changing the output. (see PID control loop) [10]

dedicated channel

point-to-point wiring. [7]

delta

the difference between the desired value and the measured value

derivative

common math function; rate of change, usually with respect to time; standard notation $\overset{1}{\times}$, x', x⁽¹⁾, dx/dt, Dx, sx+x(0⁺) [5]

design viewpoint

a DSSA view of a system.

desired

what must be achieved in order to match a plan; synonym for reference [5]

determinant

common math function; standard notation det[A], IAI [5]

deviation

difference from desired [5]

device driver

a software component that provides software platform access to a hardware platform. [7]

diagnostic

an interactive condition that occurs when the controller is in the maintenance mode. Offline diagnostics are algorithms that extensively check the entire NGC system.

September 26, 1994

SOSAS Rev. 2.6

They are intended to extensively verify all A complete system system hardware. diagnostics verification requires some or all of the NGC operations to be suspended. This will normally require a mode change at the standardized application level and - perhaps at the system level. Due to the various possible configuration of hardware and software on a NGC, the system integrator must tailor this package for the delivered environment. Health and status monitoring includes data consistency checking, and may be enabled any time system operation during for any combination of standardized applications in the system. [7]

difference equation

a mathematical relationship to model a discrete function, expressed in terms of other values in the sequence; compare with continuous-time equation, differential equation, discrete-time equation, Laplace transform, Z transform; A difference equation usually models periodic process in terms of past values; see first-order filter, integrator, second-order filter for examples

Example standard notations for difference equation are

$$y[n] = y[n-1] + y[n-2]$$

 $y_n = y_{n-1} + y_{n-2}$
 $y_{n+2} = y_{n+1} + y_n$

where y is a function of time, n denotes the current cycle, n-1 denotes the last cycle, and in general n-1 denotes the ith prior cycle. Initial conditions, like y[0]=0, are usually given or implied. Difference equations are usually derived from differential equations. [5]

differential equation

a mathematical relationship to model a continuous function, expressed in terms of derivatives; compare with continuous-time equation, difference equation, discrete-time equation, Laplace transform, Z transform. A differential equation usually models continuous-time phenomenon in terms of time derivatives; see first-order filter, integrator, second-order filter for examples.

Example standard notation for differential equations are

 $y'' - 2y' - 3y = e^{-t}$ $\ddot{y} - 2y - 3y = e^{-t}$ $y^{(2)} - 2y^{(1)} - 3y = e^{-t}$ $(d^{2}y/dt^{2}) - 2(dy/dt) - 3y = e^{-t}$ $(D^{2} - 2D - 3)y = e^{-t}$ $y - 2_ydt - 3_ydt^{2} = _e^{-t} dt^{2}$ $s^{2}y - sy(0^{+}) - y(0^{+}) - 2sy + 2y(0^{+}) - 3y = e^{-t}$

where y is a function of time t. Initial conditions, like y(0)=0 or y(0) = 1 ft/s are usually given or implied.

Differential equations are commonly used by systems engineers to model systems. The systems engineer usually converts differential equations to difference equations for specification and implementation in software. [5]

digital numerical control (DNC)

a type of machine controller designation, in which programs downloaded from elsewhere. [16]

dimension

1 a standard quantity, such as ft or mi; synonym for units [5]

2 a degree of freedom within a vector space. [17]

$$y[n] = \frac{-5}{16}e^{-n} + \frac{5}{16}e^{3n} - \frac{1}{4}ne^{-n}$$

dimensional measurement interface specification (DMIS)

standard language to program coordinate measuring machines. [7]

dimensionless

no units, such as ratios; synonym for unitless [5]

direct numerical control (DNC)

a host computer controls a group of machine tools by downloading part programs to them. Operators do not program parts at the machine. They start machines and fixture parts. This control structure is common in large shops that mass produce parts. [4]

discrete I/O

input or output data represented by on or off states. [7]

discrete time

time divided into quantized intervals; time is usually divided into equal intervals to create a periodic process; compare with continuous time [5]

discrete-time equation

a mathematical relationship to describe a function of time, expressed in terms of discrete time; compare with continuoustime equation, difference equation, differential equation, Laplace transform, Z transform; see first-order filter, unit functions for examples.

Example standard notation for discrete-time equations is [5]

distance

synonym for range; method of measurement dependent on use [5]

distributed control

distributed control means that a number of PLCs on a network can control I/O in local or remote places. They communicate over a control network like Arcnet or CANbus. In distributed control, the task of implementing the control system is divided into a number of small modular and easily managed parts. Each part of the control is performed by individual stations networked together to communicate needed information. The idea is to minimize the communication between stations to allow each of them to run autonomously. This approach works well in large complex systems, applications with many similar repeated elements, and when configuration machine upgrade and flexibility is essential. [10]

distributed database

a fully distributed database is one that has multiple data bases stored on multiple servers at physically separate locations. The computers that store, manage, and interface with users across the various locations are linked by network. A centralized distributed database utilizes a central server to access dispersed database sources. [7]

DLL

abbreviation for Dynamically Linked Library.

DML

abbr for data manipulation language. [6]

- DMS

abbr contained in definition for transaction

DMIS

abbreviation for Dimensional Measurement Interface Specification

DNC labbr for Direct Numerical Control.

2abbr for Digital Numerical Control.

domain

the set of values assigned to the independent variable of a function [11]

domain independent application

an applications that supplies general functionality and was not necessarily developed for NGC. [7]]

drawing interchange file (DXF)

the AutoCAD system's ASCII drawing interchange files describe a CAD drawing created by that system software. These files can be easily translated into formats for other CAD systems. The files and their format are of interest to companies manufacturing CNCs because customers want to use AutoCAD to design parts and then dump the DXF files into the CNCs on their machine tools and quickly cut the parts. [4]

drag

to move the tool with a dead spindle. [3]

drift

slow monotonic change in measured data [5]

drill

a machining task; to make a rounded hole or cavity in a solid by removing bits with a rotating cutting tool. [13]

drill bit

a cutting tool for boring holes in material.

drive

a mechanism for controlling a motor.

DSP

abbr contained in definition for servo motor control

dwell

to stop tool motion while the spindle is turning; to break chip, for example. [3]

DXF

abbr for Drawing Interchange File.

dynamic binding

the code associated with a given procedure call is not necessarily linked until the moment of the call at run-time. [7]

dynamic error

the difference between the true and the reported values of a machine motion variable due to error sources. [7]

- dynamic following error the distance between the commanded position and the actual position of the device under motion control. [7]

EBM

abbr for Electron Beam Machine.

ECM

abbr for ElectroChemical Machine.

EDM

abbreviation for electrodischarge machine.

EIA

abbr for Electrical/Electronic Industry Association.

effector

a device that physically performs an action or transformation function. [7]

electrical cabinet

large metal box, also called the magnetic cabinet, containing the CNC computer boards, servo amplifiers, communications boards and other wiring necessary to control the machine electronically. Some OEMs attach the operator console to the outside of this cabinet. [4]

electrochemical machine (ECM)

a machine which performs material removal via corrosive chemicals.

electrodischarge machine (EDM)

a machine which performs material removal via electric potential.

element

a constituent of the NGC system at the level directly below system level. [7]

emergency stop

a system's emergency stopping circuits and devices were designed to avoid injury to personnel and damage to the machine. Usually large, red E-stop buttons are the emergency stopping devices on the operator console, the remote jog unit, and the machine itself. [4]

enclosure

screening around the cutting surfaces of a machine to protect the operator from material and coolant that may be propelled from the cutting area during operation. Enclosures are usually metal doors with windows in them to allow the operator to see inside. Many countries have regulations defining these enclosures and the mechanisms for opening and closing them. [4]

encoder

a mechanic device for translating motion into a unique electronic signal or combination of signals. Modern machine tools use optical encoders. See also quadrature. [4]

encoder feedback

an interface that allows a controller to ... interface to incremental or Ouadrature encoders or other high speed pulse sources. The interface is capable of accepting Phase 0, Phase 90, and index pulse inputs. These signals can be TTL, single-ended or - differential (strappable) and are optically isolated. The interface provides logic to detect overflow/underflow conditions and desired position comparison triggers. The index pulse can be used to latch counter These conditions can be made to data. generate interrupts to the host processor. Inputs are conditioned by a four stage digital filter. Five jumper selectable sampling frequencies are available for filter use. Selecting the lowest frequency compatible with the highest expected input rate will maximize noise immunity. The interface accepts quadrature input and input pulse The counter register operates in rates. quadrature decoding, pulse and direction input counting, or in a pulse input up/down mode. Counter output is available to the host bus as a binary or coded decimal form. [10]

end effector

an effector, gripper, or mechanical device, located at the end of last joint, by which objects can be grasped or acted upon. [7]

endfeed grinding

a type of centerless grinding.

end mill a type of task.

end mill sharpening

a type of grinding.

endpoint control

any control scheme in which only the motion of the tip of the end effector may be commanded. [7]

end user

anyone who uses a manufacturing workstation for the execution of tasks. [7]

enterprise

the top level of a manufacturing hierarchy under which exists the factory, center, and cell [7]

enterprise information

the enterprise information is any data from the enterprise level that is communicated to lower levels. Enterprise information may include procedures, policies and schedules. [7]

environment

the aggregate of physical factors that act on the manufacturing workstations, manufacturing processes, and final products. [7]

error

difference between desired and measured data; synonym for delta; standard engineering term [5]

ESPRIT

abbr for the European Strategic Programme for Research and Development of Information Technology.

E-stop

abbr for Emergency Stop.

etch

ŧ.

ń

a task; to produce a design, usually in a metal or glass surface by covering it with an acid resistant ground through which a design is scratched with a pointed instrument, then submitting the surface to an acid bath. [13]

euler parameters

four parameters for specifying quaternions; standard symbols e1234, a, b, c, s; standard units 1; primary units 1 [5]

European Strategic Programme for Research and Development of Information Technology (ESPRIT)

a consortium of 21 European companies now working on the specification and definition of CIM-OSA. The consolidated efforts which focus on CIM has resulted in what is today the AMICE consortium. [6]

exception handling

managing deviations from the planned/nominal production or process. [7]

executive viewpoint a DSSA view of a system.

expert system

a system that captures a human expert's knowledge in a computer application for general application to provide solutions for similar problems. An expert system attempts to capture some of the intuitive reasoning that experts use to make decisions. [6]

exponential

common math function; standard notation e^{x} , exp x [5]

external communication

of manufacturing communication а workstation with the outside world through its computing platform. [7]

external interface

interfaces of a manufacturing workstation with the outside world, to cell level above, and to sensors/effectors below at the machine level. [7]

extrapolate

function to determine values from two or values in a table; usually linear but can be higher order [5]

fabrication

a term used to distinguish manufacturing operations for components, as opposed to assembly operations; to make or produce a part from raw stock. [6]

face mill cutter sharpening a type of grinding.

face plate

the part of a robot at the end of the last joint, to which an end effector can be attached.

factory

a manufacturing unit consisting of two or more centers and the materials transport, storage buffers and communications that interconnect them. [7]

failure recovery

a condition that begins with an assessment of the damage. Repair of equipment and/or replacement of components by the operator - may be required. In some instances the process may be allowed to resume after a failure recovery operation has been successfully performed, but at a lower level of performance. Failure recovery shall involve the performance of specific automatically generated and sequentially executed steps, and may require some interactive step execution as well. In most instances, a combination of automatic and interactive steps are required to facilitate a successful failure recovery. [7]

fault detection/isolation

an automatic condition that occurs when the controller is in the normal production mode. It automatically detect failures or interruptions in service, indicates location, and diagnoses cause when possible, Fault actions also recoverv аге initiated. Detection takes place when a flag is set or when some metric takes on a significantly different value. Possible metrics include the following: a process parameter that has exceeded its control limits, a statistical process control parameter, the output of a sensor fusion/integration module, or derived data such as runtime statistics that monitor performance. [7]

fault management

prevention, correction, or recovery from major deviations from current planned operations; predicts impending failures in time for corrective action; detects failures or interruptions in service; indicates location and diagnoses the cause when possible, determines and initiates recovery action. [7]

Fanuc

a controller manufacturer.

FBD

abbr for Function Block Diagram

feature

a physical attribute such as a surface, hole, or slot, used to describe a part. [7]

feature-based definition

a method of describing a part in terms of its features. [7]

feed

to provide a machine with material.

feedback

1 Input describing what is being controlled. This data is used by the CNC to adjust positioning and velocity. [4]

2 the furnishing of data concerning the output of a machine to an automatic control device or to the machine itself, so that subsequent or ongoing operations of the machine can be altered or corrected [11]

3 correcting or controlling a system by using part of the output as input; the flow of information back into the control system so that actual performance can be compared with planned performance. [6]

feedrate

feeding material into a tool. [4]

٢

feedrate override

a way, via the operator console, for the operator to manually change the feedrate.

- feeds and speeds

machine tool movements determining the rate at which a material is machined or transported. This can be in the form of stock feedrate into the tool, tool rotational speed, or tool translational rate. [7]

field replaceable unit

a hardware unit or software module that can be physically replaced on site, thus reducing downtime. [7]

filter

a device to alter a signal; software to alter a data steam; see averaging filter, band-pass filter, complementary filter, first-order filter, high-pass filter, hysteresis, Kalman filter, limiter, low-pass filter, rate limiter, secondorder filter, smoothing filter, wash-out filter; standard engineering term [5]

finish

a task; to polish a raw edge to form a smooth surface; to finish a raw edge by hemming, pinking, overcasting, or facing. [13]

first-order filter

a filter in which the output follows the input, only more slowly; It is usually implemented in software as a difference equation of period T as

$$y_n = e^{\frac{2\pi}{\tau}} y_{n-1} + (1 - e^{\frac{2\pi}{\tau}}) x_n = x_n + e^{\frac{2\pi}{\tau}} (y_{n-1} - x_n)$$

where τ is a filter constant. If the input is a unit step, then

$$y(t) = 1 - e^{\frac{\pi}{\tau}}$$

 $y(\tau) \approx 0.6321$
 $y[n] = 1 - e^{-\frac{\pi}{\tau}(n+1)}$

The Laplace transform is

$$\frac{Y(s)}{X(s)} = \frac{1}{\tau s + 1}$$

The Z transform is

$$\frac{Y(s)}{X(s)} = \frac{(1 - e^{\frac{z}{\tau}})z}{z - e^{\frac{z}{\tau}}}$$

The differential equation is

$$y = x - \tau y^{x}$$

The first-order filter is commonly used in avionics to smooth data, and to wash out transients at mode change, where typical values for τ are 1-5s. It is also used as a low-pass filter, with cutoff frequency $\omega_c = 1/\tau$.

When implementing a second-order filter on normalized variables, such as angles, the discontinuities require special treatment. [5]

fixture

any device that holds workpieces during the machining operation; a fixture by itself does not provide location information. [7]

flat end mill

a type of cutting tool.

flexible tap

a type of tapping tool in which the rotation and the z-motion of the tool are not tightly coupled. [15]

FLEXIS

performs "offline programming" of robot/manufacturing processes. Uses Grafcet language.

flute

a groove machined in a cutting tool to facilitate easy chip removal and to permit cutting fluid to reach the cutting point. [9]

following error

the difference between where a cutting tool actually is and where the control commands it to be. [4]

form cutter sharpening

a type of grinding.

form grinding

a type of grinding

function

a mathematical relation between two sets, called the domain and range, which assigns a unique value in the domain to each value in the range. i.e. f(domain coordinate) = range coordinate [17]

function block diagram (FBD)

a graphical language that allows program elements that appear as blocks to be wired together in a form analogous to a circuit diagram. FBD is well suited for applications that involve continuous flow of information or data between control. [10]

gauge vector

attribute of a tool.

September 26, 1994

geometric error

the difference between the true location of a particular end effector and the location as reported by the machinery scales. The machinery is assumed to be in a static, unloaded posture at a reference temperature. [7]

geometric modeling applications program (GMAP)

built on the results of PDDI, its goal is to geometric and identify and organize definition data nongeometric product required for the engineering, manufacturing, and logistics support of complex structured components throughout the product life cycle. [6]

geometry

and definition (measurements the relationships) of the dimensional properties of points, lines, curves, surfaces, and solids, as required to convey the shape of configuration aspects of a design in human processing terms. [6]

GM

abbr for General Motors

GMAP

abbr for Geometric Modeling Applications Program.

GMF

abbr for General Motors/Fanuc.

GMTS

abbr for General Motors Technical Staff. Authors of a Standard for Machinery and Equipment.

goal

- the end result of the manufacturing process. [7]

granularity

degree of refinement. [7]

graphical user interface (GUI)

common user interfaces applied to the controller domain include: Citect, Fix, Genesis, Labview, Lotus@Factory, Paragon, PCIM, Wonderware [10]

gravitational acceleration

acceleration caused by the force of gravity; standard symbol g', standard units lbf, kip; primary units, ML/ θ ; gravity sometimes includes effects of the earth's rotation; gravity is often treated as a constant, but for greater accuracy gravity is a function of latitude, altitude, and the phase of the moon [5]

grind

a machining task; to wear down, polish, or sharpen by friction. [13]

grind belt a type of grinding tool.

grind wheel a type of grinding tool.

gripper

a type of end effector, a device by which physical objects are grasped and held. [7]

GUI

abbr for Graphical User Interface

Hall effect sensor

highly accurate, non-contact limit switch which detects the proximity of a magnet and provides a digital output to assure an accurate position reference. [8]

hand crank

usually a wheel with a handle attached used to control the motion of an axis on a manual machine. [4]

hierarchical control

control in which tasks are decomposed into sequences of subtasks that are passed to the next lower level in the hierarchy until the lowest level is reached. [7]

high-pass filter (HPF)

a filter that allows frequencies above a cutoff to pass while attenuating frequencies below the cutoff frequency [5]

holding torque

stepper motors, when energized, hold position via a magnetic field. The holding torque is the maximum torque which can be generated before the rotor slips to the next pole location. [8]

home position

a tool may have a home position to which it traverses to get out of the way of the machine.

home switch

any of a variety of sensors which can be used to establish an accurate initial position. This may consist of a standard end-of-travel Hall sensor; a center position optointerupter with half-travel blocking vane; an index - signal on a linear encoder; a shaft coupling mounted magnet with Hall sensor; or a once-per-revolution encoder index signal. Once-per-revolution sensors will usually require a logical ORing with a linear signal if a unique home position is required. [8]

HPF

abbr for High-Pass Filter [5]

HUI

abbr for Human User Interface

human-user interface (HUI)

an information channel that conveys information between a human user and a system. It is composed of the surface layout, domain-independent applications (such as window and graphic packages), open HUI applications, and the PMSs. [7]

hysteresis

1 the error which can occur when a motion system is commanded to return to a starting position after several interim moves are made. It is attributable to lead screw reversing errors, backlash, etc. [6]

2 a function in which the algorithm for computing output changes at defined events or thresholds, such that output follows as input increases and another path as input decreases.

Hysteresis can be formalized: at initialization, select algorithm-0 if event-1 occurs, switch to algorithm-1 if event-2 occurs, switch to algorithm-2

if event-n occurs, switch to algorithm-n

¥

Hysteresis can prevent a test $(x = x_c)$ from oscillating near the transition point $(x \text{ Å } x_c)$ due to noise. Implementation is usually:

at initialization, set y = 0if $x_c + (h/2)^2 x$, then set y = 1if $x_c - (h/2) < x < x_c + (h/2)$, then let y retain its value if $x^2 xc - (h/2)$, then set y = 0 [5]

B

abbr for Information Base.

ICAM

abbr for Integrated Computed Aided Manufacturing.

identity

identity matrix; standard symbol I; standard units 1; primary units 1 [5]

IEC-1131

1 A standard PLC programming and execution environment that would facilitate the proliferation of many third party vendor, reusable software modules, and new software development tools. [21]

2 A controller language standard which is a collection of four different languages: Pascal-like, Grafcet-like, Signals on IC chip like, assembly language like. Statements in each language can be included in each other.

IGES

1 abbr for Interactive Graphics Exchange Specification

2 abbr for International Graphics Exchange Standard.

۰IL

۶.

abbr for Instruction List.

impulse

synonym for unit impulse [5]

IMW

abbr for Intelligent Machining Workstation.

Industrial Real-Time Operating System Nucleus (ITRON)

part of Japan's TRON program to develop a new operating system for use across the spectrum of computer systems and industrial machinery. [6]

infeed grinding a type of centerless grinding.

information base (IB)

an abstract storage mechanism that may consist of multiple, coordinated information sources (e.g., data bases, knowledge bases, memory maps) distributed across heterogeneous or homogeneous platforms. The IB is the conceptual structure to which all application-and service-generated requests for shared data manipulation are made. [7]

information base application

code that allows a user to interact with or manipulate specific information in a manner that particularly suits NGC IB needs, e.g., a graphical schema browser capable of taking STEP input. [7]

information base support package

a software component (often COTS) that provides specific IB functionality to be accessed through the data management services abstraction layer, e.g., a DBMS, a KBMS. [7]

initialization

1 a procedure to reset physical devices to a known state. During initialization, the device is usually not available. [5]

2 an interactive condition that occurs when the controller is in the start-up mode. Initialization establishes all late bindings and communications links between SAs, and establishes the initial state for all SAs. Initialization also retrieves the appropriate models for the selected modes determined by the configuration. [7]

in-process gauging

measure the part with a probe while the part is still on the machine. Problems are encountered due to additional cycle time required for sensing with devices such as lasers. [18]

input/output artifacts

objects, either physical or virtual, such as buttons, dials, readouts, sliders, etc., that provide specific user input or output capabilities and that can be combined to create user interface devices. [7]

instance

a distinguished occurrence of a particular object. [7]

installation

the creation of an instance. [7]

instruction list (IL)

a low level language, similar to assembler language. It is useful for smaller applications or for optimizing parts of an application. [10]

instrumentation

hardware to measure and to monitor a system [5]

integrate/configure

an interactive condition that occurs when the controller is in the start-up mode. Hardware integration, physical integration and include configuration shall installing mechanisms and computing hardware, and/or connecting wires to the hardware ports; software integration and configuration implies loading the code and data required for the controller onto the computing platform. [7]

integrated computed aided manufacturing (ICAM)

1 an activity modeling methodology that represents the functional decomposition of any activity. An activity in the model converts inputs to outputs using the defined mechanisms and under the constraints imposed by the controls, and linked by those inputs, outputs, and controls, to other activities in the model. [6]

2 a representation of structurally integrated information. A methodology for developing

a data model which expresses the structural characteristics of information. It is used as a foundation for inter-division data standards and data base design, showing data and entity relationships, and attribute identification. [6]

•

é

integration

the formation of one complete and harmonious entity or coordinated entities of different applications that will accomplish a complete process solution. [7]

integration architecture

provide the components that the interconnection structure and interface interoperating between definition applications. It is composed of the platform services, communication services, data presentation services. management management services, task management services, geometric modeler services, basic I/O services, and the data standards. [7]

integrator

a function or filter that mathematically integrates [5]

intelligent machining workstation (IMW)

a USAF Program conducted by Carnegie Mellon University, Cincinnati Milacron and Pratt & Whitney. [6]

interactive graphics exchange specification, or International Graphics Exchange Standard (IGES)

1 files conforming to this specification are generated by CAD systems to define part geometry. However, interpretation of these files by different CAD systems may vary causing variations in the parts on different systems. [4] 2 a format for the exchange of graphical information (such as lines, ellipses, and text) between dissimilar drawing systems. This data enables the electronic transmission of prints, but does not include information about the items represented by those prints. [6]

3 a CAD/CAM data exchange specification adopted by ANSI. IGES attempts to standardize communication of twodimensional drawing and geometric product information between computer systems. [6]

interchangability

a characteristic of system components that makes it possible to replace one component with another component of equivalent functionality made by a different vendor. [7]

interface

the definition of common boundary or a point or means of interaction between two or more distinct entities. It may be a shared boundary between two subsystems, which are defined by functional characteristics, common physical interconnections, signal specifications, and other characteristics. [7]

internal grinding

a type of grinding.

International Organization for Standardization (ISO)

an organization of national standards bodies from various countries established to promote development of standards to facilitate international exchange of goods and services, and develop cooperation in intellectual, scientific, technological, and economic activity. [6]

interoperability

1 a characteristic of independent system components that makes it possible for a set of components that comprise a system to intercommunicate and work properly together. [7]

2 that characteristic of a SOSAS compliant module which makes it possible for the module to operate on different computing platforms, or be integrated with differently configured NGCs. [6]

interpolate

function to determine intermediate values from two or values in a table; usually linear but can be higher order; endpoints are either extrapolated or limited [5]

interpolation

1 the process of approximating a given function by using its values at a discrete set of points [11]

2 an algorithmic process to generate intermediate points along a prescribed path or specific geometric shape between two points. For example, interpolation might be used to generate an arc with a 3.5-in. radius (circular interpolation) between points A and B. In addition to linear and circular interpolation, there are advanced algorithms to produce helices, splines, etc. [6]

interpolation block

offline system generates more than machine can handle, so machine ignore blocks coming in when its not ready for them. [20]

interrupt

interrupts are controlled with interrupt registers consisting of an end interrupt, poll, poll status, interrupt mask, priority mask, inservice, interrupt request, and interrupt status. [10]

inventory

the aggregate of tangible company property items which are held for sale in the ordinary course of business, or are in process of production for such sale, or are to be currently consumed in the production of products or services to be available for sale. [6]

inverse

standard math function; standard matrix operator; standard notation x^{-1} , X^{-1} [5]

I/O interface

different I/O types are either accessed directly from the base hardware platform via serial or parallel interfaces or may otherwise be accessed via add-on expansion modules. I/O types include: CANbus, Opto-22 PAMUX and OPTOMUX, APC Seriplex, Keithlev Metrabyte and Workhorse, TURCK Sensoplex, AB1771 Digital, GE TI/Series. One. TI/Siemens, Series Transition Technologies, Grayhill, Gordos, Burr Brown Sensorbus, Du Tec [10]

iron

the frame of a machine without any electrical equipment, such as the CNC or servo drives, attached. The iron is produced at a foundry and then the bed, tool changer, and other electrical equipment is added. [4]

I/O

abbr for Input/Output

IPM

abbr for Inches Per Minute.

IPR

abbr for Inches Per Revolution.

ISO

abbr for International Organization for Standardization.

ITRON

abbr for Industrial Real-Time Operating System Nucleus.

jerk

rate of change of acceleration, either scalar or vector, often with subscripts to denote the coordinate frame; time derivative of acceleration; standard symbol j, J; standard units ft/s³; primary units L/ θ [5]

jig

1 a mechanical construction that determines location dimensions that are going to be machined into a workpiece. A jig can be incorporated into a fixture so that both accurate location and clamping are implemented (see fixture). [7]

2 a plate, box, or open frame for guiding work and for guiding a machine tool to the work, used especially for location and spacing drilled holes. [11]

jig saw

a type of saw machine with a narrow, vertically reciprocating blade for cutting curved and irregular lines or ornamental patterns in open work. [13]

- ЛТ

abbr for Just-In-Time.

job shop

businesses the supply small quantities of parts (20 to 30) on a contract basis to larger companies. The large companies send specifications for the parts to the job shops and award the contracts to the lowest job shop bidders. These shops usually specialize in particular types of parts (e.g., springs for the automotive industry or hand cranks for knee mills). [4]

jog

1 manual control of an axis. [4]

2 an interactive condition that occurs when the controller is in the normal production mode; allows a machine axis to moved in any direction within the machine envelope or limited parameters via specified increments, or continuously until the human interface button is released. [7]

joint

a point of articulation between moveable parts. [7]

just-in-time (ЛТ)

a logistic approach to the movement of material to the necessary place at the necessary time. It allows for no buffer stock and is designed to minimize inventory during the manufacturing process. In short, synchronized production scheduling. [6]

kalman filter

a filter for combining multiple data sources, usually of different types, to produce an estimate better than any single source; compare with averaging filter, complementary filter [5]

Karel

an robot programming language, a successor to VAL. Named for the Czech playwright Karel Capek who wrote RUR.

KBMS

abbr for Knowledge Base Management System

kinematic error

the difference between the true and the reported locations of an end effector due to mechanical distortion of the machinery caused by motion. [7]

kinematic model

the controller's representation of the machine's motion.

kinematics

the branch of physics which deals with pure motion, without reference to the masses or forces involved in it [11]

knee mill machine

an inexpensive type of machine used to cut small parts in limited quantities. These machines usually do not have enclosures around the cutting area, and they have lower horsepower motors than bed mills. Knee mills may have CNCs attached, but they all
use at least one hand crank to move the bed with the part attached up to or away from the cutting tool. The "Z" axis on these mills has limited travel up and down, so the operator always needs to adjust the position of the bed. In fact, this machine received its - nickname because operators use their knees to bump the cranks and adjust the beds when their hands are busy. [4]

knowledge base

1 semantically rich set of symbols, relations, procedures, and constraints. [7]

2 the data, data rules of use, and procedures stored to support expert systems and artificial intelligence applications. It implies that some semantic (knowledge about the meaning and possible uses of the data) content is included. [6]

knowledge base management system (KBMS)

a software system (often COTS) for managing and providing reasoning mechanisms for one or more centralized or distributed knowledge bases, e.g., rule sets. [7]

knowledge base system

a knowledge base system consists of several components: a knowledge base and a mechanism for reasoning about stored information. The knowledge base contains domain knowledge, rules and heuristics. The reasoning mechanism includes the control for rule selection and execution. [6]

ladder diagram (LD)

a powerful collection of standard graphical symbols and function blocks that can be combined into complex control programs that portray the actual execution component taking place in the SFC step or action. [10]

ladder logic

1 written representation of the electrical relays required to operate a machine. There are several different forms of ladder logic (see PAL). Ladder logic in the United States looks like rungs of a ladder with contacts and relays on the vertical rungs; however, European ladder logic is written horizontally. (Also referred to as relay ladder logic.) [4]

Ì

2 a programming language utilized for PLCs; the language for discrete control originally based on relays, rails, and rungs.
[7]

ladder logic execution

ladder logic is driven (started) by a timer. It executes for each millisecond chunk and always responds to time interrupts. After ladder is evaluated, a delay time is hardcoded to give background tasks a chance to run. Background tasks include operator interface messages and other userdefined code sequences. Ladder logic has a higher priority, so it always takes place. It must be interrupted so that other processing can be executed.

Software interrupts are used to tie userdefined code to particular points of the ladder logic. Software interrupts occur at the end of every OPI state, timer interrupt, input phase, output phase, and during some particular instruction in the ladder logic. This allows a particular ladder logic command to initiate a customized program. [10]

ladder logic programming

a simple high-level graphical language for drawing control logic diagrams. Input contacts can be represented as normally Output coils can be open or closed. energized, latched, or unlatched. Boolean logic is represented by horizontal rungs - (AND function) and vertical branches (OR function) read left rail to right rail with logic conditions on the left and output instructions on the right. Current logic states are graphically displayed as connected/disconnected sections of the rung(S). If a complete unbroken path exists from the left rail of the rung to the right trail, the rung is TRUE and the output instruction is acted upon. This discussion has only illustrated a small portion of the entire ladder logic instruction set. [10]

Laplace transform

a mathematical relationship to model a continuous function in the complex frequency domain (S-plane); compare with continuous-time equation, discrete-time equation, Z transform; Laplace transforms are commonly used by systems engineers to describe systems; see first-order filter, integrator, second-order filter, unit functions for examples [5]

latency

time between the request for an action and the initiation/completion of that action. [7]

lathe

a machine used to shape a part by gripping it in a holding device and rotating it under power against a cutting tool. Lathes are commonly used for turning, boring, facing, and threading. [4] abbr for Ladder Diagram

lead

synonym for pitch.

lead error

the deviation of a leadscrew from its nominal pitch. The error is often monotonic (linear), although periodic error and thermal expansion set limits to its predictability. [8]

leadscrew

1 a mechanical device, also called a ball screw, for translating rotary motion into linear motion, consisting of an externally threaded screw and an internally threaded carriage (nut). These typically move the bed, the spindle, and the head surrounding the spindle. [4]

2 a long precision screw on the front of a lathe bed that is geared to the spindle to transmit motion of the carriage for thread cutting [9]

leadscrew mapping

leadscrew motion can be mapped by indicating the number of revolutions per inch. This technique allows for compensation in leadscrew pitch because the pitch per inch is not constant (see lead error). [4]

lead through teach

record the trajectories. Use teach points to avoid obstacles [20]

length offset attribute of a tool

life-cycle costs

the cost of a product and its related activities that occur over the entire life of the product. [7]

limiter

a filter that passes the input to the output, except that the output is limited to a minimum value and a maximum value; compare with rate limiter.

Limiters are usually implemented as

 $y_n = \max(x_{\min}, \min(x_{\max}, x_n))$

where x_{min} is the lower limits and x_{max} is the upper limit. [5]

limit switch

a sensor, typically Hall effect, optical, eddy current, or mechanical, which is used to sense the end of travel of a linear motion assembly. In addition to preventing overtravel, it is frequently used to establish a precision reference. [8]

linear force/torque

a type of control law.

linear position/velocity

a type of control law.

link part of a robot.

locking

preventing access by one transaction to an object by another transaction. Locking may prevent data modification only or all read/write access. [7]

logarithm

common math function: standard notation $\log_b x$; $\log x = \log_{10} x$ [5]

logging

shadowing system activities over time by creating a time-stamped log file of transactions and state information. [7]

look ahead

algorithm that takes into account future commands to make decisions about current motions. For example, the control will note that the cutting is going to be commanded to stop so it begins to decelerate. [4]

look and feel

surface-level representation attributes and behaviors independent of specific presentation layout, i.e., the style of presentation. [7]

low-pass filter (LPF)

a filter that allows frequencies below a cutoff frequency to pass while attenuating frequencies above the cutoff frequency; see first-order filter [5]

low rate initial production (LRIP)

normally, the phase of production used to assess readiness for full rate production. [6]

LPF

abbr for Low-Pass Filter [5]

September 26, 1994

Ì

LRIP

abbr for Low Rate Initial Production.

- lube

¢

synonym for lubricant.

lubricant

a substance used to reduce friction between the cutting tool and the part. [4]

LWP

abbr for Light-Weight Process.

M & G codes

a slang phrase describing combinations of the letters (common letters being iMî and iGî) with multiple numbers to indicate where a machine cutter should move and what it should do. These codes are used in NC programs. The codes were originally fed into a machine as a punched tape of instructions. Many other letters are used, such as iTî codes to describe tool changes. iGî codes describe positions, and iMî codes identify miscellaneous functions. [4]

machine model

a structure data representation that provides a description of a machine and simulates its properties and characteristics of interest. [7]

machine tool

a stationary power-driven device used to shape, cut, turn, bore, drill, grind, or polish solid parts, especially metal. [4]

machine under control

any machine or process being controlled by NGC. [7]

machinist

a highly skilled machine operator who makes decisions about how a part will be programmed, fixtured, and cut. [4]

MADE

abbreviation for Manufacturing Automation Design & Engineering.

magnitude

common math function; synonym for absolute value [5]

maintainability

the measure of the ability of a system or product to be retained in or restored to specified conditions when used by personnel having the specified skills, using prescribed procedures and resources. [6]

maintenance

any activity to eliminate faults or to keep hardware or programs in satisfactory condition, including tests, measurements, replacements, adjustments, and repairs. [6]

manipulator

a mechanism usually consisting of a series of articulated links, for the purpose of grasping and moving physical objects. [7]

man-machine language (MML) a user interface.

manual machine

a machine that does not use computer components to control operation of the cutting tools. Usually the operator interacts - with the machine by turning hand cranks to adjust the spindle speed and to move the cutting tools and/or bed. [4]

manufacturing

a series of interrelated actions involving the process design, material selection, planning, production, quality assurance, management and marketing of discrete consumer and desirable goods. [6]

Manufacturing Automation Protocol (MAP)

IEEE 802.4 token-bus specification. [6]

manufacturing workstation

1 a material transformer device that includes a controller, computing platform, human interface, and peripheral devices. [7]

2 a manufacturing workstation is defined as a single material transformer device and related support equipment (e.g., material handling or inspection) together with the processing and control capability for autonomous response to commands and inquiries. [6]

MAP

abbr for Manufacturing Automation Protocol.

mark

a measuring task.

materials resource planning (MRP)

scheduling for the manufacturing floor.

matrix

standard notation [a_{ii}], A [5]

maximum

common math function; standard notation $\max(x_1,...,x_n)$ [5]

metrology

science/system of weights and measures or of measurement. [13]

microstepping

a technique which, instead of switching phase currents in a stepper motor on and off, slightly decreases the current in one winding, while slightly increasing it in another. This increases the resolution. [8]

mill

1 A machining process which removes material, usually metal, from a part using one or more rotating cutting tools. [4]

2 to shape or dress by means of a rotary cutter. [13]

milling machine

a machine capable of performing the operation of cutting, shaping, or finishing a workpiece. [7]

minimum

common math function; standard notation $\min(x_1,...,x_n)[5]$

MML

- abbr for Man-Machine Language.

MMPM

abbr for Millimeters Per Minute.

MMPR

abbreviation for Millimeters Per Revolution.

modular

a standardized and flexible construction consisting of logically self-contained and discrete parts; it allows for scalability and interchangability within a NGC. [7]

mold

a form made on a milling or punch machine into which a liquid is poured that hardens into the shape of the mold. The resulting molded product may also be milled as a finishing step. [4]

motion control

control of continuous joint motion of machines. [7]

MMST

it features interoperability of a growing number of machines used in semiconductor manufacturing.

modal

a state during which an established operation remains unchanged; compare with one-shot [3]

mode

an enumerated mechanism for coordinating behavior. [7]

model

1 a structured data representation of a process or system that simulates properties and characteristics of interest. [7]

2 a mathematical representation [5]

motor

device that converts electrical energy into mechanical energy by using forces produced by magnetic fields on current-carrying conductors. There are three basic types of motors available on machine tools: DC brushed, DC brushless, and AC induction. [4]

moving column machine

in such a machine, the part remains in a fixed position while the spindle performs all motion.

MRP

abbr for Materials Resource Planning.

multiaxis mode

keeping the cutting tool perpendicular to the surface [19]

multiprocessor control

a control computing platform that contains more than one processing unit. [7]

natural frequency

see second-order filter; standard symbol ω_n ; - standard units rad/s, Hz; primary units 1/ θ [5]

NC

abbr for Numerical Control.

NCMS

abbr for National Center for Manufacturing Sciences.

network interface

Arcnet and CANbus networks allow controllers to be networked with each other and with PCs. Arcnet is a Local Area Network (LAN) industry standard protocol that has been well accepted and proven in factory control applications with over 1.67 million installed nodes in the United States. CAN could be classified as a Small Area Network (SAN). It runs on an inexpensive twisted pair bus and provides an interface to other peripherals, peer-to-peer networking of controllers, deterministic prioritized message passing, small efficient packet sizes, robust error detection, and a reduced wiring solution. CAN is an industry standard protocol supported by multiple vendors.

Peer-to-peer networking means that each node in the network is of equal status and is able to transmit at any time within the guidelines of the network access method. This is in contrast to master-slave or serverconsumer based networks in which a particular node(s) on the network has special network functions and privileges. In master-slave based networks, a slave node must wait for the master to poll it and give it permission to transmit before being able to send messages. This method can cause large latencies to exist when high priority messages such as alarms and fault conditions need to be reported. [10]

\$

neutral manufacturing language (NML)

1 a proposed language for communications between NGC applications and between a NGC and the cell. [7]

2 used by humans to communicate with hardware/software modules and for communication between such modules. NML is the language used in communicating within an NGC, between NGCs and between an NGC and other Manufacturing process plans, systems. product descriptions, processes (at several levels of detail), machine management, motion control, and other aspects of manufacturing are expressed in NML. [6]

NGC

abbr for Next Generation Controller.

NGIS

abbr for Next Generation Inspection System.

NGIS - NC Inspection Project

a CMM with an open architecture. This project will eventually add additional sensor types besides just touch probes...vision, laser, ultrasound. It will marry wrist and machine controllers. A wrist controller has a standard I/O interface and holds a variety of different sensors. This will allow the machine controller to select a sensor for a particular process, much as it currently selects a tool for a particular process. Some sensors include SAMI (a CCD based camera with software for vision, providing information on tool wear, surface conditions, etc.), laser for interferometries for surface conditions, Strudhome capacitive touch sensor (probe nears surface but does - not touch). [21]

NML

abbr for Neutral Manufacturing Language.

NIST

abbr for National Institute of Standards and Technology.

noise

part of received data that is undesired, consisting of random sinusoidal terms added to a signal; compare with signal [5]

non-uniform rational b-spline

a spline is a function that has specified values at a finite number of points and consists of segments of polynomial functions joined smoothly at those points, enabling it to be used for approximation and interpolation of functions. For example, a cubic b-spline approximates a series of m+1 control points with a curve consisting of m-2 cubic polynomial curve segments. Nonuniform indicates that the control points do not have to be evenly spaced along the curve. Rational indicates that each parameter is defined by the quotient of two b-spline polynomials. [22]

normalizer

function to restrict input to a specific range, such as restricting an angle α in radians so that $-1^{2} \alpha^{2} 1$; angles usually require normalizing following any computation; normalized variables often present problems for filters and other functions as their discontinuities [5]

numerical control (NC)

controlling the motion of machine components using numbers (M & G codes) often fed into the controller on punched tape. The controller mechanism has no intelligence and cannot be modified in any way without changing the wiring of the machine. (see also RS-274-D, CNC.) [4]

NURBS

abbr for Non-Uniform Rational B-Spline.

nut compliance

the reciprocal of the stiffness of a leadscrew/nut assembly, measured in length per axial force. [8]

object

an abstraction of an entity representing an encapsulation of its attributes and services on those attributes. [7]

object module

software that has been compiled, but not yet linked. [7]

object-oriented

a viewpoint that models data and behavior as objects. [7]

object-oriented programming

programming in a language that embodies the concepts of objects, classes, inheritance, polymorphism, and dynamic binding. [7]

object-oriented structures

two methods used to manage complexity; classification structure, which captures class/member organization; and assembly structure, which portrays whole/part - organization. [7]

odd pockets with islands milling

method of surface milling in which the tool follows an inside-out or outside-in pattern across the part surface. [15]

OEM

abbr for Original Equipment Manufacturer.

offset

the amount of compensation required to account for the difference between the actual dimensions of a tool, probe, or end effector and the dimensions that were used to program the motion. [7]

offset vector

attribute of a fixture, which describes by how much the tool must be moved to avoid the stationary fixture

one-shot

a state after which an established operation is canceled upon execution; compare with modal [3]

open applications

applications developed for NGC whose functionality is not defined by the SOSAS but whose external interfaces are specified within the SOSAS. [7]

open loop

a control system in which data flows unidirectionally, that is, only from the control to the mechanism but not from the mechanism back to the control. [2]

open system

a system that provides capabilities that enable properly implemented applications to run on a wide variety of platforms from multiple vendors, interoperate with other system applications, and present a consistent style of interaction with the user (IEEE P1003.0). [7]

open system architecture

a specification of the capabilities or services that provides the interconnection structure and defines the interface between interoperating components, thus allowing applications to be integrated into a system with a consistent style of interaction. [7]

open system interconnection

a seven-layer model of intersystem communications specified by the International Standard Organization. [7]

open technology

a technology for which there are publicly available specification allowing the reinvention, from scratch, without royalties or license fees. [7]

operating system

software that controls the execution of application programs and manages computing platform services. [7]

operational

functioning correctly [5]

operator

human who controls movements of a - machine and communicates instructions to a machine. Usually an operator is a lower skilled worker than a machinist. [4]

operator console

a device attached to a machine that allows the operator to communicate to the machine and receive message from the machine indicating its operating state. The console communicates to the control. [4]

operator interface

for the machine shop, it is best to train operators on one controller for all the machines in the shop. Plug in particular machine model for specifics on that machineís configurations, i.e. lathe model, mill model, grinder model. [18]

orient

to move the spindle to a desired rotational position

orientation

direction in reference to a coordinate frame [5]

original equipment manufacturer (OEM)

they build machine tools and fit controls, electrical devices, and operator consoles on the machines. They may completely build their own machines, or they may receive CNC-prepared iron from a manufacturer and integrate their own control or another manufacturerís control onto the iron. [4]

Open-System Interconnection (OSI)

the ISO/OSI model is a seven-layer model of intersystem communications. The model has been adopted for use in MAP. [6]

optical encoder

a linear or angular position feedback device, typically providing incremental two channel information in quadrature format (sine or square waves with a 90 phase shift between Such two channel each channel). information allows simple counter circuits to function as absolute position indicators. [8]

optimal feedrate attribute of a tool

optimal spindle speed

attribute of a tool

orthogonality

perpendicularity, degree of the or squareness, between two axes (X, Y, Z). usually measured in arc seconds. [8]

OSI

abbr for Open-System Interconnection.

override

to alter selection made automatically by software [5]

overshoot

the amount of distance a cutting tool goes past the point where it was told to stop by the control [4]

PAL

abbr for Programmable Application Logic.

pallet

a portable platform on which materials - (parts) are stacked for transportation [12]

pallet shuttle

a mechanism for transporting pallets of parts to and from the machining area.

PAMUX bus

a high-speed parallel bus. This bus can be up to 500 feet in length. It is a very reliable high-current bus that is extremely immune to noise. An individual digital I/O byte on the bus can be accessed in less than 3 microseconds, even at distances of 500 feet. Analog channels are accessed in less than 150 microseconds. A completely populated PAMUX digital bus could be read in approximately 250 microseconds. It uses a standard 40 conductor ribbon cable that is daisy-chained between I/O interface brain boards and terminated actively at each end. [10]

paperless factory

the automated and on-line control of the factory without the aid of printed reports or forms. [6]

parallels

a type of fixture; a block or strip of metal made with two parallel sides and used especially in machine shop work, as for a gauge block or for setting up work. [13] 1 end product of a milling operation on stock/workpiece. For example, a part could be a small piece used inside a motor or a large metal casting. Also called a workpiece itself. [4]

2 normally refers to a material item which is used as a component and is not an assembly. [6]

part geometry

the dimensional features of a part or an assembly; see geometry. [6]

part number

1 a number which serves to uniquely identify a component, product or raw material. [6]

2 the number that uniquely identifies each purchased, manufactured, or assembled part. It is also referred to as an item number. [6]

part program

1 definition of the machining features of a part such as process order number, part orientation and position, cutter tool types and size, depth of cuts, spindle speed, and feed rates. Then the programming system calculates the information needed to position the cutter and table. Many systems use computer graphics to show the tool path on a computer monitor. [4]

2 a manufacturing process description for a workpiece. [7]

part programming

used to create tool path. It describes an object to be made by a machine. Part programming can be accomplished on a

part

CAD/CAM system and downloaded to the machine or it can be programmed on the machine itself using through the operator console. [4]

- part transformation

the removal or addition of material, assembly, or inspection of a part. Part transformation adds value to a part. [7]

part zero

the "zero" location on a part; where the X and Y axis meet on the part. [4]

path planner

software that produces paths for an end effector and computes collision avoidance algorithms, and satisfies other constraints. [7]

PCTE

abbr for Portable Common Tool Environment.

PDD

abbr for Product Definition Data

PDDI

abbr for Product Definition Data Interchange

PDES

abbr for Product Data Exchange using STEP.

period

time of a periodic process; $1/f_s$ where f_s is the sampling frequency; standard symbol T; standard units s; primary units θ [5]

periodic

a process that executes at a fixed rate; compare with aperiodic [5]

phase current

the rated current which a stepper motor requires to generate its rated holding torque. This value is usually based on unipolar (half-coil) operation. This choice of how the motor is wired has significant impact on performance. [8]

phase sequence

the specific sequence of coil current changes used to advance a stepper motor clockwise and counterclockwise, in either full or half step modes. [8]

PHIGS

abbr for Programmer's Hierarchical Interactive Graphics System. [6]

physical resource viewpoint

a DSSA view of a system.

PID control loop

a PID control loop is used to maintain a selected process variable at a desired set point. Many control applications require closed loop real-time control to set and maintain critical process characteristics which might be expected to vary or to drift over time. The PID function constantly monitors these parameters and calculates the error from the desired set-point and outputs a control parameter to return the process to the desired set point.

September 26, 1994

SOSAS Rev. 2.6

the update time of the PID loops determines the amount of total PID loops a single program/processor can handle. Simple loops and long update times allow for more PID loops in a program.

- PID can be used in both machine and process control applications. In a process control with temperature settings, PID can monitor the thermocouples and upon a programmed action, turn on a fan, open a valve, energize a pump and more. Other functions of PID include scaling, deadband, and zero-crossing. [10]

pitch

1 for leadscrews specified in British units, the number of full rotations required to advance the nut 1". For example, a 5 pitch leadscrew has a lead of .200". Metric screws are specified by lead only, usually in millimeters.

2 an angular deviation possible in positioning system, in which the tables' leading edge rises or falls as the table translates along its direction of travel. This represents rotation around a horizontal axis, perpendicular to the direction of travel. [8]

plain mill sharpening

a type of grinding.

plan

an ordered set of actions for accomplishing a goal. [7]

plane

a machining task.

planner based

September 26, 1994

an automatic condition that occurs when the controller is in the normal production mode; all resources are focused on part transformation or process control, and use of the production resources are in direct control of the WPSA. The WPSA provides the dialog or communications coordination that allows multiple SAs within the workstation to interact and perform simultaneous operations. [7]

planning

a procedure for determining the operations or actions necessary to transform material from one state to another. It includes the preparation of detailed instructions to produce a part transformation. [7]

plate

to cover with an adherent layer, as of metal, by mechanical, chemical, or electrical means. [13]

platform

a basic structure consisting of virtual machines, operating systems, communication mechanisms and hardware. [6]

PLC

abbr for Programmable Logic Controller.

plunge grinding a type of grinding.

PMS

abbr included in definition of human-user interface

point-to-point control

a scheme whereby the inputs commands specify only points along a desired path or motion. [7]

polymorphism

in an object language, the characteristic of an operator that allows it to work on objects of different classes. [7]

portability

the ability to operate the same component on different computing platforms. [7]

position

location, either scalar or vector, often with subscripts to denote coordinated frame; time integral of velocity; standard symbol p, P, x, y, z; standard units ft; primary units L [5]

power-up

an interactive condition that occurs when the controller is in the startup mode; involves bringing the computing platform on-line via the normal startup routines. If auto-start files are integral to the controller, the NGC automatically start; otherwise, the appropriate start command must be issued to initialize the NGC environment. In either instance, power-up starting and self-test routines, as determined by the configuration, until they are successfully completed; [7]

precision

1 measure of exactness, possibly expressed in number of digits, for example, computed to the nearest millimeter; compare with accuracy [5] 2 the degree to which a given set of measurements of the same sample agree with their mean [11]

predicate

an expression that can be evaluated as either TRUE or FALSE. [6]

primary units

a standard set of four units to which all units can be resolved. The primary units are (1) mass-M, (2) length-L, (3) time, and (4) temperature-T. For example, standard units for velocity might be ft/s or m/s, but primary is always L/θ . [5]

primitive function

the lowest level of functionality of the NGC manufacturing workstation. [7]

prismatic with regard to axes.

probing

synchronize with measure and motion. Probe must be quickly brought near to part, then slowing moved in close enough to part to just sense it for measurement. Only barely touching or not touching part at all. [17]

procedure

a systematic, controlled sequence of actions or operations performed in order to achieve a specific result. [6]

process

a set of procedures required to transform resources (inputs) into specific objectives (outputs) on a returning basis. [7]

process plan

- the sequence and description of fabrication or assembly operations. [6]

Product Data Exchange using STEP (PDES)

1 a proposed data format and data exchange standard which enables the sharing of physical and functional product information among various computer systems and applications. [6]

2 an international standard for data developed due to the problems discovered with the IGES format. PDES, represented in the EXPRESS specification language, uses specific CAD translators and application specific protocols to reduce part geometry translation differences between CAD systems. [4]

product definition data (PDD)

machine interpretable product, design, and manufacturing information that includes geometric models, features, surface finish, heat treatment, materials specifications, and tolerances. [7]

product definition data interchange (PDDI)

a format that allows for interchanging of PDD between dissimilar CIM systems. [7]

product life cycle

all the stages of a product from initial concept to finish abandonment. [7]

programmable application logic

an application specific "ladder logic" that is continuously executed by the controlís executive firmware. This "ladder logic" is composed of PAL ladder language elements. These elements are combined to form ladder rungs that make up PAL ladder modules. These ladder modules are linked together to form a PAL program.

The PAL program coordinates and controls the I/O interface between the 9/240 control and the machine tool. Use PAL to perform the following functions: 1) sequence machine functions; 2) operate machine functions; 3) monitor machine functions. PAL communicates with the controlís executive firmware and discrete I/O to coordinate these machine functions.

The controlis executive firmware sends information to the PAL program through specific PAL flags and variables. The PAL program sends information back to the control through other PAL flags and variables. The PAL program can also turn on PAL messages and custom display pages that appear on the operator panel CRT. These flags are divided into the following areas:

> primary system flags manual motion flags axis mover flags offset modification flags part program selection flags part program execution flags feedrate flags G code flags M code flags tool change flags random tool flags spindle and gear change flags spindle orient flags mode select flags

status flags from the control PAL message flags display page flags digitizing flags [1]

- programmable logic controller (PLC)

1 a device for discrete control characterized by repetitive cycling typically through the ladder logic program. [7]

2 a control device designed to replace banks of relays for the repetitive cycling through a series of mechanical device control commands. Current designs are computers with sophisticated logic and communications capabilities. [6]

Programmable Universal Machine for Assembly (PUMA)

"At a luxurious press conference in 1978, General Motors unveiled the PUMA system that included conveyors, parts feeders, and robots small enough to work alongside humans. But it was the small robots in that system, supplied by Unimation, that became known as PUMAs". [2]

protocol

a set of semantic and syntactic rules that determine the behavior of functional units in achieving communication. [6]

PUMA

abbr for Programmable Universal Machine for Assembly.

punch press

a machine with a table controlled in the x and y plane used to cut or press holes

usually into sheet metal. Punches are often used to press vents into the sheet metal for the sides of electrical cabinets. [4]

quadrature

state of being separated in phase by 90° or one quarter cycle. A quadrature encoder provides both position and direction information to the control. [4]

quality

conformance to customer expectations. [6]

query

to request information from a database. [6]

query language

a language close to natural language allowing flexibility of expressing syntactical forms. It serves as the interface between the user and the mechanics of the system and enables him to query data. [6]

radial error probability (REP)

a probability that a percentage of onedimension measurements will lie on a radial (line) of given length, with the origin centered at truth or mean of the measurements; compare with circular error probability, spherical error probability; used to specify test cases for measurement errors of sensors of one dimension, such as velocity.

For example, a position error of 1 ft/s (50% REP) means that for any given measurement of position, p(distance from truth to measurement is less than 1) = 0.5.

In another example, a velocity error of 1 ft/s (50% REP) means that for any given measurement of velocity

then

$$p(\left|\vec{V}_i - \vec{V}_{\mu}\right| \le 1) = 0.5$$

 $V_i = V_v$

where \vec{V}_{μ} is the average of all \vec{V}_i or \vec{V} truth, depending upon context. [5]

radius

a straight line extending from the center of a circle or sphere to the circumference or surface [11]

RAMP

abbr for Rapid Acquisition of Manufacturing Parts. [6]

ramp

1 n. synonym for unit ramp [5]

2 v. to gradually accelerate a motor, essential if performance beyond the start/stop range is required. The slope of the ramp is a function of screw pitch, load, motor, and drive voltage and design. [8]

rapid traverse

cutting feedrate to quickly position the cutting tool for the next cut. [4]

rate

rate of change or data; derivative with respect to real time [5]

rate limiter

a filter that passes the input as the output, except that rate of change of the output is

September 26, 1994

limited to a maximum absolute value; compare with limiter.

Rate limiters are usually implemented as

$$y_n = y_{n-1} + sgn(x_n - y_{n-1})\min(x_{\max}^{\prime}, |x_n - y_{n-1}|)$$

where $\frac{\chi_{\text{max}}}{\chi_{\text{max}}}$ is the maximum absolute change per period. [5]

real time

1 time in a computational process which runs at the same rate as a physical process; for example, algorithms designed to run a fixed period T (filter time constants at set for T) and actually execute with frequency 1/T in real time [5]

2 within a strict, predictable time interval. Real-time processing often involves event waiting, timed reminders, validity intervals, and some form of approximate or progressive reasoning that allows creation of an acceptable (in some sense "best") response within the time allowed. [7]

real-time function

a function that must be carried out in a relatively brief time interval (typically a minute or less) and that is characterized by a commitment to execution. Here, any attempt to make a significant alteration in the planned course of action will result in some negative effect on the workpiece. [6]

real-time operation

an activity within a strict, predictable bounded-time interval. [7]

ream

a machining task; to finish a drilled hole to an exact size using a reamer [9]

reamer

a cutting tool used to produce a smooth accurate hole by removing a small amount of metal from a drilling hole [9]

reamer sharpening a type of grinding. reciprocating saw a type of saw machine.

recovery

restoration of IB consistency in the event of hardware or software error. [7]

reference

what must be achieved in order to match a plan; synonym for desired [5]

referential integrity

the capability of ensuring that repository data does not contain references to non existent data or definitions. It involves automatically deleting a relationship instance whenever the source or target of that instance is deleted. [6]

relative

applies to measurements, in a non-standard, moving reference, as opposed to fixed reference; compare with absolute [5]

reliability

the measure of a system's ability to perform in a stated manner for a given period of time under specified operating conditions and environments. [7]

remote jog pendant

September 26, 1994

a hand-held device with wheels and buttons used to manually move axes. Also called a teach pendant. [4]

REP

abbr for Radial Error Probability [5]

repeatability

capability of CNC software to exactly imitate accurate motions. [4]

replication

the existence of multiple instantiations of a module's functionality within an implementation. [7]

requirement

a declarative statement describing customer needs. [7]

resolution

the smallest unit of measurement that a measuring system is capable of distinguishing. [7]

resolver

analog position feedback device. [4]

resonance

1 midrange resonance: a parasitic oscillation which is endemic to stepper motors, although frictional loads may mask its effect. It typically sets in from 5 to 15 revolutions per second, and can easily cause a loss of synchronization (stalling). [8] 2 primary resonance: The rotor inertia of a stepper motor, coupled to its spring-like magnetic field characteristics, constitutes a basic spring-mass oscillator. In the absence of sufficient damping, stepper at certain frequencies may excite resonance in this - system, or resonate with the load, resulting in loss of synchrony. The addition of system damping, operation in half-step mode, or ramping through problem speeds will usually eliminate resonance. [8]

resource

any item within the manufacturing workstation used to transform a part, assist in the transformation of a part, or support other resources. These include machinesunder-control, such as robots, machine tools, and material handling devices, as well as fixtures, tools, lubricants, coolants, and workpieces. [7]

retrofitter

a person or company who installs (integrates) new CNCs and other electrical equipment on old machine tools. Some large companies have retrofitting departments that are responsible for constantly upgrading the machine tools. Small job shops usually wait until a machine is too expensive or impossible to fix or it cannot be used to fulfill a new contract for parts before trying to retrofit a new control on the machine. [4]

reusable software

software that can be used for more than one implementation. [7]

examining an existing part to determine its structure and make a copy or use that structure to create another part based on the features of the first. [4]

revolutions per minute (RPM)

this unit of measurement indicates the velocity at which a object must travel at a uniform angle to complete a circle in one minute. [4]

RMS

abbr for Root Mean Square

rigid tap

a type of tapping tool; compare flexible tap

robot

a programmable manipulator and its associated system. [7]

roll

an angular deviation from ideal straight line motion, in which the positioning table rotates around its axis of travel as it translates along that axis. [8]

rollback

return a database to its state before an uncommitted transaction was updated. In order to do this, a "before image" that reflects the state of the data items prior to update must be available for restoration. [7]

rollforward

restore updated information following a rollback. In order to do this, an "after image" that reflects the state of the data items following update must be available for restoration. [7]

root mean square (RMS)

a statistical measure of data; the square root of the mean of the square; compare with root sum square; for variables with mean of zero, the standard deviation is equal to the RMS. [5]

In discrete time, RMS is computed

$$x_{RMS} = \sqrt{\frac{\sum_{i=0}^{n} x_i^2}{n}}$$

In continuous time, RMS is computed:

$$x_{RMS} = \sqrt{\frac{1}{t}} \int_{0}^{t} x_{i}^{2}(\tau) d\tau$$

root sum square (RSS)

a statistical measure of data; the square root of the sum of the square; compare with root mean square; for a vector, its length is equal to the RSS of its scalar elements. [5]

In discrete time, RSS is computed:

$$x_{RSS} = \sqrt{\sum_{i=0}^{n} x_i^2}$$

In continuous time, RSS is computed:

$$x_{RSS} = \sqrt{\int_{0}^{t} x_{i}^{2}(\tau) d\tau}$$

rotary saw a type of saw machine. rotational force/torque a type of control law.

rotational position/velocity a type of control law.

roughing pass

a cycle of milling in which a significant margin is added to the tool motion, so that large amounts of material can be removed quickly. [15]

RPM

abbr for Revolutions Per Minute.

RS-274-D

an industry standard, part programming language for generating numerical control (NC) programs (using M & G codes). The programs describe tool path. This standard specifies the use of the EIA Standard RS-358 character code set and the basic characteristics of a numerical controller. [4]

abbr for Root Sum Square [5]

RT abbr for Real-Time.

SA

abbr for Standardized Application

saber saw

a type of saw machine which is a portable electric jigsaw. [13]

safety

safety relates to procedures to keep humans free from injury or exposed to hazards. [7]

safety envelope

- attribute of a part which describes the volume about the part in which it is safe to move the tool.

safety interlock

a safety interlock is dependent upon the design and purpose of the equipment. It provides a workstation warning and control in the event of human errors and misjudgments. It also detects component failures such as electrical overloads and shorts, valve sticking, or lack of flows and pressure. [7]

sampling frequency

rate of a periodic process; 1/T where T is the period; standard symbol fs; standard unit Hz; primary units $1/\theta$ [5]

saw

a type of machine; a machine tool used to cut hard material, usually consisting of a thin, flat blade or plate or tempered steel with a continuous series of teeth on the edge and mounted in a handle or frame. [13]

scalability

the capability to increase or decrease the functionality of a manufacturing workstation or its components. [7]

scalar

a quantity possessing only magnitude; compare with vector; standard notation x [11]

scaling

the process in the PID instruction which allows the setpoint and zero crossing values to be displayed in engineering units. Setpoint, deadband, process variable, and error may be scaled. (see PID control loop) [10]

schema

a data model that defines all abstract object data types including relationships, attributes and constraints. [7]

sculpted surface

a machined or molded object whose surface is contoured and non-linear, e.g. the shape of a light bulb. [6]

SDT

abbr for System Development Tool

SEB

abbr for Sensor/Effector Bus

second-order filter

a smoothing filter in which the output follows the input, only more slowly. It is usually implemented in software as a difference equation of period T as:

$$y_n = Ay_{n-1} - By_{n-2} + (1 - A + B)\frac{x_n + 2x_{n-1} + x_{n-2}}{4}$$

= $Ay_{n-1} - By_{n-2} + (1 - A + B)x_n$

A and B vary depending upon case:

$$A = 2e^{-\zeta \omega_n T} \cos\left(\omega_n T \sqrt{1 - \zeta^2}\right) = e^{-aT} + e^{-bT}$$
$$B = e^{-2\zeta \omega_n T} = e^{-aT} + e^{-bT}$$

where $\omega_n > 0$ (natural frequency) and $0 < \zeta < 1$ (damping ratio) are filter constants for underdamping (case 1), $\omega_n > 0$ and $\zeta = 0$ for no damping (case 2), a and b are filter constants for overdamping (case 3), and a=b or $\omega_n > 0$ and $\zeta = 1$ for critical damping (case 4).

If the input is a unit step then, the time responses are:

underdamping, no damping

$$y(t) = 1 - \frac{e^{-\zeta \omega_{nt}}}{\sqrt{1 - \zeta^2}} \sin(\omega_n \sqrt{1 - \zeta^2 t} + \cos^{-1} \zeta)$$

overdamping, a_b

$$y(t) = 1 + \frac{b}{a-b}e^{-at} + \frac{a}{b-a}e^{-bt}$$

critical damping, a=b

$$y(t) = 1 - (1 + at)e^{-at}$$

The Z transform is:

$$\frac{Y(z)}{X(z)} = \frac{(1 - A + B)z^2}{z^2 - Az + B}$$

The differential equations are:

$$y = x - \frac{2\zeta}{\omega_n} \dot{y} - \frac{1}{\omega_n^2} \ddot{y}$$
$$= x - \frac{a+b}{ab} \dot{y} - \frac{1}{ab} \ddot{y}$$
$$= x - \frac{2}{a} \dot{y} - \frac{1}{a^2} \ddot{y}$$

The Laplace transform is:

$$\frac{Y(s)}{X(s)} = \frac{\omega_n^2}{s^2 + 2\zeta\omega_n s + \omega_n^2} = \frac{ab}{(s+a)(s+b)} = \frac{a^2}{(s+a)^2}$$

When the second-order filter is used, it is commonly to smooth data, and to wash out transients at mode change. Usually, a firstorder filter suffices, and it being less expensive, is chosen over a second-order filter. Typical values for ω_n , a, and b are 0.1-2 rad/s, and ζ 0.1-0.9 (unitless). It should also be noted that two first-order filters can be chained together to form a second-order filter that is critically damped or overdamped.

When implementing a second-order filter on normalized variables, such as angles, the discontinuities require special treatment. [5]

semantics

the meaning or interpretation assigned to a group of characters in a language; compare syntax

semicircle

a measure of angle, 1 semicircle = 1 rad = 180 deg; angles from physical devices may be reported in semicircles in order to compress data [5]

sensor

1 feedback device used on a CNC machine to monitor the operation and check output. Common types of sensors are force and power sensors, touch probes, and acoustic monitors. [4]

2 a device that measures, receives, or generates data [5]

3 a transducer whose input is a physical event and whose output is a quantitative measure. [7]

sensor control

- control of a machine, based on sensor measurements. [7]

sensor/effector bus (SEB)

a shared medium for distributing electronic signals between the controllers, sensors, and effectors. [7]

sensor/effector standardized application (SESA)

connects the controller to the instrument(s) under control, i.e., it handles all input and output relating to controlled instruments on the manufacturing workstation. [7]

sensor feedback

the return of information obtained from sensors to maintain performance or to control a system or process. [7]

sensor hierarchy

a relationship of sensor information processing elements whereby the results of lower level processing are abstracted and used as inputs to higher level processing. [7]

SEP

1 abbr for Spherical Error Probability

2 abbr for Scenario-based Engineering Process

sequential function chart (SFC)

a graphical language that uses five simple elements to provide graphical а representative diagram of any sequential The basic elements are steps, process. transitions, and actions. A step consists of any number of actions or pieces of a program that are carried out until an action qualifier is met. The language supports alternative and parallel sequences. Because discrete manufacturing applications all run as a sequence of steps, SFC is an excellent way to logically decompose the process and automatically transfer the process into a structured program. SFC greatly enhances communication between the system architect. project system manager. integrator, and maintenance personnel. [10]

service

a domain-independent mechanism that provides fundamental capabilities enabling integration of independently developed, domain-specific functional entities. [7]

servo control

1 a feedback control system involving both hardware and software in which the output is some mechanical position, velocity, or acceleration. [4]

2 an interface that allows a controller to interface to industry standard servo drive amplifiers using +/- 10 volt motor commands and A quad B encoder feedback. The interface is capable of independently controlling the axis in closed-loop mode in either velocity or position modes using a DSP to perform the computations required for servo control. The DSP has a command set that eliminates most of the host CPU overhead associated with motion. Velocity profiling is supported. Loop compensation is accomplished by a digital Proportional Integral Derivative (PID) filter updating the servo at 4khz. Feedback from an encoder is received at 1 million counts/second. The interface also handles motion control applications requiring opto-isolated inputs for end-of-travel limits, marker pulse, and - open collector or differential quadrature encoders. [10]

servo-loop closure rate

in a closed-loop servo controller, the synchronous rate at which servo commands are generated, feedback sensors are sampled, control laws are computed and servo drive signals are generated. [7]

servo motor

a DC motor which produces a torque proportional to current. Precise positioning is achieved by linear or PWM (duty cycle) control of motor current or voltage, together with accurate monitoring of position via an external feedback device. [8]

SESA

abbr mentioned in the definition for controls standardized application.

SFC abbr for Sequential Function Chart

SFM

abbr for Surface Feet per Minute.

shared axis

a programmable axis that is shared by two or more axes or groups of axes; e.g. two 5axis milling heads sharing a common rail type axis that would position them. [6]

shop floor

the part of a manufacturing enterprise that has the responsibility to produce a product by performing manufacturing operations. The shop floor is the physical location through which the product passes on its way to completion. [6]

sign

common math function; returns whether the input value is positive or negative; standard notation sign x [5]

signal

part of received data that is desired; compare with noise [5]

signal-to-noise-ratio (SNR, S/N)

the ratio of magnitude of a desired signal to the magnitude of the noise received with it; standard units dB, 1; primary units 1 [5]

simultaneous axes

axes that move simultaneously at different or varying velocities as commanded by the control. These begin and end at programmed points but the simultaneous axis motion does not accommodate specific multi-axis paths in between these points. [6]

sine

common math function; standard notation sin x, sx [5]

single instruction

an interactive condition that occurs when the controller is in the normal production mode; starts a process that enables the manual jog of machine axis for tool positioning

September 26, 1994

SOSAS Rev. 2.6

purposes during work piece set-up, or allows a process parameter to be temporarily varied; manual operation is also useful for activities where the workpiece remains fixture on the machine and a cursory inspection is performed via a digital probe - or similar device; single instruction can also be an interactive condition that occurs when the controller is in the normal production mode. Examples of machine and process related macro execution include "Home Machine", "Auto Tool Change", "Set Point", "Null". Macro execution shall also involve the performance of a set of manual data input instructions intended to drive an axis or the machine to a predetermined position/location within the work envelope; or execute a predetermined collective set of process parameters. [7]

single-step normal production

an interactive condition that occurs when the controller is in the normal production mode; the machine/process pauses after each control plan segment and maintains that position until a human interface button is pressed. [7]

situation

current environment and surroundings [5]

slabbing cutter sharpening a type of grinding.

slaved axis

an axis whose path follows that of another axis but in some translated or offset amount. [6]

SMM

abbr for Surface Meters per Minute.

smoothing filter

a filter to reduce quick changes of a signal by attenuating high frequencies; see firstorder filter, second-order filter, wash-out filter [5]

S/N

abbr for Signal-to-Noise ratio [5]

SNR

abbr for Signal-to-Noise Ratio [5]

software library

a repository for software components. [7]

solution viewpoint

a DSSA view of a system.

SOSAS

abbr for Specification for an Open System Architecture Standard.

SOSAS-conformant

a characteristic of a NGC product indicating that it meets all applicable requirements in the NGC SOSAS. [7]

SPC

abbr for Statistical Process Control.

speed

the scalar component of velocity, measured in distance per unit time

spherical error probability (SEP)

SOSAS Rev. 2.6

a probability that a percentage of threedimension measurements will lie within a sphere of given radius, with the sphere centered at truth or mean of the measurements; compare with circular error probability, radial error probability

SEP specifies test cases for measurement errors of sensors of three dimensions, such as velocity X, Y, and Z. For example, a total velocity error of 1 ft/s (50% SEP) means that for any given measurement of velocity

$$\vec{V}_i = \{V_E, V_N, V_V\}$$

then

$$p(\left|\vec{V}_i - \vec{V}_{\mu}\right| \le 1) = 0.5$$

where \vec{V}_{μ} is the average of all \vec{V}_i or \vec{V} truth, depending upon context. [5]

spindle

a rotating mechanical assembly that serves as an axis for revolving tools, a workpiece, or an end effector for a machine tool. [7]

spindle lock

condition in which the spindle is stopped so that it cannot rotate

s-plane

continuous complex frequency plane; see Laplace transform; S-plane is used in control systems engineering in the design of control laws [5]

sprayer

a type of end effector.

spotface

a machining task; to machine a circular spot on the surface of a part to furnish a flat bearing surface for the head of a bolt or nut [9]

\mathbf{ST}

abbr for Structured Text.

stall speed

the maximum speed which a stepper motor, properly ramped, can achieve without loss of synchrony. This speed is a function of motor inductance, ramp slope, applied load, drive voltage and design. [8]

stand alone control

one PLC controls all of the I/O. Stand alone is perfect for a smaller application, especially one that requires a continuous process. [10]

standardized application (SA)

an application that supports current controller practices and provides a baseline controller. [7]

standard notation

common form of a mathematical expression [5]

standard units

units commonly encountered for a particular quantity [5]

start/stop speed

the maximum step rate which can be applied to a stationary stepper motor and still retain error-free performance. Also, the rate from which a stepper motor may be instantaneously stopped without overshooting. This is a function of the screw pitch, load, drive voltage and design, ... and motor. [8]

statistical process control (SPC)

the statistical techniques to analyze a process or its end product so appropriate actions can be taken to control and improve - the process. [7]

steel block

a type of fixture.

step

synonym for unit step [5]

stepper motor

a type of motor featuring two or four stator coils and a toothed permanent magnet rotor, which moves through a small angle in response to a specific sequence of coil current changes. [8]

stepper motor control

an intelligent interface that allows a controller to interface to stepper motors. The interface provides precise motion control profiling requiring microstepping rates at millions of pulses per second. Each axis is controlled by an intelligent controller which provides five output signals and seven input signals, along with inputs for incremental encoders connected in either single-ended or differential form. It provides pulse, direction CW/CCW, hold, and user defined outputs. Inputs are provided for axis limits, ramp up/down, home, alarm and user defined input. All input and output is opto-isolated for noiseimmunity and electrical isolation. Internal registers control the number of steps, acceleration/deceleration rates, starting rate, slew rate, and ramp-down point. [10]

step rate

the frequency of coil current changes, or input pulse train applied to a stepper motor, in pulses/second or hertz. For 200 step/revolution motors, the full step rate multiplied by .3 equals the rotation rate in RPM. [8]

stock

1 ahe raw material from which a part is cut. This material is often metal, but it can be any substance that can be cut such as plastic or even glass. [4]

2 any transformable material that is not currently involved in a transformation process. [7]

structured text (ST)

a high level block structured language that has a syntax that resembles PASCAL. ST can be used to express complex statements involving variables representing a wide range of data types. [10]

surface representation

the form presented to a user that represents surface function. [7]

synchronous

1 occurring at fixed time intervals (see asynchronous). [7]

2 refers to any tasks or electronic/computer events that occur at fixed intervals or periods and must be executed in a lock-step fashion (see asynchronous). [6]

syntax

set rules specified for a language. The relationships among characters or groups of characters independent of their meaning or the manner of their interpretation and use. [7]

system

an organized collection of people, machines, and procedures, required to accomplish a set of specific architectural or enterprise objectives. [6]

system development tool (SDT)

software that supports the analysis, development, integration, operation and maintenance of NGC-conformant elements. [7]

tachometer

a device for measuring angular velocity [5]

tactile sensor

a transducer that is sensitive to contact pressure. [7]

tangent

common math function; standard notation tan x [5]

tap

tape

to cut an internal screw thread in an existing hole

a type of fixture

tap sharpening a type of grinding

target position

position command by the CNC. [4]

task

an activity within an operation to accomplish a goal. [7]

task execution standardized application (TESA)

responsible for routine coordination of multiple machines within a manufacturing workstation, as well as direct invocation of sensor/effector functionality as appropriate. [7]

task schedule

the highest or master schedule for the workstation. It is the combined schedules of all activities for the workstation. [7]

TCP

abbr for Tool CenterPoint.

temperature

the average kinetic energy of a body; standard symbol T; standard units $^{\circ}C$; primary units T [11]

terminal emulation

display software that makes a computer display (CRT) and keyboard behave as a "dumb" terminal to be interfaced to a host or other computer. [6]

TESA

abbr for Task Execution Standardized Application

third party vendor

one who sells or offers for sale goods and/or services for use with equipment manufactured by other vendors. [6]

- thread

a machining task; to form an external screw thread or threads on or in material. [13]

throughfeed grinding

a type of centerless grinding.

time

time relative to some defined time base; delta time as a fixed period (usually the average time for one computational cycle); standard symbol t, T; standard units s; primary units θ [5]

time constant

constant for a first-order filter determining time at which the output of the filter reaches Å0.6321% of a step input; standard symbol τ ; standard units s; primary units θ [5]

tolerance

1 allowed error in measurements [5]

2 permissible variations in the dimensions of a machined part. A part that does not vary beyond these predefined limits is said to be "in tolerance." [4]

tool

1 a cutting tool used by a machine. [4]

2 an item that physically implements a predefined action. [7]

tool centerpoint (TCP)

a fixed point on the tool that serves as the reference point for the tool; all tool motion is expressed as motion of the TCP

tool change position

attribute of a tool, describing the point to which the tool should travel so that the tool changer can get to it.

tool changer

an automatic device to hold and dispense tools during part cutting. [4]

tool path

route a cutter takes when machining a part. A part program may display this path graphically on a computer monitor. [4]

tooth loading material removal rate

top-down methodology

a process that begins with an analysis of key objectives, data or activities at a high level and works down to the lowest level of detail. [6]

torque

Force and the direction of the force combined to determine rotation movement.
 [4]

2 the moment of a force or system of forces tending to cause rotation; torque = (net force) (length of moment arm) [11]

torque compensation

used in digital control over analog movement, associated with tuning. Should the joystick move to the ifull oni position, the system must back off the command until the engine can catch up with its RPM rate, creating a lag in response time. [21]

total quality management (TQM)

a leadership philosophy focused on customer satisfaction, achieved through empowered employees, teamwork, and continuous improvement. [6]

TQM

abbr for Total Quality Management.

transfer speed

in teaching a path through a teach pendant, the accuracy of the path generated from the path program decreases as the transfer speed is increased. This causes the teaching to be done at a slow rate, thereby hindering the testing of new algorithms. [17]

transaction

 a mechanism for ensuring that all actions associated with one or more DMSs service call will be treated as a single unit of work.
 [7]

2 a single bi-directional exchange of messages from a requestor to its responding partner and from the responder to the requestor in that order. [6]

transducer

a device that converts a physical phenomenon into an electrical signal. [7]

transient-free switch

a switch with a wash-out filter so that the output contains no transients (steps) at switch time [5]

transpose

standard matrix operator; standard notation X^{T} [5]

travel

distance the bed moves along an axis. [4]

traverse grinding

a type of grinding.

trigger

a monitor placed on a data item that initiates some action based on access to or change in value of that item. [7]

tuning

digital control over analog movement. An example is joystick control of a process (like a hydraulic pump). As soon as the joystick crosses 10%, the valve should open (since it takes a specific amount just to start the movement). Every movement of the joystick causes the valve to open more. [21]

tuple

row of data in a (database) table. Relationally speaking, tuples are constituents of relations. [6]

turn

a machining task.

two-phase commit

process in which a commit manager sends out an intent-to-commit message to all subprocesses and those subprocesses must unanimously acknowledge consent before - the commit can take place. [7]

ultrasonic machine (UM)

a machine which performs material removal via sound waves.

UM

abbr for Ultrasonic Machine.

unit

a standard quantity, such as ft or mi; synonym for dimension [5]

unit function

one of a collection of functions used as standard test cases in control systems engineering; standard notation $u_k(t)$; standard units 1; primary units 1. The primary unit functions of interest are the unit impulse, the unit step, and the unit ramp (see table in Notation section for properties) [5]

unit impulse

a function used as a standard test case in control systems engineering; a spike of "area" one at time t = 0; synonym for impulse; see unit functions; standard notation $u_0(t)$, δ (t); standard units 1; primary units 1; written $u_0(t - \tau)$ for a ramp starting at time $t = \tau$; standard engineering term [5]

unitless

no units, such as ratios; a quantity with standard units of 1; a quantity with primary units of 1; synonym for dimensionless [5]

unit ramp

a function used as a standard test case in control systems engineering; a line of slope 1 starting at zero at time t = 0; synonym for ramp; see unit functions; standard notation $u_2(t)$; standard units 1; primary units 1; written $u_2(t-\tau)$ for a ramp starting at time $t = \tau$ [5]

unit step

a function used as a standard test case in control systems engineering; a step from zero to one at time t = 0; synonym for step; see unit functions; standard notation $u_{.1}(t-\tau)$, u(t); standard units 1; primary units 1; written $u_{.1}(t-\tau)$ for a step at time $t = \tau$ [5]

upgradeability

the characteristic of enhancing speed, capacity or functionality of a software or hardware element within a manufacturing workstation (see scalability). [7]

user interface

the messages and informational and data entry screens displayed on a CRT to guide the system user in operation of the machine. Common elements of a CNC user interface are the graphic representations of parts, messages from the motion control subsystem, and part setup and programming option lists. [4]

user interface application

code that allows a user to interact with or manipulate specific information. [7]

user interface device

provides surface presentation functionality - directly to a user. [7]

user interface support package

a COTS component that provides specific HUI functionality. [7]

VAL

an robot programming language, BASIC-like in syntax.

vector

a quantity possessing both magnitude and direction, represented by an arrow the direction of which indicates the direction of the quantity and the length of which is proportional to the magnitude; compare with scalar; standard notation \bar{x} [11]

velocity

rate of change of location, either scalar or vector, often with subscripts to denote the coordinate frame; time derivative of position; time integral of acceleration; standard symbol v, V; standard units kt, ft/s; primary units L/θ [5]

verification

the methods or means used to ascertain that the SOSAS conforms to the requirements. [7]

version

a formal record used to help track an object's evolution over time; a version may

be associated with a context and loaded according to that context. [7]

vice

a type of fixture; any of various tools having two jams for holding work that close, usually by a screw, lever, or cam. [13]

view

a presentation with limited perspective based on context. [7]

virtual machine

a specification of mechanisms providing a common software interface to computing platform services that isolates software modules from the computing platform. [7]

wash-out filter

a filter to smooth a transition due to change of input source, such as when changing modes; see transient-free switch

Wash-out filters are usually designed as $y = x_i - f_{HPF}(AEx)$, where i selects a particular input x_i , $AEx = x_p - x_r$, when i changes, and f_{HPF} is a high-pass filter, often first order. Time constants are usually about 5 s. [5]

waypoint

a discrete point along a path

ways

the rails on which a large portion of the machine, such as the bed or the spindle column, moves. [4]

welder

a type of end effector.

wheel dressing

a type of grinding.

WMSA

abbr for Workstation Management Standardized Application.

working envelope

a defined boundary representing the maximum extent or reach of a machine in all directions. [7]

working range

all reachable positions within the working envelope; the range of any variable within that the system normally operates. [7]

workpiece

the material that is being machined to produce a product or the product itself. [7]

workstation

see manufacturing workstation. [7]

workstation management standardized application (WMSA)

performs the control of the manufacturing workstation by determining and rescheduling its activities. These activities include coordination of startup, calibration, diagnosing and shutdown of the machines,

and external communications. This SA provides for configuration management of the manufacturing workstation and controls its modes of operation. It monitors the and controller machines within the manufacturing workstation. predicts failures, and provides for failure mitigation. This WMSA tracks and maintains the status of all assets available within the manufacturing workstation. Furthermore. this SA provides the single point of communication with the cell. [7]

)

workstation model

the data necessary to control the manufacturing processes. It includes information such as parameters of the controlled equipment, quality control data, tool performance data, and part material characteristics. [7]

workstation planning standardized application (WPSA)

performs the preparatory planning required for the workstation. This planning includes determining executing and the manufacturing workstation operations plan, preparing a CCP to be executed by the TESA, and determining collision free path for a machine. This SA also incorporates modeling services for use in operations, planning task, and path and plan verification. [7]

world coordinate

a position relative to a frame of reference fixed on the manufacturing workstation. [7]

world model

the system's estimate and evaluation of the history, current state, and possible future

states of the world, including the states of the system being controlled. [7]

WPSA

abbr for Workstation Planning Standardized Application.

yaw

an angular deviation from ideal straight line motion, in which the positioning table rotates around the Z (vertical) axis as it translates along its travel axis. [8]

z axis

identifies the Cartesian axis that moves up and down (the "Z" plane). The movement of the "Z" axis controls the distance a cutting tool travels into a part. [4]

zero-crossing

a part of the deadband control. The PID instruction uses the deadband error range when a process variable moves into the deadband and crosses the setpoint. (see PID control loop) [10]

z-plane

discrete complex frequency plane; see z transform; z-plane is used in control systems engineering in the design of control laws [5]

z transform

#V

a mathematical relationship to model a discrete function in the complex frequency domain (z-plane); compare with continuous-time equation, difference equation, differential equation, discrete-time equation, Laplace transform; Z transforms are commonly used by systems engineers to describe systems; see first-order filter, second-order filter, unit functions for examples [5]

.

71

>

)

E.3 Notation

The table below lists units used in this dictionary. Many entries list standard units for quantities. These are the units most likely to be used by domain experts. It is not intended to serve as the final word on units. Some engineers will prefer other units.

This dictionary lists primary units for many symbols. All units can be broken down into combinations of four primary units: (1) mass (M), length (2) length (L), (3) time (θ), and (4) temperature (T). Unitless quantities have primary units of 1. The primary units are useful for checking consistency of equations.

symbol	primary units	meaning	measures
g	L/O	gravitational units	acceleration
deg	1	degrees	angle
DMS	1	degrees, minutes, seconds	angle
rad	1	radians	angle
semicircle	1	semicircles	angle
rpm	10	revolutions per minute	angle
lbf	ML/0'	pound force	force
kip	ML/0'	thousand pounds	force
Hz	1/0	hertz	frequency
ft	L	feet	length
in	L	inch	length
lbm	Μ	pound mass	mass
slg	М	slug	mass
in Hg	ML/0'L	inches mercury	pressure
mbar	ML/0'L	millibar	pressure
psi	ML/0'L	pounds per square inch	pressure
%	1	percent	ratio
dB	1	decibel	ratio of power
°C	T ·	degrees Celsius	temperature
hr	θ	hour	time
S	θ	second	time

₩.

ť

The tables below list the symbols used in this dictionary.

nomenclature	standard units	primary units	meaning
Ā	ft/s',g	L/0'	acceleration
a	ft/s'.g	L/0'	acceleration
- a	1	1	Euler parameters
ac	ft/s',g	Ľ/θ'	Coriolis acceleration
α	rad/s'	1/0'	angular acceleration
-------------------	--------------	-------	----------------------------
b	1	1	Euler parameters
с	1	1	Euler parameters
e ₁₂₃₄	1	1	Euler parameters
fs	Hz	1/0	sampling frequency
g	ft/s'	L/θ'	gravitational acceleration
- g	lbf, kip	ML/θ'	gravity
ĩ	1	1	identity
J	ft/s'	L/0'	jerk
j	ft/s'	L/0'	jerk
ω	rad/s, rpm	1/0	angular velocity
Р	ft, mi	L	position
р	rad/s, deg/s	1/0	pitch rate
p	ft, mi	L	position
q	rad/s, deg/s	1/0	roll rate
r	ft, mi	L	range (mg)
r	rad/s, deg/s	1/0	yaw rate
S	1	1	Euler parameters
Т	S	θ	period
Т	°C	Т	temperature
Т	S	θ	time
t	S	θ	time
θ	rad, deg	1	angular position
ΔV	ft/s, kt	L/0	speed error
V	kt, ft/s	L/0	velocity
v	kt, ft/s	L/0	velocity
x	ft, mi	L	position
у	ft, mi	L	position
Z	ft, mi	L	position

Greek	standard	primary	
symbol	units	units	meaning
ω	rad/s, Hz	1/0	damped frequency
ω _c	rad/s, Hz	1/0	cutoff frequency
ω _n	rad/s, Hz	1/0	natural frequency
Φ	rad, deg	1	roll
ф	rad, deg	1	roll
φ	rad/s, deg/s	1/0	roll rate
Ψ	rad, deg	1	yaw
Ψ	rad, deg	1	yaw
Ψ	rad/s, deg/s	1	yaw rate
τ	S	θ	time constant
θ	rad, deg	1	pitch
θ	rad, deg	1	pitch

.

SOSAS Rev. 2.6

•

)

θ	rad/s, deg/s	1/Ө	pitch rate
ζ	1	1	damping ratio

۹.	

(

notation	meaning
<u>S(+)</u>	unit impulse
0(1)	integral
	Integral
A	matrix
IAI	determinant
arctan x	arctangent
atan x	arctangent
[aij]	matrix
cos x	cosine
cx	cosine
Dx	derivative
det [1]	determinant
dx/dt	derivative
ex	exponential
exp x	exponential
ln x	logarithm
log x	logarithm
log _b x	logarithm
$\max(x_1,, x_n)$	maximum
$\min(x_1,,x_n)$	minimum
sgn x	sign
sin x	sine
SX	sine
$sx + x(0^{+})$	derivative
tan x	tangent
tan ⁻¹ x	arctangent
$\tan^{-1}(y,x)$	arctangent
u(t)	unit step
u ₋₁ (t)	unit step
u . ₂ (t)	unit ramp
u _k (t)	unit functions
u ₀ (t)	unit impulse
X-1	inverse
X ^T	transpose
x	scalar
x	derivative
x (1)	derivative
x ⁻¹	inverse
IxI	absolute value
x '	derivative

SOSAS Rev. 2.6

vector

subscript	meaning	example
-2	ramp	unit ramp $(u_{-2}(t))$
· -1	step	unit step $(u_{-1}(t), u(t))$
0	impulse	unit impulse $(u_0(t), \delta(t))$
b	base	$logarithm (ln x, log x, log_b x)$
с	Coriolis	Coriolis acceleration (a _c)
с	cutoff	cutoff frequency (ω _c)
k	order	unit functions $(u_k(t))$
max	maximum	limiter
min	minimum	limiter
n	cycle n	first-order, integrator, second-order filter
n	natural	natural frequency (ω_n)
S	sampling	sampling frequency (f _s)

ī

١

)

Category Index E.4

computer, communication

_ Arcnet **CANBus** dedicated channel PAMUX bus sensor/effector bus

computer science

algorithm analog input/output ASCII autonomous control batch execution binary coded decimal checkpointing commit computer cycle computing platform concurrency database database management system data bus data structure device driver discrete I/O distributed database dynamic binding exception handling graphical user interface hierarchical control human user interface information base information base application interrupt knowledge base knowledge base management system latency locking logging

operating system protocol schema semantics syntax trigger user interface

control law, categories of

linear force/torque linear position/velocity rotational force/torque rotation position/velocity

controller, concepts regarding block block cycle time canned cycle controls standardized application numerical control conversational part programming open loop data starvation diagnostic drawing interchange file feedback forward kinematics interpolation interpolation block inverse kinematics leadscrew mapping multiaxis mode offset overshoot part program part programming quadrature roughing pass servo-loop closure rate stepper motor control target position torque compensation transfer speed waypoint

controller, parts of control law interpolater kinematic model machine model path planner servo control

control, types of adaptive control binary cutter location closed loop computer numerical control continuous path control direct numerical control distributed control endpoint control motion control point-to-point control sensor control stand alone control

engineering & mathematics absolute acceleration accelerometer accuracy angle angular acceleration angular position angular velocity aperiodic asynchronous axis bias coefficient of friction coordinate correlation delta desired

deviation dimension dimensionless discrete time distance domain - drift error euler angles euler parameters function granularity gravitational acceleration impulse ierk metrology model noise non-uniform rational b-spline orientation period periodic position precision radius ramp rate real time reference relative resolution root mean square root sum square sampling frequency scalar semicircle signal signal-to-noise ratio speed s-plane standard notation standard units step synchronous tachometer temperature

time tolerance torque transducer unit unit impulse unit impulse unit step vector velocity z-axis z-plane

english

abort activity clockwise counterclockwise operational situation

error abbe error dynamic error dynamic following error following error geometric error kinematic error

filter, concepts regarding

cutoff frequency damped frequency damping ratio hysteresis natural frequency time constant transient-free switch

filter, types of averaging filter band-pass filter complementary filter first-order filter high-pass filter integrator kalman filter limiter low-pass filter rate limiter second-order filter smoothing filter wash-out filter

)

fixture, types of

bolt chuck clamp jig parallels steel block tape vice

functions, types of

absolute value arctangent continuous-time equation cosine derivative determinant difference equation differential equation discrete-time equation exponential extrapolate identity interpolate inverse Laplace transform logarithm magnitude matrix maximum minimum normalizer

sign sine tangent transpose z transform

ſ

grinding, types of

centerless grinding endfeed grinding end mill sharpening face mill cutter sharpening form cutter sharpening form grinding infeed grinding internal grinding plain mill sharpening plunge grinding reamer sharpening slabbing cutter sharpening tap sharpening throughfeed grinding traverse grinding wheel dressing

ladder logic

ladder diagram ladder logic execution ladder logic programming programmable logic controller

machine, types of

abrasive waterjet machine bed mill saw coordinate measuring machine electrochemical machine electrodischarge machine electron beam machine knee mill machine lathe manual machine milling machine moving column machine Programmable Universal Machine for Assembly punch saw ultrasonic machine universal mill machine vertical mill machine

manufacturing, concepts in 2.5 D aiding application-specific integrated circuit approach vector availability backlash chatter circular error probability close dancing computer-aided manufacturing configuration consumable control plan emergency stop encoder feedback external communication external interface feature-based definition feedrate feedrate override function block diagram geometry goal **I/O** interfaces lead lead error lead through teach look ahead materials resource planning modal network interfaces NGIS-NC Inspection Project nut compliance one-shot operator interface orthogonality part transformation part zero pitch plan planning prismatic process radial error probability rapid traverse reference frame repeatability revolutions per minute roll safety safety envelope safety interlock scalability sensor feedback spherical error probability spindle lock statistical process control tool path tooth loading travel tuning upgradeability virtual machine working envelope working range world coordinate world model yaw

manufacturing, languages in

dimensional measurement interface specification EXPRESS FLEXIS IEC1131-3 instruction list

interactive graphics exchange specification Karel ladder logic M & G codes man-machine language - neutral manufacturing language Product Data Exchange using STEP programmable application logic **RS-274-D** sequential function chart structured text VAL

manufacturing, levels of cell center enterprise factory job shop

manufacturing, objects in

actuator asset automated ground vehicle bed chip controller conventional machine tool coolant coolant mechanism coordinated joint drive effector electrical cabinet enclosure encoder end effector end user environment face plate

feature fixture gripper Hall effect sensor hand crank home switch instrumentation iron joint leadscrew limit switch link lubricant machinist manipulator manufacturing workstation mold motor operator operator console optical encoder original equipment manufacturer pallet pallet shuttle part remote jog pendant resolver resource retrofitter robot sensor spindle sprayer stock table tactile sensor tool changer ways welder workpiece workstation

motor, concepts about holding torque

microstepping phase current phase sequence ramp resonance stall speed start/stop speed step rate

¥

)

motor, types of AC induction motor brushless motor DC motor servo motor stepper motor

PID

deadband PID control loop scaling zero-crossing

saw, types of

band saw jig saw reciprocating saw rotary saw saber saw

software engineering, concepts in actor agent application program architecture description language application programming interface

attribute class commercial off-the-shelf package

component computer-aided design configuration management domain-independent application dynamically linked library - field replaceable unit holon input/output artifacts instance installation integration architecture interchangability interface interoperability life cycle costs modular object object module object-oriented object-oriented programming object-oriented structures open system open system architecture open system interconnection open technology polymorphism portability product life cycle reusable software reverse engineering service software library verification version view viewpoint

SOSAS, concepts in conformance conformance class element mode machine under control multiprocessor control

open applications primitive function product definition data product definition data interchange real-time operation recovery replication sensor/effector standardized application sensor hierarchy single instruction single-step normal production standardized application surface representation system development tool task execution standardized application task schedule transaction two-phase commit user interface application user interface device user interface support package workstation management standardized application workstation model workstation planning standardized application

task, types of area clear milling blue bore break chip broach calibration climb milling conventional milling counterbore countersink drag drill dwell etch failure recovery fault detection/isolation fault management feed finish grind initialization integrate/configure iog mark mill odd pockets with islands milling orient plane plate probing ream saw spotface tap thread turn

tool, concepts regarding cutter compensation cutter compensation plane cutter compensation radius offset flute gauge vector home position in-process gauging length offset optimal feedrate optimal spindle speed tool centerpoint tool change position

tool, types of drill bit end mill flat end mill flexible tap grind belt grind wheel rigid tap saw belt saw blade

viewpoint, types of

application viewpoint architecture viewpoint communications viewpoint component viewpoint design viewpoint executive services viewpoint physical resource viewpoint solution viewpoint

other

command context look and feel override planner based power-up reliability requirement

recently added

abstract data type American National Standards Institute artificial intelligence assembly cache canonical class hierarchy data data dictionary European Strategic Programme for Research and Development of Information Technology

expert system fabrication geometric modeling applications program integrated computed aided manufacturing intelligent machining workstation inventory International Organization for Standardization Industrial Real-Time **Operating System** Nucleus just-in-time low rate initial production maintainability maintenance manufacturing Manufacturing Automation Protocol Open System Interconnection paperless factory part geometry part number platform predicate process plan quality query query language real-time function referential integrity sculpted surface shared axis shop floor simultaneous axes slaved axis system terminal emulation third party vendor top-down methodology total quality management

tuple

¥

)

APPENDIX F --- ARCHITECTURE DESCRIPTION LANGUAGE (ADL)

The architecture description language (ADL) describes rules for composing application architectures. The language constrains responsibilities, components, and interfaces. It is a constraint-based rather than a computational language. Categories of ADL rules include:

- applicable to architectural concepts, such as messages;
- applicable to domain concepts, such as mode;
- applicable to responsibilities, such as tool verification;
- applicable to components, such as motion control.

In a general sense, the ADL may contain a rule stating that the application architecture is not complete until all resources required by components are supplied by other components in the application. However, a great deal of domain knowledge is embodied in the ADL. More domain knowledge is necessary with rules that require the selection of particular primitive components, such as the technical selection of forward and inverse kinematics.

Future efforts in language research are focusing on communication concepts that will support the dynamic construction of component-based architectures. One such language is the Adaptive Semantic Language, described below.

THE ADAPTIVE SEMANTIC LANGUAGE

Introduction

As the complexity of software systems has increased, a number of formal models have been proposed to address the decomposition of large software systems into sets of reusable components. The Object Oriented (OO) paradigm is only the latest in a collection of models designed to divide the responsibilities of systems into separate objects which can provide a service to the rest of the system through a standard set of messages. The strength of the OO paradigm is that the methods and data needed to fulfill a message request can be completely held within an object, and therefore, hidden from the rest of the system. Thus, the high-level system designer needs only to understand the message-passing aspects of the object, not how the object internally deals with the requested service. At a higher level of abstraction, the complete set of

objects making up a sub-component of a very large system can be abstracted to a single unit, an agent, which itself can provide a higher level of services to the overall system through a standard set of messages.

Both the object and the agent model have the potential of greatly reducing the amount of new - software required to build a new system by allowing a standard method for the reuse of existing components. However, neither of these models addresses the complexity of design at the high level. Without clear documentation of a component's services and message interface, the component becomes unusable, but even with good documentation the selection of components requires a complex search over a database of standard messages and services. Further, new components cannot be considered for use by the system designer until their services and message requirements are added to this database. If this database exists within a standards document the process of adding new components can be very slow and not easily automated.

A better solution than a static database which must be modified for every change and addition to the set of available components would be a method by which the components themselves could contain semantic knowledge about their abilities and interface requirements. A system could then be designed by providing a set of system-construction agents with an ability to query the components, and then, based on the set of selected components tailor the system's internal interfaces to allow these components to carry out their required communication. However, such a method would require that both the abilities and interface information of a component be held in a common semantic representation which can be communicated between components using a semantic language.

Since the required semantic language would require an adaptiveness similar to human language while maintaining the non-ambiguity of a formal language, the logical source of such a language should be found by first formally defining the syntactic and semantic features of a human language required, and then, attempting to formally define the resulting features into a knowledge representation model which will allow the programmatic storage and transmission of the model's syntactic and semantic components. To accomplish this, both the requirements of the adaptive semantic language and a model for the semantic representation must be determined. Further, a testing platform must be constructed to ensure the resulting theory can be proven to be achievable.

This appendix will provide the theoretical support for an adaptive semantic language and the n-Towers model for the semantic representation which supports its use for communication from Y

)

F-2

programmatic agent to programmatic agent and from programmatic agent to human. The method by which this model can interface to sub-language elements of the programmatic agents will also be addressed.

The n-Towers Model

(

The n-Towers model's purpose is to provide a method for computers to understand HL and to communicate with each other through a connection at the Adaptive Semantic Language (ASL) layer. The model has been based largely on the principles of Government and Binding (GB) linguistic theory. Using this foundation, the model attempts to incorporate the best features of three current semantic theories; 1) Situation Semantics, 2) Possible Worlds Semantics, and 3) Conceptual Semantics. Since no existing Knowledge Representation (KR) was found to be adequate, the model is based on the completely new KR defined as Situational Dependencies.

Figure 1 presents one tower of the of n-Towers Model. In this model there are five levels (or layers) between the Human Language (HL) Surface Structure and the Symbol Grounding Plane. The two lowest levels (Meta-Semantics and Situational Dependencies) are viewed as below linguistic knowledge. The model's breaking of HL syntactic processing into two layers (the surface and deep structure) should not be viewed as a total embracing of GB theory. The Deep Structure layer may best be viewed as a place holder for any HL information which cannot be captured at the surface layer. Thus, the contents of these two layers could depend on the requirements of any linguistic theory chosen to support the language processing above the ASL layer.

The interface between the HL layers and the lower three layers is the ASL layer. This layer provides a non-ambiguous language syntax which attempts to capture the ability to express the full semantic richness of the HL above it. Based on a GB model, an expression in ASL best maps to a logical form. The three lowest layers, as well as, the ASL and Symbol Grounding Plane will be discussed in more detail in the next four sections.



Figure F-1. A Tower of the n-Towers Model

The actual number of towers needed by a system will depend on the application being considered. For the communication of two agents within a generic system, each agent would normally need only one tower. If these agents were collocated within a discrete portion of the system, even some portions of these towers could be shared. An example of the way in which a group of agents controlling various portions of a factory floor would communicate is given in Figure 2. In this example, only the one agent whose duties include the system's man-machine interface would need the top two HL levels of the n-Towers model.

Theoretical Basis for an Adaptive Semantic Language

As stated above, the purpose of the ASL is to provide a non-ambiguous interface between the HL layers and the three lowest layers of the n-Towers model. As its name implies, ASL is viewed as semantic language. This means it theoretically falls between the syntactic D-structure and the conceptual structure represented by the three lower layers of the n-Towers model. This positioning in the HL model requires that the resolution of at least syntactic and semantic ambiguity be done in the top two layers of the model. This does not force a restriction in the expressive power of the n-Towers model as a whole since there is nothing to preclude the two top layers from generating any number of different ASL sentences for a given HL sentence.

Before the concepts of the ASL can be fully defined, a formal definition of the syntax of both a standard formal language and HL must discussed.



Figure F-2. A Factory Application of the n-Towers Model

The Standard Formal Language Model

Given a grammar G = (V, T, P, S), a standard formal language L(G) is the set:

$$\{w \mid w \in T^* \text{ and } S \stackrel{*}{\cong} w\}$$

This definition assumes that both the set of terminal symbols $\{t \mid t \in T\}$ and the set of variables $\{v \mid v \in V\}$ are finite sets of pre-defined symbols from which a potentially infinite set of sentences can be constructed. The actual number of well formed sentences which can be generated by a non-recursive set of production rules $\{p \mid p \in P\}$ depends on the total number of elements in the sets P, T and V. Any form of recursion over the set P will cause the number of well formed sentences to increase to infinity.

А

While the addition of recursion to the set P allows the standard formal language L(G) to produce an infinite number of sentences, the fact that such a L(G) depends on an increasing sentence length to increase its expressional power presents a problem in attempting to use any semantic model based on L(G). As the length of the sentence grows, the overall meaning of the sentence will begin to become computationally clouded due to the exploding size of the semantic representation. This problem becomes especially acute if the recursion is indirect, due to the lack of predictability about which portion of the representation will grow. Further, since only the syntactic meaning of sentences can be dynamically varied, the rate of increase in the expressional power of L(G) will naturally decay as the sentence length increases due to reduction in the contribution of the more powerful lexical elements of semantics.

The Human Language Model

In contrast to the grammar which generates a standard formal language, a human language HL(HG) can be defined using elements of GB theory as:

$$HG = (X_v, G_w, P_x, C, CP).$$

Here the set of variables $\{v \mid v \in V\}$ has been replaced with the set of word categories $\{x \mid x, x', x'' \in X_V\}$ which can be plugged into the X-bar rules. The set of terminal symbols $\{t \mid t \in T\}$ has been replaced with a word grammar, G_W , which using its own grammar can generate the word level elements of the top level sentence grammar. In contrast to a formal language, a set of production rules $\{p \mid p \in P\}$ is not sufficient to capture the nature of the GB rules which control syntax, so a set P_X and C are provided as a replacement. The set P_X consists of the three X-bar rules from GB theory. The set C consists of all rules generated by the other five modules of the GB theory (i.e., Binding Theory, Bounding Theory, Case Theory, Control Theory and Theta Theory). The start symbol S has been replaced by the CP symbol which is the top phrase marker of a sentence in GB theory.

The word generation grammar can be defined as $G_W = (M, LU, P, R)$ where M is a set of morphological variables, LU is a set of stored lexical units {u | u \times LU}, MP is a set morphology rules, and R is the set of word roots {r | r \times R} which can grow over time. Thus, W is the set of well formed words generated from G_W by:

$$\{w \mid w \in LU^* \text{ and } R \stackrel{*}{\Longrightarrow}_{WG}^* w\}$$

ŀ

}

As can be seen from these definitions of HL(HG) and G_w , human language allows for both the growth of the word set W (the resulting set of all possible word generations from all possible roots) and the recursive application of X-bar production rules within the set P_x under the control of C. In HL(HG), the extent of recursion of the X-bar rules within the set P is somewhat controlled by the set C which places some constraints on the application of the X-bar rules. Further, the power of the set W to grow and the resulting increase in the lexical semantics that results, reduces the need to draw additional semantic value from a more complex syntactic form. The result is a dynamic language which can adapt to changing semantic requirements without increasing the mean-length-of-utterance required for communication. In a massively parallel system such as our brains, the resulting optimization toward increased search complexity does not present a processing problem; however, in a modern computer such a complex grammar within a grammar construction for a language does present a substantial processing problem.

The Adaptive Semantic Language Model

Clearly the best language model for the syntax of the ASL, based on its role as a computer to computer communication language, would be one which is as expressive as the human language model but as algorithmically efficient as the standard formal language model. Of course, such a model cannot be constructed since these two requirements are nearly mutually exclusive, but there is a large middle ground between the two models in which to select a compromise solution. To determine exactly where the ASL model should fit between these two models, the syntactic and lexical requirements of the ASL are discussed. Based on these requirements an ASL model is proposed.

ASL Syntax Requirements

4

One of the difficulties in the processing of a HL is the complexity of the syntax necessary to provide the richness of expressions required for the socialization of humans. Since the surface structure form of a sentence needs to take in account the social environment of the utterance, a number of pragmatically different forms of a semantically identical D-structure form can be generated by a set of rules provided for this expressiveness. Further, since the human brain uses a massively parallel processor which can store a great deal of information in temporary registers and process the stored utterances almost simultaneously with the completion of input, the S-structure ordering of phrases presents little or no problem to the parsing of the utterances, as long as, this ordering falls within the range of acceptable possibilities.

Since the top two levels of the n-Towers model can handle the pragmatic aspects of a HL, the language selected for the ASL does not need a number of surface variations of the D-structure of

an utterance. Thus, a model based on the logical forms of HL expressions is sufficient to express almost all of the semantic meaning needed. The only surface marker that needs to be preserved is one which differentiates between a declarative and interrogative utterance. However, a simple pre-punctuation of the utterance will convey the information without the need of the redundant word order changes used in such languages as English.

Based on ASL's proposed position below the D-structure, the need for the ASL to allow any recursion over its set of syntactic production rules can be eliminated if the lexical forms used have sufficient semantic power. In fact, based on the reduced scope of the semantic transmission inherent in a formal language, a major portion of the structure of the language can be contained within the lexicon. This assumes that the concept of external and internal q-roles is eliminated within the lexical entries, and that the potential difficulty in parsing a SVO or OVS word order is addressed.

Since the formal language of predicate calculus is very similar to a VSO or VOS language and presents little problem to a trained user, the mapping of the lexical forms to a predicate syntax would provide both a powerful level of expression and a very simple language to parse. Since the parser would be provided with the lexical unit's identity first (as the predicate label), it could easily look up the unit in the lexicon to determine its expected syntax and the number of thematic roles and start mapping the semantics of the contents as the terms of the predicate are read in. If the complete syntactic form of a predicate in predicate calculus is maintained, recursion within the s-grids could be easily recognized and interpreted.

The ASL's Lexical Requirements

During the natural evolution of a human language, the language's lexicon constantly changes. The phonology, morphology, syntactic relation and meaning of a lexical unit will naturally change over time and lexical units will be added or lost to the corpus of the language as the need to express new sets of ideas is created. Attempts to stop this evolution have proven fruitless since human language is closely tied to human society, and social change is a natural force not easily controlled. As a result of this natural change, human language contains methods of controlling the effect of a changing lexicon which can be exploited by an n-Towers model to allow the language to change to support the expression of new sets of ideas. Since the major forces for change in a human language reside outside the language, these control methods can be exploited without fear of encountering an internal force within the n-Towers model which would cause an uncontrollable change to the ASL.

)

F-8

Research in GB theory has provided evidence that all human languages use a concept of q-grids within the their lexicon to govern sentence construction and that the types of lexical categories containing the q-grids are remarkably uniform across all human languages. However, the number of lexical categories which have associated q-grids in HL is somewhat controversial. In addition, the amount of information stored in the q-grids varies greatly within different views of GB theory. Since there is some debate as to what amount, if any, of syntactic information is stored in the q-grid, the term s-grid (author's convention) will be used to refer to syntactic information stored in the lexicon. In addition a third grid, the a-grid (again author's convention) will be added to the lexicon to support the connection to the Symbol Grounding Plane.

In summary, the supporting lexicon will need to provide 1) the internal syntactic realization for all lexical units which can vary their internal syntactic realization, 2) a method of describing the semantic linking for lexical units of governing categories and 3) a method of firing some action as a result of the occurrence of a word in the ASL sentence. This will be done using s-grids, qgrids and a-grids, respectively. Since our goal is a formal semantic language which by nature will limit the number of ways the same semantic meaning can be expressed within the ASL syntax, the storage of a good portion of the language's syntax within the lexicon should be achievable, assuming that the number of lexical categories is reduced to a small set.

Lexical Categories

(

Human languages vary in the number of lexical categories used to logically divide the lexicon. For a majority of human languages, a set of 12 lexical categories will be sufficient to separate the language's lexicon. These are:

determiner

- adjective 1)
- 2) adverb
- 3) auxiliary verb
- infinitive 7) main verb

5)

6)

- 4) conjunctions
- noun 8)
- participle 9) 10) particle
- 11) preposition
- 12) pronoun
- Although this richness in the number of categories forces a complex syntax, the disadvantage of this complexity is far outweighed by the extreme expressive power it provides. Our task is reducing the number of categories which must exist at the semantic level while maintaining all of the expressive power of HL.

Due to their semantic importance, neither the noun nor preposition category can be eliminated or reduced in any real fashion. However, in almost all human languages, prepositions are notorious for having broad overlapping meanings. This fact can be remedied in the ASL since prepositions are a closed category, thus, allowing a more careful definition which is both narrow and non-overlapping.

The first reduction which can be made is in the number of categories required to express the overall concept of a verb. The major reason why this concept is spread over the auxiliary verb, - main verb, infinitive and participle is to allow for the expression of mood, aspect, tense and voice. If given a method in which to encode a verb's usage within the lexical form, the ASL only needs to express externally the concept in the present active indicative without aspect and these categories can be reduced to a single 'verb' category.

Adverbs, in human language, normally qualify the meaning of the main verb. This qualification is normally in relative terms and often difficult to interpret in any concrete way. For these reasons, it seems best to eliminate adverbs from the list of the ASL's lexical categories. This may again force some sort of encoding of verb category or addition of verbs like 'run quickly' unless a method can be found to make adverbs act as more well behaved members of the adjective category.

Particles are sort of the waste bucket of human language lexical categories. Any concept which cannot be fully expressed by another category or any difficulty in syntactic construction may result in the use of a particle. In the ASL, the need for particle can be eliminated by a careful definition of the syntax and other lexical categories.

The discussion of pronouns is slightly more complex since their number and use varies greatly from human language to human language. For the sake of our discussion, we will take a 'standard' view based on a number of Indo-European languages. Based on this view, the personal, reflexive and relative pronouns will be treated as pronouns and other types will be treated as either adjectives or determiners.

In most human languages, a personal pronoun is normally coreferential with either a referent which has already been described in the discourse by a Noun Phrase (NP) or has been associated to a grounded symbol by some non-language method. Since we are dealing with a formal language, we can dismiss the non-language symbol grounding condition as beyond our realm of interest and assume that for our purposes all personal pronouns will be coreferential with a referent which has been described by a NP. The use of personal pronouns as external referents is driven mostly by the need to reduce the amount of redundant information being transmitted in HL. In a formal language, this need is less important and use of the personal pronoun can be eliminated. £

F-10

The reflexive pronoun can be eliminated from a formal language for the same reason as the personal pronoun. The need for the ASL to support relative clauses, and thus, relative pronouns can be eliminated, but only if some type of conjunction with a temporal meaning is supplied at the syntactic level.

- As with pronouns, adjectives serve to shorten the utterance by allowing a qualification of the NP to be directly stated within the NP. Therefore, adjectives can also be eliminated from the ASL without any loss of semantic expressive power as long as nouns with the same semantic value are provided.

Determiners on the other hand serve to quantify, not qualify, their associated NP. As such, they cannot be fully eliminated from the ASL without losing semantic value. Since their semantic value will need to be carefully defined for a proper semantic interpretation of the whole utterance, they will need to be a closed category in the ASL. This follows from human language where they are a closed category for the same reason.

The semantic value of conjunctions also cannot be eliminated from the ASL without losing semantic value. However, since the number of conjunctions needed at both the lexical and syntactic level is very small, they can be formalized in the ASL syntax.

Based on all the justifications given, the number of open categories needed in the ASL can be reduced to two; 1) nouns and 2) verbs. The number of closed categories required can be reduced to three; 1) determiners, 2) conjunctions, and 3) prepositions. Of these, verbs and prepositions will require q-grids. Only verbs and nouns will require s-grids.

Lexical Entries

L

Since the purpose of the lexicon is to support adaptive change within the ASL, the number of entries in the lexicon must be allowed to change over time. This will require some set of components in the ASL model to have meta-knowledge about how to interpret the syntactic and semantic information stored in the lexicon. The meta-knowledge about syntax can be relatively straightforward since a category will either have a built-in syntactic realization or a s-grid which can be directly read for this information. The meta-knowledge about the semantic model of the ASL will need to be a little more complex since the interpreting component will need to do more than simply read an entry's q-grid. This component will need to understand both, 1) the number of possible thematic roles and how these roles map into a semantic understanding of the

language and 2) how to read a semantic description of a verb or preposition to complete the semantic understanding.

For a verb entry, the q-grid will need to consist of the required thematic elements and their thematic roles. To aid in parsing, no optional q-roles should be permitted. To allow some optional forms of a verb, the ASL should allow two or more entries for the same verb name in the lexicon. The s-grid for the verb will provide the phrase level constituents (either NP or PP) for each actor in the q-grid. For a preposition entry, the q-grid will need to consist of the thematic role of its actor. Since only NP's can fill this role, no s-grid will be required.

For a noun entry, the s-grid will define the other elements which can be used with the noun to form a NP. These elements will include determiners, conjunctions, prepositional phrases, and other NP's.

Determiners and conjunctions will be atomic elements within the ASL, and therefore, will have neither a s-grid or q-grid. Since determiners and conjunctions are both closed categories and are tightly defined, their storage in the lexicon is optional.

The ASL Model

Based on the requirements provided above, the grammar rules and lexical syntax can be now be defined for the ASL model. This is done in the next two sections. A set of examples is then given to show how the ASL could be used.

Grammar Level Syntax

Using a predicate syntax where a verb name is the predicate label and the verb's governed q-roles are the terms, the syntactic production rules of the ASL grammar can be reduced to the following non-recursive set of rules:

S Æ SI | ?(SI) SI Æ VP | NOT VP VP Æ V | V / V | V Æ V | V > V where: V is a verb predicate, VP is a verb phrase, SI is an declarative sentence, and ?(SI) is interrogative sentence. 1

This language allows a syntactic level negation (NOT) of the sentence. In addition, the following syntactic level conjunctions are allowed across two V's:

- 1) a logical disjunction (/)
- 2) an if-then relationship (\mathcal{A})
- 3) a temporal ordering (>)

A logical conjunction (AND) is not provided at the syntactic level since there is no semantic value added to two simple sentences being ANDed together to make a complex sentence over the same two simple sentences being stated as totally separate statements.

A more powerful recursive language could also be defined as:

S Æ SI | ?(SI) SI Æ VP VP Æ V | V / VP | V Æ VP | V > VP |NOT (VP)

However, such a language would present a more difficult semantic evaluation and may be overly expressive for the average domain.

Lexical Unit Level Syntax

As stated above, only verb phrases will appear at the sentence level. The syntax of the verb phase will be controlled by the grammar and the lexical syntax of the verb. The syntax of a verb will be:

verb-name $(s_0, s_1 \dots s_N)$

where the number of s-roles (s_x) is determined by the verb's s-grid. These s-roles can either be filled by a NP or a PP. The syntax of these will be:

```
noun-name (s_0, s_1 \dots s_N)_0 [j noun-name (s_0, s_1 \dots s_N)_1
... j noun-name (s_0, s_1 \dots s_N)_N]
preposition-name (NP)
```

where the symbol, j, stands for either a logical AND (\ddot{Y}) or OR (/) operator and the number of sroles (s_x) for the noun element within the NP is determined by the noun's s-grid. The s_0 position of this s-grid can only be filled by a determiner. The s_1 position will be reserved for NP's acting as adjectives. The remaining s-roles are optional and can be either filled by a NP or a PP.

Example Expressions

Based on the above design, the following sentence examples for a NGC type domain are given:

English:	The lathe rounds the part with a cutting tool.				
ASL:	rounds(lathe(the, ø), part(the, ø), with(tool(a, cutting)))				
English:	Pick up the blue part on the green table with the stationary robot arm.				
ASL:	pick-up (part(the, blue, ø, on(table(the, green, ø, ø))), with(robot-arm(the, stationary(ø))))				
English: pocket in pa	Get the cutting tool from the tool caddie, and then, cut a 2mm deep rt.				
ASL:	<pre>get(tool(the, cutting(Ø), Ø, Ø), from(caddie(the, tool(a, Ø, Ø, Ø), Ø, Ø))) > cut(pocket(a, 2mm), part(a, Ø, Ø, Ø))</pre>				
English:	Move the tool alone the x-axis plus 5mm.				
ASL:	move-linear (tool(the, ø, ø, ø), position(relative, x-axis, +5mm))				

In the examples, the ø symbol indicates a required s-role within a lexical unit being used for which no (or NULL) information is needed to complete the English sentence being translated.

Situational Dependencies

As stated above, the Situational Dependencies layer provides the basic semantic network for the storage of the Lexical Semantics. This semantic network allows for the existence of three types of nodes: 1) verb, 2) noun and 3) preposition. As shown in Figure 3, each of these node types provide the required storage of the tag and the needed grids.

١

)



Figure F-3. Situational Dependency Nodes with Theta Relations

The expressive power of this network comes from the ability of the directed arcs (relations) to contain both type and situation information about the relation (i.e., the time and world in which the relation applies). Thus, procedural events, scenarios, and even beliefs can be directly superimposed on the lexical semantics without any risk of the information being misused by the system. As a result of this ability, the concept of long term and short term memory can be used with the n-Towers model without any of the complexity normally associated with systems that store these two types of knowledge in two different places.

As shown in Figures 3, 4, and 5, there are five different classes of situational relations define in the Situational Dependencies KR. These are:

rq relations -	defines the noun or preposition node allowed to fill a verb's
	theta role.
r _e relations -	defines the noun or preposition node allowed to fill a noun's
	epsilon role.
r _i relations -	defines a pure reciprocal relation between nodes of the same
	type.

r _n relations -	defines non-reciprocal relation between nodes of the same
r _w relations -	type. defines two relations which when taken to together produce a reciprocal relation between nodes of the same type.

- The actual number and names of the relations allowed in a relation class is defined in the Meta-Semantic level of the n-Towers model.



Figure F-4. Epsilon Relations of Situational Dependency Nodes

Meta-Semantics

The Meta-Semantics layer allows the definition of the legal relations allowed in a tower of the n-Tower model and the inferencing methods associated with those relations. Since relations must be defined as one of the five classes stated above, inferencing method can be associated with either a relation or a relation class.

The Meta-Semantics layer also handles the housekeeping for the reciprocal relations, ensuring that when a reciprocal relations is defined between Node A and Node B, the correct relation is defined between Node B and Node A. As an example of this, if the relation 'A NC-machine

7

)

always is-a machine in the real-world \dot{O} is added to the semantic knowledge, then the relation A machine always could-be-a NC-machine in the real-world \dot{O} would be added by the metasemantics if a reciprocal relation between is-a and could-be has been defined.



Figure F-5. Other Relations of Situational Dependency Nodes

Lexical Semantics

As stated above, the verbs, nouns and prepositions which can be used by the ASL will be stored in a lexicon which is free to grow as new concepts are needed to be expressed. In the n-Towers model, this lexicon will exist in the Lexical Semantic layer. The amount of information which must be stored in the Lexical Semantic layer for a lexical unit will depend on the category of the lexical unit. For all entries, a tag and the set of current situational links to the semantic representation of the domain knowledge (i.e., the other lexical units) will be required.

If the lexical unit is a verb, a q-grid containing the number and type of q-roles associated with the unit will be maintained to allow a road map to which noun and preposition nodes in the domain can be legally linked to that verb. If the lexical unit is a preposition, a single q-role will be maintained for this purpose. If the lexical unit is a noun, a e-grid containing the number and

٩

type of e-roles associated with the unit will be maintained to allow a road map to which noun and preposition nodes in the domain can be legally linked to that noun. The q-grid, q-role and the e-grid are represented as a simple ordered list of roles that corresponds to the unit's s-grid and a-grid.

- Since a verb is one of the most complex types of entry, a verb will be used as an example. Given the verb, 'rounds', as used in the sentence:

The lathe rounds the part with a cutting tool.

The following lexical entry for 'rounds' would appear in the lexicon:

Tag: I	hit			•	
q-grid:	 Agent Patient Instrument 	s-grid:	1) NP 2) NP 3) PP	a-grid:	 action X action Y action Z

Both the domain knowledge and general knowledge of a tower will be contained in a set of situational dependent links between the lexical unit elements in the network (the nodes). The Meta-Semantic layer will determine the set of inferencing methods needed for this knowledge. As a result of the language information stored in the Lexical Semantic layer, the ASL layer is able to prohibit meaningless sentences. For example, given the sentence:

English: The part rounds the lathe with a cutting tool.ASL: rounds(part(the, ø), lathe(the, ø), with(tool(a, cutting)))

using the information stored in the Lexical Semantic layer, the ASL parser would be able to determine that the q-roles for 'rounds' have been violated by the use of the NP:'the part' as a agent and the NP:'the lathe' as a patient.